

### Department of Energy

Yucca Mountain Project Office P D. Box 98608 Lac Vegas, NV 89193-8608

WBS 1.2.9 QA: N/A

1

102.8

## DEC 10 1990

Distribution

TRANSMITTAL OF "STATUS OF DATA, MAJOR RESULTS, AND PLANS FOR GEOPHYSICAL ACTIVITIES, YUCCA MOUNTAIN PROJECT" BY H. OLIVER, E. HARDIN, AND P. NELSON

The subject report is enclosed for your information. This report has frequently been referred to as the Geophysics White Paper and was written in response to U.S. Nuclear Regulatory Commission concerns about a need for integration in the geophysical program of the Yucca Mountain Site Characterization Project.

If you have any questions or desire extra copies of the report, please contact Christopher J. Fridrich at (702) 794-7587 or FTS 544-7587.

Maxwell Bunhard Carl P. Gertz, Project Manager Yucca Mountain Site Characterization Project Office

RSED: COF-1012

Enclosure: "Status of Data, Major Results, and Plans for Geophysical Activities, Yucca Mountain Project"

cc w/o encl: W. V. Macnabb, SAIC, Las Vegas, NV, 517/T-04 T. A. Grant, SAIC, Las Vegas, NV, 517/T-13 E. L. Hardin, SAIC, Las Vegas, NV, 517/T-13

9101020013 901210 PDR WASTE WM-11 PDR

YMP-5

270026

a dia series

FULL TEXT ASOIL SCAN

DEC 10 1980

Distribution--Memorandum dated

J. W. Bartlett, HQ (RW-1) FORS F. G. Peters, HQ (RW-2) FORS T. H. Isaacs, HQ (RW-4) FORS J. D. Saltzman, HQ (RW-5) FORS Samuel Rousso, HQ (RW-10) FORS J. C. Bresee, HQ (RW-10) FORS S. J. Brocoum, HQ (RW-20) FORS D. E. Shelor, HQ (RW-30) FORS G. J. Parker, HQ (RW-30) FORS R. A. Milner, HQ (RW-40) FORS F. G. Peters, HQ (RW-50) FORS D. U. Deere, NWTRB, Washington, DC (13) W. M. Hewitt, Weston, Washington, DC Technical Information Center, Weston, Washington, DC A. E. Gurrola, RSN, Las Vegas, NV, M/S 519 J. C. Calovini, RSN, Las Vegas, NV, M/S 519 D. L. Lockwood, RSN, Las Vegas, NV, M/S 403 R. L. Bullock, RSN, Las Vegas, NV, M/S 403 R. J. Herbst, LANL, LOE Alamos, NM (4) H. N. Kalia, LANL, Las Vegas, NV B. M. Crowe, LANL, Los Alamos, NM D. E. Broxton, LANL, Los Alamos, NM (4) L. J. Jardine, LLNL, Livermore, CA (3) D. G. Wilder, LLNL, Livermore, CA (3) R. W. Lynch, SNL, 6300, Albuquerque, NM T. E. Blejwas, SNL, 6310, Albuquerque, NM (5) Glenda Sweatt, SNL, 6310, Albuquerque, NM (2) L. E. Shepard, SNL, 6313, Albuquerque, NM (10) P. H. Nelson, USGS, Denver, CO (20) K. W. Causseaux, USGS, Denver, CO R. V. Watkins, USGS, Denver, CO V. M. Glanzman, USGS, Denver, CO (100) J. F. Divine, USGS, Reston, VA L. R. Hayes, USGS, Las Vegas, NV (6) D. L. Zesiger, USGS, Las Vegas, NV H. W. Oliver, USGS, Menlo Park, CA (100) J. N. Stellavato, SAIC, Las Vegas, NV (2) J. H. Nelson, SAIC, Las Vegas, NV SAIC-TEMSS Library, SAIC, Las Vegas, NV (2) C. D. Wolss, SAIC, Las Vegas, NV R. E. Lowder, MACTEC, Las Vegas, NV Project Directorate, Rep. Lic. & QA, NRC, Washington, DC Sr. Project Mgr., Yueca Mountain Repository Project, NRC, Washington, DC NRC Document Contro Desk, NRC, Washington, DC P. T. Prestholt, NRC, Las Vegas, NV E. P. Binnall, LBL, Berkeley, CA Center for Nuclear Waste Regulatory Analyses, San Antonio, TX E. C. Ezra, EG&G, Las Vegas, NV C. F. Costa, EPA, Las Vegas, NV ONWI Library, Battelle, Columbus, OH Thomas Hay, Office of the Governor, Carson City, NV R. R. Loux, NWPO, Carson City, NV (3) C. H. Johnson, NWPO, Carson City, NV

## DEC 1 0 1990

#### Distribution-Memorandum dated

John Fordham, DRI, Reno, NV Martin Mifflin, DRI, Las Vegas, NV (2) Eric Anderson, Mountain West Research-Southwest, Inc., Phoenix, AZ Department of Comprehensive Planning, Clark County, Las Vegas, NV Planning Department, Nye County, Tonopah, NV Lincoln County Commission, Lincoln County, Pioche, NV Judy Foremaster, Caliente, NV (5) Economic Development Department, City of Las Vegas, NV Community Planning and Development, City of North Las Vegas, NV City Manager, Henderson, NV Director of Community Planning, City of Boulder City, NV Commission of the European Communities, BELGIUM I. B. Alterman, National Academy of Sciences, Washington, DC (19) Charlotte Abrams, Advisory Council on Nuclear Waste, Washington, Dan John Price, Nevada Bureau of Minesand Geology, UNR, Reno, NV Dave Brune, UNR/Seismic, Reno, NV C. G. Bell, UNLV, Las Vegas, NV Technical Information Office, DOE, NV (12) D. J. Bales, DOE, Oak Ridge, TN P. K. Fitzsimmons, HPD, NV C. L. West, OEA, NV D. L. Fraser, REECo, Las Vegas, NV R. F. Pritchett, REECo, Las Vegas, NV C. P. Gertz, YMP, NV (5)

D. G. Horton, YMP, NV



YMP/90-38

YUCCA MOUNTAIN PROJECT

## STATUS OF DATA, MAJOR RESULTS, AND PLANS FOR GEOPHYSICAL ACTIVITIES, YUCCA MOUNTAIN PROJECT

THE NT OF STREET, WILLING

JULY 1990

6

UNITED STATES DEPARTMENT OF ENERGY

STATUS OF DATA, MAJOR RESULTS, AND PLANS FOR GEOPHYSICAL ACTIVITIES, YUCCA MOUNTAIN PROJECT

## Editors

H.W. Oliver (USGS, Menlo Park, California) E.L. Hardin (SAIC, Las Vegas, Nevada) F.H. Nelson (USGS, Denver, Colorado)

¢.

July 1990

Prepared for the U.S. Department of Energy by the United States Geological Survey and Science Applications International Corporation.

.

#### DISCLAIMER

This report was prepared as an account of work sponsored by the United States Government. Neither the United States nor the United States Department of Energy, nor any of their employees, makes any warranty, expressed or implied, or assumes any legal liability or responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights. Reference herein to any specific commercial product, process, or service by trade name, mark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof.

Printed in the United States of America

DOE and DOE contractors may acquire additional copies from:

Office of Scientific and Technical Information P.O. Box 62 Oak Ridge, Tennessee 37831

Others can order from:

National Technical Information Service U.S. Department of Commerce 5285 Port Royal Road Springfield, Virginia 22161

Price: Printed Copy Contact OSTI or NTIS Microfiche "

iii.

## TABLE OF CONTENTS

	an a	of C.
1.0	INTRODUCTION	1
	Scope of This Report	1
	Maps and Geologic Units	2
2.0	PAST GEOPHYSICAL ACTIVITIES	5
2.1	Gravity Investigations	5
	Objectives	5
	Data Coverage and Quality (Gravity Measurements)	ě.
	Data Coverage and Guality (Deneity Data)	11
	Cummary of Deciles	11
	Summery of Mesures ( ) , , , , , , , , , , , , , , , , , ,	4.4
	protrodrabul	73
2.2	Magnetic and Paleomagnetic Investigations	17
	Objectives	17
	Data Coverage and Ouality	17
	Ground Magnetic Data	02
	Palantamatia and back Mannatia Studies	50
	Cummert of Deputes	00
	Dibligeraby	60
	ereregraphy	28
2.3	Geoelectric Surveys	33
	Objectives	33
	Data Coverage and Quality	22
	Summary of Results	24
	Ribliography	AE
	www.wogewpag.co.co.co.co.co.co.co.co.co.co.co.co.co.	4.0
2.4	Seismic Refraction Surveys	47
	Objectives	47
	Data Coverage and Quality	47
	Summary of Results	52
	Bibliography	5.5
2.5	Seismic Reflection Surveys	57
	2.5.1 Shallow, High-Resolution Reflection Profiling	57
	Objectives	57
	Data Coverage and Quality	60
	Summary of Results	65
	2.5.2 IntermediaterDenth Deflection Current	00
	OPPORTION AND AND AND AND AND AND AND AND AND AN	00
	Objectives and outline and a state of the st	00
	Data coverage and Quality	66
	Summary of Results	70
	2.5.2 Deep Reflection Profiles	70
	Objectives	70
	Data Coverage and Quality	71
	Summary of Results	72
	Bibliography	73

.

## TABLE OF CONTENTS (continued)

.

		Page
2.6	Remote Sensing and Aerial Gamma-Ray Surveys	75
	2.6.1 Remote Sensing	75
	Objectives	75
	Data Coverage and Quality	75
	Summary of Results	76
	Bibliography	77
	2.6.2 Aerial Gamma-Ray Surveys Over the Nevada Test Site	81
	Objectives	81
	Data Coverage and Quality	81
	Summary of Results	82
	Bibliography	82
2.7	Borehole Geophysics and Petrophysical Measurements	85
	Objectives	85
	Data Coverage and Quality	85
	Summary of Results	86
	Fracture Detection/Characterization	93
	Neutron Moisture Meter Logging	93
	Other Slim-Hole Moisture Logging	94
	Related Infiltration Data from Borehole Geophysics	95
	Bibliography	95
5		
2.8	Teleseismic Tomography Surveys	99
	Objectives	99
	Data Coverage and Quality	99
	Summary of Results	101
	Bibliography	102
3.0	PLANNED GEOPHYSICAL ACTIVITIES	105
3.1	Disquesion of Dispused Coophysical Asticulture	
w 1.4	2 1 1 Neestursted Jane Budgelenia Tempetiesting	115
	Far-Field Characteries from Westigstions	115
	Formation Characteristics from vertical Seismic Profiling .	115
	obveice) logging	
	Borphole Ornephole and Curfegete-benchels Mathematic	110
	Remote Sensing	118
	Remove pending	118
	Evelopation of the Jarge Budgeulie Creditat to	119
	Investigation of Docharge in Charge Machine	119
	Fracture Characteriastics is the Seturated Sec.	123
	Pagional Characterization of Comparis and Dilinguish and	124
	3 1 3 Geophysical Characterization of Viewold and Paleozoic Rocks .	125
	Percurase	
	3.1.4 Geophysical Studies for Content and Tenterio	126
	Decional Combusies for Georogy and Tectonics	130
	Perional Geophysical Studies	130
	Detoction (Charactering of Decker	131
	Mountain Parian	
	Characterization of Volcovic Boston and a sector	132
	UNALAULELIZATION OF VOICANIC FEATURES	137

6

1

## TABLE OF CONTENTS (continued)

		Page
	Teleseismic	137
	Support of Paleoclimate Studies	138
	3.1.5 Geophysical Studies in the Site Area, for Geology and	
	Tectonics	139
	Structure and Stratigraphy	139
	Detailed Gravity	139
	Aeromagnetic and Ground Magnetic Surveys	142
	Geoelectric Methods	143
	Radioactivity and Remote Sensing	143
	Seismic Reflection	144
	Seismic Refraction	145
	Characterization of Fractures and Faults in the Repository	
	Block	146
	Borehole Geophysics	146
	EM Tomography	146
	Vertical Seismic Profiling	147
	3*D Characterization of Lithology and Hydrologic State of	
	the Repository Block	148
	Borencle Geophysics	148
	5.1.6 Engineering Geophysics for Surface Facilities Siting	149
	Trenching and Ground Penetrating Kadar in Midway Valley	150
	Exploration of the Alluvium-Bedrock Contact	150
	In Situ Engineering Froperties	101
	References	TDT
3.2	Geophysical Integration and Prioritization of Activities	155
	Interfaces Between Studies	156
	NRC Site Characterization Analysis (SCA) Concerns	156
	Activity Prioritization	165
	Subsurface Structure of the Yucca Mountain Site Area and	
	Region	166
	Distribution of Volcanic Deposits	168
	Exploration of the Large Hydraulic Gradient Area	169
	Characterization of Fractures and Fault Zones in the Repository	
	Block	170
	Exploration of the Proposed Exploratory Shaft Location	171
	Assessment of the Potential for Natural Resources	171
	Feasibility Test Plan - Geophysics Integration and Feasibility	
	Testing	171
	Summary ,	172
3.3	Feasibility Tests	175
3.4	Oh Requirements for Using Data from Past Geophysical Activities	1.91
	Introduction	181
	NRC-Defined Data Qualification Methods	182
	The Qualification Process	182
	Candidate Data Sets for Oualification	184
	References	186

## TABLE OF CONTENTS (continued)

## Page

4.0 SUMM	ARY AND	CONCL	USION	IS .		4 A		÷,	÷	× . ;	4.5		i de s		4.1	d,	4	189
5.0 ACKN	OWLEDGME	INTS		, ,						÷	•			4		1		191
APPENDIX CHARACT DECEMBE	A - TABI ERIZATIC R 1988)	LE OF ON PLA	CONTE N (SE	NTS	FOR N 8	. vo	LUMI - SI	SS I SCTI	V AL ON	ND V 8.3.	OF 1.1	THE 7.5)	SITE (DOE	C/RV	(-01 	99,		A-1
INDEX										4.4					1			1-1

## LIST OF FIGURES

Figure	Title	Page
2.1-1	Distribution of gravity stations in the Yucca Mountain region	7
2.1-2	Distribution of gravity stations in the Yucca Mountain site area	9
2.2-1	Aeromagnetic index map of regional study area showing flight line spacing of available data	19
2.2-2	Aeromagnetic index map of site area and vicinity showing the locations of available surveys	21
2.3-1	Location of geoelectric surveys in the Yucca Mountain region, including the Nevada Test Site, southern Nevada	35
2.3-2	Location of geoelectric traverses in the Yucca Mountain site area	37
2.3-3	Location of Schlumberger resistivity soundings acquired by Greenhaus and Zablocki (1982)	43
2.4-1	Map of Yucca Mountain refraction lines from Pankratz (1982) .	49
2.4-2	Map of the Yucca Mountain region showing the locations of five high resolution upper-crustal seismic refraction profiles	53
2.5-1	Location of seismic reflection surveys reported by McGovern (1983) for the Yucca Mountain site area	63
2.5=2	Location of seismic reflection lines in the Yucca Mountain region, acquired by the Yucca Mountain Project, or by the oil-and-gas industry for speculative purposes	67
2.7-1	Map of boreholes in the Yucca Mountain site area from which geophysical logs have been acquired by the Yucca Mountain Project	87
2.8=1	Maps of seismograph stations used in teleseismic tomography studies	100
3.1-1	Map of Yucca Mountain, showing the location of planned magnetotelluric soundings, which will be used to investigate the large hydraulic gradient	121
3.1~2	Planned regional geophysical traverses with tentative deep and intermediate depth seismic reflection lines added	127
3.1-3	Planned shallow seismic reflection (Mini-Sosie) lines for assessment of method, and for investigation of faults in the Yucca Mountain region	133

## LIST OF FIGURES (continued)

with .

ò

, e.

1.00

8

L. Sara

68

Figure	Title	Page
3.1-4	Tentatively planned deep (and intermediate-depth) seismic reflection profiles across Yucca Mountain, and in the	105
	runeardre Arcrurch	135
3.1-5	Approximate coverage for planned detailed gravity and detailed aeromagnetic surveys	141
3.2-1	Matrix correlating categories of geophysical methods, with SCP studies wherein geophysical data will be collected and used .	157

### LIST OF TABLES

Table	Title	Page
2.1-1	Regional gravity maps of various areas within the regional study area	6
2.1-2	Density and thickness of selected geologic units in the Yucca Mountain area	12
2.2-1	Aeromagnetic surveys all or partly within the Yucca Mountain site area and vicinity	18
2.2-2	Magnetic properties of volcanic stratigraphy at the site	26
2.3-1	Geoelectric investigations on Yucca Mountain	39
2.5-1	Existing seismic reflection data and past surveys	58
2.7+1	Log types in each drillhole	89
2.7-2	Petrophysical data from Yucca Mountain boreholes	91
3.1-1	Summary of planned geophysical activities	107
3.2-1	Interfaces between geophysical activities, arranged by SCP activity	158
3.2-2	Summary of high-priority geophysical activities	173
3.3-1	Geophysical feasibility tests	177



#### REPORT ON GEOPHYSICAL ACTIVITIES FOR THE YUCCA MOUNTAIN PROJECT

#### 1.0 INTRODUCTION

This report describes past and planned geophysical activities associated with the Yucca Mountain Project and is intended to serve as a starting point for integration of geophysical activities. Geophysical surveys were conducted at Yucca Mountain as early as 1978, when repository siting investigations in the Nevada Test Site (NTS) area were begun. This report relates past results to site characterization plans, as presented in the Yucca Mountain Site Characterization Plan (SCP). As indicated in the SCP, many geophysical activities have not been planned explicitly or in detail because of uncertainty as to the applicability of various methods. A characterization activity was incorporated in the SCF to structure the evaluation and planning of geophysical activities during site characterization (SCP Section 8.3.1.4.1.2). This integration activity is tasked with reducing the uncertainty attendant to the application of geophysical methods. This report ("white paper") is a preparation for that activity. Whereas this report identifies some new exploration concepts and elaborates on some activity descriptions in the SCP, if changes are made to the scope of work described by the SCP, they will be made in accordance with change-control procedures.

Importantly, this report does not present geophysical data or interpretation. Rather, only survey coverage, data quality, and applicability of results to site characterization are discussed, as a means to relate past and planned activities. Extensive references to data and interpretive reports are provided, including many not directly cited in this report. Several such reports are currently in preparation and could not be referenced, including one summarizing regional geophysics, one summarizing geophysical logging at Yucca Mountain, and one describing teleseismic tomography based on data collected in 1982.

The SCP contains a number of studies and activities that will use relatively new geophysical methods, or methods that have not been applied in volcano-tectonic settings such as Yucca Mountain. The need to try these is based on the prospect that they may yield information of sufficient quality to be of significant value in site characterization. Both the SCP and this report emphasize plans for feasibility testing, on the basis that the cost of such testing is outweighed by the potential gain in added confidence of characterization of site conditions.

#### SCOPE OF THIS REPORT

This report discusses seismic exploration, potential field methods, geoelectrical methods, teleseismic data collection and velocity structural modeling, and remote sensing. The following are important areas of investigation that are discussed in the SCP, but are excluded from direct consideration in this report for reasons of expediency: geodesy and strain monitoring, periodic remeasurement of gravity stations over the long term, teleseismic monitoring for ground-motion studies (as opposed to tomography, which is included), heat flow measurements, and borehole stress measurement. This report discusses surface-based, airborne, borehole, surface-to-borehole, crosshole, and Exploratory Shaft Facility-related activities. The data described in this paper, and the publications discussed, have been selected based on several considerations: location with respect to Yucca Mountain, whether the success or failure of geophysical data is important to future activities, elucidation of features of interest, and judgment as to the likelihood that the method will produce information that is important for site characterization.

#### MAPS AND GEOLOGIC UNITS

The maps used in this report are, with one exception, presented on consistent regional and site area base maps. The regional base was prepared by considering a 100 km radius around the site, then extending the map boundaries to the next 0.5 degree increment of latitude and longitude (see for example, Figure 2.1=1). Physiographic information was included by shading within selected elevation contours, which encompass particular features, and labeling those features. Various different elevation contours were used. The site area base (e.g., Figure 2.3=2) covers from 116°22'W to 116°30'W, and from 36°45'N to 36°55'N. This quadrangle includes the conceptual boundaries of the repository perimeter drift and the controlled area. Topographic 50-m contours were obtained from the U.S. Geological Survey (USGS) 1:100,000 Beatty CA/NV quadrangle, and labels were provided for certain physiographic features.

These boundaries were adopted for purposes of presenting planned site characterization activities in this report. Geophysical activities will investigate areas beyond arbitrary boundaries as necessary.

The stratigraphic units used in this report are consistent with those of the SCP, particularly Section 1.2.2 (lithostratigraphy) and Section 3.9 (hydrogeologic stratigraphy). Lithostratigraphic information is also contained in Tables 2.1-2 and 2.2-2 of this report. In general, this report refers to lithostratigraphy when discussing geologic and tectonic applications, and hydrogeologic stratigraphy when discussing hydrologic applications. (The discussion is not particularly sensitive to the differences between the stratigraphic systems.)

A description of the stratigraphy and general characteristics of the unsaturated, repository block is provided here for the convenience of the reader. Briefly, the tuff sequence consists of several tuff units that were erupted from nearby volcanic centers in Miocene time, and have since undergone structural deformation. The total thickness of tuff varies, but is more than 1,200 m over most of the site area. Tuff lithology ranges from nonwelded to densely welded, corresponding to a matrix porosity range of roughly 30% to 5%, respectively. Mineralogical differences are superimposed on welding, depending on source magma composition, rapidity of cooling, flow thickness, ar secondary alteration that is probably associated with ground water. Much or the welded tuff is pervaded by mineralized, interconnected macropores (17 hophysae). These were formed from the action of volatiles during cooling, and comprise from zero to 30% bulk porosity. Finally, cooling fractures occur in the welded and partially welded units, especially the relatively nonlithophysal zones. The available data suggest that all tuff units at Yucca Mountain contain tectonic fractures. The porcus nonwelded units generally contain far fewer such fractures.

The caprock at Yucca Mountain consists of about 140 m of welded tuff known as the Tiva Canyon Member of the Paintbrush Tuff, with prevalent cooling fractures and lithophysae. Canyons on the eastern flank of Yucca Mountain are infilled with a few meters of alluvium; the minimum thickness of Tiva Canyon under these canyons is several tens of meters. Immediately underlying is a sequence referred to as the nonwelded beds of the Paintbrush Tuff, consitting of individual ash flow and ash fall units totaling roughly 30 m, but varying significantly in thickness and continuity of constituent beds within the site area. The porosity of these nonwelded tuffs is at least 30%, and the saturated matrix conductivity is several orders greater than welded tuff. Underlying these beds is the Topopah Spring Member of the Paintbrush Tuff, consisting of about 300 m of welded tuff. This unit may be further segregated into thick zones corresponding to variations in welding, devitrification, lithophysal porosity, and other alteration. The candidate host rock has been identified as a relatively nonlithophysal, densely welded zone near the top of the lower third of the Topopah Spring Member.

Between the candidate repository horizon and the water table is the lower third of the Topopah Spring Member, and the upper part of the nonwelded tuffaceous beds of the Calic: Hills (hereinafter called the Calico Hills unit). The lower part of the Topopah Spring Member consists of fractured welded tuff, some partially welded strata, and a basal vitrophyre approximately 10 m thick. The Calico Hills unit is comprised of several major ash flows and ash falls. The vitric matrix of the Calico Hills has been substantially altered to zeolites in the northern part of the site area, thus changing the rock fabric and the hydrologic properties. Fracturing and faulting in the Calico Hills unit are believed to resemble that observed in the thick, nonwelded tunnel beds of Rainier Mesa on the NTS, which have been explored by extensive tunneling.

A number of welded and nonwelded tuff units lie below the Calico Hills unit in the saturated zone. Extrusive and shallow intrusive volcanic rocks have been encountered in boreholes penetrating the lower part of the tuff section. Below the tuffs is a few hundred meters of older, poorly known volcanic rocks and sediments of volcanic and other origin. The tuffs and sediments in this lower part of the section are generally more altered than overlying units, and contain secondary clay, zeolites, and carbonate minerals. Underlying this are Paleozoic carbonates, which have been mapped throughout the region.

Yucca Mountain is an upland area mostly surrounded by deep, downfaulted, alluvial filled basins. North-south normal far ts transect Yucca Mountain and have given rise to gentle (6 to 10°) eartwild in three whom most of the site area. The repository horizon follows the tilted stratigraphy, meaning that the repository height above the water table and the intervening stratigraphy vary significantly. Tuff unite at the site, particularly those comprising the unsaturated zone, exhibit late is variability corresponding to the inferred distances to the eruptive center for the units. Thus at Busted Butte situated southeast of the site, the nonwelded beds of the Paintbrush Tuff and the Calico Hills unit are substantially thinner than to the north. Also, the Topopah Spring Member is generally thinner and less densely welded.

Alternative conceptual models have been developed to describe and explain the structural setting of Yucca Mountain, as presented in SCP Section 8.3.1.8 (Tables 8.3.1.8-7 and -8) and Section 8.3.1.17 (Tables 8.3.1.17-7 and -8). These models pertain to such topics and features as the significance of Crater Flat, the subsurface geometry of faults, and the presence of a detachment below Yucca Mountain. One of the major applications of geophysical methods in site characterization will be to further elucidate these features in the subsurface.

#### 2.0 PAST GEOPHYSICAL ACTIVITIES

This section reviews past and ongoing geophysical activities performed for the Yucca Mountain Project, or performed in the vicinity of the Yucca Mountain site. As stated in Section 1.0, only the location, methodology, and applicability of past activities are discussed. The reader is referred to the various cited references for presentation of geophysical data. The scope of this section is limited as discussed in Section 1.0, and notably does not include teleseismic ground motion characterization (but does include teleseismic tomography), heat flow studies, or stress measurements.

#### 2.1 GRAVITY INVESTIGATIONS

#### OBJECTIVES

Gravity investigations were begun at Yucca Mountain in about 1978 to characterize the general geologic and tectonic setting of the area. Gravity studies are particularly useful for (1) characterizing the general configuration of the regional pre-Cenozoic basement, (2) detecting concealed or unrecognized faults, (3) estimating the offset or extent of known faults, and (4) detecting and characterizing igneous features such as calderas and plutons. Gravity methods can detect shallow as well as deep features that juxtapose rocks of significantly different densities. With appropriate horizontal and vertical controls, gravity data also can reveal undulations of the base of the crust, which occur at a depth of approximately 33 km beneath Yucca Mountain.

#### DATA COVERAGE AND QUALITY (Gravity measurements)

Figure 2.1-1 shows the location and distribution of all gravity measurements from available data sources located within the area described in Chapter 1.0. About 33,000 gravity measurements have been made in this area, and all have been adjusted to a common gravity datum and recompiled. The accuracies of the gravity measurements themselves are generally 0.1 mgal, but Bouquer and residual anomalies derived from these measurements are less accurate, particularly in mountainous terrain where terrain corrections are required. Anomalies calculated from the data in Figure 2.1-1 generally have an uncertainty of about 0.5 mgal. Regional gravity anomalies are generally in the range of 5 to 50 mgal (see, for example, Hildenbrand et al., 1988, Figure 2.11), so calculated anomalies are sufficiently accurate for regional studies. Figure 2.1-2 shows the same information at about twelve times enlargement for the Yucca Mountain site area and vicinity. Gravity maps based on these data have been compiled for a number of areas at various scales (Table 2.1-1). The principal facts of all the data have been released on magnetic tape for Nevada (Saltus, 1988c) and California (Snyder et al., 1981).

Description	Scale	Reference
Bouguer and residual gravity map of Southern Great Basin	1:2,500,000	Hildenbrand et al., 1988, Fig. 2.8 & 2.11
Bouguer gravity map of Nevada	1:750,000	Saltus, 1988a
Bouguer gravity map of California	1:750,000	Oliver et al., 1980
Residual gravity map of California	1:750,000	Roberts et al., 1981
Residual gravity map of Nevada	1:1,000,000	Saltus, 1988b
Bouguer gravity map of Death Valley Sheet	1:250,000	Healey et al., 1980b
Bouguer gravity map of Goldfield	1:250,000	Healey et al., 1980a
Bouguer gravity map of Caliente Sheet	1:250,000	Healey et al., 1981
Bouguer gravity map of Las Vegas Sheet	1:250,000	Kane et al., 1979
Complete Bouguer gravity map of NTS	1:100,000	Healey et al., 1988c
Isostatic residual gravity map of NTS	1:100,000	Ponce et al., 1988
Residual gravity map of Yucca Mt. and vicinity	1:48,000	Snyder & Carr, 1982

# Table 2.1-1. Regional Gravity Maps of Various Areas within the Regional Study Area (Figure 2.1-1)

1.3





a

9012210167-0/





9012210167-02

0.0

.

-

#### DATA COVERAGE AND QUALITY (Density data)

Rock densities at the NTS and vicinity can be separated into three major groups: pre-Cenozoic sedimentary rocks and intrusive rocks with an average density of about 2.67 gm/cc, Cenozoic volcanic rocks with a density of about 2.4 gm/cc, and nonwelded and partially welded ash-flow tuffs and alluvium with a density of about 2.0 gm/cc. There are three primary sources of rock density information from the NTS and vicinity: rock samples (including core samples), borehole gravity meter surveys, and borehole density logs. Table 2.1-2 lists in stratigraphic sequence the geologic units in the Yucca Mountain area that are important for gravity interpretation. This table also gives a range of thickness and representative values for density of each unit, and provides a reference for geologic unit names used in the following discussion.

#### SUMMARY OF RESULTS

Many regional subsurface geologic structures at the NTS and vicinity were initially identified by the gravity method. A notable result was the prediction of the depth to pre-Cenozoic basement rocks at the location of drillhole UE25 p#1, about 5 km north of Busted Butte (Figure 2.1-2). Snyder and Carr (1982, p. 27) used gravity modeling to estimate the depth to Paleozoic basement to be about 3,500 ft at Busted Butte and about 4,750 ft at the nearby gravity saddle ("s" in Figure 2.1-2). On this basis, the depth to basement at p#1 was estimated to be 4,000 ft. Drilling revealed dolomitic basement at a depth of 4,080 ft (Carr et al., 1986, p. 17). A threedimensional gravity model suggests that pre-Cenozoic basement increases in depth westward directly under the conceptual repository location and reaches 10,000 ft under Crater Flat (Snyder and Carr, 1984). This basement model is apparently consistent with recent seismic refraction data, but has not been tested by drilling. Drillhole UE25 p#1 (Figure 2.1-2) is the only hole in the Yucca Mountain area to reach basement; the deepest drillholes in the Yucca Mountain-Crater Flat area other than p#1 are 6,000 ft deep and bottom. in Miocene volcanic rocks.

Gravity methods also helped to identify the Silent Canyon caldera underlying Pahute Mesa to the north of the site area, where gravity data indicate a volcanic section at least 16,000 ft thick (Healey, 1968, p. 153). Drilling to 13,686 ft within the gravity anomaly, and surface mapping (Byers et al., 1976) have since confirmed the gravity model. Similarly, Kane et al. (1981) interpreted gravity data from the Timber Mountain area and determined that (1) a broad gravity high over the southeast side of Timber Mountain is associated with exposed Miocene intrusive rocks and suggests that this part of the caldera is underlain by such rocks, an important conclusion if the Timber Mountain area were to be considered for radioactive waste storage; and (2) the Timber Mountain caldera truncates the southern edge of the older Silent Canyon caldera.

Another useful application of gravity methods is in measuring vertical movement of subsurface density layers associated with major earthquakes (Oliver et al., 1975). Initial absolute and high-precision measurements were recently made in the Yucca Mountain site area for this application and

		Approx.	Average Density
λge	Unit	(M)	(gm/cc)
Quaternary &	alluvium	0-300	1.6-2.0
Tertiary	basalt Mistar Mountain Must	0=200	2.9
TellTellà	limper Mountain luii	0-150	1 0
44.0 Mar	Rainier Mesa Mbr, undivided	0-100	4.9
	Paintbrush Tuff		
12.6 Ma*	Tiva Canyon Mbr	120	2.1
	Yucca Mountain Mbr	0=60	1.9
	Pah Canyon Mbr	0=70	1.9
13.1 Ma*	Topopah Spring Mbr	300	2.2
13.4 Ma*	Rhyolite lavas and tuff of Calico Hills	10-200	1.9
	Crater Flat Turi		
	FIOW Fass MDI	100	2.1
74.0 145	Bulling Mbr	150	4.1
	Tram Mor	300	2.25
	Knyodacite Lavas	0=200	4.35
	Lichic Kidge Tull	300	2.35
York Palazzaia	Ash ilow and bedded tull	300+	2.40
nace leteozoto	units	2000	2.62
Middle and early	various limestone,	4000	2.72
Paleozoic	dolomite, quartzite, & other sedimentary units		
Early Paleozoic & Frecambrian	quartzite, and other sedimentary units	3500+	2.65

Table 2.1-2. Density and Thickness of Selected Geologic Units in the Yucca Mountain Area (after Snyder and Carr, 1984)

\* Radiometric ages from Marvin et al. (1970).

calibration purposes (2umberge et al., 1988; Harris and Ponce, 1988), and future remeasurements should be able to detect changes of less than 5 cm in the absolute elevation of measurement points.

Gravity surveys are also useful for inexpensive study of tectonic structures, particularly those which offset the basement or cause variations in the depth to basement. However, the interpretation of gravity alone does not produce unique models of the subsurface. It is therefore important that gravity data be collected along traverses and analyzed with other gravity data measurements including seismic reflection and refraction, magnetic, and geoelectric data. It is also important to obtain density data from surface measurements, gamma-gamma logs, and gravity meter borehole measurements to constrain gravity models.

#### BIBLIOGRAPHY FOR SECTION 2.1

- Byers, F.M., Jr., W.J. Carr, P.P. Orkild, W.D. Quinlivan, and K.A. Sargent, 1976. <u>Volcanic Suites and Related Cauldrons of Timber Mountain-Oasis</u> <u>Valley Caldera Complex, Southern Nevada</u>, U.S. Geological Survey Professional Paper 919, 70 p.
- Carr, M.D., S.J. Waddell, G.S. Vick, J.M. Stock, S.A. Monsen, A.G. Harris, B.W. Cork, and F.M. Byers, 1986. Geology of Drillhole UE25p#1: A Test Hole into Pre-Tertiary Rocks Near Yucca Mountain, Southern Nevada, U.S. Geological Survey Open File Report 86-175, 87 p.
- Harris, R.N., and D.A. Ponce, 1988. <u>High-Precision Gravity Network to</u> <u>Monitor Temporal Variations in Gravity Across Yucca Mountain, Nevada</u>, U.S. Geological Survey Open File Report 88-243, 19 p. (in press).
- Harris, R.N., D.L. Healey, D.A. Fonce, and H.W. Oliver, \_\_\_\_. Principal facts for 15,355 Gravity Stations in the Nevada Test Site and Vicinity.
- Healey, D.L., 1968. "Application of Gravity Data to Geologic Problems at the Nevada Test Site," in Eckel, E.B. (ed.), Nevada Test Site, Geological Society of America Memoir 110, p. 147-156.
- Healey, D.L., 1970. "Calculated In Situ Bulk Densities from Subsurface Gravity Observations and Density Logs, Nevada Test Site and Hot Creek Valley, Nye County, Nevada," in <u>Geological Survey Research</u>, U.S. Geological Survey Professional Paper 700 B, p. 852-862.
- Healey, D.L., 1983. "Gravity Investigations," in <u>Geologic and Geophysical</u> <u>Investigations of Climax Stock Intrusive, Nevada</u>, U.S. Geological Survey Open File Report 83-377, 76 p.
- Healey, D.L., 1986. Borehole Gravity Meter Survey in Drillhole USW G-4, Yucca Mountain Area, Nye County, Nevada, U.S. Geological Survey Open File Report 86-205, 18 p.
- Healey, D.L., F.G. Clustom, and D.A. Glover, 1984. <u>Borehole Gravity Meter</u> <u>Surveys in Drillholes USW G-3, UE-25p#1, Yucca Mountain Area, Nevada</u>, U.S. Geological Survey Open File Report 84-672, 16 p.
- Healey, D.L., R.N. Harris, D.A. Ponce, and H.W. Oliver, 1988c. <u>Complete</u> <u>Bouquer Gravity Map of the Nevada Test Site and Vicinity</u>, U.S. Geological Survey Open File Report 87-506, scale 1:100,000.
- Healey, D.L., D.B. Snyder, R.R. Wahl, and F.E. Currey, 1981. "Bouguer Gravity Map of Nevada, Caliente Sheet," Nevada Bureau of Mines and Geology Map 70, scale 1:250,000.

- Healey, D.L., R.R. Wahl, and F.E. Currey, 1980a. "Bouguer Gravity Map of Nevada, Goldfield and Mariposa Sheets," Nevada Bureau of Mines and Geology Map 68, scale 1:250,000.
- Healey, D.L., R.R. Wahl, and H.W. Oliver, 1980b. "Bouguer Gravity Map of Nevada, Death Valley Sheet," Nevada Bureau of Mines and Geology Map 69, scale 1:250,000.
- Hildenbrand, T.G., A.M. Rogers, H.W. Oliver, S.C. Harmsen, J.K. Nakata, D.S. Aitken, R.N. Harris, and M.D. Carr, 1988. "Regional Geologic and Geophysical Maps of Southern Great Basin," in M.D. Carr and J.C. Yount (eds.), <u>Geologic and Hydrologic Investigations of a Potential Nuclear Waste</u> <u>Disposal Site at Yucca Mountain</u>, <u>Southern Nevada</u>, U.S. Geological Survey Bulletin 1790, p. 3-21.
- Izett, G.A., 1960. "Granite" Exploration Hole, Area 15, Nevada Test Site, Nye County, Nevada - Interim report, Part C. Physical Properties, U.S. Geological Survey Open File Report TEM-836-c, 36 p.
- Jansma, P.E., D.B. Snyder, and D.A. Ponce, 1982. Principal Facts of Gravity Stations with Gravity and Magnetic Profiles from the Southwest Nevada Test Site, Nye County, Nevada, as of January 1982, U.S. Geological Survey Open File Report 82-1041.
- Johnson, R.B., and I.R. Ege, 1964. <u>Geology of the Pluto Site, Area 401,</u> <u>Nevada Test Site, Nye County, Nevada</u>, U.S. Geological Survey Trace Elements Investigations Report TEI-841, 127 p

a

0 .

- Kane, M.F., D.L. Healey, D.L. Peterson, H.E. Kaufmann, and D. Reidy, 1979. "Bouguer Gravity Map of Nevada, Las Vegas Sheet," Nevada Bureau of Mines and Geology Map 61, scale 1:250,000.
- Kane, M.F., M.W. Webring, and B.K. Bhattacharyya, 1981. A Preliminary Analysis of Gravity and Aeromagnetic Surveys of the Timber Mountain Area, Southern Nevada, U.S. Geological Survey Open File Report 81-189, 40 p.
- Keller, G.V., 1959. "Porosity, Density, and Fluid Permeability of the Oak Spring Formation," in Diment, W.H. et al., <u>Properties of the Oak Spring</u> <u>Formation in Area 12 at the Nevada Test Site</u>, U.S. Geological Survey Trace Elements Investigations Report TEI-672.
- Marvin, R.F., F.M. Byers, Jr., H.H. Mehnert, P.P. Orkild, and T.W. Stern, 1970. "Radiometric Ages and Stratigraphic Sequence of Volcanic and Plutonic Rocks, Southern Nye and Western Lincoln Counties, Nevada," Geological Society of America Bulletin, vol. 81, p. 2657-2676.
- Morelli, C. (ed.), 1974. The International Gravity Standardization Net 1971, International Association Geodesy Special Publ., no. 4, 194 p.
- Oliver, H.W., R.H. Chapman, S. Biehler, A. Griscom, E.A. Silver, L.A. Beyer, and S.L. Robbins, 1980. "Gravity Map of California and Its Continental Margin," California Division of Mines and Geology Special Map, scale 1:750,000.

- Oliver, H.W., S.L. Robbins, R.B. Grannell, R.W. Alewine, and S. Biehler, 1975. "Surface and Subsurface Movements Determined by Remeasuring Gravity," in G.B. Oakshott (ed.), <u>San Fernando Earthquake of February 9</u>, 1971, California Division of Mines and Geology Bulletin 196, p. 195-211.
- Ponce, D.A., 1981. Preliminary Gravity Investigations of the Wahmonie Site, <u>Nevada Test Site, Nye County, Nevada</u>, U.S. Geological Survey Open File Report 81-522, 64 p.
- Ponce, D.A., R.N. Harris, and H.W. Oliver, 1988. Isostatic Gravity Map of the Nevada Test Site and Vicinity, Nevada, U.S. Geological Survey Open File Report 88-664, scale 1:100,000 (in press).
- Roberts, C.W., R.C. Jachens, and H.W. Oliver, 1981. <u>Isostatic Residual</u> <u>Cavity Map of California</u>, U.S. Geological Survey Open File Report 81-573, scale 1:750,000.
- Saltus, R.W., 1988a. "Bouguer Gravity Anomaly Map of Nevada," Nevada Bureau of Mines and Geology Map 94A, scale 1:750,000.
- Saltus, R.W., 1988b. "Regional Residual and Derivative Gravity Maps of Nevada," Nevada Bureau of Mines and Geology Map 94B, four at scale 1:1,000,000 and two at scale 1:2,000.000.
- Saltus, R.W., 1988c. Gravity Data for the State of Nevada on Magnetic Tape, U.S. Geological Survey Open File Report 88-433, 20 p.
- Snyder, D.B., and W.J. Corr, 1982. Preliminary Results of Gravity Investigations at Yucca Mountain and Vicinity, Southern Nye County, Nevada, U.S. Geological Survey Open File Report 82-701, 36 p.
- Snyder, D.B. and W.J. Carr. 1984. "Interpretation of Gravity Data in a Complex Volcano-Tectonic Setting, Southwestern Nevada," Journal of Geophysical Research, vol. 89, no. B12, p. 10,193-10,206.
- Snyder, D.B., C.W. Roberts, R.W. Saltus, and R.F. Sikora, 1981. <u>Magnetic</u> <u>Tape Containing the Principal Facts of 64,402 Gravity Stations in the State</u> <u>of California</u>, U.S. Geological Survey Report, 30 p.; available from National Technical Information Service, U.S. Department of Commerce, Springfield, VA 22161, NTIS PB-82168287.
- Zumberge, M.A., R.N. Harris, H.W. Oliver, G.S. Sasagawa and D.A. Ponce, 1988. Preliminary Results of Absolute and High-Precision Gravity Measurements at the Nevada Test Site and Vicinity, Nevada, U.S. Geological Survey Open File Report 88-242, 29 p.

 $\mathbf{\Omega}$ 



#### 2.2 MAGNETIC AND PALEOMAGNETIC INVESTIGATIONS

#### OBJECTIVES

Aeromagnetic, ground magnetic, paleomagnetic, and magnetic property measurements have been made intermittently at the NTS and vicinity in support of subsurface structural studies, Curie temperature isotherm analysis, correlation of volcanic strata, and studies of structural rotation about a vertical axis. These data are also needed to locate and estimate the volume of buried Quaternary basalts for determining the probability of future eruptions and to locate concealed faults which offset strata within Yucca Mountain (Oliver and Ponce, 1987; DOE, 1988).

#### DATA COVERAGE AND QUALITY

Figure 2.2-1 shows the existing coverage of aeromagnetic data within a radius of about 100 km of Yucca Mountain, and Figure 2.2-2 shows the outlines of the six aeromagnetic surveys listed in Table 2.2-1 which cover various parts of the proposed Yucca Mountain site area and vicinity. (The location of Figure 2.2-2 is shown on Figure 2.2-1 for reference). These index maps indicate the availability of a considerable amount of magnetic data and help delineate areas where more data are needed. For example, the present coverage of the site area and vicinity to a distance of about 15 km consists of draped profiles (flown at a constant elevation above terrain) with a spacing of 400 m (1/4 mi) and 800 m (1/2 mi) except for the area beginning only 5 km to the northwest of the site area for which only barometric Hata (flown at a constant elevation above sea level) are available (Figure 2.2-1). The draped data are generally 120 m (400 ft) above ground, whereas the barometric data were flown at a constant elevation of 2,440 m (8,000 ft) and are therefore much less detailed because the average elevation of terrain is about 4,000 ft. The Death Valley area southwest of the site area was surveyed along flightlines at 120 m (400 ft) above ground and spaced 1.6 km apart. This survey was intended for reconnaissance exploration of uranium under the NURE (National Uranium Resource Evaluation) program (Figure 2.2-2). The flightlines are too far apart for such low-level flights, and this coverage misses about a third of buried magnetic structures; thus, the NURE data are regarded as inadequate for structural studies.

Compilations of regional aeromagnetic data indexed in Figure 2.2-1 have been made at a scale of 1:2,500,000 by computer continuation of all data sets to a common surface of 3,800 m (12,500 ft) above sea level (Hildenbrand et al., 1988). A larger scale regional aeromagnetic map was also prepared by continuing these data either downward or upward to a level of 305 m (1,000 ft) above terrain at 1:750,000 (Hildenbrand and Kucks, 1988). Mosaics of original contract data are being assembled at 1:250,000 for 1° by 2° areas surrounding Yucca Mountain. Those to the east and south are complete (Saltus and Sryder, 1986; Saltus and Ponce, 1988), and those to the west and north are in progress.

Aeromagnetic compilations at 1:100,000-scale are being planned in order to be consistent with four new 1:1,000,000-scale 1/2° x 1° topographic maps

			NEV	ADA (see	e E	rwin et al	., 1980; and	Hill, 1986	i)		
A	rea name	Year flown	Contractor	Elev (ft)		Spacing (mi)	Direction	Scale	Gradient Removed	Digital	Reference
1	Topopah Spring	1961	USGS	8,000	b	1/2	E-W	1:62,500	no	no	Boynton & Vargo, 1963
2	Goldfield	1967	LKB	9,000	b b	1	E-W	1:250,000	no	no	USGS, 1971 <sup>b</sup>
2	Goldfield	1967	LKB	9,000	b	1	E-W E-W	1:62,500	no IGRF	a.	USGS, 1967 <sup>b</sup> USGS, 1979
3	Lathrop	1977	Aero	400	d	1/4-1/2	E-W N-S	1:62,500	IGRF	28	USGS, 1978
5	NURE (Death	1979	GEO-LIFE	400	d	1	N-S	1:500,000	IGRF	yes	DOE, 1979
6	Yucca Mtn.	1982	HLQEB	400	d	1/4	N-S	1:62,500	IGRF	n.a.	USGS, 1984

Table 2.2-1.	Aeromagnetic surveys all or p	artly within	the	Yucca	Mountain	Site	Area	and	vicinity	
	(refer to Figure 2.2-1) <sup>a</sup>									

<sup>a</sup> Key:	Aero	Aero Service
	AG	Applied Geophysics Inc.
	b	barometric
	d	drape
	Е	east
	HLQEB	High-Life QEB
	IGRF	International Geomagnetic Reference Field
	LKB	Lockwood, Kessler & Bartlett, Inc.
	N	north
	n.a.	not available
	NURE	National Uranium Resource Evaluation
	S	south
	USGS	United States Geological Survey

.

W west bUSGS (1971) and USGS (1967) were reports on the same survey at different compilation scales. 100



9.



ŵ

## YUCCA MOUNTAIN PROJECT

125.74

FIGURE 2.2-1. Aeromagnetic Index Map of Regional Study Area Showing Flight Line Spacing of Available Data.



#### LEGEND

823	1/4 & 1/2 MILE SPACING DRAPED
IIIII	1/2 MILE SPACING BAROMETRIC
	1 MILE SPACING DRAPED
	1 MILE SPACING BAROMETRIC
823	2 MILE SPACING BAROMETRIC
	1 & 3 MILE SPACING NURE (Entire map area covered)
833	SITE AREA SI
$\sim$	HIGHWAY APERTURE
	IMPROVED ROAD
N	RAILROAD
111	STATE BOUNDARY Also Available On
N	COUNTY BOUNDARY Aperture Card
	ADMINISTRATIVE BOUNDARY



CSEGEG YWP-89-068.1

9012210167-03




that surround Yucca Mountain. A 20-nT (nanoTesla) contour aeromagnetic map of the site area and vicinity (Figure 2.2-2) is available at 1:48,000 (Kane and Bracken, 1983). For a detailed index and listing of all aeromagnetic surveys in southern Nevada being incorporated into these maps, see Erwin et al. (1980) and Hill (1986).

# Ground Magnetic Data

Measurements of total-field magnetic intensity have been made along selected traverses at Yucca Mountain, Lathrop Wells (to the south), and Wahmonie/Calico Hills (to the east). Most of the measurements were made on foot with a ground magnetometer read at 3 to 30 m intervals depending on the horizontal gradient of the magnetic field. Some measurements at Wahmonie and Calico Hills were made with a truck-mounted flux-gate magnetometer (G. Bath, written communication, 1980).

The ground data at Yucca Mountain consist of five N-S profiles from one to several km in length. These data were used along with aeromagnetic data to test the feasibility of magnetic methods for locating concealed faults and possible intrusions (Bath and Jahren, 1984; Bath, 1985).

Ground measurements were made near Lathrop Wells in an area apparently underlain by reversely magnetized Quaternary basalts (Kane and Bracken, 1983; Crowe et al., 1986). The measurements consist of E-W and N-S lines through the southern anomalous low of the dipole anomaly at intervals of 1 to 10 m as needed to define the anomaly.

At Wahmonie and Calico Hills, N-S ground profiles were obtained to better delineate aeromagnetic highs thought to be associated with buried intrusions and magnetite-rich altered argillite, respectively (Ponce, 1984; Snyder and Oliver, 1981).

All these ground measurements were made with a Geometrics protonprecession magnetometer, which has a reading accuracy of 1 nT. Diurnal corrections were made using continuous magnetic measurements made at a local magnetic base station with Geometrics equipment.

### Paleomagnetic and Rock Magnetic Studies

There are three important objectives of the paleomagnetic and rock magnetic studies at Yucca Mountain: (1) to support structural geologic studies by providing paleomagnetic data bearing on vertical-axis rotation; (2) to aid studies of volcanic stratigraphy; and (3) to provide constraints for the modeling of airborne, surface, and borehole magnetic data. In addition, the direction of remanent magnetization of samples from drill core provides a means of obtaining azimuthal orientation of core segments from boreholes at Yucca Mountain (Rosenbaum and Rivers, 1985).

Laboratory measurements of magnetic properties (e.g., remanent magnetization and magnetic susceptibility) have been obtained on numerous samples from surface outcrops and from drill core. In the immediate vicinity of Yucca Mountain, a total of about 65 surface sites have been collected and analyzed from the Crater Flat, Paintbrush, and Timber Mountain tuffs. These sites lie within the area from 36°40'N to 36°55'N latitude, and from 116°22'W to 116°35'W longitude, roughly corresponding to the designated site area. Data also exist for about 70 sites in the area of Yucca Flat. Most of these sites are within the same geologic strata as those at Yucca Mountain, although some are in older units. In addition, about 25 other paleomagnetic sites from a variety of geologic units exist within the NTS. The directional and susceptibility data from these sites are of high quality. Samples were collected at roughly 3 m intervals throughout drillholes USW G-1, USW G-2, USW G-3, and USW GU-3, and at somewhat greater intervals from holes VH-1 and VH-2. These samples were oriented with respect to the drill core axis. More closely spaced fully-oriented samples were collected from oriented core runs in these and other holes (e.g., USW H-1, UE25a#4, a#5, a#6, and a#7).

Directions of remanent magnetization from 30 surface sites show that the region between the north and south ends of Yucca Mountain have undergone about 30° of vertical-axis rotation since emplacement of the Tiva Canyon Member. Although rotations about a horizontal axis (tilts) are easily recognized visually in many layered rocks, comparison of declinations of remanent magnetization is often the only technique that can provide a measure of rotation about a vertical axis. On a regional scale, relative vertical-axis rotations between sites may be obtained by comparison of paleomagnetic directions from individual volcanic flows or cooling units. Such studies require that (1) individual volcanic units can be unequivocally identified, (2) structural attitudes can be accurately determined, and (3) the volcanic units are reliable paleomagnetic recorders. Geologic mapping and studies of volcanic stratigraphy in the Yucca Mountain area have helped satisfy the first two requirements. Detailed paleomagnetic studies of several ash-flow sheets, using samples from boreholes at Yucca Mountain, have demonstrated that the Tiva Canyon Member of the Paintbrush Tuff, and the Prow Pass and Bullfrog Members of the Crater Flat Tuff are reliable paleomagnetic recorders, but that the Topopah Spring Member of the Paintbrush Tuff is not (Rosenbaum and Snyder, 1985; Rosenbaum, 1986). Directions of remanent magnetization from numerous surface sites in the three paleomagnetically reliable ash-flow tuff sheets demonstrate about 30° of vertical-axis rotation between the north and south ends of Yucca Mountain since emplacement of the Tiva Canyon Member (Scott and Rosenbaum, 1986; Rosenbaum and Hudson, 1988).

Borehole magnetic field and magnetic susceptibility logs have been used at the NTS to aid correlations among volcanic strata penetrated by drillholes. Interpretation of these logs has been largely empirical. At Yucca Mountain such logs have been obtained from several holes, and for the first time these logs can be compared to laboratory rock magnetic data from drill core samples. Laboratory data demonstrate that remanent magnetization is generally much larger than induced magnetization (Koenigsberger ratio, Q>5) so that the character of magnetic field logs is determined by variations in remanent magnetization. Therefore, contacts between ash-flow sheets of opposite magnetic polarity, such as the reversely magnetized Tram Member and the normally magnetized Bullfrog Member, can be easily recognized from the magnetic field logs. Both the log data and the laboratory data show high-amplitude (often order of magnitude) systematic variations in remanent intensity and susceptibility. These variations occur within the interiors of individual ash-flow sheets as well as at the margins. These variations provide a means of using borehole data to map zones within the ash-flow sheets throughout the repository area, and to thereby contribute directly to

knowledge about the lateral continuity of the various beds and of structures offsetting those beds.

Studies of magnetic properties have concentrated on the relationship among the observed systematic variations in remanent magnetization and magnetic susceptibility and the depositional and cooling histories of the various ash-flow sheets (Hagstrum et al., 1980; Rosenbaum and Snyder, 1985; Rosenbaum and Spengler, 1986). Bath (1968) noted a correlation between the degree of welding and values of both remanent magnetization and susceptibility. Results from Yucca Mountain confirm that nonwelded to poorly welded tuffs are generally characterized by low susceptibilities and magnetizations. However, large variations in both remanent magnetization and susceptibility occur within the thick densely-welded parts of individual cooling units penetrated by boreholes at Yucca Mountain. Studies of rock magnetic properties and Fe-oxide microcrystals have demonstrated that increases in susceptibility and remanent magnetization away from cooling breaks are produced by the growth of Feroxide microcrystals away from the quenched margins of the ash-flow sheets (Rosenbaum and Schlinger, 1987; Schlinger et al., 1988). In addition, there is a high degree of correlation between the location of depositional breaks within the thick cooling units (which may or may not correlate with changes in the degree of welding) and susceptibility minima (Rosenbaum and Spengler, 1986). Understanding the sources of variations in remanent magnetization and susceptibility will provide the basis for improved interpretation of borehole magnetic field and susceptibility logs, and thereby for mapping of tuff layers at Yucca Mountain. Knowledge of the lateral continuity of these layers is essential to determining the structural integrity and hydrologic character of the repository block.

Magnetic property data are also useful for the interpretation of ground magnetic and aeromagnetic data (Hagstrum et al., 1980; Rosenbaum and Snyder, 1985). Four areally extensive ash-flow sheets possess moderate to strong magnetizations throughout large stratigraphic thicknesses, and are thus considered likely sources of aeromagnetic anomalies. Remanent magnetization is more important than induced magnetization (Q>5); therefore, anomalies reflect magnetic polarity of the ash-flow sheets. The Tiva Canyon and Pah Canyon Members of the Paintbrush Tuff and the Tram Member of the Crater Flat Tuff are reversed, and the Topopah Spring Member of the Paintbrush Tuff and the Bullfrog Member of the Crater Flat Tuff are normal (Table 2.2-2) (Rosenbaum and Snyder, 1985).

Paleomagnetic studies on surface outcrops of the young basaltic rocks of the Crater Flat area have also begun. These studies, combined with K-Ar and Ar40/Ar39 dating of basalts, are intended to evaluate the episodic nature of late Tertiary volcanic activity in the vicinity of Yucca Mountain. Paleomagnetic measurements and interpretation will improve the resolution of separate volcanic events in the geologic record.

#### SUMMARY OF RESULTS

One of the most important results is the discovery of a number of magnetic anomalies over alluvial areas in Crater Flat and the Amargosa Desert

K-Ar Age (Ma)	e Rock I Unit I	Magnetic Polarity	J <sub>NRM</sub> (Am <sup>-1</sup> )	Suscept. (10 <sup>-3</sup> SIV)	Q	J <sub>Total</sub> (Am <sup>-1</sup> )	Thickness (m)
10.2	Basalt dikes						
	Timber Mtn Tuff Rainier Mesa Mbr bedded tuff						0-46 0-61
12.5	Paintbrush Tuff Tiva Canyon Mbr	R	0.4-10	2-11	2-38	0.3-10	69-148
	bedded tuff Yucca Mtn Mbr bedded tuff	R	0.1-0.4	5-6	1-2	0.1-0.2	1-15 0-29 0-47
	Pah Canyon Mbr	R	1.8-3.3	4-6	9-17	1.6-3.1	0-71
13.1	Topopah Spring M bedded tuff	or N	0.2-1.3	1~5	2-10	0.2-1.4	287-359 1-17
13.4	Tuffaceous heds Calico Hi is bedded tuff	of	0.1	1-2	2	0.1-0.2	27-289 0-21
	Crater Flat Tuff						
	Prow Pass Mbr bedded tuff	N	0.1-0.5	1-3	2-20	0.1-0.6	80-193 2-10
13.5	Bullfrog Mbr bedded tuff	N	0.1-2.8	1-8	5-16	0.2-5.5	68-187
	Tram Member bedded tuff	R	0.1-2.2	1-5	3-13	0.1+2.0	190-369 3-50
	Dacite lava and fl breccia bedded tuff	ow N	0.1-3.1	3-30	0.3-9	0.2-3.8	0-249 0-14
	Lithic Ridge Tuff bedded tuff	I	0.2-0.3	3-5	1	0.1-0.3	185-304 3-7
13.9	Older volcanic roc	ks -	0.1-1.3	1-43	1-27	0.1-3.0	345+

Table 2.2-2. Magnetic properties of volcanic stratigraphy at the site\*

<sup>a</sup>Age is in millions of years before present. Polarity symbols: N = normal, R = reversed, I = intermediate.  $J_{\rm NRM}$  is the range of intensity of natural magnetization, Suscept. is the range of SI susceptibility, Q is the range of the Koenigsberger ratio of  $J_{\rm NRM}$  to induced magnetization,  $J_{\rm Total}$  is the range of intensity of total magnetization using the direction of remanent magnetization. Thickness is the range in meters of rock units from surface or drill core from the Yucca Mountain area. Mbr means Member. The table was compiled from Table 3 of USGS (1984b) and Tables 2, 5, 6, 8, 9, 11, 13, 15, 16, 18, and 20 of Rosenbaum and Snyder (1985). just west and south of Yucca Mountain. These magnetic anomalies are very similar to those associated with exposed Quaternary cinder cones and basalt flows. If this correlation is confirmed by planned drilling, the magnetic data could be used in combination with other methods and drilling to estimate the total volume of buried volcanic rocks that must be considered in evaluating the probability for future eruptions that could affect repository performance (Kane and Bracken, 1983; Crowe et al., 1986).

Another significant finding is the presence of an east-west trending magnetic high over northern Yucca Mountain, extending eastward across Calico Hills (Boynton et al., 1963a). This magnetic high is associated with magnetite-bearing, thermally altered argillaceous basement rocks of the Eleana Formation (USGS, 1984b, Figure 23; Bath and Jahren, 1984). The Eleana Formation may be altered in this location because it is at the margin of the Timber Mountain caldera (Kane et al., 1981). The magnetic high is similar in areal size and direction to the anomaly known to be associated with Mesozoic intrusive rocks at the north edge of Yucca Valley (Bath et al., 1983). The possible relation of this feature to the large hydraulic gradient area is discussed in Section 3.1.2.

Elongate magnetic highs and associated lows are associated with mapped faults in the site area, some of which are only partially exposed (Kane and Bracken, 1983). The extent and continuity of such faults with depth are important concerns for site characterization, and it appears that high-resolution magnetic surveys combined with other geophysical methods may provide useful resolution of these features.

Analysis of the statistical properties of magnetic anomalies in the Yucca Mountain region has produced estimates of the depth within the earth's crust to the Curie-temperature isotherm - about 580°C depending chiefly on the amount of titanium present in magnetic minerals. Results from Curie isotherm analysis indicate that the isotherm depth in the vicinity of Yucca Mountain is about 15 km, and that the depth increases northward to a value of about 30 km coinciding with an anomalously low heat flow near 38°N, 116°W east of Tonopah, Nevada (Blakely, 1988, Figure 10). This interpretation conflicts with the conclusion of Sass et al. (1971) that low heat flow in this area is caused by near-surface hydrologic phenomena.

The southern tip of Yucca Mountain has undergone about 30° of clockwise vertical-axis rotation since emplacement of the Tiva Canyon Member of the Paintbrush Tuff. This conclusion is based on declinations of remanent magnetization from 32 sites in the reversely magnetized Tiva Canyon Member that display a systematic southward increase over the 25 km north-south extent of Yucca Mountain. The observed rotations could reflect either oroflexure above a deep-seated right-lateral shear zone or shear related to differential extension within hanging-wall rocks of a regional detachment system.

Borehole magnetic field and magnetic susceptibility logs as well as laboratory magnetic property measurements demonstrate the existence of high-amplitude, systematic variations in both remanent magnetization and susceptibility through thick volcanic sections penetrated by boreholes at Yucca Mountain. The magnetic property variations are intimately related to the depositional and cooling histories of the various ash-flow sheets. These results provide a basis for the future use of borehole magnetic logs to map volcanic strata throughout the repository area.

BIBLIOGRAPHY FOR SECTION 2.2

- Bath, G.D., 1968. <u>Aeromagnetic Anomalies Related to Remanent Magnetization</u> <u>in Volcanic Rock, Nevada Test Site</u>, Geological Society of America Memoir 110, p. 135-146.
- Bath, G.D., and C.E. Jahren, 1984. Interpretations of Magnetic Anomalies at a Potential Repository Site Located in the Yucca Mountain Area, Nevada Test Site, U.S. Geological Survey Open File Report 84-120, 40 p.
- Bath, G.D., 1985. Investigation of an Aeromagnetic Anomaly on the West Side of Yucca Mountain, Nye County, Nevada, U.S. Geological Survey Open File Report 85-459, 24 p.
- Bath, G.D., C.E. Jahren, J.G. Rosenbaum, and M.J. Baldwin, 1983. "Magnetic Investigations," p. 40-77 in <u>Geologic and Geophysical Investigations of</u> <u>Climax Stock Intrusive, Nevada</u>, U.S. Geological Survey Open File Report 83-377, 82 p.
- Blakely, R.J., 1988. "Curie Temperature Isotherm Analysis and Tectonic Implications of Aeromagnetic Data from Nevada," <u>Journal of Geophysical</u> Research, vol. 93, no. B10, pp. 11,817-11,832.
- Boynton, G.R., J.L. Meuschke, and J.L. Vargo, 1963a. "Aeromagnetic Map of the Timber Mountain Quadrangle and Part of the Silent Canyon Quadrangle, Nye County, Nevada," U.S. Geological Survey Geophysical Investigations Map GP-443, scale 1:62,500.
- Boynton, G.R., and J.L. Vargo, 1963. "Aeromagnetic Map of the Topopah Spring Quadrangle and Part of the Bare Mountain Quadrangle, Nye County, Nevada," U.S. Geological Survey Map GP-440, scale 1:62,500.
- Carr, M.D., and J.C. Yount (eds), 1988. Geologic and Hydrologic Investigations of a Potential Nuclear Waste Disposal Site at Yucca Mountain, Southern Nevada, U.S. Geological Survey Bulletin 1790, 152 p. plus 2 plates, scale 1:2,500,000.
- Crowe, B.M., K.H. Wohletz, D.H. Vaniman, E. Gladney, and N. Bower, 1986. <u>Status of Volcanic Hazard Studies for the Nevada Nuclear Waste Storage</u> Investigations, Los Alamos National Laboratory Report LA-9325-MS, vol. 2.
- DOE (U.S. Department of Energy), 1979. Airborne Gamma-Ray Spectrometer and Magnetometer Survey, Death Valley Quadrangle, California, Nevada, U.S.

Department of Energy Open File Report GJBX=164 (79), vol. 2, variously paged.

- DOE (U.S. Department of Energy), 1988. Site Characterization Plan, Yucca <u>Mountain Site, Nevada Research and Development Area, Nevada</u>, U.S. Department of Energy Report RW-0199, 9 volumes.
- Erwin, J.W., S.L. Nichols, R.H. Godson, and P.L. Hill, 1980. "Aeromagnetic Map Index of Nevada," Nevada Bureau of Mines and Geology Map 62, scale 1:1,000,000.
- Hagstrum, J.T., J.J. Daniels, and J.H. Scott, 1980. <u>Analysis of the Magnetic</u> <u>Susceptibility Well Log in Drillhole UE25a-5</u>, Yucca Mountain, Nevada Test <u>Site</u>, U.S. Geological Survey Open File Report 80 1263, <u>33</u> p.
- Hildenbrand, T.G., and R.P. Kucks, 1988. "Total Intensity Magnetic Anomaly Map of Nevada," Nevada Bureau of Mines and Geology Map 93A, scale 1:750,000.
- Hildenbrand, T.G., A.M. Rogers, H.W. Oliver, S.C. Harmsen, J.K. Nakata, and M.D. Carr, 1988. "Regional Geological and Geophysical Maps of the Southern Great Basin," p. 3-21 in Carr, M.D., and J.C. Yount, <u>Geologic and Hydrologic Investigations of a Potential Nuclear Waste Disposal Site at Yucca Mountain, Southern Nevada</u>, U.S. Geological Survey Bulletin 1790, 152 p. plus 2 clear film plates.
- Hill, P.L., 1986. <u>Bibliography, Index, and Location Maps of Airborne Surveys</u> <u>In and West of the Rocky Mountain States</u>, U.S. Geological Survey Den File Report 86-525A, 200 p.
- Kane, M.F., and R.E. Bracken, 1983. "Aeromagnetic Map of Yucca Mountain and Surrounding Regions, Southwest Nevada: U.S. Geological Survey Open File Report 83-616, scale 1:48,000.
- Kane, M.F., M.W. Webring, and B.K. Bhattacharyya, 1981. A Preliminary Analysis of Gravity and Aeromagnetic Surveys of the Timber Mountain Area, Southern Nevada: U.S. Geological Survey Open File Report 81-189, 40 p. plus 6 illustrations.
- Oliver, H.W., and D.A. Ponce, 1987. U.S. Geological Survey Scientific Investigation Plan - Geophysical Investigations, Gravity and Magnetic Methods, U.S. Geological Survey Administrative Report 3221G-01R0, 27 p. plus 5 illus.
- Ponce, D.A., 1984. "Gravity and Magnetic Evidence for a Granitic Intrusion Near Wahmonie Site, Nevada Test Site, Nevada," <u>Journal of Geophysical</u> Research, vol. 89, no. B11, p. 9401-9413.
- Rosenbaum, J.G., 1986. "Paleomagnetic Directional Dispersion Produced by Plastic Deformation in a Thick Miocene Welded Tuff, Southern Nevada: Implications for Welding Temperatures," Journal of Geophysical Research, vol. 91, no. B12, p. 12,817-12,834.

- Rosenbaum, J.G., and M.R. Hudson, 1988. "Paleomagnetic Investigation of Tertiary Rotations, Southwest Nevada-Yucca Mountain" (abs), EOS, vol. 69, no. 44, p. 1164.
- Rosenbaum J.G., and W.C. Rivers, 1985. <u>Paleomagnetic Orientation of Core</u> from Drillhole USW GU-3, Yucca Mountain, Nevada: Tiva Canyon Member of the Paintbrush Tuff, U.S. Geological Survey Open File Report 85-48, 116 p.
- Rosenbaum, J.G., and C.M. Schlinger, 1987. "Magnetic Property Variations and the Origin of PaleCmagnetic Carriers in an Ash Flow Sheet at Yucca Mountain, Nevada" (abs), EOS (Transactions of the American Geophysical Union), v. 68, p. 1250.
- Rosenbaum. J.G., and D.B. Snyder, 1985. Preliminary Interpretation of Paleomagnetic and Magnetic Property Data from Drillholes USW G-1, G-2, GU-3, G-3, VH-1, and Surface Localities in the Vicinity of Yucca Mountain, Nye County, Nevada, U.S. Geological Survey Open File Report 85-49, 73 p.
- Rosenbaum, J.G., and R.W. Spengler, 1986. "Variation in Magnetic Properties in Thick Section of the Crater Flat Tuff, Southern Nevada" (abs.), EOS, v. 67, p. 924.
- Saltus, R.W., and D.A. Ponce, 1988. "Aeromagnetic Map of Nevada, Las Vegas Sheet," Nevada Bureau of Mines and Geology, Map 95, scale 1:250,000, includes Merged Aeromagnetic Map, scale 1:1,000,000.
- Saltus, R.W., and D.B. Snyder, 1986. "Aeromagnetic Map of Nevada, Caliente Sheet," Nevada Bureau of Mines and Geology Map 89, scale 1:250,000, includes Merged Aeromagnetic Map, scale 1:1,000,000.
- Sass, J.H., A.H. Lachenbruch, R.J. Monroe, G.W. Greene, and T.H. Moses, Jr., 1971. "Heat Flow in the Western United States," Journal of Geophysical Research, vol. 76, p. 6376-6413.
- Schlinger, C.M., J.G. Rosenbaum, and D.R. Veblen, 1988. "FE-Oxide Microcrystals in Welded Tuff from Southern Nevada: Origin of Remanence Carriers by Precipitation in Volcanic Glass," Geology, vol. 16, p. 556-559.
- Scott, R.B., and J.G. Rosenbaum, 1986. "Evidence of Rotation about a Vertical Axis during Extension at Yucca Mountain, Southern Nevada" (abs.)," EOS, vol. 67, no. 16, p. 358.
- Snyder, D.B., and H.W. Oliver, 1981. Preliminary Results of Gravity Investigations of the Calico Hills, Nevada Test Site, Nye County, Nevada, U.S. Geological Survey Open File Report 81-101, 77 p.
- USGS (U.S. Geological Survey), 1967. <u>Aeromagnetic Map of Goldfield and</u> <u>Cactus Peak, Nevada</u>, U.S. Geological Survey Open File Report 67-239, scale 1:62.500.
- USGS (U.S. Geological Survey), 1971. "Aeromagnetic Map of Parts of the Goldfield, Mariposa, and Death Valley 1° BY 2° Quadrangles, Nevada-California," U.S. Geological Survey Geophysical Investigations Map GP-753, scale 1:250,000.

- USGS (U.S. Geological Survey), 1978. Aeromagnetic Map of the Lathrop Wells Area, Nevada, U.S. Geological Survey Open File Report 78-1103, scale 1:62,500.
- USGS (U.S. Geological Survey), 1978. Aeromagnetic Map of the Timber Mountain Area, Nevada, U.S. Geological Survey Open File Report 79-587, scale 1:62,500.
- USGS (U.S. Geological Survey), 1984a. <u>Aeromagnetic Map of the Yucca Mountain</u> <u>Area, Nevada</u>, U.S. Geological Survey Open File Report 84-206, scale 1:62,500.
- USGS (U.S. Geological Survey), 1984b. A Summary of Geologic Studies through January 1, 1983, of a Potential High-Level Radioactive Waste Repository Site at Yucca Mountain, Southern Nye County, Nevada, U.S. Geological Survey Open File Report 84-792, 103 p.



# 2.3 GEOELECTRIC SURVEYS

### OBJECTIVES

The objectives of geoelectric work are to support interpretations on the existence, location, and geometry of (1) faults and altered zones, (2) magma chambers and other thermally-related deep crustal rock units, (3) economic sulfide mineralization, (4) thickness of alluvium, and (5) aquitards. Electrical resistivity and electrical polarization data are acquired with ground-based surveys. Geological interpretations are derived from geoelectric models constrained by known petrophysical relationships and borehole logs.

## DATA COVERAGE AND QUALITY

Rock resistivity is affected by fluid content and mineral composition. Minerals may be conductive (e.g., most economic minerals, but few common ones), or they may render pore fluids conductive by ionic exchange. Increased porosity, rock alteration, and elevated temperature are major factors associated with decreased resistivity. Fault zones and altered rocks are geoelectric targets because they provide increased porosity for conductive fluids, and mineralogy with high ionic-exchange capacity. Compact igneous intrusions hosted by fractured or altered rock, or by sedimentary rock, are geoelectric targets because they are more resistive than the surroundings. Structural or stratigraphic features that are enriched in clay-minerals are geoelectric targets, as are melted rocks, thermal anomalies in the crust, or zones of sulfide mineralization.

One of the earlier geoelectric surveys of the Yucca Mountain area involved magnetotelluric (MT) soundings, some of which are near seismic monitoring stations of the Southern Great Basin Seismic Network (Furgerson, 1982). MT soundings provide a means of detecting and delineating resistivity contrasts corresponding to major features such as the depth and fabric of the crystalline basement, or deep structural and lithologic contacts. Structural fabric and lithologic changes are inferred from MT resistivity structure, analogous to inferences from teleseismic velocity structure. MT data interpretation can thus reduce nonuniqueness of the resulting geophysical models of earth structure. The principal objective of crustal MT sounding and profiling is characterization of deep structure of the Yucca Mountain area, using the unique aspects of the method to test geologic structural models. For shallower exploration objectives, particularly where laterally heterogeneous resistivity structure is known to exist near the surface, audio-frequency MT and telluric profiling have been applied.

Resistivity (and IP) surveys have been performed at the site area using the Schlumberger sounding method, the dipole-dipole section-profiling method, and various controlled-source electromagnetic methods. Principal objectives for these surveys were (1) fault detection and delineation such as investigation of the inferred faults in Midway Valley and Drill Hole Wash, and (2) sounding the thickness of alluvium, the distribution of conductive lithologic units in the volcanic section, and the depth to the Paleozoic basement. Profiles were modeled and interpreted in two dimensions.

The extent of geoelectric surveys in the vicinity of Yucca Mountain is indicated on Figure 2.3-1. Deep MT soundings are widely spaced and are shown pointwise. Geoelectrical methods with shallow depth of penetration were carried out at closer spacings within smaller areas; the general locations of these surveys are shown by rectangular patterned areas in Figure 2.3-1.

Geoelectric surveys in areas A and B of Figure 2.3-1 were used to screen Candidate repository sites prior to the selection of Yucca Mountain (Hoover et al., 1982a; Hoover et al., 1982b). A variety of electrical surveys obtained in the immediate Project area (area C of Figure 2.3-1) are summarized on Figure 2.3-2 and explained and referenced in Table 2.3-1.

Magnetotelluric and telluric surveys in area D (Figure 2.3-1) consist of six audio-frequency MT (4 to 10,000 Hz) soundings at 3-km spacings, eight telluric soundings (J-ratio analysis) at 5- to 10-km spacing, and a single telluric-ratio line. Data in area E consists of 52 Schlumberger soundings, and data in area F consists of 136 Schlumberger soundings.

The 1982 MT data (coverage shown in Figure 2.3-1) are of high quality. analysis of the 1985 MT data is incomplete; modern interpretive techniques (e.g., tensor impedance modeling) are being used to quantify data variability and applicability to site characterization. The 1966-1979 MT data exhibit substantial scatter. Most of the remaining geoelectric data appear to support reliable interpretation, being relatively smooth over the domain of measurement, and producing estimates of geoelectric structure that are consistent with reasonable geologic models. These data can be further evaluated by cross-comparisons between different data, and by comparison with drillhole logs.

#### SUMMARY OF RESULTS

Deep MT data have identified a regional, crustal anisotropy aligned with the "Walker Lane," as well as a mid-crustal conductor at approximately 20 km depth or less. The anisotropy has been deduced from the azimuthal orientation of multicomponent, tensorial conductivity measurements using the MT method at distributed stations. Published interpretation (Furgerson, 1982) and unpublished data (Klein, personal communication, 1989) indicate that the magnetotelluric strike is northwest, or west-northwest for most stations. The midcrustal low resistivity layer implies the effect of deep pore fluids and/or enhanced temperatures, and may indicate a transition to crustal ductility. A report that describes and interprets deep MT data acquired more recently from the Yucca Mountain region (from 1982 to 1985) is currently in preparation.

MT data also indicate variations in the resistivity of the upper crust (2 to 5 km depth) that provide information on the distribution of sedimentary and volcanic rocks, with implications for the distribution of deeppenetrating, high-angle faults as well as low-angle faults. Indications on





# YUCCA MOUNTAIN PROJECT

FIGURE 2.3-1. Location of Geoelectric Surveys in the Yucca Mountain Region. Including the Nevada Test Site, Southern Nevada.



9012210167-05







SEGES YMPB9044 2

S. S. H. D. MAN

012210167 -M

Table 2.3-1. Geoelectric Investigations on Yucca Mountain

A. Investigations with depth control provided by multi-frequency or variable-spacing source-receiver configurations. (The locations of lines E1-E13 are shown on Fig. 2.3-2.)

METHOD	APPROXIMATE DEPTH RANGE	LOCATION	REMARKS	REFERENCE
Schlumberger soundings (max. electrode separation 1,200 m)	1 - 600 m	14 soundings variably spaced along Lines El, E2 of Figure 2.3-2; 14 km of profile	1-D modeling contoured to show resistivity contrasts related to faults and horizontal con- trasts indicating variation in pore-water lithology with depth	Senterfit et al., 1982
Dipole-dipole resistivity/IP (61-m dipoles)	60 - 180 m	Lines E3, E4, E5 of Figure 2.3-2; 4.4 km of profile	2-D models incorporating topography show resistivity contrasts related to faults and lithologic variation	Smith and Ross, 1982
Dipole-dipole resistivity/IP (152-m dipoles)	150 - 450 m	Lines E6, E7, E8 of Figure 2.3-2; 8.8 km of profile	2-D models as above	Smith and Ross, 1982
Dipole-dipole resistivity/IP (305-m dipoles)	300 - 930 m	Lines E9, E11, E12 of Figure 2.3-2; 16.4 km of profile	2-D models as above except models for lines Ell and El2 do not incorporate topography	Smith and Ross, 1982; Ross and Lunbeck, 1978
EM soundings (time-domain, central receiver)	200 - 1,200 m	Lines E10, E13 of Figure 2.3-2 (250-m spacings); 3.9 km of profile	1-D modeling composited to show fault-controlled lateral resistivity contrasts as well as lithology and pore-water controlled borizontal contrasts	Frischknecht and Raab, 1984

# Table 2.3-1. Geoelectric Investigations on Yucca Mountain (continued)

METHOD	APPROXIMATE DEPTH RANGE	LOCATION	REMARKS	REFERENCE
Borehole-to- Surface DC resistivity 1981	30 - 300 m	an area of 300 m x 500 m just NW of drillhole UE25b-1H; sources in drillholes UE25 a#4, a#5, a#6; dipole receivers on 25 - 50 m grid	apparent resistivity contoured; 3-D ellipsoidal body modeled	Daniels and Scott, 1981
Audiomagneto- telluric soundings (6-250 Hz)	200 - 2,000 m	4 soundings ( <b>#</b> 's 100 - 400) along 1-km profile	1-D and 2-D modeling shows general consistency with time-domain interpretation above	Unpublished data, USGS, 1981
Magneto- telluric soundings (.001-10 Hz)	2 - 20 km	soundings roughly along lines E1 and E2 of Figure 2.3-2	1-D modeling shows variations in conductance of upper crust related to variations in thickness and lithology of volcanic and sedimentary rocks; also a mid-crustal low resistivity layer, apparently related to crustal fluids and enhanced temperature.	Furgerson, 1982, and unpublished contract report, USGS, 1979

# Table 2.3-1. Geoelectric Investigations on Yucca Mountain (continued)

B. Profile/mapping investigations with equivocal depth control. (Lines are not shown on Fig. 2.3-2.)

	METHOD	APPROXIMATE DEPTH RANGE	LOCATION	REMARKS	REFERENCE
	"Slingram" (dual-loop EM profiling; 222- 3555 Hz, fixed separation)	8-150 m	16.6 km of profile	lateral apparent resistivity contrasts indicate variations in alluvial thickness and bedrock faults	Flanigan, 1981
	VLF (magnetic variation of fields related to 18.6 kHz navigation beacon)	10-60 m	4 km profile, parallel to Slingram profile (see above) over alluvium	results ambiguous due to insufficient penetration in conductive overburden	Flanigan, 1981
41	TURAM (magnetic variation of fields from EM line-source; 900 m source)	20-200 m	1.1 km of profile; over alluvium-covered fault detected by Slingram profile	results were ambiguous due to insufficient penetration or signal resolut:	Flanigan, 1981
	MAGNETOMETRIC (magnetic variation of field from 1 Hz line-source)	200-2,000 m	2 lines; 4.2 km of profile over alluvium-covered fault	identifica apparent resistivity contrast of fault	Fitterman, 1982
	Telluric-ratio (.02505 Hz, 500-m dipoles)	1 to 10 km	2 lines; 30 km of profile across northern part of Fortymile Wash	detected apparent resistivity variations related to bedrock faults	Hoover et al., 1982c

how such structural features can be detected are provided by Hoover et al. (1982a,b).

Shallower penetrating geoelectric methods applied at various locations throughout the Yucca Mountain region have indicated faults, intrusive rocks, zones of altered rocks, and the thickness of alluvial cover. The reports of Hoover at al. (1982a, b), although pertaining to Syncline Ridge and the Wahmonie/Calico Hills areas and not Yucca Mountain, indicate that shallower penetrating methods have appropriate uses in site characterization. The 136 Schlumberger resistivity soundings acquired at distributed locations throughout the Amargosa Desert, as indicated in Figure 2.3-3 (Greenhaus and Zablocki, 1982), have been used for mapping the depth to the Paleozoic basement (Czarnecki and Oatfield, 1987). Aeromagnetic data were used in conjunction with the Schlumberger data to establish the extent of Tertiary volcanics intervening between alluvium and the Paleozcic section. Apparent depths to the Paleozoic contact vary from about 200 m to as much as 1,500 m; deeper values are aligned in apparent fault-controlled basin structures in the northern part of the Amargosa Desert. In addition to depth-to-basement interpretation, inferred differences in resistivity of the upper 75 m of alluvium were used in conjunction with driller's logs to identify areas with coarser and probably more transmissive sediments.

The numerous survey lines on the northeast flank of Yucca Mountain were situated mostly on alluvial surface cover of variable thickness. Some lines extended on to tuff units, mostly of the Paintbrush Tuff (Lines E1 and E13 in Figure 2.3-2). Most of these data were in the vicinity of Drill Hole Wash northeast of the repository site. The results were interpreted to indicate various mapped and unmapped faults as lateral contrasts associated with zones of low resistivity. Most of the data sets (Figure 2.3-2) were interpreted independently from one another for purposes of fault detection. Interpretations were typically based on characteristic responses to conductive zones, or on models derived from one-dimensional (layered earth) algorithms, the composited results of which provided guasi-2-D sections. A notable exception is the dipole-dipole data set, a portion of which was modeled with a 2-D algorithm that incorporated topography as well as subsurface bodies.

In overview (Klein, personal communication, July 1989), the available geoelectric data suggest structural complexity in the vicinity of Drill Hole Wash, but uniformly high resistivity in the repository block. However, the data sets have not been incorporated into a consistent interpretive synthesis using the best available geologic control, as was done for the area of Syncline Ridge and the Wahmonie/Calico Hills areas (Hoover et al., 1982a,b). This task is highly desirable for four reasons: (1) incorporation of the current geologic control, (2) delineation of structures and lithological units that are reliably defined among the various data sets, (3) elimination of the inconsistencies among the various topical interpretations, and (4) evaluation of the merits of the various geoelectrical approaches.

The 2-D models of the dipole-dipole profiles (Smith and Poss, 1982) show an unrealistic complexity that impedes a geologic interpretation. Studies into part of these data near Coyote Wash (D. Klein, USGS, and E. Hardin, SAIC/Las Vegas, unpublished work, July 1989) indicate that the lack of incorporating geologic control and the lack of accounting for the effects of alluvial contacts may have contributed to the complexity of the models.



atore Rosses



# YUCCA MOUNTAIN PROJECT

FIGURE 2.3-3. Location of Schlumberger Resistivity Soundings Acquired by Uneenhaus and Zablocki (1982).



# LEGEND

4	SOUNDING LOCATIONS (Azimuths not indicated)
N	HIGHWAY
14	IMPROVED ROAD
14	RAILROAD
141	STATE BOUNDARY
141	COUNTY BOUNDARY
	ADMINISTRATIVE BOUNDARY



Aiso Available Or Aperture Card



-07

NS<sub>1</sub>1

However, major features of the models appear to approximate the geologic situation known from drillhole control not available to Smith and Ross (1982). Such features include a modeled low resistivity unit corresponding to a zone of nonwelded tuff that separates the Tiva Canyon and Topopah Spring Members of the Paintbrush Tuff, and a modeled high resistivity unit corresponding to a major portion of the Topopah Spring Member. Further interpretative work is called for to identify and separate the signatures of lateral lithologic variations from the signatures of faults.

# BIBLIOGRAPHY FOR SECTION 2.3

- Czarnecki, J.B., and W.J. Oatfield, 1987. "Use of Drillers' Logs and Geophysical Surveys to Define the Hydrogeologic Framework of the Amargosa Desert, Southern Nevada," EOS, Transactions, American Geophysical Union, vol. 68, no. 16, p. 302.
- Daniels, J.J., and J.H. Scott, 1981. Interpretation of Hole-to-Surface Resistivity Measurements at Yucca Mountain, Nevada Test Site, U.S. Geological Survey Open File Report 81-1336, 23 p.
- Fitterman, D.V., 1982. <u>Magnetometric Resistivity Survey near Fortymile Wash,</u> <u>Nevada Test Site, Nevada</u>, U.S. Geological Survey Open File Report 82-401, 27 p.
- Flanigan, V.J., 1981. A Slingram Survey at Yucca Mountain on the Nevada Test Site, U.S. Geological Survey Open File Report 81-980, 38 p.
- Frischknecht, F.C., and P.V. Raab, 1984. "Time-Domain Electromagnetic Soundings at the Nevada Test Site, Nevada," <u>Geophysics</u>, vol. 49, no. 7, July 1984, p. 981-992.
- Furgerson, R.B., 1982. <u>Remote-Reference Magnetotelluric Survey</u>, Nevada Test <u>Site and Vicinity</u>, Nevada and California, U.S. Geological Survey Open File Report 82-465.
- Greenhaus, M.R., and C.J. Zablocki, 1982. <u>A Schlumberger Resistivity Survey</u> of the Amargosa Desert, Southern Nevada, U.S. Geological Survey Open File Report 82-897.
- Hoover, D.B., W.F. Hanna, L.A. Anderson, V.J. Flanigan, and L.W. Pankratz, 1982a. <u>Geophysical Studies of the Syncline Ridge Area Nevada Test Site</u>, <u>Ny</u> County, Nevada, U.S. Geological Survey Open File Report 82-145, 55 p.
- Hoover, D.B., M.F. Chornack, K.H. Nervick, and M.M. Broker, 1982b. <u>Electrical Studies at the Proposed Wahmonie and Calico Hills Nuclear Waste</u> <u>Sites, Nevada Test Site, Nye County, Nevada</u>, U.S. Geological Survey Open File Report 82-446, 45 p.

Hoover, D.B., M.P. Chornack, M.M. Broker, 1982c. <u>E-Field Ratio Telluric</u> <u>Traverses Near Fortymile Wash, Nevada Test Site, Nevada</u>, U.S. Geological Survey Open File Report 82-1042.

- Ross, H.P., and James Lunbeck, 1978. Interpretation of Resistivity and Induced Polarization Profiles, Calico Hills and Yucca Mountain Areas, Nevada Test Site, University of Utah Research Institute report ESL-UURI-8.
- Senterfil, R.M., D.B. Hoover, and M.P. Chornack, 1982. Resistivity Sounding Investigation by the Schlumberger Method in the Yucca Mountain and Jackass Flats Area, Nevada Test Site, Nevada, U.S. Geological Survey Open File Report 82\*1043.
- Smith, C., and H.P. Ross, 1982. Interpretation of Resistivity and Induced Polarization Profiles with Severe Topographic Effects, Yucca Mountain Area, Nevada Test Site, Nevada, U.S. Geological Survey Open File Report 82-182.

.

# 2.4 SEISMIC REFRACTION SURVEYS

#### OBJECTIVES.

Saismic refraction surveys have been used in the Yucca Mountain region for investigations that range from measurements of crustal thickness (regional refraction profiles; e.g., Hoffman and Mooney, 1983) to the thickness of surficial alluvial and volcanic deposits (shallow refraction profiles, or engineering refraction profiles; e.g., Pankratz, 1982). The objectives for the work discussed below thus include (1) mapping shallow velocity structure at the proposed site of the repository surface facilities; (2) obtaining velocity information for input to processing of seismic reflection sections; (3) investigating upper-crustal structure in the site area and its relation to the structure of the surrounding region; and (4) investigating the structure of the middle and lower crust, and the refraction Moho.

High-resolution upper-crustal refraction methods and results are particularly relevant and are summarized here in more detail. Such profiles are intermediate in scale and provide the link between shallow and regional studies. High-resolution upper-crustal refraction profiles can be tied to the deepest drillholes in the vicinity of Yucca Mountain, and can be jointly interpreted with gravity, aeromagnetic, and geoelectric data. In terms of depth of penetration (0-5 km) and regional extent (40-60 km long lines), upper-crustal refraction profiles and seismic reflection profiles provide information on geologic structure and tectonic processes affecting performance of the Yucca Mountain site.

#### DATA COVERAGE AND QUALITY

The following discussion covers all refraction studies conducted by the Yucca Mountain Project, in the site area and regionally. The discussion is organized as follows: (1) shallow profiles in the vicinity of Yucca Mountain are discussed first (Pankratz, 1982), followed by (2) additional low-energy surveys to measure near-surface velocities (Rodriguez and Yount, in USGS, 1988), (3) an early study involving high-explosive (HE) and underground nuclear explosion (UNE) sources (Hoffman and Mooney, 1983), and (4) high-resolution upper-crustal profiles in the Yucca Mountain region (Mooney and Schapper, written communication; Ackermann et al., in USGS, 1988).

Three shallow, HE-source, reversed refraction profiles were run in 1982 in the vicinity of Yucca Mountain (Pankratz, 1982). One of the profiles tied to borehole UE25 a#1 in Drill Hole Wash, and extended partway across Midway Valley (Figure 2.4-1). An east-west line crossed Midway Valley and the southern part of Exile Hill, and a third line transected Midway Valley in a northwest orientation slightly north of Exile Hill. Each 2.7-km spread was acquired using 24 single geophones spaced 120 m apart, and HE shothole charges Values in size from 4 kg to 180 kg. A significant increase in the size of explosive charge was required for the Yucca Moult in lines compared to similar lines run in the Calico Hills and Wahmonie areas (Pankratz, 1982). Data were analyzed using an interactive ray-tracing program (Pankratz, 1982; Ackermann, 1979) to associate critically refracted rays with first arrivals. Of the three profiles, two yielded arrival data that were interpretable by this method. Interpretation of the velocity section for the profile that was tied to borehole UE25 a#1 deviates significantly from the stratigraphic section and the downhole velocity survey from this borehole. The author attributes this discrepancy to various causes, including poor picks of first arrivals, low-velocity layers, and the possible existence of a major off-axis vertical velocity discontinuity. The line crossing the Paintbrush Canyon fault on the eastern margin of Midway Valley indicates a significant offset east of the surface scarp, suggesting the existence of another fault within the Fran Ridge/Alice Hill horst block. Additional drilling has been conducted in Midway Valley since the Pankratz (1982) surveys were run, which could potentially provide additional constraints for interpretation of these lines.

The velocities determined from the seismic data indicate that the in situ compressional velocity is on the order of 1.0 km/sec for the surface alluvium, and only 1.7 to 2.6 km/sec for the Tiva Canyon and part of the upper part of the Topopah Spring Members. Note that the intact-rock velocities reported for the large volume core samples by Anderson (1984) for these units are about 4.5 km/sec for the Tiva Canyon Member, and 3.0 to 4.5 km/sec for the upper part of the Topopah Spring.

A survey of compressional refraction velocities, for alluvial units at five locations distributed across the Yucca Mountain region, was reported by Rodriguez and Yount (in USGS, 1988). The study compared the velocities of alluvial layers of various age at survey areas in Rock Valley, Crater Flat, Beatty area, Fortymile Wash area, and Frenchman Flat. Using a sledgehammer source and closely-spaced geophones, a number of short profiles were acquired in areas where the alluvial deposits were exposed in mapped trenches that were excavated for tectonics and other studies. A strong correlation was observed between compressional velocity and age (particularly Holocene vs. Quaternary). Some older alluvial units showed development of caliche-rich cemented horizons at the upper boundary; these features evidently gave rise to higher velocity, leading to speculation about the characteristics of an older unit from which the upper carbonate - cemented horizon had been eroded away, but which was otherwise present. This guestion was not resolved by the study. The alluvium velocities reported are low, in the range of 0.3 to 0.5 km/sec for the youngest (Q1 series) Quaternary materials, 0.7 to 1.1 km/sec for older (Q2 series) Quaternary materials, and 1.2 to 1.8 km/sec for pre-Quaternary alluvial materials.

Refraction studies using UNE and HE sources to penetrate the crust and upper crust were conducted by the USGS in 1980 and 1981 (Hoffman and Mooney, 1983). Up to 100 portable seismographs were arrayed along lines across Yucca Mountain and Jackass Flats, northward from the Amargosa Desert through Crater Flat, and from Yucca Mountain south to southern Death Valley. The lines were set up to coincide with three UNE shots on the NTS, and one HE shot detonated near Beatty, Nevada. The information provided from the HE profile pertained mostly to the upper crust, and the first arrivals from the UNE shots travelled along the seismic basement, i.e., a crustal layer with a seismic velocity of about 6 km/sec.





9012210167-08

Travel times (reduced to 6 km/sec) for the UNE events revealed delays (proportional to greater basement depth) of 1.2 to 1.4 sec for Crater Flat and Yucca Mountain, 0.8 to 1.2 sec for Jackass Flats, and 0.4 to 0.8 sec for Bare Mountain and Skull Mountain. The P-wave delays were interpreted by adapting the density-depth model of Snyder and Carr (1982) to velocity, producing a model of monotonically increasing velocity with depth. The results of multiple profiles, and the velocities reported by Pankratz (1982), were used to constrain the velocity model. On this basis, the depth to the Paleozoic basement was estimated to be 3.2 km beneath Crater Flat, and 1.1 km beneath Jackass Flats. The largest P-wave delays were observed in Crater Flat and northern Yucca Mountain, where the tuff section may be thickest. The UNE work also revealed a strong midcrustal reflector at about 15 km depth, from rays bottoming beneath Yucca Mountain and eastern Crater Flat. Deeper crustal reflectors were observed at 24 and 30 km, and total crustal thickness was estimated to be 35 km.

Travel times for the upper-crustal HE profile were modeled to within about 50 msec of observed values using a ray-tracing method. Unlike the gravity model of Crater Flat, the HE-source refraction profile shows evidence for a buried bench-like structure between the Bare Mountain range front and Crater Flat. (This part of the profile also corresponds to certain irregularities in the survey geometry.)

The present coverage of high-resolution upper-crustal seismic refraction profiles is indicated on Figure 2.4-2. Initial field tests, including the evaluation of shot sizes, noise levels, and seismic energy propagation characteristic were started in 1980 (Hoffman and Mooney, 1983). Two detailed profiles were collected in 1983: one in a north-south direction in Crater Flat (Crater Flat profile) and a second in an east-west direction from northern Crater Flat, south of Beatty, Nevada, to the Grapevine Mountains (Beatty profile). Interpretation of these profiles is presented by Ackermann et al. (in USGS, 1988). Three additional profiles were acquired in 1985 (Figure 2.4-2): (1) an east-west profile across Yucca Mountain from the northern Amargosa Valley to Jackass Flats (Yucca Mountain profile); (2) a north-south profile along Fortymile Wash, east of Yucca Mountain (Fortymile Wash profile; and (3) an east-west profile within the Amargosa Valley south of Yucca Mountain (Amargosa profile). Interpretation of these profiles is presented by south of Yucca Mountain (Amargosa Profile).

The data were collected by a seismic crew from the USGS, Menlo Park, California. Surveying was accomplished with topographic maps and a laser electronic-distance-measuring instrument. Energy sources consisted of high-yield chemical charges of up to 1,800 kg, in 50 m drillholes. Shot holes were spaced at approximately 8 to 15 km intervals along the profiles. Data were recorded at an effective sampling rate of 200 Hz, using 120 portable event recorders, each equipped with a 2 Hz vertical-component seismometer. Data quality on all profiles is very good to excellent. Recordings were made at night when wind and cultural noise were minimal, and large shot sizes were used. Specifics about these surveys, such as shot point and recorder locations and plots of ground motion data, are presented by Sutton (1984, 1985).

In order to avoid uncertainties inherent in trial-and-error modeling, these refraction profiles were recorded with field parameters satisfying the

requirements of a method for direct inversion of the data (Ackermann et al., 1982). The primary requirements for the data inversion method are close spatial sampling, and frequency of shot points along the profile. The attributes of these high-resolution upper-crustal profiles easily exceed the minimum requirements. As a result, the derived crustal models are considered to be reliable: seismic velocities and layer depths are estimated to be accurate to about 5 to 10% based on comparison with drillhole stratigraphic control and the mis-tie of crossing profiles. However, there are limitations to the survey methods and interpretation as listed below:

- 1. Since shot hole sources were used, it is not always possible to space the shots uniformly along the profiles because of limited access. (Seismographs are self-contained and portable, and can be transported by helicopter to any location.) A missed or displaced shot point can introduce interpretive uncertainty in the vicinity. For the profiles described, it was not possible to place shots on top of either Bare Mountain or Yucca Mountain, leading to some ambiguity regarding steep interfaces: (a) on the east flank of Bare Mountain, and (b) on the west flank of Yucca Mountain. Closely spaced gravity data (e.g., Snyder and Carr, 1984) have been used to reduce these ambiguities.
- 2. Because of the scattering of seismic energy within the thick volcanic sequence, it is often difficult to identify secondary reflected arrivals in the seismic data. Although the arrivals of secondary phases cannot be determined as accurately as first arrivals, secondary arrivals can often confirm layer depths determined from first arrival analysis. For the profiles described above, the lack of clear secondary arrivals has been compensated for by tying the five profiles at cross-points (Figure 2.4-2), and by locating the profiles as close as possible to deep drillholes (e.g., Scott and Castellanos, 1984) for better stratigraphic and velocity control.
- 3. There is no unique relationship between seismic velocities determined from the seismic refraction profiles and composition (i.e., geologic interpretation) or density (i.e., as needed for gravity modeling). This means there are ambiguities in the structural and geologic interpretations of seismic velocity models. For the profiles described above, these ambiguities have been minimized by using stratigraphic control and density logs obtained from deep drillholes coincident with or adjacent to the seismic refraction profiles.

## SUMMARY OF RESULTS

The shallow refraction study reported by Pankratz (1982) suggests that the method is potentially effective at Yucca Mountain, and produced profiles across Midway Valley that represent the type of structural information needed elsewhere in the site area. As pointed out early in the report, the usefulness of the method depends on resolution of the discrepancy between the velocity model developed, and downhole velocity surveys at nearby boreholes.





# YUCCA MOUNTAIN PROJECT

FIGURE 2.4-2. Map of Yucco Mountain Region Showing the Locations of Five High Resolution Upper-Crustal Seismic Refraction Profiles. The Three Profiles that Trend East-West are Referred to as the Beatty, Yucca Mountain, and Amargosa Profiles. The Two Profiles that Trend North-South are Referred to as the Crater Fia? and Fortymile Wash Profiles. Diamonds indicate Shot Points. The Location of the Proposed Repository is Shown in the Center of the Figure by a Shaded Rectangle.





Also, the method used to invert the travel time data has not been analyzed with respect to sensitivity of the results to travel time error, or other sources of interference and noise that may be present at the site area.

High-resolution, upper-crustal seismic refraction profiles (and deep seismic reflection profiles) provide the most useful information regarding geologic structure of the Yucca Mountain site, relative to other types of refraction studies, in terms of resolution, depth of penetration (0-5 km), and regional extent (40-60 km lines). These profiles can potentially be tied to control from the deepest exploratory drillholes, and can be reliably interpreted in conjunction with other geophysical data (i.e., gravity, aeromagnetic, and geoelectric surveys).

High-resolution seismic refraction profiles recorded across and around Yucca Mountain have provided reconnaissance of the general upper-crustal (0-5 km) structure near the Yucca Mountain site. Interpretation suggests seven distinct upper-crustal refracting layers, corresponding to successive alluvial, volcanic, and pre-volcanic (Paleozoic) units. This result is in agreement with velocity data from other surveys at the site area, and with borshole control.

Interpretation of the five available refraction profiles defines the proposed repository site as overlying complex extensional features manifested in the pre-Tertiary strata. The velocity contrast between the Tertiary (mainly volcanic) and pre-Tertiary sections is large enough that, when interpreted together with the borehole and gravity data, a more reliable representation of the pre-Tertiary surface is produced. A large structural depression is apparent beneath Crater Flat; it is an asymmetrical, westward-deepening structure that extends from the Bare Mountain front to the eastern flank of Yucca Mountain.

#### BIBLIOGRAPHY FOR SECTION 2.4

- Ackermann, H.D., L.W. Pancratz, and D.A. Dansereau, 1982. <u>A Comprehensive</u> System for Interpreting Seismic Refraction Arrival-Time Data Using <u>Interactive Computer Methods</u>, U.S. Geological Survey Open File Report 82-1065.
- Ackermann, H.D., W.D. Mooney, D.B. Snyder, and V.D. Sutton, 1988. "Preliminary Interpretation of Seismic Refraction and Gravity Studies West of Yuoca Mountain, Nevada and California," in Carr, M.D., and J.C. Yount, (eds), <u>Short Contributions to the Geology and Hydrology of a Potential</u> <u>Nuclear Waste Site at Yuoca Mountain, Southern Nevada</u>, U.S. Geological Survey Bulletin 1790, p. 23-33.
- Anderson, L.A., 1984. <u>Rock Property Measurements on Large-Volume Core</u> <u>Samples from Yucca Mountain USW GU-3/G-3 and USW G-4 Boreholes, Nevada Test</u> Site, Nevada, U.S. Geological Survey Open File Report 84-552.
- Hamilton, W.B., 1987. "Crustal Extension in the Basin and Range Province, Southwestern United States," in Coward, M.P., J.F. Dewey, and Society Special Publication no. 28, p. 155-176.

- Hoffman, L.R., and W.D. Mooney, 1983. A Seismic Study of Yucca Mountain and Vicinity, Southern Nevada: Data Report and Preliminary Results, U.S. Geological Survey Open File Report 83-588, 50 p., 1 plate.
- Mooney, L.R., and S.G. Schapper, written communication. "Report on the Status of Regional Geophysics, NNWSI Program: Seismic Refraction Profiles," 27 p., 8 figures (technical report submitted November 1988 to USGS/NNWSI management).
- Pankratz, L.W., 1982. Reconnaissance Seismic Refraction Studies of Calico Hills, Wahmonie, and Yucca Mountain, Southwest Nevada Test Site, Nye County, Nevada, U.S. Geological Survey Open File Report 82-478, 25 p.
- Rodriguez, E.A., and J.C. Yount, 1988. <u>Relation Between P-Wave Velocity and Stratigraphy of Late Cenozoic Deposits in Southern Nevada</u>, U.S. Geological Survey Bulletin 1790, p. 139-146.
- Scott, R.B., and M. Castellanos, 1984. Stratigraphic and Structural Relations of Volcanic Rocks in Drillholes USW GU-3 and USW G-3, Yucca Mountain, Nye County, Nevada, U.S. Geological Survey Open File Report 84-491, 121 p., 1 plate.
- Snyder, D.B., and W.J. Carr, 1984. "Interpretation of Gravity Data in a Complex Volcano-Tectonic Setting, Southwestern Nevada," <u>Journal of</u> Geophysical Research, vol. 89, p. 10,193-10,206.
- Sutton, V.D., 1984. Data Report for the 1983 Seismic Refraction Experiment at Yucca Mountain, Beatty, and Vicinity, Southwestern Nevada, U.S. Geological Survey Open File Report 84-661, 66 p., 11 plates.
- Sutton, V.D., 1985. Data Report for the 1985 Seismic Refraction Experiment at Yucca Mountain, Beatty, and Vicinity, Southwestern Nevada, U.S. Geological Survey Open File Report 85-591, 96 pp., 17 plates.

### 2.5 SEISMIC REFLECTION SURVEYS

Seismic reflection surveys relevant to site characterization may be broadly categorized as shallow, intermediate=depth, or deep methods:

- Shallow penetrating (up to about 1 sec), high-resolution profiling performed in the Yucca Mountain site area using vibrator, explosive, or weight-drop energy sources; and reflection profiling performed at locations distributed across the Yucca Mountain region using the Mini-Sosie method.
- Intermediate-depth profiles (up to several seconds) from areas of the NTS explored as prospective areas for underground nuclear testing, and from resource-related speculative lines in the Yucca Mountain region.
- Deep seismic reflection (up to about 15 sec) such as the test line acquired recently in Amargosa Valley, and other regional profiles run in Death Valley and in various parts of the Basin and Rango Province.

This section will summarize the objectives, data quality and coverage, and results from past work, organized according to these categories. All of the existing seismic reflection data or past reflection surveys of any type, known to be relevant to site characterization, are listed in Table 2.5-1.

### 2.5.1 SHALLOW, HIGH=RESOLUTION REFLECTION PROFILING

#### **OBJECTIVES**

The following discussion summarizes past uses for shallow, high-resolution profiling methods, thereby describing applicability to future studies as well. Seismic reflection methods are planned for use in establishing the continuity and regularity of important stratigraphic contacts, such as the upper and lower surfaces of the Calico Hills unit. Although reflections were empected from such contacts in past studies, no coherent reflections have been observed (see discussion of data quality below). Volcanic ash flows and ash fall deposits can exhibit lateral discontinuity depending on the conditions of deposition, such as pre-eruptive topography and the timing of eruption. Similarly, reflection methods may also be used for detection and delineation of buried volcanic deposits such as basaltic dikes or sills, which are sampled and studied as components of the geologic history of the site.

Seismic reflection methods have found important application to detection of buried faults, mapping the extent of fault zones, and investigating the subsurface geometry of fault zones. Fault detection problems may be approached using photogeology or other remote sensing techniques, but subsurface characterization may be required to determine whether a surface feature is caused by a fault (e.g., see USGS, 1988). Faults may be detected from discontinuity of reflections, or from offsets. Estimates for total fault offset can be produced for certain types of faults by seismic Table 2.5+1. Existing Seismic Reflection Data and Past Surveys

Shallow Crustal (0=2 sec) Profiles

 Crater Flat/Amargosa Desert = high=resolution Mini+Sosie reflection profiles acquired and processed by the USGS (USSS, 1988)

Quality - fair to good Objectives - image faults/fault offsets in Quaternary alluvium Results - partly to fully successful imaging of fault offsets

- Death Valley high-resolution Mini-Sosie reflection line (L. Serpa, University of New Orleans, personal communication, 1988)
- Yucca Mountain various high-resolution profiles commissioned by the USGS discussed by McGovern et al. (1983)

Intermediate Crustal (0=5 sec) Profiles

 Mid Valley/Nevada Test Site - land air-gun profiling of the upper 3-5 sec of the crust (McArthur and Burkhard, 1986)

Quality - good to excellent Objectives - map depth to Paleozoic basement in Tertiary basins Results - successful imaging of basin fill and subhorizontal detachment faults

 Yucca Flat/Nevada Test Site - land air-gun profiling of the upper 3-5 sec of the crust (N. Burkhard, personal communication, 1988)

Quality - good to excellent Objectives - map depth to Paleozoic basement in Tertiary basins Results - successful imaging of basin fill and subhorizontal detachment faults

 Frenchman Flat/Nevada Test Site - land air-gun profiling of the upper 3-5 sec of the crust (N. Burkhard, personal communication, 1988)

Quality - good to excellent Objectives - map depth to Paleozoic basement in Tertiary basins Results - successful imaging of basin fill and subhorizontal detachment isuits

- Seisdata speculative lines obtained with vibroseis and explosive sources
  - a. Las Vegas shear zone (Lines 8 and 8a of Wasatch Cordilleran hingeline reconnaissance survey)

Quality = poor to fair Objectives = map Tertiary basins for hydrocarbon exploration Results = successfully obtained reflections from only the upper 1 sec
Table 2.5-1. Existing Seismic Reflection Data and Past Surveys (continued)

- Seisdrea speculative lines obtained with vibroseis and explosive sources (continued)
  - b. Pahrump Valley (Lines 12 and 17 of Sandy speculative survey)
- Geophysical Service Inc. speculative lines obtained with vibroseis and explosive sources
  - a. Big Smokey Valley, Nevada
  - b. Monitor Valley, Nevada
  - c. Railroad Valley, Nevada
  - d. Tikaboo Valley, Nevada

for a-d:

Quality = poor to excellent Objectives = image Tertiary basins for hydrocarbon exploration Results = variable success in imaging sedimentary fill of Tertiary basins

9. Other speculative data - unknown source types

a. White River Valley, Nevada
 b. Garden Valley, Nevada

Quality - unknown Objectives - image Tertiary basins for hydrocarbon exploration Results - unknown

Deep Crustal (0-15 sec) Profiles

 Amargosa Desert near Beatty and Lathrop Wells - deep crustal Vibroseis and explosive feasibility study (Brocher et al., 1989)

Quality = good to excellent Objectives = image entire crust down to the Moho Results = successful imaging of all portions of the crust

 Death Valley/SW Amargosa Desert/S. Pahrump Valley - deep crustal Vibroseis survey by COCORP (Serpa et al., 1988)

Quality - poor to fair Objectives - image entire crust down to the Moho Results - partly successful in mapping mid- to lower crustal structure reflection methods. High-resolution is needed because the targets are shallow, and offset distances or fault zone dimensions are typically a fraction of a wavelength. A principal objective for site characterization is the geometry of N=S striking faults with depth, particularly in the immediate vicinity of the site. The most significant information needed from seismic surveys is the attitude of the subsurface extensions of faults proximal to Yucca Mountain, i.e., whether they are coplanar with surface indications, or listric, at depths on the order of 1 to 3 km. A related objective is to characterize the nature of tilting and warping associated with faulting. (Results obtained so far from Yucca Mountain have not been of appropriate quality for interpreting subsurface fault geometry and related kinematics.)

#### DATA COVERAGE AND QUALITY (Shallow, High-Resolution Reflection Profiling)

Several field surveys have been conducted in association with the Yucca Mountain Project, using shallow reflection methods with portable equipment and limited geophone deployment. Much of the work has been performed using the Mini-Sosie method (Barbier, 1983). The salient aspects of this method are that it uses one or more portable, soil compaction vibrators as the source, and a closely spaced array of single geophones. Data quality is achieved by stacking a large number of traces, each triggered from a single vibrator cycle. Randomization of the vibrator repeat period suppresses contributions from extraneous vibrator cycles. The processing steps used in the work described here typically included common-depth-point (CDP) sorting, constant velocity analysis, normal movement correction, spectral whitening, deconvolution, bandpass filtering, datum and residual statics, and final CDP stack migration (Barbier, 1983; USGS, 1988). The method can produce useful reflections at depths corresponding to two-way travel times of about 0.3 to 1 sec, depending on site conditions. Note that the uppermost 50 to 150 m (depending on velocity) cannot be imaged by this method.

The following items summarize the shallow, Mini-Sosie reflection studies performed to date in association with the Yucca Mountain Project:

- 1. A profile was acquired over the Carpetbag fault and the Yucca fault, using the Mini-Sosie method. The portion of the line across the Carpetbag fault produced good results, interpreted as the Rainier Mesa tuff dipping steeply into the Yucca Flat basin, with no indication of fault offset. This leads to speculation that at least at the profile location, the post-event faulting observed in association with UNEs was caused by compaction of alluvium, and slip on the steeply dipping Rainier Mesa Member of the Timber Mountain Tuff. The portion of the line crossing the Yucca fault yielded no useful information, probably because the Rainier Mesa Member is below the 1 sec two-way travel time, and there are no interpretable reflecting horizons in the overlying alluvium.
- 2. An east-west line approximately 1.1 km in length was run across the Beatty scarp, about 4 km south of Beatty, Nevada (USGS, 1988). The profile shows two bands of reflections, which are coherent beneath the Beatty scarp. The relative incoherency of the section was attributed to low contrasts between geologic layers, residual

statics problems, and scattering by boulders near the surface (USGS, 1988). Extension of a fault from the surface scarp to a portion of the section that suggests offset would require reverse motion on a fault plane dipping 30 degrees to the west. This is inconsistent with the regional predominance of normal slip on north trending faults, therefore the investigator opines that the Beatty scarp is not fault related.

- 3. An east-west line was run across a fault scarp (Windy Wash fault) in Crater Flat that had been investigated by means of several nearby trenches (USGS, 1988; Swadley and Hoover, 1983). The near-surface velocity structure was found to be complex. A CDP sorting and constant velocity analysis was used to determine residual statics, which were correlated somewhat with the surface distribution of Quaternary alluvial materials and extrusive Tertiary volcanics. A single reflecting horizon was traced across the section, and exhibited dipping portions and fault offset. Subsurface faulting indications were related to, but different from, those observed in nearby surface trenches. The results were interpreted not to support normal faulting mechanisms, and to be consistent with strike-slip faulting.
- 4. An east-west line was run from the east slope of Bare Mountain, between the two volcanic cones in central Crater Flat, then further into the basin toward Yucca Mountain. Some reflected energy was observed; however, the data were of insufficient quality to substantially support or challenge the published structural model by W. Carr (USGS, 1988).
- 5. An east-west line was run from about 1 km up Tarantula Canyon in the Bare Mountain block, down into the valley toward Crater Flat. This line was run to see if the range front fault could be imaged, and detected at the mouth of Tarantula Canyon. However, no coherent reflections were observed at this portion of the line. Another portion does indicate a near-surface anticlinal feature oriented on-strike with a zone of faults mapped by M. Reheis (see Figure 8.1 of USGS, 1988).
- 6. A line was run along the old railroad bed east of Lathrop Wells, Nevada. These data are perhaps least informative of any yet acquired, in that the only feature to the section is a general change in character from east to west.

In addition to the Mini-Sosie work, several reflection studies of a more conventional type were performed in the Yucca Mountain site area during the 1980-82 time period. These are described in a summary report (McGovern, 1983) as three distinct surveys undertaken by different organizations: Colorado School of Mines (1980), Birdwell (1981), and Seisdata (1982). The recording windows for these surveys generally extended beyond 1 sec, but they are discussed here because (1) high frequency sources were used (small explosive charges; vibrator sweeps to >100 Hz), (2) the receiver arrays were small, and (3) the lines were located in the immediate site area. These surveys were thus designed to investigate shallow reflectors, of which there may be several in the first 1 sec. The summary report presents negative results from these surveys, and states that useful seismic reflection data cannot be acquired from the site area. The following is a brief description of each of these studies:

- The Colorado School of Mines (CSM) ran two lines on the eastern flank of Yucca Mountain in 1980, as shown in Figure 2.5-1. One of the lines crossed Midway Valley (perpendicular to strike) and the other extended part of the way up Drill Hole Wash (along strike) terminating in a side canyon. These were in-line type surveys with a single (8 to 120 Hz) vibrator source on 55 ft intervals (55 to 1,320 ft offsets), and single geophones placed at the same interval. No interpretable data were acquired from these lines (see also Barry, 1980).
- 2. A single reflection line was conducted by Birdwell, Inc., during 1981, in Drill Hole Wash (along strike) in the vicinity of drillholes USW G-1 and H-1. Based on several noise tests, a group array of 120 geophones was used in a 100 by 200 ft pattern. Three in-line vibrators (48 to 6 Hz) were deployed on 50 ft stations (500 to 2,850 ft offsets). No reflections were observed on the raw reflection records, and improvement was not achieved from processing. Coherent noise was evident throughout the acquisition window, and when removed by velocity filtering, produced a record with no coherence basis for residual statics analysis (McGovern, 1983).
- 3. The Seisdata (1982) survey consisted of a series of noise studies, and four swath-type profiles. Ground roll and direct noise were observed in the noise studies, and the combined receiver and vibrator source arrays, group patterns, and source-receiver offsets were designed using the noise characteristics according to standard industry practice developed for sedimentary terrane. Surface Primacord explosive sources were tested in addition to vibrator sources, and found to produce significantly more noise than vibrators (but were used predominantly in the surveys because of vibrator mechanical failures). The survey also included an array of 10-25 lb. exposure shots at 200 ft depths. The swath profiles were variously configured up to 8 stations wide and 24 stations long, with a maximum CDP fold of 384. As indicated in Figure 2.5-1, two of the lines were situated in canyons on the eastern flank of Yucca Mountain (roughly perpendicular to strike), one parallel to Exile Hill (along strike of the Bow Ridge fault), and one along the crest of a northwest trending ridge north of the site area. The processed sections presented in the summary report are the result of constant velocity moveout analysis, and sorting to maximize CDP fold. No other sorting strategy for the swath data was effective for improving coherency. Automatic residual statics were not relied upon, so as to avoid artificial lineups. One of the lines was processed by two separate contract service companies, proving uninterpretable in both cases.

Further information on the array designs, acquisition, and processing for these three studies, as well as presentation of the final processed sections, may be found in the summary report (McGovern, 1983).



In addition to the surveys described above, several tests of other shallow reflection methods have been performed in association with the Yucca Mountain Project. Two reflection lines were run in the tunnels of Rainier Mesa with blasting caps as sources. While reflected energy was observed, its origin was unknown (reflection from below or above the tunnel, or both). The data are therefore virtually uninterpretable, and are unpublished.

Two shear wave tests have been conducted, using a sledgehammer shear wave source and closely spaced, three-component geophone arrays. One line was run at the mouth of Drill Hole Wash, in a setting where high-resolution work using P-wave sources yielded no interpretable records (McGovern, 1983; discussed below). The source consisted of a sledgehammer impinging on a fixture anchored into the ground surface. The distance between geophones was 1 m. This test confirmed the absence of interference from generated Love waves, and produced a coherent set of arrivals that were apparently reflected from depths ranging to about 200 m (Rasbrouck, 1987).

The same shear wave method was tested in an east-west profile on the flank of Fran Ridge, near Yucca Mountain. The method demonstrated some facility for locating faults without reliance on offset reflectors, and for characterizing the average distribution of fractures. The test was located in fractured tuff of the Topopah Spring Member, lithologically similar to the repository host rock. Ten records acquired exhibit differences between seismograms produced by SV and SH methods, indicating possible birefringence related to fracturing. Although transmission was strongly attenuated by a relatively highly fractured portion of the outcrop, seismic arrivals could be detected to 36 meters from the source in this region. All of the records acquired (P, SV, SH) indicate that this portion of the profile crossed an unmapped fault (Hasbrouck, 1988).

Another set of shallow reflection lines was run in the Exile Hill area, at the proposed location for repository surface facilities (Reynolds, 1985). A total of five lines were run using a short-offset, short-spread configuration with a weight drop source and a moveable streamer-type receiver array with gimballed geophones on 66-ft stations. The array was towed by a truck that also operated the weight drop source. Signal enhancement was possible by stacking traces, but was not generally used in this survey.

A frequency-wavenumber (f-k) filter was applied to each gather, to nominally reject events with phase velocities less than 5,000 m/sec. The records were then CDP stacked to six-fold, and datum corrections were applied using velocities derived from nearby refraction surveys. A "dip filter" was applied whereby nine adjacent traces were examined for coherency at dips up to 20 degrees. Events identified as coherent were odded at mean amplitude back to unfiltered data at the position of the central trace of the nine (Reynolds, 1985).

Some of the reported results are discussed here for perspective on the survey. The dip slip offset of the Bow Ridge fault in the vicinity of Trench 14 was estimated to be 250 to 300 ft. Variations in apparent velocity indicate that the valley east of Exile Hill may be composed of a number of small faulted blocks, filled in by alluvium. Low velocity, possibly associated with fracturing and weathering, was found to be more prevalent on the east side of Exile Hill than to the west. These conclusions are consistent with, and correspond to, the known geology of the location.

The Exile Hill survey raises a number of questions that are relevant to future use of the method, and possibly to future exploration at this location. Low velocities such as those observed are often associated with severe attenuation. However, coherent reflections were reported with travel times up to 1 sec corresponding to reflector depth of 1,500 ft or more. Frequency-wavenumber filtering can introduce artificial lineups, as mentioned in the report (Reynolds, 1985), as shown for random noise by Howard and Danbom (1983), and as discussed by Serpa et al. (1988). The use of f=k filtering may have produced artifacts because of clipping and spatial aliasing. Refraction data and downhole velocity surveys showed the shear wave velocity to be on the order of 400 to 550 m/sec, which was the approximate phase velocity of ground roll. With receiver spacing of 20 m and source band of roughly 20 to 60 Hz, the ground roll wavelength range of 6 to 25 m was inadequately sampled for f-k filtering. As pointed out by Jones et al. (1987), f~k filtering can provide significant noise rejection, but introduces stringent conditions on data guality and sampling. The nonlinear dip filter used subsequent to f-k filtering (Reynolds, 1985) probably accentuated the artifacts caused by aliasing.

SUMMARY OF RESULTS (Shallow, High-Resolution Reflection Profiling)

Experience gained in acquiring, processing, and interpreting highresolution reflection data in southern Nevada, particularly using the Mini-Sosie method, is judged to be applicable to site characterization. Improvement has been obtained by grouping geophones in clusters, and by processing in the field for adjustment of acquisition parameters.

For future fault detection and characterization work, the Mini-Sosie investigator (Harding, in USGS, 1988) recommends multiple lines in network configuration, rather than single profiles. Accordingly, additional lines would be run across the Windy Wash fault in Crater Flat, and at the mouth of Tarantula Canyon, to substantiate the existing interpretations. Satisfactory high-resolution reflection data can probably be acquired from some areas of Crater Flat, but there are areas where the alluvial cover is too deep, or the volcanic bedrock too irregular for interpretable profiling by shallow reflection methods.

The summary report on the 1980, 1981, and 1982 reflection surveys (McGovern, 1983) recommends against "...any additional reflection surveys in the Yucca Mountain complex" on the basis of the work described above. The report discounts the 1980 and 1981 surveys because the designs were limited and clearly vulnerable to noise. The negative conclusion is thus based principally on the 1982 survey. However, the 1982 survey was also limited, and the results very likely do not represent all methods available for seismic reflection exploration. Additional discussion of the merits of the 1982 survey is presented in Section 3.1 of this paper

Though the shear wave tests were not extensive, they indicate that shallow-penetrating shear wave techniques may be effective in the tuff

sequence, despite the mixed results obtained with more conventional survey methods, using explosive and vibrator sources. The shallow reflection lines in the vicinity of Exile Hill produced some useful results (e.g., velocities); however, the presence of artifacts in the processed reflection sections is likely.

#### 2.5.2 INTERMEDIATE-DEPTH REFLECTION SURVEYS

#### OBJECTIVES

This category includes surveys where the two-way acquisition window extends to roughly 5 sec (+/~), and target structures lie between about 0.3 and 10 km depth. Historically, this type of survey has been used mostly for resource exploration in sedimentary terranes. Structural features that represent the tectonic setting of Yucca Mountain are likely to be found in this depth range. Older features (not necessarily inactive) in the Yucca Mountain site area may be found below the Tertiary volcanics. Intermediatedepth methods are more specific to the site area relative to surrounding areas, and thus may be useful for assessing the relation of deeper structure to shallower, more easily recognizable features of which the effects on site performance are better understood.

The clarity of interpretation that can result from intermediate-depth seismic reflection in certain geologic settings is preferable to that which is typically obtained from other geophysical methods. Clarity is important for investigation of such complex topics as the origin of Crater Flat, and the conformation of the Paleozoic-Tertiary contact beneath Yusca Mountain.

#### DATA COVERAGE AND QUALITY (Intermediate-Depth Reflection Surveys)

A significant number of intermediate-depth (0-5 sec) lines has been acquired regionally, especially to the southeast of Yucca Mountain (Figure 2.5-2). These surveys have used shallow explosive, Vibroseis, and land air-gun sources. The guality of the data is variable, and reflects the care with which the surveys were designed and conducted. As shown in the figure, an extensive grid of lines has been run in the Pahrump Valley, west of Las Vegas. This grid is aligned with the strike of major structures such as the Las Vegas shear zone, the Spring Mountains, and Death Valley. Additional lines have been run across the Las Vegas valley and the Spring Mountains, and a northwest trending line has been run following the Las Vegas shear zone. The area has been considered for cil and gas exploration because thick Paleozoic carbonate rocks have been thrust over Mesozoic sandstone and shale. The data are proprietary and have not been reviewed by the Yucca Mountain Project. However, the total number of lines suggests that the quality of the data favors useful interpretation. Efforts will be made to obtain access to these data for review and possible acquisition, and for possible application of the same methods closer to the site.

A number of reflection lines have been run on the NTS in support of weapons testing programs (see Table 2.5-1). These include a network of lines





19. 1

2.5-2. Location of Seismic Reflection Lines in the Yucca Mountain Region, Acquired by the Yucca Mountain Project, or by the Oil-and-Gas Industry for Speculative



# LEGEND



in Yucca Flat (Figure 2.5-2), lines in Frenchman Flat, and three lines in Mid Valley. The geology of Yucca Flat differs somewhat from the Yucca Mountain site area, and consists of tuffaceous and mixed alluvium (originating from Cenozcic and Paleozoic deposits) overlying tuffs, and upper Paleozoic clastics and carbonates over a large area. Velocity structure at Yucca Flat and Frenchman Flat differs significantly from conditions in the Yucca Mountain site area. However, methodological and interpretive findings from reflection profiling in these areas may be applicable to exploration of areas adjacent to Yucca Mountain, such as Jackass Flats.

Strong reflections from the upper 2 to 3 km can be readily obtained in Yucca Flat and Frenchman Flat, where there is excellent borehole control for use in interpretation and data processing. Impedance contrast suitable for mappable reflections is typically observed at the tuffaceous alluvium=mixed alluvium contact, and the Faleozoic=Tertiary contact. The tuffaceous alluvium/tuff contact is not generally a mappable reflector, nor is the mixed alluvium/Paleozoic contact. In addition some of the vitrophyres in the tuff section can be mapped as reflectors. Borehole control in Yucca Flat has provided a water table database which confirms that the water table is not a mappable reflector. Much of the seismic data for these areas on the NTS are unclassified, but the data sections, the details of acquisition, seismic interpretation, and integration with other geophysical and geological data have not been published (N. Burkhard, personal communication, 1989).

Another survey run for the weapons-related program on the NTS consisted of three intersecting reflection lines, and two exploratory boreholes, in the southern part of Mid Valley (McArthur and Burkhard, 1986). A total of 18 km of profiling was done using a land air-gun source for the first time on the NTS. The survey lines were designed based on gravity data, to better characterize the target area with minimal effort. The lines were located so as to pass near the boreholes. A linear source-receiver array was used for each line, with 48-fold CDP coverage and 25 m group separation.

Several coherent events were found, allowing interpretation of the tuff-alluvium contact, the basal member of the Timber Mountain Tuff, and the Thirsty Canyon Tuff embedded in the alluvium (McArthur and Burkhard, 1986). Lithologic work on drill core from the calibrating boreholes, in conjunction with downhole geophone surveys, provided the basis for stratigraphic interpretation of the seismic profiles. In addition the Paleozoic-Tertiary contact, and structure deeper in the section (possibly a subhorizontal detachment), were inferred. Various faults were observed, and were fully consistent with reasonable geologic interpretations for the formation of Mid Valley. Little or no evidence for listric faults was observed down to at least 1.5 sec two-way travel time.

A high-quality, east-west reflection line was recently acquired by the USGS south of Lathrop Wells, to evaluate methods for deep, regional crustal exploration (Brocher et al., 1989). Although the survey was conducted primarily to test methods for deep, regional profiling as described below, extensive information was also obtained from intermediate-depth. Vibrator and explosive sources were compared using in-line source-receiver geometry. The dynamite and vibrator-source sections are similar in the first 5 sec, with good to excellent reflections (Brocher and Hart, 1988). Numerous fault blocks are well-delineated, and intrabasin reflections are common.

#### SUMMARY OF RESULTS (Intermediate=Depth Reflection Surveys)

When data are appropriately collected, the reflection method has apparently revealed the crystalline basement, reflections from the tuff-alluvium contact, reflections from various tuff units, and the Paleozoic-Tertiary contact. The Amargosa Valley test line (Brocher et al., 1989) shows that methods based on both vibrator and explosive sources can provide useful intermediate-depth information in a geologic setting that is comparable to, although not the same as, the site area. (Further improvement might be obtained by adjusting survey parameters to optimize response from intermediate depth features.) The Mid Valley survey (McArthur and Burkhard, 1986) and the Amargosa Valley test line will be further examined as described in Site Characterization Plan Activity 8.3.1.17.4.7.1 (Evaluate intermediate-depth reflection and refraction methods and plan potential application of these methods within the site area). Reflection profiling for the weapons testing program on the NTS has been extensive and successful, and aspects of the geologic setting in these areas are similar to areas in the vicinity of Yucca Mountain. Information on the acquisition and interpretation of these profiles is accessible and will be incorporated into Project activities to the extent practicable.

A fundamental obstacle to the imaging of subsurface structure at intermediate-depth is that travel times for reflected events correspond to travel times for the strongest noise. Available data and interpretation indicate that useful intermediate-depth reflection data can be obtained for the Yucca Mountain site area through exercise of appropriate controls, notwithstanding the past negative results described above in the section on shallow high-resolution profiling. Appropriate controls include choice of sources (including sources with sufficient power); orientation of lines with respect to the fabric of the tectonic setting; location of the lines with respect to complex structure, particularly off-axis structure; survey design in conjunction with gravity and magnetic interpretations; attention to statics and the near-surface velocity structure, whereby travel time correction uncertainty is reduced to acceptable levels; use of existing or future boreholes for calibration of the reflection response; and use of downhole geophone surveys for advance estimation of seismic response.

#### 2.5.3 DEEP REFLECTION PROFILES

#### OBJECTIVES

Extensive deep, regional reflection profiling has been undertaken by the GUCCTP (Consortium for Continental Reflection Profiling), supported mostly by the National Science Foundation. General objectives of this profiling are to further understand the tectonic setting of major geologic provinces, to identify and delineate active processes, and evaluate alternate tectonic models. More specific objectives include investigation of (1) subsurface geometry of major faults; (2) possible concealed features such as mid-crustal detachments or magma bodies; (3) the relation of surface mapped faults to concealed features; (4) variations in the seismic basement and transitions in Moho depth; (5) deep subsurface conditions associated with prominent potential field anomalies; (6) structural differences across boundaries



5.4



# YUCCA MOUNTAIN PROJECT

2.5-2. Location of Seismic Reflection Lines in the Yucca Mountain Region, Acquired by the Yucca Mountain Project, or by the Oil-and-Gas Industry for Speculative Purposes, and by CORCORP.



### LEGEND

SEISMIC REFLECTION LINES HIGHWAY INFROVED ROAD RAILROAD STATE BOUNDARY 121 / COUNTY BOUNDARY ADMINISTRATIVE BOUNDARY APERTURE NEVADA 1393 2148 9012210167-16 in Yucca Flat (Figure 2.5-2), lines in Frenchman Flat, and three lines in Mid Valley. The geology of Yucca Flat differs somewhat from the Yucca Mountain site area, and consists of tuffaceous and mixed alluvium (originating from Cenozoic and Paleozoic deposits) overlying tuffs, and upper Paleozoic clastics and carbonates over a large area. Velocity structure at Yucca Flat and Frenchman Flat differs significantly from conditions in the Yucca Mountain site area. However, methodological and interpretive findings from reflection profiling in these areas may be applicable to exploration of areas adjacent to Yucca Mountain, such as Jackass Flats.

Strong reflections from the upper 2 to 3 km can be readily obtained in Yucca Flat and Frenchman Flat, where there is excellent borehole control for use in interpretation and data processing. Impedance contrast suitable for mappable reflections is typically observed at the tuffaceous alluvium-mixed alluvium contact, and the Paleozoic-Tertiary contact. The tuffaceous alluvium/tuff contact is not generally a mappable reflector, not is the mixed alluvium/Paleozoic contact. In addition some of the vitrophyres in the tuff section can be mapped as reflectors. Borehole control in Yucca Flat has provided a water table database which confirms that the water table is not a mappable reflector. Much of the seismic data for these areas on the NTS are unclassified, but the data sections, the details of acquisition, seismic interpretation, and integration with other geophysical and geological data have not been published (N. Burkhard, personal communication, 1989).

Another survey run for the weapons-related program on the NTS consisted of three intersecting reflection lines, and two exploratory boreholes, in the southern part of Mid Valley (McArthur and Burkhard, 1986). A total of 18 km or profiling was done using a land air-gun source for the first time on the NTS. The survey lines were designed based on gravity data, to better characterize the target area with minimal effort. The lines were located so as to pass near the boreholes. A linear source-receiver array was used for each line, with 48-fold CDP coverage and 25 m group separation.

Several coherent events were found, allowing interpretation of the tuff-alluvium contact, the basal member of the Timber Mountain Tuff, and the Thirsty Canyon Tuff embedded in the alluvium (McArthur and Burkhard, 1986). Lithologic work on drill core from the calibrating boreholes, in conjunction with downhole geophone surveys, provided the basis for stratigraphic interpretation of the seismic profiles. In addition the Paleozoic-Tertiary contact, and structure deeper in the section (possibly a subhorizontal detachment), were inferred. Various faults were observed, and were fully consistent with reasonable geologic interpretations for the formation of Mid Valley. Little or no evidence for listric faults was observed down to at least 1.5 sec two-way travel time.

A high-quality, east-west reflection line was recently acquired by the USGS south of Lathrop Wells, to evaluate methods for deep, regional crustal exploration (Brocher et al., 1989). Although the survey was conducted primarily to test methods for deep, regional profiling as described below, extensive information was also obtained from intermediate-depth. Vibrator and explosive sources were compared using in-line source-receiver geometry. The dynamite and vibrator-source sections are similar in the first 5 sec, with good to excellent reflections (Brocher and Hart, 1988). Numerous fault blocks are well-delineated, and intrabasin reflections are common.

#### SUMMARY OF RESULTS (Intermediate-Depth Reflection Surveys)

When data are appropriately collected, the reflection method has apparently revealed the crystalline basement, reflections from the tuff-alluvium contact, reflections from various tuff units, and the Paleozoic-Tertiary contact. The Amargosa Valley test line (Brocher et al., 1989) shows that methods based on both vibrator and explosive sources can provide useful intermediate-depth information in a geologic setting that is comparable to, although not the same as, the site area. (Further improvement might be obtained by adjusting survey parameters to optimize response from intermediate depth features.) The Mid Valley survey (McArthur and Burkhard, 1986) and the Amargosa Valley test line will be further examined as described in Site Characterization Plan Activity 8.3.1.17.4.7.1 (Evaluate intermediate-depth reflection and refraction methods and plan potential application of these methods within the site area). Reflection profiling for the weapons testing program on the NTS has been extensive and successful, and aspects of the geologic setting in these areas are similar to areas in the vicinity of Yucca Mountain. Information on the acquisition and interpretation of these profiles is accessible and will be incorporated into Project activities to the extent practicable.

A fundamental obstacle to the imaging of subsurface structure at intermediate-depth is that travel times for reflected events correspond to travel times for the strongest noise. Available data and interpretation indicate that useful intermediate-depth reflection data can be obtained for the Yucca Mountain site area through exercise of appropriate controls, notwithstanding the past negative results described above in the section on shallow high-resolution profiling. Appropriate controls include choice of sources (including sources with sufficient power); orientation of lines with respect to the fabric of the tectonic setting; location of the lines with respect to complex structure, particularly off-axis structure; survey design in conjunction with gravity and magnetic interpretations; attention to statics and the near-surface velocity structure, whereby travel time correction uncertairty is reduced to acceptable levels; use of existing or future boreholes for calibration of the reflection response; and use of downhole geophone surveys for advance estimation of seismic response.

#### 2.5.3 DEEP REFLECTION PROFILES

#### OBJECTIVES

Extensive deep, regional reflection profiling has been undertaken by the COCORP (Consortium for Continental Reflection Profiling), supported mostly by the National Science Foundation. General objectives of this profiling are to further understand the tectonic setting of major geologic provinces, to identify and delineate active processes, and evaluate alternate tectonic models. More specific objectives include investigation of (1) subsurface geometry of major faults; (2) possible concealed features such as mid-crustal detachments or magma bodies; (3) the relation of surface mapped faults to corcealed features; (4) variations in the seismic basement and transitions in Moho depth; (5) deep subsurface conditions associated with prominent potential field anomalies; (6) structural differences across boundaries between crustal provinces; and (7) paleotectonic history of the continent, and the influence of pre-Tertiary structure on active processes.

#### DATA COVERAGE AND QUALITY (Deep Reflection Profiles)

About 250 km of deep, regional reflection profiles have been acquired by COCORP in the Mojave Desert and Death Valley (Serpa et al., 1988; deVoogd et al., 1986). The COCORP Death Valley Survey consisted of five intersecting profiles, as shown in Figure 2.5-2. Other COCORP profiles in adjacent areas include an east-north-east trending transect across central Arizona (Houser et al., 1987) and a series of connected east-west lines across Nevada at about 40°N (overview by Allmendinger et al., 1987).

COCORP data have typically been acquired using Vibroseis sources, with offsets up to 10 km, and receiver arrays designed according to noise conditions. The lines referenced above were run with in-line spreads. Most have used continuous profiling with high-fold CDP coverage; off-end configurations were used for the Death Valley lines (Serpa et al., 1987), resulting in lower-fold but fewer source stations. COCORP profiles are typically acquired with a sampling interval of 4 to 8 msec, recording window of 0 to 15 sec, 8 to 32 Hz sweep, and up to 8 sweeps stacked per source station. Processing steps are designed to avoid artificial events (Serpa et al., 1987; Allmendinger et al., 1987), and may sacrifice high-frequency content for interpretability consistent with the deep profiling objectives. The use of velocity (f-k) filtering has been investigated for many sections (Allmendinger et al., 1987), but is usually not used for the final section. Where coherence is a problem or higher resolution is needed, single-source, single-fold, near-vertical incidence sections have been generated (deVoogd et al., 1986; Klemperer et al., 1986). Data quality of the COCORP profiles could probably be increased by more attention to testing of acquisition parameters, and characterization of noise sources in the field.

COCORP lines in the Yucca Mountain region, including the profile at 40°N (about 300 km north of Yucca Mountain), have contributed significantly to understanding of the tectonic provinces and paleotectonic history of western North America. Previously unknown crustal features have been detected such as (1) the possible feeder dike structure associated with the Wingate Wash fault in southern Death Valley (deVoogd et al., 1986; Serpa et al., 1988), which supports a new interpretation of laterally continuous, mid-crustal reflections as active or cooled magmatic intrusions; and (2) an offset in the reflection Moho at the transition between the Basin and Range, and the Colorado plateau (Hauser et al., 1987). Association of mid-crustal reflections with models for deep detachment surfaces supports the reinterpretation of structures that had previously been inferred from surface observations. The nature of the reflection Moho correlates with the morphology of the Basin and Range, and is most poorly expressed where low-angle normal faults are best developed in the mid-crust (Allmendinger et al., 1987).

There are two factors that probably contribute to the success of the COCORP surveys. The lines have been carefully located and criented with respect to target structures, and structures that could interfere with deep

profiling. In addition, off-axis scatter and surface wave noise problems are reduced for deep profiles because of the long travel time of deep reflections, the apparent low seismic attenuation of the middle and lower crust, and the relatively rapid dissipation of energy confined near the surface.

A high-quality, 27-km long, east-west reflection test line was recently acquired by the USGS in the Amargosa Valley south of Yucca Mountain (Figure 2.5-2). Vibrator reflection data were acquired by continuous profiling at 60-fold, and explosive-source data were acquired in single-fold, split-spread configuration. The comparison of explosive and Vibroseis sources indicates that explosive sources in shot holes provide high-quality images of the lower crust and Moho, due to the high energy levels and better coupling to the earth. The explosive source data are more interpretable below about 5 sec two-way travel time, despite the high fold of the vibrator data. The explosive source data clearly indicate near-horizontal discontinuous (e.g., 1 km length) reflections below 5 sec.

Surface conditions along the Amargosa Valley line are relatively uniform, consisting of faulted blocks mostly buried by alluvium. The structure of the faulted, buried terrane is evident from the reflection profile down to about 5 sec two-way travel time. Below this is a mid-crustal reflector at about 6 to 8 sec, which has been observed in Death Valley COCORP results (Serpa et al., 1987) and explained variously as an extensive magma body, cooled intrusions, deep detachment, or brittle-ductile crustal transition. Abundant reflections from below this depth are less readily interpretable, but may represent detachment surfaces and major pre-Tertiary extensional or thrust faults.

A change in character below about 10 sec vaguely suggests a demarcation such as the base of the crust, corresponding to the reflection Moho identified from COCORP profiles across central Nevada by Klemperer et al. (1986). The change is recognized principally by a lack of reflections at greater travel times, similar to that observed by Knuepfer et al. (1987) from a COCORP profile across the Walker Lane at about 40°N.

#### SUMMARY OF RESULTS (Deep Reflection Profiles)

COCORF lines have been run across the Walker Lane in west-central Nevada (Knuepfer et al., 1987), across the central Basin and Range at the 40th parallel (Allmendinger et al., 1987), and in the vicinity of Death Valley (Serra et al., 1988). Interpretable results have been obtained from these lines, such as mapping of the reflection Moho, mapping of strong mid-crustal reflections postulated to be regional detachments (Knuepfer et al., 1987), and detection of probable features such as magma bodies and Moho offset. It is reasonable to assume that such results could be obtained on a large scale for the Yucca Mountain region, and results from reflection testing in the Amargosa Valley support this view. This topic will receive further consideration in a review planned for SCP Activity 8.3.1.17.4.3.1.

#### BIBLIOGRAPHY FOR SECTION 2.5

- Allmendinger, R.W., et al., 1987. <u>Overview of the COCORP 40°N Transect</u>, <u>Western United States: The Fabric of an Orogenic Belt</u>, Geological Society of America Bulletin, vol. 98, p. 308-319.
- Barbier, M.G., 1983. The Mini-Sosie Method, International Human Resources Development Corp, Boston, Massachusetts.
- Barry, C.T., 1980. A Study of the Vertical Vibrator Source for Seismic Profiling in the Yucca Mountain Area of the Nevada Test Site, Master of Engineering thesis, ER 2429, Colorado School of Mines, 58 p.
- Brocher, T.M., and P.E. Hart, 1988. "Direct Comparison of Dynamite and Vibroseis Methods for Deep Crustal Seismic Reflection Studies in the Southwestern Basin and Range, Nevada, USA" (poster), EOS, Transactions, American Geophysical Union, vol. 69, no. 44, p. 1313.
- Brocher, T.M., P.E. Hart, and S.F. Carle, 1989. Feasibility Study of the Seismic Reflection Method in Amargosa Desert, Nye County, Nevada, U.S. Geological Survey Open File Report, in preparation.
- deVoogd, B., et al., 1986. "Death Valley Bright Spot: A Midcrustal Magma Body in the Southern Great Basin, California?," Geclogy, vol. 14, p. 64-67, January 1986.
- Hasbrouck, W.P., 1987. "Hammer-Impact, Shear-Wave Studies," in Danbron, S.H., and S.N. Domenico (eds.). <u>Shear Wave Exploration</u>, Geophysical Development Series, vol. 2, Society of Exploration Geophysicists, p. 97-121.
- Hasbrousk, W.P., 1988. Five Shallow Seismic Feasibility Surveys, Expanded Abstracts, Society of Exploration Geophysicists, 58th Annual International Meeting and Exposition, Oct. 30-Nov. 3, 1988, p. 286-289.
- Hauser, E.C., et al., 1987. "COCORP Arizona Transect: Strong Crustal Reflections and Offset Moho Beneath the Transition Zone," <u>Geology</u>, vol. 15, p. 1103-1106.
- Howard, M.S., and S.H. Danbom, 1983. "Random Noise Example," in Bailey, A.W. (ed.), <u>Seismic Expressions of Structural Styles</u>, Vol. I, American Association of Petroleum Geologists Studies in Geology, ser. 15, p. 25.
- Klemperer, S.L., T.A. Hauge, E.C. Hauser, J.E. Oliver, and C.J. Potter, 1986. The Moho in the Northern Basin and Range Province, Nevada, Along the COCORP 40°N Seismic Reflection Transect, Geological Society of America Bulletin, vol. 97, p. 603-618.
- Knuepfer, P.L.K., et al., 1987. Crustal Structure of the Basin and <u>Range-Sierra Nevada Transition from COCORP Deep Seismic-Reflection</u> <u>Profiling</u>, Geological Society of America Bulletin, vol. 98, p. 488-496.

- McArthur, R.D., and N.R. Burkhard, 1986. <u>Geological and Geophysical</u> <u>Investigations of Mid Valley</u>, Lawrence Livermore National Laboratory, UCID-20740, 92 p.
- McGovern, T.F., L.W. Pankratz, and H.D. Ackermann, 1983. An Evaluation of Seismic Reflection Studies in the Yucca Mountain Area, Nevada Test Site, U.S. Geological Survey Open File Report 83-912, 57 p.
- Serpa, L., B. de Voogd, L. Wright, J. Willemin, J. Oliver, E. Hauser, and B. Troxel, 1988. Structure of the Central Death Valley Pull-Apart Basin from <u>COCORP profiles in the Southern Great Basin</u>, Geological Society of America Bulletin, vol. 100, p. 1437-1450.
- USGS (U.S. Geological Survey), 1988. <u>Geologic and Hydrologic Investigations</u> of a Potential Nuclear Waste Disposal Site at Yucca Mountain, Southern <u>Nevada</u>, M.D. Carr and J.C. Yount (eds.), U.S. Geological Survey Bulletin 1790, 152 p.

.

-0 )0

#### 2.6 REMOTE SENSING AND AERIAL GAMMA-RAY SURVEYS

This section addresses the two topics separately; each topic is followed by its own bibliography.

#### 2.6.1 REMOTE SENSING

#### OBJECTIVES

Remote sensing techniques are based on airborne or satellite-based imaging of the surface, at electromagnetic frequency levels ranging from gamma ray through the visible, and thermal infrared to radio frequencies. Digital image data are acquired and processed by computer into various data sets, including synthetic-aperture radar image mosaics for geomorphic and structural mapping, and Landsat Multispectral Scanner (MSS) and thematic mapper images for spectral discrimination for lithologic mapping. Major objectives for the use of remote sensing in the Yucca Mountain Project are (1) characterization of surface hydrology, (2) exploration for indications of hydrothermal mineralization, and (3) fracture-pattern mapping for structural and tectonic studies.

#### DATA COVERAGE AND QUALITY

A list of currently available remote sensing data and interpretive work of different types, applicable to studies of the Yucca Mountain region, follows. The text refers to map quadrangles which correspond to the generally available, 1° x 2° series of maps at 1:250,000 distributed by the USGS.

- The USGS Remote Sensing Tape Library (item 36 of bibliography) includes a complete set of digital tapes of all spectral bands of the Landsat MSS coverage of Nevada. The digital tapes are of high quality with little or no cloud cover for most of the State. Concatenation and Applicon or Calcomp printing of MSS scenes for four 2° quadrangles at 1:250,000 (Caliente, Las Vegas, Death Valley, and Goldfield) constitute a database suitable for spectral band ratioing for research purposes.
- 2. The USGS Remote Sensing Tape Library also includes a complete set of digital Landsat Thematic Mapper (TM) tapes of the four quadrangles listed above, consisting of six scenes, mostly cloud free and of good quality. They are suitable for concatenation to fit selected quadrangles at scales greater than 1:250,000 (e.g., suitable for fitting the Beatty quadrangle at 1:100,000 or greater). Concomitant spectral band-ratio techniques are available for the TM data sets.
- 3. Selected airborne thermal infrared and multispectral (TIMS) data are available for a very few selected areas within the State and could be useful. See items (6) (Osgood Mountains), and (10) and (18) (Carlin district) from the bibliography; good volcanic terrain data

are available; see (18) for the El Dorado Mountains southeast of Las Vegas.

- 4. Publications on the occurrences of limonitic surface materials and, in places, their relationship to thermal and hydrothermal alteration zones, include items (1), (5), (8), (9), (19), (20), (22), (23), (26), (27), (28), (30), and (34) of the bibliography. Many of these items cover the Walker Lake 2° quadrangle. A composite map of the occurrences of limonitic surface materials displayed on a Landsat MSS mosaic for the entire State of Nevada, without discrimination as to genesis of the iron-oxide bearing surfaces, has been completed as a map data set and printed experimentally, but has not yet been published.
- Publications on lineamentation, and more specifically on fracturepattern analyses and their relationship to geologic structure and mineralization, include items (13), (14), (15), (16), (17,, (23), (29), (32), and (33) of the bibliography.
- Publications dealing with spectral discrimination of rock types and related topics include items (1), (2), (3), (4), (5), (7), (8), (9), (10), (18), (19), (20), (21), (22), and (34) of the bibliography.
- Publications on remote sensing technology and computer techniques include items (11), (12), (24), (25), and (35) of the bibliography.
- 8. Complete synthetic aperture radar (SAR) mosaics are available through the USGS Branch of Geophysics for the Caliente and Las Vegas 2° quadrangles at 1:250,000 with the exception of classified areas. The mosaics are fitted to USGS map bases at 1:250,000 and are of high clarity; they are of exceptional value in geomorphic and structural mapping. Although slant-to-ground range corrections have been made, radar shadows and layover problems remain in areas of high relief.

#### SUMMARY OF RESULTS

The following materials and data sets, currently available to the Branch of Geophysics, are suitable for interpretative investigations of the Yucca Mountain region.

- Complete coverage of all spectral bands of MSS coverage of the entire region, suitable for concatenation, spectral band ratioing, and discrimination of iron-oxide surface materials at scales of 1:500,000 with clear detail, and at 1:250,000 at the margin of good resolution.
- TM coverage of the entire region, with space ' resolution at two times that of the MSS coverage (hence well-sur. 4 for geologic mapping and fracture pattern investigations, as wel' as spectral discrimination of desert varnish surfaces).

3. Flightline radar (SAR) data tapes for all four 2° 1:250,000 quadrangles: Caliente, Las Vegas, Death Valley, and Goldfield. Synthetic aperture radar mosaics are available for the Las Vegas quadrangle, and can be concatenated for the Caliente quadrangle. The mosaics are not yet available for Death Valley and Goldfield quadrangles.

Digital multispectral and stereo coverage tapes exist for at least the Yucca Mountain regions; their acquisition by the Yucca Mountain remote Insing project would entail expenditures of more than \$10,000. Small areal coverage of airborne thermal infrared coverage exists, but the acquisition of new thermal infrared coverage is best if carefully planned and integrated with ground control and experimentation (e.g., aerial IR coverage could be suitable for mapping areas of infiltration and discharge if it is acquired during night or very early morning hours within a few hours after a sizable rainfall event).

Interpretative studies already published are cited in the bibliography.

#### BIBLIOGRAPHY FOR SECTION 2.6.1

- Abrams, M.J., R.P. Ashley, L.C. Rowan, A.F.H. Goetz and A.B. Kahle, 1977. "Mapping of Hydrothermal Alteration in the Cuprite Mining District, Nevada, Using Aircraft Scanner Images for the Spectral Region 0.46 to 2.36 m," <u>Geology</u>, vol. 5, no. 12, p. 713-718.
- Bailey, G.B., J.L. Dwyer, and M.H. Podwysocki, 1985. "Evaluation of Landsat Thematic Mapper Data for Geologic Mapping in Semi-Arid Terrains," in <u>International Symposium on Remote Sensing of Environment</u>, Thematic Conference on Remote Sensing for Exploration Geology, 4th, Colorado, 1985, Proceedings, Ann Arbor, Michigan, Environmental Research Institute of Michigan, p. 781-782.
- 3. Brickey, D.W., 1987. "The Use of Thematic Mapper Imagery for Mineral Exploration in the Sedimentary Terrane of the Spring Mountains, Nevada," in International Symposium on Remote Sensing of Environment, Thematic Conference on Remote Sensing for Exploration Geology, 5th, Reno, Nevada, Sept. 29-Oct. 2, 1986, Proceedings, Ann Arbor, Michigan, Environmental Research Institute of Michigan, p. 607-613.
- 4. Hunt, G.R., 1979. Spectra of Rocks and Soils from the Eastern Shoshone Range, Nevada, U.S. Geological Survey Open File Report 79-707, 4 p.
- 5. Knepper, D.H., Jr., 1988. "Distribution of Exposed Limonitic Rocks and Soils from Landsat Multispectral Scanner Data on the Southern Ute Indian Reservation, Southwestern Colorado," in Fassett, J.E. (ed.), <u>Geology and Coal-Bed Methane Resources of the Northern San Juan Basin, Colorado and New Mexico</u>, Rocky Mountain Association of Geologists Guidebook, p. 295-296.

- Krohn, M.D., 1984. "Interpretation of Thermal-Infrared Multispectral Scanner (TIMS) Images of the Osgood Mountains, Nevada," in <u>International</u> <u>Symposium on Remote Sensing of Environment</u>, Thematic Conference on Remote Sensing for Exploration Geology, Colorado Springs, Colorado, April 16-19, 1984, Proceedings (ext. abs.), Ann Arbor, Michigan, Environmental Research Institute of Michigan, vol. 2, p. 735-737.
- Krohn, M.D., 1986. "Spectral Properties (0.4 to 25 microns) of Selected Rocks Associated with Disseminated Gold and Silver Deposits in Nevada and 1. tho," Journal of Geophysical Research, B. 91, p. 767-783.
- Krohn, M.D., M.J. Abrams, and L.C. Rowan, 1978. <u>Discrimination of</u> <u>Hydrothermally Altered Rocks Along the Battle Mountain-Eureka, Nevada,</u> <u>Mineral Belt Using Landsat Images</u>, U.S. Geological Survey Open File <u>Report 78-585, 63 p.</u>
- 9. Kruse, F.A., 1986. "Digital Mapping of Alteration Zones in a Hydrothermal System Using Landsat Thematic Mapper Data -- An example from the Northern Grapevine Mountains, Nevada/California," in <u>International Symposium on Remote Sensing of Environment</u>, Thematic Conference on Remote Sensing for Exploration Geology, 5th, Reno, Nevada, Sept. 29-Oct. 2, 1986, Summaries, Ann Arbor, Michigan, Environmental Research Institute of Michigan, p. 51.
- 10. Kruse, F.A., S. Hummer-Miller, and K. Watson, 1987. Thermal Infrared Rem te Sensing of the Carlin Disseminated Gold Deposit, Eureka County, Nev. A, Symposium on Bulk Mineable Precious Metal Deposits of the Western United States, Sparks, Nevada, April 6-8.
- 11. Kruse, F.A., and G.L. Raines, 1984. "A Technique for Enhancing Digital Color Images by Contrast Stretching in Munsell Color Space," in <u>International Symposium on Remote Sensing of Environment</u>, Thematic Conference on Remote Sensing for Exploration Geology, 3d, Colorado, 1984, Proceedings, Ann Arbor, Michigan, Environmental Research Institute of Michigan, vol. 2, p. 755-760.
- Niblack, W., 1981. <u>Control Point Library Building System</u>, Photogrammetric Engineering and Remote Sensing, vol. 47, no. 12, p. 1709-1750.
- 13. O'Leary, D.W., and J.D. Friedman, 1981. "Toward a Workable Lineament Symbology," in O'Leary, D.W., and J.L. Earle (eds.), <u>International</u> <u>Conference (1) Basement Tectonics</u>, 3d, Proceedings, Denver, Colorado, Basement Tectonics committee, Publication 3, p. 29-31.
- O'Leary, D.W., J.D. Friedman, and H.A Pohn, 1976. Lineament, Linear, Lineation--Some Proposed New Standards for Old Terms, Geological Society of America Bulletin, vol. 87, p. 1463-1469.
- 15. O'Leary, D.W., J.D. Friedman, and H.A. Pohn, 1976. Lineament, Linear, Lineation--Some Proposed New Standards for Old Terms: Discussion and Reply, Geological Society of America Bulletin, vol. 88, p. 159-160.

- 16. O'Leary, D.W., J.D. Friedman, and H.A. Pohn, 1979. "Lineament and Linear, A Terminological Reappraisal," in <u>2d International Conference on</u> <u>Basement Tectonics</u>, Proceedings, Basement Tectonics Committee, Denver, <u>Colorado</u>, p. 571-576, 22 ref.
- 17. O'Leary, D.W., and H.A. Pohn, 1978. A Photogeologic Comparison of Skylab and Landsat Images of Southwestern Nevada and Southeastern California, National Technical Information Service N76 20598/LL.
- 18. Podwysocki, M.H., W.J. Ehmann, and D.W. Brickey, 1987. "Application of Combined Landsat Thematic Mapper and Airborne Thermal Infrared Multispectral Scanner Data to Lithologic Mapping in Nevada," in International Symposium on the Remote Sensing of Environment, 21st, Ann Arbor, Michigan, Oct. 26-30, Summaries (abs.), Environmental Research Institute of Michigan, p. 20.
- Podwysocki, M.H., M.S. Power, J.W. Salisbury, and O.D. Jones, 1984.
  "Evaluation of Low-Sun Illuminated Landsat-4 Thematic Mapper Data for Mapping Hydrothermally Altered Rocks in Southern Nevada," in <u>International Symposium on Remote Sensing of Environment</u>, Thematic Conference on Remote Sensing for Exploration Geology, 3d, Proceedings, Ann Arbor, Michigan, Environmental Research Institute of Michigan, vol. 2, p. 541-551.
- 20. Purdy, T.L., G.B. Bailey, and J.L. Dwyer, 1984. "Alteration and Geologic Mapping South-Central Nevada Using Thematic Mapper Data," in <u>International Symposium on Remote Sensing of Environment</u>, Thematic Conference on Remote Sensing for Exploration Geology, 4th, San Francisco, California, 1985, Proceedings (abs.), Ann Arbor, Michigan, Environmental Research Institute of Michigan.
- 21. Purdy, T.L., G.B. Bailey, and J.L. Dwyer, 1985. Landsat Multispectral Scanner and Landsat Thematic Mapper Images and Interpretations of Hannapah and Royston Hills, Nevada, U.S. Geological Survey Open File Report 85-461, 3 p., 21 slides.
- 22. Purdy, T.L., G.F. Brem, and W.E. Collins, 1984. Use of High-Spectral Resolution Airborne Spectroradiometer Data to Map Hydrothermally Altered Rocks in the Gabbs Valley Range, Nevada, U.S. Geological Survey Open File Report 84-563, 8 p.
- 23. Purdy, T.L. and L.C. Rowan, 1985. "Map Showing Distribution of Altered Rocks, Faults, and Linear Features in the Walker Lake 1° x 2° Quadrangle, Nevada and California," U.S. Geological Survey Miscellaneous Investigation Map I-1643.
- 24. Raines, G.L., 1977. "Digital Color Analysis of Color-Ratio Composite Landsat Scenes," in <u>International Symposium on Remote Sensing of</u> <u>Environment</u>, 11th, Proceedings, Ann Arbor, Michigan, Environmental Research Institute of Michigan, p. 1463-1472.
- Robinove, C.J., 1982. "Computation with Physical Values from Landsat Digital Data," <u>Photogrammetric Engineering and Remote Sensing</u>, vol. 48, no. 5, p. 781-784.

- 26. Rowan, L.C., M.D. Krohn, and T.L. Purdy, 1980. <u>Generalized Map of</u> <u>Occurrences of Limonitic Rocks in the Walker Lake 1° by 2° Quadrangle,</u> <u>Nevada-California</u>, U.S. Geological Survey Open File Report 80-232, scale 1:250,000.
- 27. Rowan, L.C., and T.L. Purdy, 1980. Preliminary Map Showing Distribution of Altered Rocks and Limonitic Unaltered Rocks in the Walker Lake 1° x 2° Quadrangle, Nevada-California, U.S. Geological Survey Open File Report 80-931, scale 1:250,000.
- 28. Rowan, L.C., and T.L. Purdy, 1981. "Definition of Alteration Belts in the Walker Lake, Nevada-California 1° x 2° Quadrangle (Summary)," in <u>International Geoscience and Remote Sensing Symposium Digest</u>, Institute of Electrical and Electronics Engineers, vol. 1, p. 325-330.
- 29. Rowan, L.C., and T.L. Purdy, 1984. "Map of the Walker Lake 1° x 2° Quadrangle, California and Nevada, Showing the Regional Distribution of Linear Features," U.S. Geological Survey Miscellaneous Field Studies Map MF-1382-F, 22 p. text, scale 1:250,000.
- 30. Rowan, L.C., and T.L. Purdy, 1984. "Map of the Walker Lake 1° x 2° Quadrangle, California and Nevada, Showing the Regional Distribution of Hydrothermally Altered Rocks," U.S. Geological Survey Miscellaneous Field Studies Map MF-1382-Q, 19 p. text, scale 1:250,000.
- 31. Rowan, L.C., and P.H. Wetlaufer, 1973. "Structural Geologic Analysis for Nevada Using ERTS-1 Images: A Preliminary Report," in <u>Symposium on</u> <u>Significant Results Obtained from Earth Resources Technology</u> <u>Satellite-1</u>, Proceedings, National Aeronautics and Space Administration, NASA SP-327, vol. 1, sec. A, p. 413-424.
- 32. Rowan, L.C., and P.H. Wetlaufer, 1979. <u>Geologic Evaluation of Major Landsat Lineaments in Nevada and Their Relationship to Ore Deposits</u>, U.S. Geological Survey Open File Report 79-544, 87 p.
- 33. Rowan, L.C., and P.H. Wetlaufer, 1981. "Relationship Between Regional Lineament System and Structural Zones in Nevada," <u>American Association</u> of Petroleum Geologists, vol. 65, no. 8, p. 1414-1432.
- 34. Rowan, L.C., P.H. Wetlaufer, A.F.H. Goetz, F.C. Billingsley and J.H. Stewart, 1974. Discrimination of Rock Types and Detection of <u>Hydrothermally Altered Areas in South-Central Nevada by the Use of</u> <u>Computer-Enhanced ERTS Images</u>, U.S. Geological Survey Professional Paper 883, 35 p.
- 35. Sawatzky, D.L., 1985. Programmer's Guide to REMAPP, Remote Sensing Array Processing Procedures, U.S. Geological Survey Open File Report 85-231, 16 p.
- 36. USGS (U.S. Geological Survey). <u>Geophysics Remote Sensing Tape Library</u>, U.S. Geological Survey, Branch of Geophysics, Denver, Colorado.

#### 2.6.2 AERIAL GAMMA-RAY SURVEYS OVER THE NEVADA TEST SITE

#### OBJECTIVES

Aerial and ground gamma-ray surveys measure the gamma radiation emitted by radicisotopes at and near the surface of the ground. The measurements are sensitive to radioisotopes that result from natural processes, primarily members of the uranium-238 decay series, thorium-232 decay series, and potassium-10, and to radicisotopes such as cesium-137 and cobalt-60 that result from nuclear activities. The distribution of natural radioisotopes reflects the geologic processes that formed that distribution and thereby enables the use of natural gamma-ray measurements in geologic mapping and mineral exploration. The distribution of artificial radioisotopes reflects the activities that formed them and the weather that controlled their distribution. Major objectives for the use of aerial and ground gamma-ray measurements with the Yucca Mountain Project are (1) preparation of natural radioelement (uranium, potassium, and thorium) maps from the existing USGS=DOE database; (2) map interpretation to determine their geologic content, including detection of fault traces where radon-222 anomalies may occur and detection of areas of hydrothermal alteration; (3) selective ground measurements with a portable quantitatively calibrated spectrometer at sites determined from interpretation of the aerial data; and (4) acquisition of aerial gamma-ray data concurrent with aeromagnetic surveying for selected areas where natural radioactivity targets are known to exist. Concealed faults may be revealed in natural gamma-ray data by the detection of lithologic discontinuities and by anomalous radon-222 occurring at the surface.

#### DATA COVERAGE AND QUALITY

From about 1955 through the Sedan tests of 1962, the USGS operated a total count gamma-ray system in aerial surveillance of Atomic Energy Commission (AEC) nuclear activities on the NTS. Results of the surveys were quickly supplied to AEC personnel by means of informal written communications and no formal publications or even open file reports were prepared. Davis and Reinnardt (items 3 and 4 in the bibliography) discuss the USGS total count system and calibration. EG&G began aerial surveillance of NTS nuclear activities in the early 1960's (items 1, 5, and 18), and continues today.

The DOE during 1974 to 1981 accomplished the National Uranium Resource Evaluation (NURE) program for the conterminous 48 states and Alaska, which included aerial gamma-ray spectrometry surveying for uranium exploration. The surveying was keyed to the National Topographic Map Series (NTMS) 1° x 2° quadrangles. Flightline spacing varied from 1 to 6 miles (1.6 to 9.6 km), dependent on known or probable occurrences of uranium mineralization. Of the four NTMS quadrangles that include the NTS, the Death Valley quadrangle (item 6) has 1-mile (1.6-km) spaced north-south flightlines (and includes the Project), and the Caliente (item 9), Goldfield (item 7), and Las Vegas (item 8) quadrangles have 3-mile (4.8-km) spaced east-west flightlines, excluding the southwest quarter of the Caliente quadrangle. The USGS has compiled all NURE gamma-ray data for the lower 48 states into a database that includes flightline and 3-km grid values for the apparent surface concentrations of uranium, potassium, and thorium. The conversion of aerial gamma-ray measurements to radioelement concentrations is enabled by calibration of aerial survey systems at sites of known radioelement concentrations (items 15, 17, and 26). The database permits map preparation for any feature that has geographic definition. Maps prepared to date include those for Nevada (item 11), New Mexico (items 12 and 13), Ohio (item 10), and Arizona (item 22). For this study flightline data from the USGS database would be used to prepare radioelement maps for the NTS area at 1.5-mile (2.4+km) grid cell size and for the Project area at 0.5-mile (0.8-km) grid cell size. Interpretation of maps for Nevada made from the database is discussed by (item 14).

The helicopter gamma-ray spectrometer system used by EG&G for NTS nuclear surveillance acquires spectral data with artificial and natural radioisotope components. These complex spectra are readily separated and reduced by digital computer. Repeat surveys to monitor leakage or other contamination at a repository site would supply data that would be used to determine whether atmospheric pressure-induced radon anomalies occur along faults in the vicinity of Yucca Mountain.

Aerial gamma-ray measurements have been used in geologic studies as described by Darnley (item 2) and Pitkin (item 21), and in mineral exploration as described by Gnojek and Prichystal (item 16), Killeen (item 19), and Yeates and others (item 25). Ground follow-up of aerial surveys is described by Pitkin and Huffman (items 23 and 24), and calibration of portable spectrometers for ground surveys is described by Grasty and Darnley (item 17) and Lovberg and Mose (item 20).

#### SUMMARY OF RESULTS

Airborne gamma ray data collected by the NURE project, and for NTS nuclear surveillance, are available over Yucca Mountain and surrounding areas. These data should be examined to determine their geologic content and to determine locations for ground surveys if warranted.

#### BIBLIOGRAPHY FOR SECTION 2.6.2

- Burson, Z.G., 1974. <u>Research Progress in Airborne Surveys of</u> <u>Terrestrial Gamma Radiation</u>, Workshop on Environmental Radiation, USAEC HASL, New York.
- Darnley, A.G., 1972. "Airborne Gamma-Ray Survey Techniques," in Bowie, S.H.U., M. Davis, and D. Ostle, (eds.), <u>Uranium Prospecting Handbook</u>, Institute of Mining and Metallurgy, London, p. 174-211.
- Davis, F.J., and P.W. Reinhardt, 1957. "Instrumentation in Aircraft for Radiation Measurements," <u>Nuclear Science and Engineering</u>, vol. 2, no. 6, p. 713-727.

- Davis, F.J., 1962. "Radiation Measurements Over Simulated Plane Sources," Health Physics, vol. 8, p. 233-243.
- Deal, L.J., et al., 1972. "Locating the Lost Athena Missile in Mexico by the Aerial Radiological Measuring System (ARMS)," <u>Health Physics</u>, vol. 23, p. 95-98.
- DOE (U.S. Department of Energy), 1979a. Aerial Radiometric and Magnetic Survey, Death Valley National Topographic Map, California and Nevada, U.S. Department of Energy Open File Report GJBX-164(79), vol. 2, 386 p.
- DOE (U.S. Department of Energy), 1979b. Aerial Radiometric and Magnetic Survey, Goldfield National Topographic Map, California and Nevada, U.S. Department of Energy Open File Report GJBX-66, vol. 2, 73 p.
- DOE (U.S. Department of Energy), 1979c. Airborne Gamma-Ray Spectrometer and Magnetometer Survey, Kingman and Las Vegas Quadrangles, California and Nevada, U.S. Department of Energy Open File Report GJBX-59(80), vol. 2, 192 p.
- 9. DOE (U.S. Department of Energy), 1980. Aerial Radiometric and Magnetic Survey, Caliente National Topographic Map, Nevada and Utah, U.S. Department of Energy Open File Report-52(80), vol. 2, 198 p.
- Duval, J.S., 1987. "Aerial Radiometric Color Contour Maps and Composite Color Map of Regional Surface Concentrations of Uranium, Potassium, and Thorium in Ohio," U.S. Geological Survey Geophysical Investigations Map GP-966.
- 11. Duval, J.S., 1988a. "Aerial Gamma-Ray Contour Maps of Regional Surface Concentrations of Potassium, Uranium, and Thorium in Nevada," U.S. Geological Survey Geophysical Investigations Map GP-982, 3 sheets, scale 1:750,000.
- Duval, J.S., 1988b. "Aerial Gamma-Ray Contour Maps of Regional Surface Concentrations of Uranium, Potassium, and Thorium in New Mexico," U.S. Geological Survey Geophysical Investigations Map GP-979, 3 sheets, scale 1:750,000.
- Duval, J.S., 1989. "Aerial Gamma-Ray Color Contour Maps of Regional Surface Concentrations of Potassium, Uranium, Thorium, and Composite-Color Maps of Uranium, Potassium, Thorium, and Their Ratios in New Mexico," U.S. Geological Survey Geophysical Investigations Map GP-980, 4 sheets, scale 1:1,000,000.
- 14. Duval, J.S., and J.A. Pitkin, 1988. Interpretation of Aerial Gamma-Ray Data for Nevada, U.S. Geological Survey Open File Report 88-288, 12 p., 9 sheets, scale 1:1,000,000.
- Geodata International, Inc., 1977. Lake Mead Dynamic Test Range for Calibration of Airborne Gamma Radiation Measuring Systems, U.S. Department of Energy Open File Report GJBX-46 (77), 83 p.

- Gnojek, Ivan, and A. Prichystal, 1985. "A New Zinc Mineralization Detected by Airborne Gamma-Ray Spectrometry in Northern Moravia (Czechoslovakia)," Geoexploration, vol. 23, no. 4, p. 491-502.
- Grasty, R.L., and A.G. Darnley, 1971. <u>The Calibration of Gamma-Ray</u> <u>Spectrometers for Ground and Airborne Use</u>, Geological Survey of Canada Report 71-17, 27 p.
- 18. Guillou, R.B., 1964. "The Aerial Radiological Measuring Surveys (ARMS) Program," in Adams, J.A.S., and W.M. Lowder, eds., <u>The Natural Radiation</u> Environments, Chicago University Press, p. 705-721.
- 19. Killeen, P.G., 1979. "Gamma-Ray Spectrometric Methods in Uranium Exploration - Application and Interpretation," in Hood, P.J. (ed.), <u>Geophysics and Geochemistry in the Search for Metallic Ores</u>, Geological Survey of Canada Economic Geology Report 31, p. 163-229.
- Lovberg, L., M. Lovberg, and E. Lovberg, 1987. "Counting Statistics in Radioelement Assaying with a Portable Spectrometer," <u>Geophysics</u>, vol. 52, no. 4, p. 555-563.
- Pitkin, J.A., 1968. <u>Airborne Measurement of Terrestrial Radioactivity</u> <u>as an Aid to Geologic Mapping</u>, U.S. Geological Survey Professional Paper 516-F, 29 p.
- 22. Pitkin, J.A., and J.S. Duval, in press. "Aerial Gamma-Ray Contour Maps of Regional Surface Concentrations of Uranium, Potassium, and Thorium] in Arizona," U.S. Geological Survey Geophysical Investigations Map GP-nnnn, 3 sheets, scale 1:1,000,000.
- 23. Pitkin, J.A., and A.C. Huffman, Jr., 1986. <u>Geophysical and Geological Investigations of Aerial Radiometric Anomalies in the Tabuk Formation, Kingdom of Saudi Arabia, U.S. Geological Survey, Saudi Arabian Mission Open File Report USGS-OF-05-10, 20 p., 3 plates, scale 1:250,000.</u>
- 24. Pitkin, J.A., 1989. <u>Geophysical and Geological Investigations of Aerial</u> <u>Radiometric Anomalies in the Phanerozoic Cover Rocks of Southern Saudi</u> <u>Arabia</u>, U.S. Geological Survey, Saudi Arabian Mission Open File Report USGS-OF-07-7, 17 p. 8 plates, scale 1:250,000.
- 25. Yeates, A.N., B.W. Wyatt, and D.H. Tucker, 1982. "Application of Gamma-Ray Spectrometry to Prospecting for Tin and Tungsten Granites, Particularly within the Lachlan Fold Belt, New South Wales," <u>Economic</u> <u>Geology</u>, vol. 77, p. 1725-1738.
- 26. Ward, D.L., 1978. Construction of Calibration Pads Facility, Walker Field, Grand Junction, Colorado, U.S. Department of Energy Open File Report GJBX-37(78), 57 p.

#### 2.7 BOREHOLE GEOPHYSICS AND PETROPHYSICAL MEASUREMENTS

#### OBJECTIVES

The objectives for borehole geophysical surveys, as stated in Section 8.3.1.4.2.1.3 of the Site Characterization Plan (SCP), are (1) to aid in the definition and refinement of the geometry and character of lithostratigraphic units and contacts between units, and (2) to determine the distribution of rock characteristics within lithostratigraphic units. Geophysical logs may be used as index tools for correlation between drillholes without reprocessing of log responses, or different log responses may be combined in a computed log that represents some directly applicable site parameter such as volumetric moisture content.

#### DATA COVERAGE AND QUALITY

More than 500 geophysical logs have been acquired since 1978 by the Yucca Mountain Project, from 40 holes in Yucca Mountain and vicinity (Figure 2.7-1). Table 2.7-1 lists the types of logs that are available for each existing borehole. Most of these were acquired by the contracted logging service operating on the NTS (Birdwell or its successor company, Dresser-Atlas). Logs have been presented for boreholes UE25a#1 (Spengler et al., 1979), USW G-1 (Muller and Kibler, 1983), USW G-4 (Spengler et al., 1984), UE-25p#1 (Muller and Kibler, 1984) and 15 boreholes in the WT series (Muller and Kibler, 1984). In addition, magnetometer logs have been acquired by Lawrence Livermore National Laboratory using methods described by Douglas and Millett (1978). Magnetic susceptibility and induced polarization logs have been acquired by the USGS in boreholes UE25a#1 (Hagstrum et al., 1980a), UE25a#5 (Hagstrum et al., 1980b), and UE25a#4, a#5, a#6, and a#7 (Daniels and Scott, 1981). Borehole gravity measurements have been acquired by the USGS in boreholes USW H-1 (Robbins et al., 1982), UE25p#1 and UE25c#1 (Healey et al., 1984), and USW G-4 (Healey et al., 1986). The compilation and presentation of logs from all 40 boreholes listed in Table 2.7-1 is the subject of a report which is in progress.

Acquisition and interpretation of geophysical logs from drillholes in the Yucca Mountain site area is complicated by several factors:

- 1. Drillholes penetrate both saturated and unsaturated rocks. Some logs require fluid in the wellbore, and cannot be run effectively in the unsaturated zone. Others operate with reduced effectiveness, or have never been calibrated for unsaturated conditions in tuffaceous rocks. For example, compensated neutron logging tools were generally designed to operate in fluid-filled holes, where stray neutrons are readily captured. As a result, the conventional compensated neutron tool has been used almost exclusively in the saturated zone at the site.
- Drillholes are often quite rugose (i.e., borehole geometry varies considerably from a cylindrical opening at the nominal bit diameter) in welded units such as the Topopah Spring Member. Drilling dry, or

with foam circulation in the unsaturated zone, may produce more severe rugosity because of the lack of supporting wellbore fluid. Also, for certain types of logging tools, the effects of rugosity on tool response may be accentuated without wellbore fluid. For example, the compensated density log is closely tied to formation bulk density, but contains low-density "spikes" caused primarily by rugosity. Severe rugosity is an environmental condition that adversely affects the response of most logging tools.

3. Drilling progresses in several stages to accommodate casing and measurement programs. Thus, the logs are acquired in stages, above or below the static water level, and often in holes of differing diameter. This requires added effort to select and splice logs from the various logging runs, and may introduce uncertainty with respect to tool response under varying logging conditions. This factor is represented in the various reports on the edited logs (e.g., Muller and Kibler, 1983, 1984a, and 1984b; Spengler et al., 1984; Hagstrum et al., 1980a).

Petrophysical measurements of core properties, on samples retrieved from logged drillholes, are very important to the logging program. Measurements of density, porosity, velocity, electrical properties, magnetic properties, and moisture saturation are generally needed to calibrate log measurements. Such a are available from core or cuttings samples, as applicable, from the nine boreholes listed in Table 2.7-2. In addition to the physical property data, mineralogical data on the tuffs acquired by X-ray diffraction has been summarized by Bish and Chipera (1989) for 17 boreholes. The mineralogy is expected to be quite valuable in relating log response to rock alteration.

The USGS Nuclear Hydrology Project has maintained a program of periodic relogging of shallow borings at Yucca Mountain since 1984, using neutron moisture meter tools. There currently are 74 such boreholes in the program, distributed among hydrologic settings that sample the variation of surficial materials and topography at the site. The depth of these borings is generally about 10 m, although many are shallower and a few are deeper. Different types of neutron tools have been used, and are run without centralization. The holes are logged once per month, and also after or during major precipitation, runoff, or snowmelt events. A program of crosshole gamma logging has also been carried out on a preliminary basis, in addition to the neutron logging. Crosshole logging is done in cased, parallel borings spaced about 2 m apart.

#### SUMMARY OF RESULTS

The first objective stated above consists of the task of correlation between drillholes using log character. Logs have been routinely used to confirm the location of bed boundaries based on changes in log response due to physical property changes. The compensated density log has been particularly useful, because density varies in a generally systematic way between welded and nonwelded units, and because the log is available for all drillholes in both the saturated and unsaturated zones. The magnetic





A Sparses

and the second

Hole	CAL	GR	SP	DBC	NBC	CVL	IND	RES	ENP	NNL	DIEL	KUT	MAG	MS	IP	FLOW
G-1	*	*	*	*		*	*	*	*	*						
6-2	*	*	*	*	*	*	*	*	*	*		*				
G-3	*	*	*	*	*	*	*	*	*	*		*	*	*	*	
G-4	*	*	*		*	*	*			*		*				*
B-1B	*	*	*	*	*	*	*	*	*			*				*
P-1	*	*	*	*	*	*	*	*	*		*	*	*			*
H-1	*	*	*	*	*	*	*	*	*	*		*	*			*
н-3	*	*	*	*	*	*	*	*	*	*		*				*
H-4	*	*	*	*	*		*	*	*			*	*			*
H-5	*	*	*	*	*		*	*	*			*	*			*
H-6	*	*	*	*	*		*	*	*			*	*			*
C-1	*	*	*	*	*	*	*	*	*		*	*				
C-2	*	*	*	*	*	*	*	*	*		*	*				
C-3	*	*	*	*	*	*	*	*	*		*	*				
WT-1	*	*	*	*			*	*	*		*	*				
WT-2	*	*	*	*			*	*	*		*	*				
WT-3	*	*	*	*			*	*	*		*	*				
WT-4	*	*	*	*			*	*	*		*	*				
WT-6	*	*	*	*			*	*	*		*	*				
WT-7	*	*	*	*			*	*	*		*	*				
WT-10	*	*	*	*			*	*	*		*	*				
WT-11	*	*	*	*			*	×	*		*	*				
WT-12	*	*	*	*			*	*	*		*	*				
WT-13	*	*	*	*			*	*	*		*	*				
WT-14	*	*	*	*			*	*	*		*	*	*	*		
WT-15	*	*	*	*			*	*	*		*	*				
WT-16	*	*	*	*			*	*	*		*	*				
WT-17	*	*	*	*			*	*	*		*	*				
WT-18	*	*	*	*			*	*	*		*	*				
UZ-1	*	*	*	*			*	*	*		*	*				
112-6	*	*	*	*			*	*	*		*	*	*	*		

Table 2.7-1. Log types in each drillhole

Hole		CAL	GR	SP	DBC	NBC	CVL	IND	RES	ENP	NNL	DIEL	KUT	MAG	MS	IP	FLOW
25-A1		*	*	*	*		*		*						*	*	
25-A4		*	*		*			*	*	*					*	*	
25-A5		*	*	*	*			*	*	*					*	*	
25-A6		*	*	*	*				*						*	*	
25-A7		*	*	*	*			*	*	*					*	*	
J-13		*	*	*			*		*		*						
A-3																	
VH-1		*	*	*	*		*	*	*	*	*						
VH-2		*	*	*	*		*	*	*	*			*				
Key:	CAL CVL DBC DIEL ENP FLOW GR IND IP KUT MAG MS NBC NNL BEC	Caliper Compensated sonic velocity log Density borehole compensated Dielectric log Epithermal neutron porosity Flowmeter log (saturated zone) Gamma ray Induction Induced polarization Spectral gamma log Total magnetic intensity Magnetic susceptibility Neutron borehole compensated Neutron-neutron log Electrical resistivity															

# Table 2.7-1. Log types in each drillhole (Continued)
Hole	Bulk density	Grain density	Porosity	Water saturation	Resistivity	Induced polarization	Magnetic properties	Sonic velocity
G-1	*	*	*				*	
G-2	*	*	*				*	
G-3/GU-3	3 *	*	*		*	*	٠	*
G-4	*	*	*		*	•		•
b∰lH	*	*	*	*				
a#1	*	*	*		*	*	*	*
H-1	*	*	*	*				
VH-1							*	
a#3	*	*	*		*	*	*	*

idule 2.1 2. Ittlophysical abta trong of the	Tab]	le 2	.7-	-2.	Petro	ophy	/sical	data	from	core	anal	LYS1
--	------	------	-----	-----	-------	------	--------	------	------	------	------	------

intensity and magnetic susceptibility logs provide unique lithostratigraphic information corresponding to strong remanent magnetization, and laterally contiguous compositional variability in the host rock and other ash flows. Large reversed polarization signatures in the Bullfrog Member of the Crater Flat Tuff and normal polarization signatures in the Tram Member serve as marker horizons which can be traced from one drillhole to the next. Magnetic susceptibility logs are sensitive to the amount of magnetite and the size of magnetite crystals (Hagstrum et al., 1980b), and apparently reveal distinct eruptive events within thick ash flow sheets. The magnetic logs have been run on an experimental basis at Yucca Mountain, so few logs are presently available.

The gamma-ray log, which serves for lithologic correlation in sedimentary basins, is not as effective in the tuffs. It shows that the Topopah Spring Member of the Paintbrush Tuff is characterized by a relatively lather and unvarying gamma-ray signature, and that the Prow Pass Member of the Crass Flat Tuff can be correlated between drillholes by a distinctive, blocky signature near the top of the unit.

The degree of welding, which can be described from observation of core, produces characteristic responses in the density and resistivity logs. The density log almost always indicates low density in bedded tuffs and in nonto partially-welded ash flow tuffs. Density increases as welding becomes moderate or dense. Superposed on the response to welding is an increase of density with increasing depth.

Resistivity of the Topopah Spring Member is high, usually greater than 200 ohm-meters, not only because the unit is unsaturated, but also because it is densely welded with consequent restriction in pore connectivity. By contrast, both the density and resistivity of the underlying Calico Hills unit are significantly lower. The relative differences between welded and nonwelded units are evident whether measurements are made above or below the present water table, and are also consistent in drillholes distributed in the northern and southern portions of the site area. The nature of alteration may also produce characteristic log responses in the various units, particularly the Calico Hills tuffs, but present understanding of this phenomenon does not permit logs to be used to characterize alteration.

The second objective described above involves quantification of physical properties that can be derived from logs, particularly density, velocity, resistivity, porosity, moisture content, and related characteristics such as pore connectivity, which may affect transport properties. Much use has already been made of geophysical logs in an unsystematic manner, as input for interpretation of surface or airborne geophysical surveys. A quantitative, systematic approach to log analysis is computationally intensive because corrections are needed for environmental effects, logs must be compared with core measurements, and logs must be compared together to ensure consistent interpretation. Work towards this second objective is currently in progress.

Extensive measurements of bulk and grain density, porosity, compressional velocity, resistivity, and induced polarization have been made on rock cores from eight drillholes at Yucca Mountain (Anderson, 1981a, 1981b, and 1984). Water permeability, magnetic susceptibility, and remanent magnetization have also been measured for samples from certain drillholes. Bulk density, resistivity, and sonic velocity are correlated to porcesity, indicating the dependence of these properties on textural rat. at than compositional differences (Anderson, 1982). There are significant departures from simple dependence, probably associated with the effects of alteration processes such as devitrification and zeolitization. Resistivity measurements on resaturated core samples show that the natural pore waters are quite fresh. Polarization measurements show that frequency effect cannot be relied upon as an indicator of clay or zeolite content. Low porosity tuff with few conduction paths and low content of polarizable material may exhibit polarization response comparable to high porosity tuff rich in clay minerals (Anderson, 1984). Specific capacity (or the frequency domain equivalent "metal factor") is a more reliable indicator of clay or zeolite content, but not without apparent conduction path effects.

### Fracture Detection/Characterization

Fracture location and orientation data have been obtained from boreholes UE-25 c#1 and UE-25 c#2 using downhole video camera and borehole accustic televiewer (BATV) logs. In UE-25 c#3 only the BATV log was obtained. Boreholes drilled dry or with air foam at the site have generally provided sufficient optical clarity for use of borehole television both above and below the water table. Flow data obtained under pumping conditions (Erickson et al., 1985) and nonpumping conditions (Galloway and Erickson, 1985) indicate that fluid production occurs from fractures, but the majority of fractures observed do not contribute measurable flow. Flow is characterized by relatively large quantities of water entering the boreholes at a few discrete points or intervals.

Data on fracture locations and orientations have been obtained from downhole video camera and BATV logs in boreholes USW H-1, USW H-3, USW H+4, USW H-5, USW H-6, UE-25 b#1, and UE-25 p#1 (Craig and Robison, 1984; Erickson and Waddell, 1985; Lahond et al., 1984; Rush et al., 1983; Whitfield et al., 1985). Relations between fractures and intraborehole flow at these test wells are similar qualitatively to relations at boreholes UE+25 c#1, c#2, and c#3. As efforts to relate fracture data to hydrologic information at c#1, c#2, and c#3 prove successful, these efforts will be extended to other test wells near Yucca Mountain Borehole Gravity.

Borehole gravity data have been acquired from five drillholes (UE25 p#1, UE25 c#1, USW G-3, USW G-4, and USW H-1) for the purpose of delineating large-scale density contrasts and refining density information used in gravity modeling (Healey et al., 1984, 1986; Robbins et al., 1982). Gravity stations have been acquired at 7.6-m stations over much of the length of these drillholes, and is of sufficient resolution and accuracy (in conjunction with free-air and terrain corrections) such that 10% or more lithophysal porosity in a layer as thin as 7.6 m should be detectable.

### Neutron Moisture Meter Logging

As many as 74 shallow borings have been logged periodically over the past five years, principally using hand-held borehole neutron moisture meters. The purpose of this logging program is to monitor the frequency and extent of natural infiltration in different surface hydrogeologic settings at the site. The tools are read on station at 10 to 20 cm depth increments, and at monthly intervals or more frequently in response to precipitation events that are judged likely to produce measurable infiltration.

The borings were drilled dry, with 125-mm steel casing uncemented except for a grout seal at the ground surface. Although smaller diameter borings and aluminum casing are standard practice in moisture logging, steel casing was selected for compatibility with the available ODEX drilling system. This system is an effective method for dry drilling in variable alluvium with abundant boulders. Calibration tank studies discussed below show that useful sensitivity is retained with the method used. Recently, other drilling methods have been tried, or become available, including small diameter ODEX tools. These provide smaller borings and support the use of thinner wall casing. Steel casing remains a requirement for strength, but does not appear to degrade performance significantly. The moisture meter is run uncentralized, and consists of a 50 mCi Am/Be source with a closely spaced detector of thermal neutrons. The moisture meter effectively provides a point measurement of the moderating and capture cross sections of the formation, which are predominantly controlled by moisture content in unsaturated volcanic tuff.

Calibration tank studies and independent moisture measurements in boreholes indicate that, as applied in the program, the logging method can resolve changes in neutron counts on the order of 2%, which translates roughly into resolution of about 2% on the volumetric amount of water present.

The following general observations apply to moisture transients, observed primarily in alluvium: (1) transients corresponding to precipitation or snowmelt events can be observed in many borings, (2) the donward progress of such transients can typically be monitored over tens to hundreds of days, and (3) the magnitude of such transients typically diminishes with depth. Indications of more rapid infiltration in response to precipitation events have been obtained from a small minority of the existing neutron holes. Although the number of observations is limited, instances of water collecting at the bottom of neutron holes are consistent with perching above the bedrock-alluvium contact. (This paragraph derives from verbal communication, A. Flint, 2-12-90.)

The maximum measured depth of penetration of moisture transients is typically 5 to 10 m for the existing borings, and no transients have been detected below 12 m. It is important to observe that most of the existing neutron holes are collared in alluvium, and extend just beyond the bedrock-alluvium contact. The depth of neutron holes thus varies from less than 10 m to 15 m or more. Crosshole gamma-gamma surveys have been run periodically using a small number of closely spaced borings. The response of single-hole tools is apparently consistent with crosshole measurements, which are sensitive to a larger volume.

### Other Slim-Hole Moisture Logging

Several of the shallow infiltration monitoring boreholes have been logged with a Mount Sopris single-conductor logging system to investigate through-casing logging methods suitable for the UZ hydrology program. Three tools were used: short- and long-spacing neutron tools, and a singledetector density tool. These were found to be inadequate for measuring water content and density with useful accuracy. As expected, the single-detector density tool was sensitive to borewall irregularity behind the casing. Data from the short-spacing neutron tool compared favorably to that from the neutron moisture meter logging program, and exhibited similar borehole effects. The long-spacing neutron tool also exhibited both sensitivity to moisture content and borehole effects. (A dual-spacing neutron tool is available from this source and will be tried.)

### Related Infiltration Data from Borehole Geophysics

Infiltration penetrating deeper than 12 m has been inferred for only one borehole, from geotemperature data. (Note: heat flow studies are not covered by this paper.) Sass et al. (1988) discussed repeat temperature logs from borehole UE25 a#7, in the bottom of Drill Hole Wash about 0.5 km east of the planned location of the Exploratory Shaft Facility. This borehole wa: drilled in August 1979 with mud, some of which was lost to the formation. The hole was completed with a string of water-filled steel tubing for temperature logging. Repeated temperature logs (March, 1981; April, 1981; December, 1981; March, 1983; and March, 1984; see Sass et al., 1988) show the hole to have equilibrated conductively from the effects of drilling (a typical response for Yucca Mountain boreholes) until early 1983. The tomperature profile acquired in March 1983, shortly after a major precipitation event, is disturbed down to about 120 m depth. The March 1984 profile shows the disturbance to have equilibrated, such that the profile resembles that acquired soon after drilling. Fluid evidently entered the borehole, which is situated within the ephemeral channel in Drill Hole Wash. However, it is not known whether fluid entered the borehole through (1) the top of the surface casing as a result of overtopping by the runoff: (2) the alluvium at the bottom of the surface casing, above the bedrock contact; or (3) fracture pathways beneath the channel and intersecting the borehole at depth.

### BIBLIOGRAPHY FOR SECTION 2.7

- Anderson, L.A., 1981. Rock Property Analysis of Core Samples from the Calico <u>Hills UE25 a#3 Borehole, Nevada Test Site, Nevada</u>, U.S. Geological Survey Open File Report 81-1337.
- Anderson, L.P., 1981. <u>Rock Property Analysis of Core Samples from the Yucca</u> <u>Mountain UE2 a#1 Borehole, Nevada Test Site, Nevada</u>, U.S. Geological Survey Open File Report 81-1338.
- Anderson, L.A., 1982. <u>Rock Property Analysis for Core Samples from Yucca</u> <u>Mountain Boreholes, Nevada Test Site, Nevada (abs), EOS Transactions of the</u> Fall, 1982 meeting of the American Geophysical Unior, San Francisco, California, p. 1111.
- Anderson, L.A., 1984. Rock Property Measurements on Large-Volume Core Samples from Yucca Mountain USW GU-3/G-3 and USW G-4 Boreholes, Nevada Test Site, Nevada, U.S. Geological Survey Open File Report 84-552, 39 pp.

- Bish, D.L., and S.J. Chipera, 1989. Revised Mineralogic Summary of Yucca Mountain, Nevada, Los Alamos National Laboratory Report LA-11497-MS.
- Craig, R.W., and J.H. Robison, 1984. <u>Geohydrology of Rocks Penetrated by</u> <u>Test Well UE25 p#1, Yucca Mountain Area, Nye County, Nevada</u>, U.S. <u>Geological Survey Open File Report 84-4248, 57 p.</u>
- Daniels, J.J., and J.H. Scott, 1980. Borehole Geophysical Measurement for Hole UE25a-3, Nevada Test Site, Nuclear Wasto Isolation Program, U.S. Geological Survey Open File Report 80-126, 30 p.
- Daniels, J.J., J.H. Scott, and J.T. Hagstrum, 1981. <u>Interpretation of</u> <u>Geophysical Well-Log Measurements in Drillholes UE25 a-4, -5, -6, and -7,</u> <u>Yucca Mountain, Nevada Test Site</u>, U.S. Geological Survey Open File Report 81-615.
- Douglas, A.C., and M.R. Millett, 1978. Total Intensity Magnetometer Logging as a Stratigraphic Tool in Tertiary Volcanic Rock, Lawrence Livermore National Laboratory Report UCRL=52617.
- Erickson, J.R., D.L. Gall way, and K. Karasaki, 1985. Interpretation of Falling-Head Injection lest Data for Fractured Volcanic Tuffs, Yucca Mountain, Nevada Test Site, Geological Society of America, Fall 1985 Teeting (Orlando, Florida).
- Son, J.R., and R.K. Waddell, 1985. Identification and Characterization Hydrologic Properties of Fractured Tuff Using Hydraulic and Tracer Tests --Test Well USW H-4, Yucca Mountain, Nye County, Nevada, U.S. Geological Survey Water Resources Investigations Report 85-4066, 55 p.
- Galloway, D.L., and J.R. Erickson, 1985. Tracer Test for Evaluating Nonpumping Intraborehole Flow in Fractured Media, Transactions, American Nuclear Society, vol. 50, Nuclear Techniques for Hydrogeological Studies, p. 182-192.
- Hagstr m, J.T., J.J. Daniels, and J.H. Scott, 1980a. <u>Interpretation of</u> <u>Geophysical Well-Log Measurements in Drillhole UE25a-1, Nevada Test Site,</u> <u>Radioactive Waste Program</u>, U.S. Geological Survey Open File Report 80-941, 32 p.
- Hagstrum, J.T., J.J. Daniels, and J.H. Scott, 1980b. Analysis of the Magnetic Susceptibility Well Log in Drillhole UE25 a-5, Yucca Mountain, Nevada Test Site, U.S. Geological Survey Open File Report 80-1263, 32 pp.
- Healey, D.L., F.G. Clutsom, and D.A. Glover, 1984. Borehole Gravity Meter Surveys in Drillholes USW G-3, UE25p#1, and UE25c#1, Yucca Mountain Area, Nevada, U.S. Geological Survey Open File Report 84-672, 16 p.
- Healey, D.L., F.G. Clutsom, and D.A. Glover, 1986. <u>Borehole Gravity Meter</u> <u>Survey in Drillholes USW G-4, Yucca Mountain Area, Nye County, Nevada</u>, U.S. Geological Survey Open File Report 86-205, 18 p.
- Lahund, R.G., D.H. Lobmeyer, and M.S. Whitfield, Jr., 1984. Geohydrology of Science Tuff Penetrated by Test Well UE-25 b#1, Yucca Mourcain, Nye

County, Nevada, U.S. Geological Survey Water Resources Investigation Report 84-4253, 44 p.

- Muller, D.C., and J.E. Kibler, 1983. <u>Commercial Geophysical Logs from the</u> <u>USW G-1 Drillhole, Hevada Test Site, Nevada</u>, U.S. Geological Survey Open File Report 83-321, 7 p.
- Muller, D.C., and J.E. Kibler, 1984a. Preliminary Analysis of Geophysical Logs from Drillhole UE25 p#1, Yucca Mountain, Nye County, Nevada, U.S. Geological Survey Open File Report 83-321, 7 p.
- Muller, D.C., and J.E. Kibler, 1984b. Preliminary Analysis of Geophysical Logs from the WT Series of Drillholes, Yucca Mountain, Nye County, Nevada: U.S. Geological Survey Open File Report 86-46, 30 p.
- Nimick, F.B., and B.M. Schwartz, 1987. Bulk, Thermal, and Mechanical Properties of the Topopah Spring Member of the Paintbrush Tuff, Yucca Mountain, Nevada, Sandia National Laboratories Report SAND85-0762, UC-70.
- Robbins, S.L., J.W. Schmoker, and T.C. Hester, 1982. Principal Facts and Density Estimates for Borehole Gravity Stations in Exploratory Wells Ue4ah, Ue7j, Ue1h, Ue1g, Ue2co, and USW H-1 at the Nevada Test Site, Nye County, Nevada, U.S. Geological Survey Open File Report 82-277, 33 p.
- Rush, F.E., W. Thordarson, and L. Bruckheimer, 1983. <u>Geohydrology of Test</u> <u>Well USW H-1, Yucca Mountain, Nye County, Nevada</u>, U.S. Geological Survey Water Resources Investigation Report 83-4032, 38 p.
- Sass, J.H., A.H. Lachenbruch, W.W. Dudley, Jr., S.S. Priest, and R.J. Munroe, 1988. Temperature, Thermal Conductivity, and Heat Flow Near Yucca Mountain, Nevada: Some Tectonic and Hydrologic Implications, U.S. Geological Survey Open File Report 87=649.
- Spengler, R.W., D.C. Muller, and R.B. Livermore, 1979. Preliminary Report on the Geology and Geophysics of Drillhole UE25a-1, Yucca Mountain, Nevada Test Site, U.S. Geological Survey Open File Report 79-1244, 43 p.
- Spengler, R.W., M.P. Chornack, D.C. Muller, and J.E. Kibler, 1984. Stratigraphic and Structural Characteristics of Volcanic Rocks in Core Hole USW G-4, Yucca Mountain, Nye County, Nevada with a Section on Geophysical Logs, U.S. Geological Survey Open File Report 84-789, 77 p.

Whitfield, M.S., Jr., E.F. Eshom, W. Thordarson, and D.H. Schaeter, 1985. <u>Geohydrology of Rocks Penetrated by Test Well USW H-4, Yucca Mountain, Nye</u> <u>County, Nevada</u>, U.S. Geological Survey Water Resources Investigation Report 85-4030, 33 p.



## OBJECTIVES

Teleseismic tomography has been used in the Yucca Mountain site area and region primarily for detection of extant magma chambers larger than about 4 km across, and to help map deep tectonic structures that might be important to repository performance. These studies are directly pertinent to NRC staff concerns expressed to the DOE (e.g., Comments #49 and #51 of the "Point Papers" response to the Consultation Draft Site Characterization Plan). Teleseismic tomography is routinely successful for delineating silicic magma chambers in the crust and partial melts in the upper mantle. Iyer (1987) reviews teleseismic tomography results and other seismic studies for a large range of volcanic systems.

Structures affecting the repository may be of almost any size. Regional teleseismic tomography delineates upper-mantle structures that may be related to plate motions and large-scale magmatism, such as the Miccene calderas of the Timber Mountain-Silent Canyon complex. Regions of upper-mantle partial melt that could supply basalt lava to eruptions near the repository can be delineated at this scale as well. At finer scales, major faults offsetting the Paleozoic basement beneath the tuff section have discernible teleseismic signatures because of the strong velocity contrast. Using this signature, buried faults can be located. Crustal silicic magma chambers larger than about 4 km across also can be imaged using fine-scale data. Magma chambers this size and larger have been imaged routinely below active silicic volcanic centers in the western U.S. The absence of low-velocity features this size at Yucca Mountain would be a positive result indicating diminished likelihood of silicic eruptions in the site area.

### DATA COVERAGE AND QUALITY

Data are available for the Yucca Mountain area at three scales (Figure 2.8+1). First, data from the regional Southern Great Basin (SGB) network provide a large array aperture, and therefore great maximum depth of imaging, but the stations are far apart, limiting the resolving power of the network. The data of Monfort and Evans (1982), combined with a completed but unpublished data set acquired by J.R. Evans, provide 26-km lateral resolution to a depth of about 345 km in all of southern Nevada and neighboring parts of southeast California. The major findings of Monfort and Evans (1982) are not changed by these new data, but resolution characteristics are improved.

Second, later addition to the SGB network of six stations around Yucca Mountain itself (Figure 2.8-1c, five solid symbols in site area and one just west), combined with temporary deployment of 20 portable seismographs (Figure 2.8-1c, open symbols), created a circular array of seismographs with mean station spacing of about 4 km within a 15-km radius around the repository. This array can resolve objects as small as 4 km across and probe to Moho depth or slightly below. This temporary deployment was specifically designed to provide information on any small (>4 km) crustal silicic magma chambers in the site area.



Figure 2.8-1. Maps of seismograph stations used in teleseismic tomography studies. Permanent stations of the SGB network are shown as solid symbols; portable stations installed around Yucca Mountain in 1982 are shown by open symbols. Site area hachured. (a) Map showing stations used in regional ("SGB") inversions of data from the whole SGB network. "LV" is Las Vegas. (b) Mid-scale map showing stations used in mid-scale ("NTS") inversions. (c) Map showing stations used in fine-scale (Yucca") inversions.

Last, the SGB network is denser in the NTS area. Hence, intermediatescale images with up to 12-km resolution and imageable depth of 120 km are possible in this region. However, this resolution decreases toward the edges of the array more rapidly than is usual for teleseismic images. Images at both of these finer scales have now been produced. The new images and the revised images of regional-scale features are discussed in an interpretive report and a data report, both now in preparation by J.R. Evans and M. Smith.

The resolution values given above apply to the shallower parts of the images in the well-sampled parts of the region. Resolution degrades slowly with increasing depth and rapidly eyond the edges of the seismograph array. Vertical resolution is not as good as horizontal; in particular, vertical resolution of the finest-scale images is about 12 km near the surface. A special class of vertically limited but horizontally extensive objects may go completely unobserved or be resolved only at their edges. Additionally, the number of readings and the distribution on the globe of teleseismic sources used in the experiment can reduce the effective resolution or cause artifacts that must be evaluated expertly. Nevertheless, magma chambers comparable to those under Long Valley and the Mono Craters, California (Dawson et al., 1989; Achauer et al., 1986), or San Francisco Mountain, Arizona (Stauber, 1982), and within about 10 km of the repository would be resolved with these data.

A related data set has been produced from evaluation of refracted arrivals, from underground nuclear explosions (UNEs). These data are reduced in terms of travel time resi uals for stations located throughout Yucca Mountain and vicinity (Walch and Phillips, 1989; Walch, 1988; Hoffman and Mooney, 1983). Because the UNE first arrivals are refracted along a crustal horizon, the residuals apply only to the velocity structure of the upper and middle crust.

### SUMMARY OF RESULTS

Monfort and Evans (1982) summarize results for data collected from the regional SGB network in 1979 and 1980. Combined with 1982 data (currently unpublished), the major features seen are (1) a high-velocity upper-mantle anomaly beneath the northern part of the NTS and the Silent Canyon caldera; (2) a low-velocity upper-mantle region beneath the southern part of the NTS and eastward; (3) crustal low-velocity anomalies beneath the Silent Canyon caldera, the southern part of the NTS, Crater Flat, and several other stations; (4) high crustal velocity beneath stations on and near Paleozoic outcrops; and (5) a high-velocity upper-mantle feature beneath the Funeral Mountains-Death Valley region.

Though the high-velocity upper-mantle anomaly beneath the Silent Canyon caldera is better resolved by current data than by any previous study, it also has been observed by Spence (1974), Minster et al. (1981), and Taylor (1983), using teleseismic recordings of NTS tests. The anomaly may be residuum of a magma chamber associated with the Miocene volcanism that generated thick layers of tuff in this region. If so, this result implies that this system is inactive at its roots. The low-velocity upper-mantle anomaly beneath the southern part of the NTS is of unknown origin, but may be a region of partial melt associated with recent basaltic volcanism south of the repository site. However, other interpretations are equally plausible. This matter merits further investigation.

Preliminary data from the dense 1982 temporary array were used by Evans and Oliver (1987) to infer a strong correspondence between the portion of the teleseismic residuals caused by shallow (<5 km) structure and isostatic residual gravity. This correspondence implies that the thickness of tuff overlying Paleozoic basement rocks is mirrored in these data; a combined interpretation of gravity and teleseismic data may clarify the geometry of this interface. Both types of data indicate that an important structural boundary is present beneath or very near the repository block, where the eastern edge of the basin beneath Crater Flat apparently runs east of teleseismic stations in Solitario Canyon. This boundary should be located more precisely; a high-resolution active-source analogue of teleseismic tomography (Achauer et al., 1988; Evans and Zucca, 1988) can accomplish this goal and should be considered.

The significance of Crater Flat is an issue of some importance to structural models of the Yucca Mountain region. It may be one or more Miocene calderas, or simply a graben, possibly related to Miocene volcanism north of Crater Flat in the Timber Mountain-Oasis Valley caldera complex. The new teleseismic data suggest the former, in that an apparent low-velocity anomaly is present beneath Crater Flat in the middle and lower crust. (The velocity model was corrected for reasonably well-known, near-surface velocity structure such as the volcanic section in Crater Flat; details of this correction will be provided in the forthcoming report.) This columnar low may extend into the upper crust as well, but current data do not resolve the junction region between it and the shallow basin known to exist beneath Crater Flat. These two features may be either connected or separated. The low-velocity anomaly may be caused by Miocene piston-like collapse of a caldera over a now-solidified magma source near the Moho, or it may represent a modern thermal disturbance, possibly related to Quaternary basaltic volcanism.

### BIBLIOGRAPHY FOR SECTION 2.8

- Achauer, U., J.R. Evans, and D.A. Stauber, 1988. "High-Resolution Seismic Tomography of Compressional Wave Velocity Structure at Newberry Volcano, Oregon Cascade Range," <u>Journal of Geophysical Research</u>, vol. 93, p. 10,135-10,147.
- Achauer, U., L. Greene, J.R. Evans, and H.M. Iyer, 1986. "Nature of the Magma Chamber Underlying the Mono Craters Area, Eastern California, As Determined from Teleseismic Travel Time Residuals," <u>Journal of Geophysical</u> <u>Research</u>, vol. 91, p. 13,873-13,891.
- Dawson, P.B., H.M. Iyer, and J.R. Evans, 1989. "Teleseismic Tomography of the Compressional Wave Velocity Structure Beneath the Long Valley Region, California," Journal of Geophysical Research (in JGR review).

- Evans, J.R., and H.W. Oliver, 1987. "Comparison of Timber Mountain Caldera Complex, Nevada, with Yellowstone: Speculations on Mechanism" (extended abstract, Abstract Volume, <u>Hawaii Symposium on How Volcances Work</u>, 67, Hawaii Volcano Observatory, Hilo, Hawaii.
- Evans, J.R., and J.J. Zucca, 1988. "Active High-Resolution Seismic Tomography of Compressional Wave Velocity and Attenuation Structure at Medicine Lake Volcano, Northern California Cascade Range," <u>Journal of</u> Geophysical Research, vol. 93, p. 15,016-15,036.
- Hoffman, L.R., and W.D. Mooney, 1983. A Seismic Study of Yucca Mountain and Vicinity, Southern Nevada: Data Report and Preliminary Results, U.S. Geological Survey Open File Report 83-588, 50 p., 1 plate.
- Iyer, H.M., 1988. "Seismological Detection and Delineation of Magma Chambers Beneath Intraplate Volcanic Centers in Western U.S.A.," in <u>Modeling of</u> <u>Volcanic Processes</u>, edited by C.Y. King and R. Scarpa, 1-56, Friedr. Vieweg und Sohn, Wiesbaden.
- Minster, J.B., J.M. Savino, W.L. Rodi, J.F. Masso, and T.H. Jordan, 1981. "Three-Dimensional Velocity Structure of the Crust and Upper Mantle Beneath the Nevada Test Site" (abstract), EOS, Trans. AGU, 62, 972.
- Monfort, M.E., and J.R. Evans, 1982. <u>Three-Dimensional Modeling of the</u> <u>Nevada Test Site and Vicinity from Teleseismic P-Wave Residuals</u>, U.S. Geological Survey Open File Report 82-409, 66 p.
- Spence, W., 1974. "P-Wave Residual Differences and Inferences on an Upper Mantle Source for the Silent Canyon Volcanic Centre, Southern Great Basin," Geophysical J. R. Astron. Soc., 38, 505-524.
- Stauber, D.A., 1982. "Two-Dimensional Compressional Wave Velocity Structure Under San Francisco Volcanic Field, Arizona, from Teleseismic P Residual Measurements," Journal Geophysical Research, vol. 87, p. 5451-5459.
- Taylor, S.R., 1983. "Three-Dimensional Crust and Upper Mantle Structure at the Nevada Test Site," <u>Journal of Geophysical Research</u>, vol. 88, p. 2220-2232.
- Walch, M.C., and J.S. Phillips, 1989. <u>Two-Dimensional Velocity Models for</u> <u>Paths from Pahute Mesa and Yucca Flat to Yucca Mountain</u>, SAND88-3033, Sandia National Laboratories, Albuquerque, New Mexico.
- Walch, M.C., 1988. "Modeling of Anomalous Ground Motion Observed at Jackass Flats, Nevada Test Site" abstract, Seism. Res. Lett., vol. 59, no. 31.



### 3.0 PLANNED GEOPHYSICAL ACTIVITIES

This chapter presents planned geophysical activities in a manner that is organized around specific applications and needed information. This complements the Site Characterization Plan (SCP), which does not present a distinct geophysics program per se. In the SCP, plans for geophysical activities are typically intermingled with non-geophysical studies, and similar geophysical activities are discussed in different sections. This chapter also associates plans for future geophysical activities with results obtained from past activities, to the extent that these results havused in the planning process.

A summary of planned activities is presented in Table 3.1-1. This summary includes information from similar tables in the SCP (Table 8.3.1.4-4), with some additional detail provided.

A report by Jones et al. (1987) and a paper by Wynn and Roseboom (1987) on geophysical techniques for site characterization present useful summaries of the general attributes of various geophysical techniques; this information will not be repeated here. Rather, this section elaborates upon specific applications of geophysics at the Yucca Mountain site. The report by Jones et al. identified a number of methods that have not been previously used at the site. Accordingly, present plans include testing of various methods prior to full-scale implementation for site characterization.

This chapter contains some new insights from the editors, and from the Project investigators responsible for geophysical activities. For the most part, this chapter is consistent with the SCP and recently drafted Study Plans. Some of the relevant content of the SCP from Investigations 8.3.1.17.4 and 8.3.1.4.2 is reiterated for clarity. Activity descriptions in the SCP and Study Plans, and information in the Site Characterization Technical Planning Basis (DOE, 1989), are controlled by the DOE. This report, particularly Section 3, represents a working position on the application of geophysics, and is not part of a controlled planning basis. Discrepancies between information contained in this report and in the documents listed above should be regarded in favor of the controlled sources. Whereas this report identifies some new exploration concepts and elaborates on some activity descriptions in the SCP, changes to the scope of work described by the SCP and Study Plans, or to the information contained in the technical baseline (DOE, 1989), need to be reviewed and approved in accordance with change control procedures before becoming part of the controlled planning basis for site characterization. Also, it is important to recognize that a systematic effort to set priorities for surface-based testing, in a manner that will provide data for evaluating siting criteria of 10 CFR 60.122, is underway at the time of publication of this report.

Geophysical participation in site characterization is discussed in the following subsections for hydrologic, geologic, tectonic, and engineering applications (Sections 3.1.1 to 3.1.6). A description of activities related to the geophysics integration activity (SCP Section 8.3.1.4.1.2) including recommendations of a feasibility testing program and a preliminary prioritization of geophysical activities is given in Section 3.2. The section on feasibility studies (Section 3.3) describes a means for timely acquisition of certain limited but important information that is needed to

further develop the geophysics program. The important question of integrating older geophysical data with newer data sets is considered in Section 3.4.

Table 3.1-1. Summary of Flanned Geophysical Activities (page 1	OL	03
--	----	----

Marbod	Box inits	Insting	Enorga	Decision	Comments
Method	ACCEVITY	LOCATION	scope	poince	Comment 3
			SEISHOLOGY		
Deep refraction	8.3.1.17.4.3.1	E-W transect Indian Springs-Stowepipe Wells (SCP Figure 8.3.1.17-12)	Reversed profiles and cross-profiles, shot- points 8 to 20 km spacing.	Rone	Existing surveys discussed in Section 2.4.
Shallow (Bison) refraction and shear wave refrac- tion and reflection	8.3.1.17.4.4, C.3.1.2.2.1.1 and others	Quaternary faults, Yucca Mountain region and site area	250 to 500 m traverses, portable instruments, sledgehammer energy source. Shear wave method uses 12 or more geophones, 3 m spacing.	Number and location 780*,	Maximum depts of penetration -100 m. Used to detect fault offset in surficial deposits, and depth to bedrock. Shear wave method capable of detecting 30 cm offset.
Evaluation of proposed deep reflection survey	8.3.1.17.4.3.1	Planned lines across Yucca Mountain and Crater Flat; also extending across Amargosa Desert to Death Valley; subject to peer review	T90	DTP <sup>b</sup> after evaluation of preliminary tests (15 km reconnaissance line) and peer review	COCCMP <sup>®</sup> survey extending northward into southern Death Valley produced marginal quality data, although data in the upper one second are locally good. Reflections at 6-s-conds were imaged with local continu.cy. See Section 2.5.
Evaluation of intermediate- depth reflection and refraction	8.3.1.17.4.7.1 9.3.1.4.2.1.2	Site area, including one or more E-W lines across Yucca Mountain	Evaluate previous results, assess potential for appli- cation of methods to Yucca Mountain, conduct planned line across Yucca Mountain, plan other applications as appropriate.	None	This is a planning activity only. Previous reflection survey using Vibroreis at Yucca Mountain failed (McGovern, 1983). More recent surveys using air gun at Mid Valley produced useful results (McArthur and Burkhard, 1986).
Shallow (Mini-Sosie) reflection	8.3.1.17.4.7.8	Yucca Mountain, Crater Flat, Jackass Flats, Amargosa Desert, discharge areas	Planned work includes 7 to 15 profiles, 1 to 5 km in long: hand- carried instruments. Source energy from hand-operated tampers.	DTF after evaluation of two preliminary profiles.	Maximum depth of penetration 1 km. mini- mum depth about 100 m. Used to map shallow structural and stratigraphic features with possible application to investigation of large hydraulic gradient area at Yucca Mou. 2011.
Vertical seismic profiling/ tomography	8.3.1.4.2.2.5 See also 8.3.1.2.2.3 and 8.3.1.2.2.3.2	Repository block and vicinity	As many as 15 to 25 geotomographic pro- files, 0.2 to 2 km in length; also cross- hole surveys if feasible. Various shear and compressional sources will be compare	DIP after feasibility test, and after calibration in shaft and drifts.	Used to map 3-dimensional network of rock mass fractures. Fixel dimension as small as 20 m.

Method	Activity	Location	Scope	Decision points	Comment s
Shallow seismic refraction and reflection	8.3.1.14.2.3.3 8.3.1.17.4.2	Viciality of repository surface facilities	TBD	None	Will provide data for modeling soil- structure interaction and local site effects on vibratory ground motion; also to investigate for alluvium offset indicative of faulting.
		GRA	WITY INVESTIGATIONS		
Regional maps	8.3.1.17.4.12.1	Yucca Mountain region	Beatty 1:100,000 guad- rangle, Pahute Mesa 1:100,000 guadrangle, Nevada Test Site 1:100,000 map area, Yucca Mountain 1:48,000 map area	None	Field work complete, compilation rom- plete; drafts available. See Section 2.1.
Site area map	0.3.1,17.4.7.2	Site area	1:24,000 map of site and vicinity, 200-ft spacing of stations along E-W lines spaced 500-ft apart (where topography permits)	None	Will require about 7,500 additional stations. Useful for establishing stratigraphic variability of repository host rock and fault location and offset. See Sections 2.1 and 3.1.5.
Detailed surveys, deep reflection profiles	8.3.1.17.4.3.1	Along traverses between Stovepipe Wells, Yucca Mountain, and Indian Springs. Subject to peer review (with deep seismic refraction and reflection).	Grawity stations along profiles at 500-ft spacing	None	Augment seismic data for joint interpretation.
			MAGNETIC METHODS		
Regional aeromagnatic maps	8.3.1.17.4.12.1	Yucca Mountain	Beatty-, Pahute Mesa-, Indian Springs-, and Pahranagat 1:100,000 guadrangles to be com- piled from existing surveys	Nose	Field investigations complete, compila- tion nearly complete. See Section 2.2.
Site area aeromagnetic map	8.3.1.17.4.7.3	Site area	1:12,000 scale map of site and vicinity, continuous aeromag- netic survey along E-W flight lines spaced 1/16 mile apart.	None	1:62,500 scale map complete. See Sections 2.2 and 3.1.5.

# Table 3.1-1. Summary of Planned Geophysical Activities (page 2 of 8)

## Table 3.1-1. Summary of Planned Geophysical Activities (page 3 of 8)

Method	Activity	Location	Scope	Decision points	Comment #
Ground magnetic survey (with deep reflection profiles and shallow reflection profiles)	8.3.1.17.4.3.1	Along traverses between Stovepipe Wells, Yucca Mountain, and Indian Springs	Magnetic intensity stations along pro- files at 10 to 20-ft spacing where acces- sible by truck, 50 to 100-ft spacing elsewhere	DTP only if seismic surveys run.	Augment seismic data for joint interpretation.
Site ground magnetic surveys	8.3.1.17.4.7.4	Site area and wicinity	Ground magnetic surveys at (1) known and inferred structures, (2) vicinity of drill- holes, and (3) anomalies detected in site aeromagnetic map. Surveys to be semicon- tinuous (10 to 20-ft spacing)	Number and location TBD.	To locate concealed extensions of faults and buried volcanics.
Curie isotherm	8.3.1.8.5.2.1	Yucca Mountain region	Mnalysis of regional aeromagnetic data	lione	Map configuration of computed Curie isothermal surface, and to com- pare areas of shallow isotherms with areas of high heat flow and recent volcanism. See Connard et al. (1983) Nevada study complete (Blakely, 1988): further work will focus on Yucca Mountain region using detailed data coverage.
			LECTRICAL METHODS		
Regional magnetotelluric (MT)	8.3.1.17.4.3.1	Yucca Wountain, Crater Flat, Jackass Flats, Amargosa Desert, Death Valley (SCP Figure 8.3.1.17-8)	Detailed survey with stations at 3 to 5-km spacing along tra- werses across Tucca Mountain, Crater Flat, and northern Amargosa Desert: reconnaissance survey with stations at 10-km spacing in remainder of area	None	Previous work shows mappable con- ductivity contrasts in 1 to 15 km depth range. See Section 2.3.

# Table 3.1-1. Summary of Planned Geophysical Activities (page 4 of 8)

Method	Activity	Location	Scope	Decision points	Comment a
Surface and airborne geo- electric investigations (possibly including air- borne EM, slingram, VLF, DC res.stivity, VBF sound- ings, tensor audio magneto- tellurics, and telleric profiling)	8.3.1.17.4.7.5	Site area; also Amargosa Desert discharge areas	Assess potential for application of these methods, evaluate pre- vious results, plan new applications as appropriate.	DTP with feasibility testing of selected methods only, as determined by review.	Possible application to structural and stratigraphic problems at the site, water-table depth estima- tion in regional ground-water discharge areas, and reconnais- sance of the large hydraulic gradient area at Yucca Mountain. See Section 2.3.
Induced Polarization (IP)	8.3.1.17.4.7.5	Site area, Shoshone Mountain area; exact line loca- tions TBD	Investigate possible extension of altera- tion, and minerali- ration from Calico Bills west to repository area. Large dipole spacing.	DTP as determined by review.	Assist interpretation of potential mineralization at northern end of site area.
Surface electrical resistivity methods EB and EM	8.3.1.17.4.7.5	Site area	To supplement existing or planned surveys where data are needed to test for a particu- lar mineral deposit	DTP only after data review for applica- bility to minerals assessment.	Assist in identification of deposit characteristics.
		RADIOWET	RIC AND REMOTE SENSING METRO	os	
Surface and airborne gamma ray investigations	8.3.1.17.4.7.6	Site area and Yucca Mountain region	Assess application of these methods with preliminary survey over known faults, using static ground measurements	DTP with major application of air- borne methods only if warcanted by feasibility test results.	Could detect percolation of radom through fault zones. Perform feasibility study in time to coordinate airborne survey with aeromagnetic survey.
Thermal infrared investiga- tions	8.3.1.17.4.7.7 8.3.1.2	Site area and Yucca Mountain region	Assess potential for application of air- borne and satellite thermal infrared imagery for mapping fracture networks in exposed bedrock, and infiltration processes in surface materials.	DTP based on feasibility testing, and eveluation of cost versus expected results.	Depends on detection of surface temperature variation, which is related to soil moisture content, which is partly related to infiltration.

the second se					
Hetbod	Activity	Location	Scope	Decision points	Comment a
Thematic Mapper satellite imagery	8.3.1.17.4.3.5 8.3.1.17.4.9.1	Yucca Mountain and wicinity	Tapes of the four Them- atic Mapper V scenes encompassing the Yucca Moustain region will be used to produce spectral and spectral ratio maps.	Sone	Used to define structural domains (including lineaments), areas of well-developed desert varnish, areas of hydrothermal siteration, and areas of vegetative cover.
			PALEOMAGNETISM		
Regional outcrop studies	6.3.1.17.4.3.2	Yucca Mountain, Little Skull Mountain, Crater Flat, Skull Mountain, southern Yucca Moun- tain, eastern Yucca Flat	5 to 6 sites at Yucca Mountain will be sam- pled. If useful results are obtained, other sites as listed may be sampled.	DTP only if useful results obtained at fucca Mountain, and if suitable strata are present.	Preliminary results at Yucca Mountain suggest 30* rotation. See Section 2,7.
Site area outcrop and drill core studies	8.3.1.4.2.1.5	Yucca Mountain	Orient drill core as it becomes available. Establish reference orientation through study of outcrop samples. Determine magnetic character of outcrop samples to aid in interpretation of aeromagnetic data.	Scne	
		BCREH	DLE GEOPHYSICAL METHODS		
Geophysical logging					
Borehole gravimetry	8.2.1.4.2.1.3	Site area and vicinity	Survey 15 existing water-table drill- holes, eristing deep holes that can be made available, and all new holes that reach the base of the Topopah Spring Member in the site area	None	Already have data in drillholes H-1, P-1, G-1, G-3, and G-4. Data will b used to model structure in the immediate vicinity of each borehole, to study lithophysal zones, and to model the Paleozoic surface beneath Yucca Mountain.

# Table 3.1-1. Summary of Planned Geophysical Activities (page 5 of 8)

the second se	the second s	the second s		and the second	the second se
Method	Activity	Location	Scope	Decision points	Comment a
Borebole magnetic logs	8.3,1.4.2.1.3	Site area and wicinity	Survey 15 existing water-table drill- holes, and new drill- holes prior to casing operations.	None	Used to determine mappable magnetic events for studying structural and stratigraphic continuity of Yucca Mountain, and to supplement paleomagnetic and lithophysal studies.
Induced polarization logs	8.3.1.4.2.1.3	Site area	<pre>Inst in one or two boreholes; investi- gate log response to authigenic tuff alts;ation, and to sulfide mineralization in boreholes.</pre>	Evaluate for effectiv, ss.	Feasibility study to determine if the method can be used to detect reolitization, and to assess the abundance of sulfides.
Commercially available logs	4.3.1.4.2.1.3	Yucca Mountzin	Acquire a standard suite of logs in all existing unlogged drillholes, and all new holes. Relog selected holes not previously logged for IP.	Nose	To obtain parameters for hydrologic, geologic, and geophysical models, and to determine uniformity and lateral distribution of rock prop- erties within the stratigraphic units.
Time-domain reflectometry (TDR)	8.3.1.2.2.4.2	Exploratory Shaft Facility	Measure changes in rock moisture content for infiltration test in the ESF.	None	Prototype testing required.
Borehole nuclear- geophysical logs, and neutron moisture meter logging	8.3 1.2.2.1, 8.3.1.2.2.3, 8.3.1.2.2.4	Site area, and Exploratory Shaft Facility	Unsaturated-zone boreholes.	None	Neutron scatter and gamma ray attenuation for volumetric moisture content.
Temperature loge	8,3.1,2,1,3	Yucca Mountain region	Selected boreholes.	None	For geothermal, heat flow, and hydrologic studies.
Acoustic televiewer logs and TV camera logs	8.3.1.4.2.2.3	Site area and vicinity	Ali drillholes; in the SZ and UZ as applicable.	None	For fracture and fault zone detection (television and televiewer), and stratigraphic and lithologic correlation (television).
Large spacing electro- magnetic and resisti- vity logs	8.3.1.4.2.1.3	Yucca Mountain	Selected drillholes.	After evaluation of surface and borehole data	To determine accurate large-volume in situ values for studying fracture and lithophysal zones, and for interpreting anomalies detected with surface and borehole data.

# Table 3.1-1. Summary of Planned Geophysical Activities (page 6 of P)

# Table 3.1-1. Summary of Planned Geophysical Activities (page 7 of 8)

				Decision	
Hethod	Activity	Location	Scope	points	Comment 3
Borehole gamma-ray spectrometry	8.3.1.4.2.2.3	Site area boreholes not already logged	Investigate presence of anomalous K, U, Th, and migration of these elements.	DTF after review for applicability to minerals assessment.	To look for evidence of hydrothermal activity or mineralization.
		BOREH	DLE-TO-SURFACE METHODS		
Resistivity and elec- tromagnetic methods	8.3.1.4.2.2.3	Yucca Mountain	Selected drillholes.	After evaluation of surface and borehole data, and crosshole studies.	Obtain bulk resistivity and dielectric information in areas of anomalous surface geophysical data or where faults are projected into the subsurface.
High resolution P and S wave seismic	0.3.1.4.2.1.3	Yucca Mountain	Selected drillholes.	After evaluation of surface and borehole surveys.	Obtain bulk properties in areas of anomalous seismic character or where faults are projected into the subsurface. Also, to obtain parameters for designing effective deeper-penetrating seismic surveys.
Surface-to-bole seismic refraction	0.3.1.4.2.1.3	Yucca Mountain	Selected drillholes.	After evaluation of surface and borehole surveys.	Possible application to fault detection/delineation, and bed tracing.
Op-hole and down-hole seismic	8.9.1.24.2.3.3	Vicinity of surface facilities	TBD	After evaluation of shallow seismic refraction survey.	Will provide data for modeling soil- structure interaction and local site effects on wibratory ground motion.
			ROSSHOLE METHODS		
Screhole-to-borshole methods	8.3.1.4.2.2.5	Fucca Mountain close- spaced holes for hydrologic testing and for surface facilities studies	Selected drillholes.	Sone	Possible feasibility studies to investi- gate the mappability of features that intersect the deilholes, such as fractures or somes of high water content, using resistivity, EM, and high resolution selamic methods.
Cross-bule seismic	8.3.1.14.2.3.3	Vicinity of surface facilities	TEU	After evaluation of shallow seismic refraction survey.	Will provide data for modeling soil- structure interaction and local size effects on wibratory ground motion.

and the

Method	Activity	Location	3cope	Decision points	Comment s
			PETROPHYSICS		
Petrophysics	8.3.1.4.2.1.4	Tucca Mountain	Selected samples from new corenoles.	None	To verify geophysical log accuracy, calibrate computed logs, measure properties not measured in situ, and to support modeling and inter- pretation of surface geophysical studies.

## Table 3.1-1. Summary of Planned Geophysical Activities (page 8 of 8)

\*TBD = to be determined. PDTP = gacision to proceed. \*CONCORP = Consortium for Continental Beflectic: Profiling.

114

### 3.1 DISCUSSION OF PLANNED GEOPHYSICAL ACTIVITIES

# 3.1.1 UNSATURATED ZONE HYDROLOGIC INVESTIGATIONS FAR-FIELD CHARACTERISTICS FROM VERTICAL SEISMIC PROFILING (VSP)

VSP will be tested early in the site program to determine its utility for characterizing variation in bulk fracture properties. It is thought that seismic methods may be capable of imaging large-scale (tens of meters) trends in average or bulk rock characteristics that are attributable to fracturing. Also, some features such as stratigraphic contacts or faults may be imaged as distinct seismic structures. Seismic profiles will be compared to results from pneumatic packer testing in boreholes and long-term monitoring of natural processes in the unsaturated zone (UZ), to evaluate the importance of observable seismic trends with respect to site behavior (Activity 8.3.1.2.2.3.2).

The use of VSP for hydrologic studies is based on fundamental models for seismic propagation in fractured rock. A system of near-vertical fractures tends to produce transverse anisotropy in compressional and shear wave responses. Velocity, anisotropy, and apparent attenuation will be interpreted in terms of contributions of fracturing to bulk deformability, or alternatively in terms of multiple scattering from distributed fracture interfaces. The surveys will be designed with coverage to support imaging with resolution as fine as about one seismic wavelength (which may be as small as 20 m). Consideration will be given to the structural fabric of the site to design surveys which can identify anisotropy or lateral heterogeneity of the seismic structure that may exist in proximity to known faults. Seismic sources will be evaluated using field trials and limited scoping studies; sources to be considered include shear vibrators, shear-impact sources, land air-gun, and explosives.

Two VSP experiments are currently planned: one at the Exploratory Shaft Facility (ESF), and another at the USW UZ-9 complex of UZ hydrology boreholes. The ESF study (SCP Activity 8.3.1.4.2.2.5) will require installation of three component geophones at intervals in the lining of each exploratory shaft, and along some workings at the main test level. Seismic data will be compared to the abundant information from geologic mapping and other testing in the ESF. In this way the relation between actual rock characteristics and seismic response will be evaluated in detail.

The USW U2-9 complex of boreholes will be constructed about 0.5 km southeast of the conceptual repository perimeter (Activity 8.3.1.2.2.3.2). These boreholes will investigate the UZ down to the water table. A prefabricated cable containing three component geophones at intervals of about 5 m, to a depth of about 400 m, will be cemented into a dedicated borehole. Present plans call for fully-cemented geophones to be used at the USW U2-9 complex VSP borehole. The same method will be used in boreholes at other locations for Activity 8.3.1.2.2.3.2, in lieu of a wall-locking (removable) VSP tool, if it can be shown that fully-cemented geophones satisfy the study objectives reliably. Information on seismic characteristics (i.e., P- and S-velocities, velocity, anisotropy, apparent attenuation, and reflections) will be compared to information obtained from drilling, logging, and pneumatic testing in a cluster of UZ boreholes. Additional VSP surveys may be conducted at other locations in the site area (particularly USW UZ-6, and systematic drilling program boreholes) depending on the results from the initial tests at the ESF and at the USW UZ-9 borehole complex.

At present, no preliminary VSP data are available for Yucca Mountain which would be of use in the design and planning of VSP studies in the ESF or at the USW U2-9 complex. Simple downhole velocity survers have been conducted using a single-component, wall-locking geophone tool and a vibrator source at zero offset. However, these data are available only on paper records, and are poorly suited for resolving velocity structure and evaluating attenuation and reflection coefficients. Physical model and computer simulations are being conducted using existing knowledge of structure and stratigraphy at the site, to support design of VSP experiments and reduction of data from the U2-9 complex. This is an important topic for examination by the geophysical integration activity (SCP Section 8.3.1.4.1.2).

### FORMATION CHARACTERISTICS NEAR THE BOREHOLE, FROM GEOPHYSICAL LOGGING

A suite of geophysical wireline logs will be run in the UZ in each new borehole constructed at Yucca Mountain for percolation studies (Activity 8.3.1.2.2.3.2) and the systematic drilling program (Activity 8.3.1.4.3.1.1). A somewhat different suite of logs is intended for the saturated zone (SZ). The borehole geophysics activity (Activity 8.3.1.4.2.2.3) will involve logging of all new boreholes, limited relogging of existing boreholes, and development of lithostratigraphic and hydrologic interpretations.

Logs will be acquired in the U2 for lithology and mineralogy (density, gamma-ray, spectral gamma-ray, magnetic intensity, and magnetic susceptibility), fracture characterization (axial and side-view television), porosity (gamma density), and moisture detection (epithermal neutron, neutron moisture meter, dielectric). Logs acquired for moisture detection will be compared to moisture content and potential from core and cuttings testing (Activity 8.3.1.2.2.3.1), and to other moisture-sensitive information such as long-term in situ hydrologic monitoring (Activity 8.3.1.2.2.3.2). The logs are expected to provide information on moisture conditions during and after drilling, and changes in these conditions over time as the holes are tested, and stemmed or shut-in.

Standard logs including caliper, temperature, and deviation surveys will also be acquired from the UZ. Although analysis of existing logs from Yucca Mountain is not yet complete, there is evidence that some of the common logs are not effective in the UZ. The compensated neutron porosity tool is of limited use in dry holes, apparently because of source-detector interference, which diminishes response to changes in formation moisture content (Serra, 1984). Also, the compensation algorithm may be unsuited for dry holes at the source-detector spacings commonly used. Experience at Yucca Mountain indicates better sensitivity to moisture content from single detector (shielded thermal detector) epithermal tools, used in non-centered configuration, than from standard compensated neutron porosity logs.

Electrical logging methods are commonly used to determine water saturation, but require some adaptation to the air-filled boreholes at Yucca Mountain. In the highly resistive (>100 ohm-m) tuffs of the UZ, particularly welded tuff, the induction log response is small and measurement uncertainty increases with increasing resistivity. If the industive tool could be improved to provide reliable readings in the highly resistive UZ, then estimates of water saturation could be obtained. Another approach to determining water saturation is provided by the dielectric tool. Experience with a 47-MHz dielectric tool at Yucca Mountain shows that the log appears to respond to variations in water content with useful depth of investigation (e.g., 10 cm or more). This response requires substantiation with laboratory measurement of dielectric permittivity on cores of determined mineralogy and as a function of water content. In addition, the dielectric tool will be run in cored holes and compared to core-based measurement of water content. The feasibility of dielectric and induction logs, as well as the relatively new nuclear magnetic resonance log, for moisture content in the UZ will be evaluated by a planned feasibility test of logging methods for UZ studies (see Section 3.3 of this report).

The need for one or more calibration boreholes has been identified in relation to geophysical logging (Activity 8.3.1.4.2.1.3) and logging for surface infiltration studies (Activities 8.3.1.2.2.1.2 and 8.3.1.2.2.1.3). The objective is to have a facility where borehole geophysical logs can be rerun as necessary during site characterization, as a calibration check that takes into account effects not generally addressed by standard calibration procedures.

Dedicated calibration boreholes are also desirable for "benchmarking" the responses of tools that should produce similar logs (e.g., identical tools from the same manufacturer, or similar tools from different manufacturers). Existing boreholes and currently planned boreholes are either unsuitable or are committed to other uses.

The facility would be designed to stabilize formation characteristics, so that long-term repeated logging would be expected to produce repeatable tool response. This and other aspects of calibration borehole design would be unique, and justify feasibility testing. Two holes, each about 30 m deep, have been tentatively proposed for the vicinity of Fran Ridge about 5 km east of the site. Each hole would sample the same part of the geologic section in a different hydrologic setting (e.g., rock slope and alluvium-covered wash). The ODEX dry drilling method would be used, and the holes could be completed with thin-wall steel, aluminum, or plastic casing over at least part of their length, to prevent them from gradually drying. Continuous core would be acquired using the dry method that was demonstrated in the existing shallow UZ holes (i.e., USW UZ-4, UZ-5, UZ-7, UZ-13). Core would be used for independent characterization of (1) lithology and hydrologic properties, and (2) baseline moisture conditions. Core data would provide a basis for inferring undisturbed conditions, which would be compared to log responses over time. Further design of the calibration borehole test will be addressed by the geophysics integration activity (see Section 3.3 of this report).

### BOREHOLE, CROSSHOLE, AND SURFACE-TO-BOREHOLE METHODS

The crosshole gamma technique has been used on a preliminary basis for monitoring infiltration into the uppermost 2 to 10 m of soil or rock, in different settings at the site (SCP Activity 8.3.1.2.2.1.2). The limit for satisfactory transmission and detection of 0.66 MeV gamma rays between steel-cased borings appears to be about 2 to 5 m. These measurements continue to be made periodically for natural and artificial infiltration studies. The technique may also be used in the clustered USW U2-9 boreholes. The spacing between boreholes in this cluster will be determined from the results of pneumatic injectivity tests and observed fracture distribution, so that the intervening distance corresponds to a lineal dimension for a characteristic volume of the rock mass, if practicable. If this distance is approximately 5 m or less, then crosshole gamma attenuation data will be acquired from the adjacent boreholds. This survey will be run once, prior to installation of borehole instrumentation (as described in the SCP for Activity 8.3.1.2.2.3.2). Accordingly, high-accuracy information on the variation of inter-borehole distance is required to infer moisture content from the attenuation data. A high-precision (inertial) directional survey will be run in each of the adjacent boreholes to provide this information.

At the present time, borehole seismic sources that can be used in dry holes have ranges that are limited to a few tens of meters. There are no immediate plans to attempt crosshole seismic surveys for UZ studies, but such surveys may be appropriate depending on the results from other techniques such as VSP.

A test of the feasibility of a number of borehole, surface-to-borehole, and possibly crosshole geophysical methods for characterizing fracturing and fault zones in the UZ and SZ is proposed in Section 3.3 of this report. This testing will be conducted using existing UZ boreholes, the existing "c-series" boreholes, or the proposed UZ-9 complex of boreholes, as appropriate. Methods identified for testing include large-spacing EM and ER logs, possible surface-to-borehole configuration of these methods, and possible crosshole VHF tomography (applicability of VHF methods is discussed further in Section 3.1.5).

### REMOTE SENSING

Airborne thermal infrared imaging will be examined for its ability to detect the results of percolation of meteoric water through fractures, faults, or porous surficial units (Activity 8.3.1.17.4.7.7). Spatial or temporal surface-temperature anomalies may be associated with differing moisture content of surficial or near-surface materials in response to precipitation or other moisture migration phenomena. Surface temperature can be detected from satellites and aircraft. This is a planning activity only, involving scoping analysis and examination of existing data from the site and similar areas. Feasibility testing of airborne thermal infrared detection for UZ infiltration studies may be justified based on the results from this activity.

## 3.1.2 SATURATED ZONE HYDROLOGIC INVESTIGATIONS

Geophysical surveys have been used in conjunction with hydrogeologic investigations for providing essential data used to construct and interpret the hydrogeologic framework of the ground-water flow system at Yucca Mountain and vicinity. The principal advantages of geophysical surveys in this context are that geophysical data can reduce the need for drilling, and that large areas can be readily explored.

There are several special problems in characterizing the ground-water flow system, which may be addressed through the use of geophysical methods. These are (1) characterization of the cause and extent of the large hydraulic gradient area north of the site, (2) estimation of the amount and lateral extent of natural recharge occurring in Fortymile Wash east of Yucca Mountain and in other major washes, (3) characterizing the frequency and interconnectivity of fractures in tuff units at Yucca Mountain, and (4) characterizing the lithology and structure of Cenozoic and Paleozoic rocks in the regional ground-water flow system. Each of these problems is discussed below.

### EXPLORATION OF THE LARGE HYDRAULIC GRADIENT AREA

A large hydraulic gradient (300~m head change in 2 km horizontal distance) has been identified from water level in drillholes to the north and west of Yucca Mountain (Robison, 1984; Robison et al., 1988), and from stem tests (e.g., Rush et al., 1983; Craig and Johnson, 1984). The cause of the gradient is unknown, but may be related to the following: (1) faults that contain nontransmissive gouge or that juxtapose transmissive tuff against nontransmissive tuff; (2) the presence of a different lithology that is less subject to fracturing, such as rhyolite or argillite, or the presence of an intrusive body such as a volcanic dike; (3) topography; or (4) a change in the direction of the regional stress field and a resultant change in the frequency, interconnectivity, and orientation of open fractures on either side of the large hydraulic gradient.

Several investigators have constructed models of ground-water flow of Yucca Mountain and vicinity, and in the process have relied to some degree on geophysical surveys to provide insight into the hydrogeologic framework of the hydrologic system. Perhaps the most comprehensive of these studies was that of Winograd and Thordarson (1975) in their report on the hydrogeologic and hydrochemical framework of the NTS. Their work relied on geologic characterization that, in turn, relied on geophysical surveys. Among those geophysical surveys was a regional gravity survey of the Death Valley region compiled by Mabey (1963).

The first numerical model of regional ground-water flow that includes Yucca Mountain was developed by Waddell (1982). This model provided the preliminary framework for the refined model of Czarnecki and Waddell (1984), which refers to extensive geophysical investigations of the NTS used for developing the conceptual model of the hydrologic system. Results from gravity and magnetic surveys are presented in Healey and Miller (1971), Kane et al. (1981), Ponce and Hanna (1982), and Snyder and Carr (1982). Resistivity studies of the area that were used in the model assessment were done by Smith et al. (1981), Fitterman (1982), and Greenhaus and Zablocki (1982). Reconnaissance seismic-refraction studies were presented in Pankratz (1982). Of these studies, gravity and magnetic surveys provided the most useful information for delineating different lithologies, particularly for differentiating volcanic rocks from carbonate rocks. Geophysical anomalies in the large hydraulic gradient area were identified by the magnetic surveys reported by Bath and Jahren (1983), and the resistivity survey of Senterfit et al. (1982).

The water table does not appear to be a dependable target for direct detection by means of electrical, gravity, or seismic methods. Density and seismic velocity or impedance contrasts across the water table are small because the tuff matrix in the UZ contains significant water. Resistivity soundings and profiles reported by Frischknecht and Raab (1984), and Senterfit et al. (1982), suggest that the water table may be associated with a resistivity change from roughly 300 to 100 ohm-meters. However, the observed contrast is diffuse, and available methods appear to be sensitive to variability in near-surface materials, and other factors that make the water table contrast difficult to trace, and produce discrepancy with water level observations in boreholes. Although it is remotely conceivable to geophysically map the water table in rugged terrain north of the site with borehole control, this type of information still might not reveal the cause of the large hydraulic gradient. Exploration will therefore emphasize reconnaissance for geophysical detection of geologic features giving rise to the gradient, which is potentially a more straightforward exploration problem.

A preliminary scoping analysis of the use of detailed gravity data for investigating the large hydraulic gradient has been performed (H.W. Oliver and S.F. Carle, written communication). Existing gravity data covering the area northwest of the site area were examined for correlation with the potentiometric surface, and for the presence of a structure affecting the potentiometric surface. Gravity data collected along Drill Hole Wash and along a parallel northwest-southeast profile were modeled in two dimensions. The resulting density model was adjusted for agreement of calculated gravity with observed gravity. This could be accomplished by including offsets for a number of known faults crossed by the profile, and by adjusting the thickness of the oldest tuff units and the depth to the Paleozoic-Tertiary contact. Thus the existing gravity data appear to correlate to deep structure and are probably not significantly affected by the potentiometric surface or structure with similar geometry. The feasibility of using gravity data that are more detailed than the existing data, for investigation of the large hydraulic gradient, will be examined in a proposed feasibility test wherein detailed data will be acquired along northwest profiles across the large hydraulic gradient area (see Section 3.3 of this report). The location of these profiles will roughly coincide with the magnetotelluric profiles shown in Figure 3.1-1, depending on the details of available elevation control in this area.

Another gravity method that will be considered and possibly tested in the field is high-resolution temporal gravity monitoring by means of a cryogenic-type gravimeter. This instrument can resolve  $10^{-1}$  µgal  $(10^{-10}g)$ , with drift of about 10 µgal  $(10^{-8}g)$  per year. One objective for gravity monitoring would be to correlate water well level observations with



atmospheric fluctuations, tidal acceleration, earth tides, and other geophysical phenomena at the site. Another application would be comparison of continuous gravity recordings with theoretical models for gravity signals that might result from active processes such as rising magma or fault slip. The applicability of continuous gravity monitoring with cryogenic-type gravimeters (as well as by other means) will be evaluated in a scoping study of geophysical monitoring capabilities and application (see Section 3.3 of this report).

The deep structure which is apparently the cause of a gravity anomaly of 8 to 10 mgal beneath the northern portion of Yucca Mountain (see Section 2.1) is roughly associated with a gradient in total aeromagnetic intensity, in the vicinity of the large hydraulic gradient (see Section 2.2). Planned detailed gravity and detailed aeromagnetic surveys will encompass this area for reconnaissance, and enable further modeling and interpretation (SCP Activities 8.3.1.17.4.7.2 and 8.3.1.17.4.7.4, respectively). The geophysics integration activity will ensure through planning and interaction between investigators that detailed gravity and aeromagnetic surveys (see Activities 8.3.1.17.4.7.2 and 8.3.1.17.4.7.4) are appropriate for reconnaissance of the large hydraulic gradient area.

A program of closely spaced magnetotelluric and audio-magnetotelluric soundings, oriented along northwest profiles across the large hydraulic gradient, has been proposed as a feasibility test. The objectives will be to detect large scale structures, and lithologic changes (as manifested in electrical resistivity variation), that may be associated with the cause of the large hydraulic gradient (see Section 3.3 of this report). The frequency band used will be approximately 0.01 to 100 Hz, which is sensitive to resistivity contrast in the upper 3 km. Current plans call for 20 fivecomponent soundings at the locations shown on Figure 3.1-1. The response of orthogonal MT components will be used to evaluate changes in apparent resistivity anisotropy, and to correlate such changes with known geologic information to identify possible anomalies that could indicate structure or facies changes that may give rise to the large hydraulic gradient. The planned program of MT soundings is subject to the review of geoelectric methods that is also planned (Activity 8.3.1.17.4.7.5). Current plans call for the review prior to additional field surveys.

Data from the following surveys, described in various parts of the SCP as noted, will also be analyzed to characterize the large gradient area, and to identify anomalies that may be associated with the cause of the gradient and identify targets for drilling or more intensive geophysical exploration. These activities are mostly planned in conjunction with geology and tectonics studies:

- 1. Mini-Sosie seismic reflection surveys (Activity 8.3.1.4.2.1.2).
- Analysis of in situ stress information obtained from fracture indications recorded by the borehole acoustic televiewer (Activity 8.3.1.4.2.2.3).
- 3. Ground magnetic surveys (Activity 8.3.1.17.4.7.3).

### Upper-crustal high-resolution seismic refraction (Activity 8.3.1.4.2.1.2).

These methods will be applied along traverses oriented to test specific hypotheses related to formation of the large gradient, including the existence of a major lateral discontinuity in buried Paleozoic rocks, a buried fault, or an igneous intrusion. This work will be done in a relatively roadless area with rugged terrain. Application of seismic reflection will depend on testing and evaluation that is planned for other locations (see Section 3.1.4).

In summary, application of geophysics to study of the large hydraulic gradient area involves feasibility testing, and integration will other planned geophysical testing and exploration activities. Coord in between technical groups and Project participants applying and using information from geophysical surveys is the function of the planned geophysics integration activity.

### INVESTIGATION OF RECHARGE IN STREAM WASHES

Recharge occurring in Fortymile Wash and other washes in the vicinity of the site area is probably the major component of water movement through the UZ to the SZ in the vicinity of Yucca Mountain. Recharge has been documented in Pagany Wash through repeated soil moisture logging in specially constructed boreholes (see Section 2.7). Czarnecki (1985) showed from modeling of the subregional ground-water flow system that the water table altitude beneath the site area is more sensitive to the nature of recharge at Fortymile Wash than to other recharge parameters used in the model.

Factor, affecting recharge (net infiltration) include (1) thickness and permeability of unsaturated channel fill or alluvium; (2) the presence of impermeable strata such as caliche within alluvial sediments; (3) the presence of vertical, open fractures beneath the channel surface; and (4) ambient soil moisture content.

To gain a better understanding of the hydrogeologic framework of the Amargosa Desert, Czarnecki and Oatfield (1987) compared estimated depth to bedrock from Schlumberger resistivity soundings (Greenhaus and Zablocki, 1982) to the estimates from gravity interpretation for the region (Healey and Miller, 1971), and the seismic refraction lines discussed by Ackermann et al. (in USGS, 1988). Depth to bedrock was estimated by Greenhaus and Zablocki (1982) to be as much as 1,800 m in parts of the Amargosa Desert. Relative differences in resistivity of the upper 75 m of sediments were used in conjunction with lithologic descriptions from test borings and irrigation wells to identify areas with coarser and potentially more transmissive sediment. The resulting alluvial thickness model is important and will be used in ongoing numerics. models of the ground-water flow system. Past results indicate that the approach to mapping alluvial thickness from resistivity, gravity, and seismic refraction data could be applied north of the Amargosa Desert in areas such as Fortymile Wash and Jackass Flats. Existing gravity data (Section 2.1) and an east-west upper-crustal seismic refraction profile (Sutton, 1985; also see Section 2.4) are available for this region.

Detecting the presence of impermeable strata in borings within the valley fill may best be accomplished by coring, but in lieu of this procedure borehole geophysical logging will be used. This method has been applied at Yucca Mountain, using a neutron moisture meter tool in a steel-cased boring. The distribution of moisture following recharge events can be used to identify and infer the hydrologic behavior of stratigraphic features. This approach is less costly than coring, and has detected infiltration to as much as 12 m depth (see Section 2.7). The same methodology would be used in major recharge areas such as Fortymile Warn, but extended to depths of hundreds of m ters. Development work is ongoing to improve neutron moisture logging tool response and calibration. The logging approach also permits repeated measurements during possible recharge events.

### FRACTURE CHARACTERIZATION IN THE SATURATED ZONE

Ground-water flow through Tertiary volcanic units in the SZ at Yucca Mountain is probably dominated by the effects of fractures, and moisture movement through the UZ may be dominated by fractures depending on the moisture flux.

Fracture data will be used in the interpretation of falling-head injection tests and other hydraulic-stress tests previously conducted at the existing "c-series" boreholes (SCP Activity 8.3.1.2.3.1.3). Fracture data will be used in conjunction with interborehole flow tests to select combinations of test intervals for multi-well interference tests (SCP Activity 8.3.1.2.3.1.4) and tracer tests (see SCP Activity 8.3.1.2.3.1.5). Fracture network models (SCP Activity 8.3.1.2.3.1.4) used to interpret the results of multi-well tests will be based partly on borehole logs and (to the extent practicable) the results of crosshole and surface-to-borehole geophysical surveys.

Fracture characteristics that principally affect ground-water mobility are spatial distribution, frequency, length, connectivity, and transmissivity. Of these, distribution and frequency can be characterized geophysically. Length and connectivity are geometric characteristics for which geophysical methods provide useful information only in the unusual case that geophysical responses of individual fractures can be recognized. Measurement of transmissivity requires a hydraulic test to induce fluid transport, with measurement of induced flow or potential. Electrical current might be substituted as an indicator of hydraulic response, to the extent that current flows along the same pathways, and sufficient resistivity contrast exists between the fractures and the rock matrix. This possibility will be tested during prototype testing early in site characterization, as discussed below.

Fracture data obtained prior to site characterization by borehole geophysical logging of the SZ in the "c-series" boreholes, and other boreholes at Yucca Mountain, will be used in several hydrologic investigations. Borehole television, the borehole acoustic televiewer (BATV), and the full waveform acoustic log (FWAL) will be used to obtain fracture frequency and fracture attitude in the SZ (Activity 8.3.1.4.2.2.3). Fracture information from limited available core, and lithologic information from cuttings, will also be used.

Fractures are difficult to characterize from vertical boreholes because most fractures in the tuff units are nearly vertical. Additional information on fracture characteristics in the vicinity of boreholes would be useful to evaluate the bias inherent to borehole observation. Several geophysical methods are planned to characterize the spatial variability of fracture characteristics on a volume average basis throughout the rock mass near certain existing exploratory boreholes. Feasibility testing (see Section 3.3 of this report) will be performed in both the UZ and SZ, using existing borehole USW G-4, the existing "c-series" boreholes (spaced about 60 m apart), and the planned USW UZ-9 cluster (spaced roughly 5 to 20 m apart). Tests will assess the capability to trace features that are observed to intersect the boreholes. Based on the results, the following methods may be extended to other boreholes in the site area for characterization of fractures in the SZ: (1) conventional and shear-wave VSF, (2) large-spacing ER and EM borehole logs, (3) surface-to-borehole measurements designed to extend the borehole surveys, and (4) crosshole VHF tomography.

The hydrophone-VSP method (e.g., Beydoun et al., 1984) for detecting transmissive fractures by means of tube wave generation in the SZ will be considered as a candidate method for feasibility testing. Performance of the method depends on the availability of a suitable seismic source. A broad-band source is needed that can penetrate the rock mass intervening between the source point (e.g., at the surface) and fracture intersections with the well bore in the SZ.

### REGIONAL CHARACTERIZATION OF CENOZOIC AND PALEOZOIC ROCKS

For the purpose of characterizing regional ground-water flow, more information is needed about the depth to top, thickness, and areal distribution of pre-Tertiary rocks. One borehole (UE25 p#1) penetrates the pre-Tertiary section at Yucca Mountain at a depth of 1,244 m. Because of the information needed for regional hydrologic characterization, and the expense of deep drilling, geophysical methods are needed to characterize subsurface structure, particularly of the Paleozoic section where it is buried beneath the Tertiary cover.

Recent seismic refraction surveys (Ackermann et al. in USGS, 1988) have provided reasonably good definition of pre-Tertiary units in the Yucca Mountain area, but these surveys do not cover the entire area of interest for regional hydrology. Magnetic surveys (e.g., Bath and Jahren, 1984) and gravity surveys (Ponce, 1981; Snyder and Carr, 1982) have provided additional information on the structure of Tertiary and pre-Tertiary units. While not decisive in determining either geometry or lithology, or regional distribution of Paleozoic rocks, these results provide some basis of the necessary information needed to characterize the regional ground-water flow system framework. Additional geophysical surveys will provide useful data for defining the regional hydrogeologic framework in the vicinity of the site.

The objectives of geophysical surveys associated with regional SZ studies are (1) to determine the areal distribution of thickness of Quaternary and Tertiary rocks in the vicinity of the site; (2) to determine the areal distribution, depth to top, and thickness of Paleozoic rocks in the vicinity of the site; and (3) to determine the structural configuration of the rocks in the regional ground-water flow system, including the nature of the contact between Tertiary and Paleozoic rocks. Plans to achieve these objectives are several fold. One possibility includes expansion of the currently existing seismic refraction surveys previously discussed in Section 2.4 and Figure 2.4-2 of this report. Such expansion would include two or three new east-west lines crossing the Ash Meadows and Amargosa River lineaments, and extensions of the existing reflection and refraction lines for more complete coverage of the regional ground-water flow system (Figure 3.1-2). Gravity, magnetic, and resistivity surveys are recommended within a 5-km-wide zone coincident with both reflection and refraction lines. Finally, supplemental gravity, magnetic, and geoelectric surveys are needed where gaps in coverage exist. Coordination for design of the geophysical surveys, with respect to SZ hydrology studies, will be undertaken by the geophysics integration activity (SCP Section 8.3.1.4.1.2).

## 3.1.3 GEOPHYSICAL CHARACTERIZATION OF MINERAL AND ENERGY RESOURCES

Application of geophysics to mineral resource assessment will be closely integrated with corresponding geological and geochemical studies throughout the assessment process. Geophysical data have relevance at all scales of investigation, from the regional (for identification of favorable terranes and large structures) to deposit-scale and borehole geophysics. Geophysical methods provide direct and indirect measurement of a wide range of physical properties or contrasts, and are a necessary complement to geologic and geochemical studies.

There are relatively few geophysical methods which directly detect particular types of mineral deposits. Gamma-ray spectrometry for uranium is a notable example, if the deposit is near the surface or intercepted by a borehole. Induced polarization for sulfide mineralization is another example of direct detection. However, most other methods provide only indirect




# YUCCA MOUNTAIN PROJECT

FIGURE 3.1-2. Planned Regional Geophysical Traverses (after SCP Figure 8.3.1.17-12, with Tentative Deep Reflection Lines Added).



## LEGEND





67-12

evidence for the presence of mineral deposits, which when combined with geological and geochemical information, can identify terrains or regions where the probability of mineral occurrence is favorable. Geophysics can identify structures, lithologies, alteration, the presence of some mineral groups (iron oxides, sulfides, clays, zeolites, etc.), radioactive elements (K, U, Th), and density anomalies, which may then be used indirectly to infer potential mineralization. Geophysical methods are especially important at Yucca Mountain because significant mineralization, if present, is probably below the volcanic units.

Modern assessment and exploration work is being aided by the generation of genetic and descriptive mineral deposit models. Descriptive models such as those presented by Cox and Singer (1986) are believed to contain the essential descriptive features of various kinds of deposits, but do not necessarily reflect genetic parameters. These models, as well as grade-tonnage rodels, will play an important part in the assessment of mineral and energy resources at Yucca Mountain. Descriptive models published to date contain limited information on geophysical attributes of the individual deposit models. However, a number of geophysical attributes have been identified for many deposit types. The USGS is beginning to correlate geophysical attributes with other mineral deposit parameters in order to improve the utility of descriptive models for assessment of resources at Yucca Mountain.

Deposit models, which include geophysical attributes once developed, will support quantitative evaluation of the geophysical methods for reconnaissance (i.e., gravity and magnetic surveys, seismic and geoelectric profiling, and remote sensing) and for detailed investigation of suspected targets. Additional geophysical data, if needed, will be acquired as part of existing geophysical activities planned or used as a basis for new activities in association with the geophysics integration activity (SCP Section 8.3.1.4.1.2).

Methods used in minerals assessment are similar to those used in other site characterization investigations. Resource assessment can use results from a number of planned geophysical activities and interpret them in terms of potential mineral deposits. For example, borehole logs acquired principally as lithostratigraphic indicators will be inspected for the remnants of hydrothermal systems. Subsequent investigations would focus on particular mineralization processes or types of deposits.

There are significant problems with identifying high-value, small-volume deposits which might be present in Paleozoic units below the volcanic section at Yucca Mountain. Surface or airborne geophysical surveys may not detect small deposits beneath the thick volcanic section, and only one drillhole at Yucca Mountain presently penetrates through the volcanic sequence. Sampling and geophysical logs from existing and planned deep drillholes will be used to the extent practicable.

Structural/stratigraphic studies as discussed below for geology and tectonics, will also be applied to mineral and energy resource assessment. Such information will support evaluation of the resource potential of faults and other structures; evaluation of traps, seals and reservoirs; and comparison of structure to that at known resource occurrences in the Great Basin of Nevada (SCP Activity 8.3.1.9.2.1.4). (These evaluations will also rely on other data such as geochemical analysis, source rock evaluation, thermal maturation studies, etc.) Requirements for geophysical exploration related to evaluation of mineral and energy resources evaluation will be addressed in the geophysics integration activity (SCP Section 8.3.1.4.1.2).

Opportunity to combine refraction and reflection surveys will be considered in planning the regional geophysical lines. The refraction lines will be acquired using programmed remote recorders, distributed over a distance that is much larger than a typical reflection spread. The number of large (e.g., 1000 kg) explosive shots available for refraction survey is limited, so possibilities for piggy-backing surveys are limited. Data from previous refraction surveys (described in Section 2.4) suggest that under certain conditions useful records could be obtained from a long reflection-type geophone spread during a typical refraction experiment. To be useful for site characterization, the spread would need to be layed out over a refraction objective, proximal to shot points, when the refraction survey is performed.

Additional boreholes may prove necessary for mineral or energy resource assessment. Deep boreholes are not currently planned because specific site information is unavailable. Geophysics will contribute to siting decisions should the need arise, as addressed by the geophysics integration activity (SCP Section 8.3.1.4.1.2).

#### 3.1.4 GEOPHYSICAL STUDIES FOR GEOLOGY AND TECTONICS

### REGIONAL GEOPHYSICAL STUDIES

Study of geologic structure and tectonic processes in the Yucca Mountain region contributes to the development of tectonic models. Geophysical data will be used for evaluation of alternate hypotheses of past and present processes which have formed, and continue to affect the Yucca Mountain site. Use of regional geophysical data will involve comparison of the site area to terranes with known tectonic style and rate of activity, and direct investigation for the occurrence of features that could be important with respect to site performance. The product, a developed model for regional tectonics, is needed to assess the potential for ground motion, volcanism, magmatic activity, and faulting that could affect the design or the performance of repository facilities.

Regional geophysical studies involve deep surveys utilizing seismic, gravity, magnetic, and magnetotelluric methods for study of major structures (SCP Activity 8.3.1.17.4.3.1), and also characterization of faulting from surface indications by means of remote sensing, gamma-ray surveys, and other techniques (Activity 8.3.1.17.4.3.5). (Note that regional gravity and aeromagnetic mapping has already been done as described in Sections 2.1 and 2.2, respectively.) The following summary of regional geophysics refers to features and alternative models that are discussed in SCP Section 8.3.1.17.4.3; the reader is referred there for additional detail.

#### Regional Geophysical Lines

The regional geophysical exploration program is designed to acquire specific types of data (i.e., seismic, geoelectric, gravity, magnetic, remote sensing), which will be analyzed with other geophysical data (i.e., heat flow, stress), surface geologic data, and recorded seismicity.

Several geophysical transects are planned to cross the Furnace Creek fault, Walker Lane, and Yucca Mountain (Figure 3.1-2). The objectives for these regional geophysical lines, as explained in SCP Section 8.3.1.17.4.3.1, include investigation of the following:

- Width and subsurface geometry of extensions of the Furnace Creek fault zone.
- 2. The possibility of an incipient rift at Crater Flat.
- The relation of the Furnace Creek fault zone and other features to detachment faulting and Quaternary faulting in the vicinity of the site.
- Lateral heterogeneity of crustal structure in the Yucca Mountain area.
- 5. The relation of density, magnetic, and geoelectric structure with velocity structure for the Yucca Mountain region.
- Possible magma bodies in the vicinity of Yucca Mountain (see also Activity 8.3.1.8.1.1.3).

The description for SCP Activity 8.3.1.17.4.3.1 presents a compact summary of regional geophysical methods to be applied along regional traverses, which is further summarized below. Seismic refraction profiles will be used with gravity and magnetic lines to characterize upper-crustal structure in the Yucca Mountain area, and to identify major discontinuities in that structure. Refraction work is planned to complete a major east-west traverse across Yucca Mountain, and two intersecting north-south lines in the site vicinity (Figure 3.1-2). Existing seismic refraction data provide some evidence for the presence of a low-angle velocity contrast, possibly a detachment fault, extending from the Grapevine Mountains to the Yucca Mountain region. Detachment faults are believed to accommodate Cenozoic crustal extension of at least 100% in the Yucca Mountain region (Hamilton, in USGS, 1988). The existing high-resolution refraction data (Section 2.4) pertain mostly to the upper 4-5 km, which may be above the depth of detachment beneath Yucca Mountain. Additional, longer profiles are needed to follow faults into the Yucca Mountain region from exposure in the Grapevine and Funeral Mountains. Preliminary results from the southwest end of the Beatty profile (Figure 2.4-2) are encouraging, but this was a reconnaissance profile and is incomplete. Planned refraction lines will extend fully between Death Valley and the site area.

Seismic reflection will be used in lieu of, or in addition to, refraction and potential-field methods where higher resolution is required. Results from several recent reflection surveys of different types (Knuepfer

et al., 1987; Serpa et al., 1986; Brocher et al., 1989) indicate that structural features of importance to tectonic assessment of Yucca Mountain can be imaged (see Section 2.5). The method will be applied to features of interest, including (1) a 6-sec (two-way travel time) reflection in Death Valley that was interpreted as a brittle-ductile transition (deVoogd et al., 1986); (2) the subsurface extension of parallel northwest trending faults in the northern part of the Walker Lane (west-central Nevada), the Furnace Creek fault zone, Walker Lane; (3) a mylonitic detachment bounding the upper surface of the Precambrian rocks exposed in the northern part of the Amargosa Desert; and (4) the Paleozoic-Miocene contact. Based on peer review of the reflection feasibility test results (Activity 8.3.1.17.4.3.1) and the application of the planned regional traverses to site characterization, the method reported by Brocher et al. (1989), supplemented by gravity and magnetic profiling, will be used to characterize selected features that could be associated with significant ground motion, rupture, and magmatism within 100 km of the site. Current plans call for two east-west reflection lines crossing Yucca Mountain at the site area, and possibly another line extending from Death Valley across the Walker Lane to the vicinity of Yucca Mountain (Figures 3.1-2, -3, and -4).

Magnetotelluric soundings in the Yucca Mountain region indicate substantial conductivity contrasts between Precambrian crystalline rocks, Paleozoic aquifers and aquitards, argillaceous Paleozoic units, Miocene volcanics, and Quaternary basin fill. The method permits local characterization of gross features of crustal structure at low cost. Planned traverses will coincide with the refraction, gravity, and magnetics traverses described above, with additional intersecting traverses (Figure 3.1-2). In addition, a reconnaissance profile at larger spacing between soundings will extend from southern Death Valley where COCORP lines were located (Sarpa et al., 1988; deVoogd et al., 1986) northward to Yucca Mountain.

Detection/Characterization of Faults in the Yucca Mountain Region

Regional tectonic studies as planned in the SCP will focus on the relationship of the Death Valley-Furnace Creek fault zone and the Walker Lane to the Yucca Mountain site, but other potentially significant Quaternary faults lie within 100 km of the site. Because of the intensity of previous geologic investigations near the site, the density of known Quaternary faults there exceeds that for the surrounding region. A wide range of geophysical methods have been proposed to obtain more representative data on faulting within 100 km of the repository site, as indicated in Table 3.1-1. Each method has the potential to contribute interpretable information, but certain methods have been demonstrated to be more effective than others as indicated in the table by different plans for implementation.

Faulting within 100 km of the site will be characterized by review of existing data, air-photo interpretation, and ground reconnaissance. Also, evaluation of structural domains within the 100 km distance is the objective of Activity 8.3.1.17.4.3.5, whereby Landsat V thematic mapper imagery will be purchased for the quadrangles covering the Yucca Mountain region. These images will be analyzed for evidence of lineaments that may be associated with concealed faults and fracture patterns, and for evidence of hydrothermal deposits associated with igneous activity. The same thematic mapper data set







will also be examined in Activity 8.3.1.17.4.9.1 for indications of desert varnish that may be associated with tectonically stable areas.

Regional geophysical activities will involve the detailed study of a number of specific faults in the Yucca Mountain region. Some of these studies are ongoing, or are substantially complete as discussed below and in USGS (1988). The following paragraphs describe plans for further study of several faults, in a manner that is similar to previous studies. Also, several feasibility tests are described that will determine the applicability of shallow seismic refraction and Mini-Sosie reflection methods for fault characterization.

An activity (8.3.1.17.4.7.8) is planned to further evaluate the application of the Mini-Sosie method of shallow reflection at and proximal to the site area. Past results are mixed for imaging of faults in alluvium because the faulting is typically complex, and the reflecting horizons are sparse and weak. Harding (in USGS, 1988) reports moderate success at Crater Flat and the Beatty scarp, but some uncertainty remains as to application of the method for characterizing amount of fault offset and other aspects. Accordingly, this activity calls for nine profiles within the site area and at other selected features, of which two are designated preliminary traverses (the Jackass Flats and Fran Ridge traverses of Figure 3.1-3) and will be performed before the others. A decision to proceed with all profiles, or to apply the method more selectively, will be based on preliminary results.

Subsurface expression of the Rock Valley fault will be mapped using shallow refraction and reflection (Activity 8.3.1.17.4.4.1). This activity is 90% complete (see USGS, 1988). Quaternary displacement along the main fault trace was investigated using low-energy seismic refraction. Refracting horizons reported by Rodriguez and Yount (in USGS, 1988) suggest that it is possible to detect fault offset and other characteristics from refraction continuity and moveout. Deeper faulting (up to several hundred meters) was examined using the Mini-Sosie method (Harding, in USGS, 1988). The southern extension of the Rock Valley fault system south of the Amargosa Valley fault will be investigated using the same methods. The objectives are to estimate fault width, detect Quaternary displacement, and detect horizontal and vertical components of slip. The same techniques used to evaluate the main trace will be used to evaluate whether the faulting events that have been identified involve the southern extension of the fault zone.

The Mine Mountain fault zone will be evaluated primarily by synthesis of available geologic information (Activity 8.3.1.17.4.4.2). The seismic reflection data of McArthur and Burkhard (1986) will be reviewed. Mini-Sosie shallow reflection data will be acquired depending on whether the method provides useful information on fault offset and configuration (Activity 8.3.1.17.4.7.8). In a similar study, the Cane Spring fault system, proximal to the Mine Mountain and Rock Valley faults, will be investigated using shallow seismic refraction to evaluate the origin of known lineaments which may be associated with faulting of Quaternary deposits (Activity 8.3.1.17.4.4.4).

The Stagecoach Road fault south of the site will be investigated to estimate the nature and rate of faulting, and the relationship to the Paintbrush Canvon-Busted Butte fault zone (Activity 8.3.1.17.4.4.3). This fault may be important to estimates of ground motion at the site because of its proximity to the repository site, and because models suggest that rupture on this fault may occur concurrently with activity on other faults. Mini-Sosie data also will be acquired depending on a determination in Activity 8.3.1.17.4.7.8 that the method provides useful information on fault offset and configuration.

## Characterization of Volcanic Features

Geophysics is an important part of planned studies of volcanic and intrusive igneous features and processes in the Yucca Mountain region. Aeromagnetic, seismic, and gravity data generated in other studies of the SCP (notably Studies 8.3.1.17.4.3 and 8.3.1.17.4.7) will be used to evaluate structural controls on basaltic volcanic activity (Study 8.3.1.8.1.1). Cluster analysis will be done on the location of surface and subsurface volcanic deposits, based partly on aeromagnetic data. The presence of magma bodies beneath the site will be evaluated using a range of geophysical methods, each of which is planned for a different study in the SCP. Such methods include high-resolution upper-crustal seismic refraction and deep seismic reflection (Activity 8.3.1.17.4.3.1), gravity (8.3.1.17.4.7.2), magnetics (8.3.1.17.4.7.3), magnetotellurics (8.3.1.17.4.3.1), Curie isotherm interpretation (8.3.1.8.5.2.1), and teleseismic tomography (8.3.1.8.1.1.3).

Four aeromagnetic anomalies already identified in Crater Flat and Amargosa Valley (Kane and Bracken, 1983) will be resurveyed at ground level or along flightlines 100 m above the surface and spaced 100 m apart. A detailed gravity survey will be acquired on a grid of about 30 m dimension (Study 8.3.1.8.5.1). These data, which will be collected under another study (see Activity 8.3.1.17.4.7.4), will be used to estimate volumes of buried volcanics and shallow intrusions, and to site boreholes. A feasibility test is planned (see Section 3.3 of this report) for evaluating the use of geophysics for detection and delineation of volcanic deposits. Paleomagnetic orientation from drill cores will be used to estimate the age of lava flows and shallow intrusions (see Activity 8.3.1.4.2.1.5). The Curie isotherm will be reinterpreted in more detail (see Section 2.2 of this report) as new detailed aeromagnetic data become available, for investigation of the possibility of upwelling magma (see Activity 8.3.1.8.5.2.1). Reinterpretation of the Curie isotherm will include appropriate consideration of edge effects in the analysis method, as they may affect the quality of interpretation throughout the Yucca Mountain region.

#### Teleseismic

Teleseismic tomography is included as an analysis technique in SCP Activity 8.3.1.8.1.1.3 (Presence of magma bodies in the vicinity of the site). Data for tomographic analysis are collected under Activity 8.3.1.17.4.1.2 (monitor current seismicity). A related higher resolution technique called "NeHT" tomography (pronounced "net"; Evans and Zucca, 1988) is not included in current plans, but would be useful in several studies (see item 4 below). Other activities in which these techniques may be useful include (1) Activity 8.3.1.8.2.1.3 (Probability and rate of faulting), (2) Investigation 8.3.1.17.1 (Studies to provide required information on volcanic activity that could affect repository design or performance), and (3) Study 8.3.1.17.4.5 (Detachment faults at or proximal to Yucca Mountain). The relation of teleseismic tomography to other types of surveys will be examined by the geophysics integration activity (SCP Section 8.3.1.4.1.2).

A number of structural hypotheses that are important to site characterization can be addressed by teleseismic tomography. The regional-scale study and the north fringe of the fine-scale study indicate that complex structures are present beneath the Timber Mountain-Oasis Valley caldera complex. These structures may include low-velocity lows such as those beneath Crater Flat. The low-velocity upper-mantle anomaly beneath the southern part of the NTS and the overlying crustal structure should be imaged at higher teleseismic resolution to investigate its tectonic significance.

Based on preliminary results discussed in Section 2.8 of this report, four restricted-array seismic tomography studies are under consideration:

- Higher resolution P-wave teleseismic tomography studies of the southern NTS/Yucca Mountain region to image the upper-mantle velocity-low and its Moho-depth connections, if any, to the Crater Flat columnar velocity-low. (This study will clarify the nature of the basalt magma system, possibly identifying the size and location of the magma source and any major active intrusion paths.)
- 2. S-wave teleseismic tomography and  $Q_p$  teleseismic tomography of the Crater Flat columnar velocity-low to permit better detection and definition of melting in the middle and lower crust. These parameters can be obtained from any array of intermediate-period seismometers. P-wave velocity alone is ambiguous in this regard because many things can cause small variations in that single parameter. Knowledge of both velocities and  $Q_p$  permits a better interpretation.
- 3. A higher resolution P-wave teleseismic tomography study of the Timber Mountain/Silent Canyon area would provide detailed crustal and upper-mantle information, principally for comparison with Crater Flat. Existing data suggest that a columnar velocity-low exists under the Timber Mountain caldera, which would imply that this is a common feature of inactive silicic calderas, and thus support an alternative interpretation of the Crater Flat velocity-low.
- 4. An active-source high resolution "NeHT" tomography study of the Yucca Mountain site area can image the tuff/Paleozoic boundary with 500 m or better accuracy, resolving the location and orientation of the Crater Flat east-boundary structure. The same experiment can evaluate the occurrence of upper-crustal silicic magma chambers as small as 500 m across. This technique is applicable to Study 8.3.1.17.4.7 (Subsurface geometry and concealed extensions of Quaternary faults at Yucca Mountain) and Activity 8.3.1.8.1.1.3 (Presence of magma bodies in the vicinity of the site).

#### Support of Paleoclimate Studies

A shallow (<1.0 sec travel time) high-resolution seismic reflection method, based on the feasibility testing of various techniques (Activity 8.3.1.4.1.2), may be utilized in the Paleoclimate Study (8.3.1.5.1.2) to

analyze the stratigraphy-sedimentology of marsh, lacustrine, and playa deposits (Activity 8.3.1.5.1.2.2). This activity requires establishing the three-dimensional lithostratigraphic framework of various marsh-lacustrineplaya environments, which are characterized by sequences of fine-grained clastic deposits. Seismic reflection profiles will be used to define this framework, and to provide guidance for establishing the optimal locations for drilling sites. Seismic reflection properties of these fine-grained clastic sequences may be substantially different than the reflection properties of the highly fractured Miocene volcanic sequences comprising Yucca Mountain; consequently, different instrumentation may be required. Various reflection systems (e.g., Mini-Sosie, sledgehammer shear wave source, vibrator source, land air-gun) will be considered for possible application, in an effort to obtain optimum results.

#### 3.1.5 GEOPHYSICAL STUDIES IN THE SITE AREA, FOR GEOLOGY AND TECTONICS

#### STRUCTURE AND STRATIGRAPHY

The work described jointly by SCP Studies 8.3.1.4.2.1, 8.3.1.4.2.2, and 8.3.1.17.4.7 will involve extensive collection of geophysical information from the site area in support of the site tectonic model. Data on the distribution of mass, geoelectric features, seismic velocity structure, magnetization, and seismic reflectors will be used to detect and delineate features with possible tectonic significance. Data also are needed to map known Quaternary faults where concealed by surficial deposits, to evaluate whether Quaternary faults exposed as high-angle faults continue as such to depth or flatten and merge with one or more low-angle faults, and to evaluate the continuity of rock units within the repository block and controlled area.

Surface geophysics will also be used to increase confidence in stratigraphic models of Yucca Mountain. Detailed gravity, ground magnetics, EM soundings, high-resolution seismic reflection, and upper-crustal high-resolution seismic refraction will be used, as appropriate, to establish and interpret sites for drillholes (Activity 8.3.1.4.2.1.1). Much of the planned geophysic work that is applicable to site stratigraphy and the geologic model is described for Study 8.3.1.17.4.7. The following is a discussion of planned activities that will provide information on the geologic setting, and the nature of faulting in the site area.

## Detailed Gravity

Detailed gravity data will be collected to supplement the currently available gravity maps for the site area, which are constructed from observations too widely spaced for discrimination of faults or detection of discontinuous rock units (Activity 8.3.1.17.4.7.2). A 1:24,000-scale map will be developed from new data collected at 60-m spacing along east-west lines spaced 150 m to 300 m apart, topography permitting (Figure 3.1-5). A total of approximately 7,500 additional gravity stations are planned for this effort. Additional high-resolution gravity work is planned for assessing buried volcanics at selected locations in the Amargosa Desert (Study 8.3.1.8.5), and for investigating the large hydraulic gradient northwest of the site as discussed in Section 3.1.2 of this report. The following paragraphs discuss the capability of detailed gravity to detect faults in the site area. Because density generally increases with depth (Snyder and Carr, 1982), terrain-corrected gravity is generally lower on the downthrown side, and higher on the upthrown side of a fault with vertical offset. To examine the relative magnitude of this effect, a two-dimensional profile across the site area was postulated, using density data from borehole gravity and gamma-gamma density log surveys. Basic features of the model were taken from surface geologic mapping. A density of 2.8 gm/cc was used below the Paleozoic basement.

Hypothetical faults with offsets of 40, 80, 160 and 500 m cause gravity anomalies with amplitudes of 0.2, 0.4, 0.9, and 3 mgal, respectively, and spatial extent about 500 m. Hence, station spacing should be about 100 m perpendicular to the expected strike and the gravitational effect of faulting may be masked by terrain effects, which are difficult to correct because of non-uniform terrain density.

Where terrain effect is minimal, fault interpretation depends on data uncertainty from other sources including gravity measurement, station location, and reduction parameters. Measurement errors of less than 0.3 mgal can be achieved if high precision measurement techniques and drift correction methods are followed. Location errors affect vertical control producing gravity uncertainty of 0.2 mgal per meter of elevation. Elevation can be controlled to 0.10 m using an electronic distance measuring device, limiting the error to about 0.02 mgal.

Much of the uncertainty in the reduction procedure is caused by the terrain correction, which is generally assumed to be accurate to within 5% of its total value. Typically, terrain corrections at the site area are as large as 2 mgal, with 0.10 mgal uncertainty. However, with the expected availability of a gridded (30 m) digital elevation model, terrain corrections should be more accurate, perhaps to 0.02 mgal. The sum of errors from measurement, elevation control, and corrections gives a total expected uncertainty of less than 0.07 mgal. Thus the detailed gravity method should detect steeply dipping faults with vertical offset on the order of 40 m in gentle terrain, provided that the terrain effect is minimal, adequate station spacing is maintained, and profiles are oriented perpendicular to strike.

Borehole gravimetry can be used to obtain bulk density, and detect contacts between different media in the immediate vicinity of the borehole. In principle, borehole gravity measurements are sensitive to density differences in a region extending in every direction for about three times the spacing between measurements in the borehole (for typical station spacings). Thus the method might be used to infer contacts between different stratigraphic units such as highly lithophysal and nonlithophysal tuffs over a large area of the repository block, given sufficient density contrast and sufficiently accurate terrain corrections. This possibility will be investigated using presently available data from four boreholes at Yucca Mountain (Healey, 1984, 1986) and terrain corrections calculated with high-resolution digital topographic. Borehole gravity will also provide bulk density information for terrain corrections of surface gravity stations. The existing borehole gravity data, and possibly data collected in the future from other boreholes, will be used for this purpose.



-

#### Aeromagnetic and Ground Magnetic Surveys

Aeromagnetic maps of the site that are currently available are based on flightlines spaced 1/4 mile apart (see Section 2.2), too widely spaced to resolve ambiguities in fault continuity or to demonstrate continuity of rock units in the site area. A 1:12,000-scale map will be developed from data to be collected along flightlines spaced 1/16 mile apart, and draped over the topography at 400 ft (Activity 8.3.1.17.4.7.3). The area to be covered by this survey will be comparable to previous surveys of the site area discussed in Section 2.2; the actual area will be determined by the Principal Investigator with input from the geophysics integration activity, and will reflect data needed for various applications including assessment of volcanic hazards and mineral resources.

Ground magnetic surveys of specific features within the site area are planned to measure variation in magnetization of specific strata and surface deposits (Activity 8.3.1.17.4.7.4). Ground magnetic data will be collected on foot, at 3- to 60-m intervals, to characterize the location and continuity of rock units, and to investigate known and inferred structures in the vicinity of certain faults, shafts, surface facilities, drillholes, and aeromagnetic anomalies. A recording base-station magnetometer will be used during the surveys as appropriate. Not all such features will be surveyed; the number and location will be determined by the Principal Investigator with input from the geophysics integration activity. Ground magnetic data also will be collected along geophysical traverses including the detailed gravity lines (Figure 3.1-5), and other profiles intended to detect concealed faults in the site area or to investigate the steep hydrologic gradient beneath northern Yucca Mountain.

As an independent test of the feasibility of using magnetics to detect concealed faults, it is proposed that truck magnetometer profiles be collected along roads and in some off-road areas of southern Yucca Mountain. The off-road profiles would be collected along closely spaced (30 m) parallel traverses perpendicular to fault trends. Positioning could be accomplished using range-range radio navigation. Magnetic gradients traced across known and inferred fault traces would provide information for evaluation of the magnetic method for characterizing buried faults and contacts.

Ground magnetic traverses will provide estimates of the resolution of ground magnetic data to detect buried faults and lithologic contacts in the Yucca Mountain area. Simple upward continuation of these profiles will be used to evaluate the trade-off between increased signal-to-noise ratio and decreased short-wavelength resolution in aeromagnetic data, as a function of terrain clearance.

Detailed ground magnetic and gravity surveys will be performed near USW G-5, G-6, and G-7, associated with these planned 1,500-m cored boreholes. Information on magnetic and density variation will be used to map the local vertical and lateral distribution of stratigraphic units in order to augment the structural and stratigraphic information obtained from the cored boreholes.

#### Geoelectric Methods

The results of electrical surveys previously carried out at Yucca Mountain (Section 2.3 and Table 2.3-1) must be evaluated before planning future surveys, as stipulated in Activity 8.3.1.17.4.7.5. Surveys over Yucca Mountain . . be designed to detect lateral electrical boundaries associated with faults and lithological variations in a bedrock environment which is generally resistive (greater than 300 ohm-meters) at the surface and becomes more conductive with increasing depth. Survey design and interpretation must contend with rugged topography and near-surface variability in the weathered zone, both of which produce changes of first order in the electrical data. In order to account for such three-dimensional effects, the Project requires complementary overlapping data sets that use the capabilities of different methods to achieve maximum depth of penetration with minimum distortion by lateral effects. The interpretation methods and software available for each electrical method play a major role in the ultimate uscuracy and resolution of the final interpretation, and hence are factors in survey design.

Recent advances in field and interpretative time-domain technology make the time-domain method a primary candidate for deep sounding (greater than 1,000 m) and profiling, supplemented by a loop-loop frequency-domain electromagnetic method for shallower (less than 200 m) depths. The audio-magnetotelluric method, supplemented by telluric ratio measurements, is also important for deep sounding and profiling because the data are highly amenable to the interpretation of three-dimensional structures. Several east-west lines over Yucca Mountain utilizing combinations of these techniques will provide insight on structures related to the Yucca Mountain block.

The same techniques are also useful for fault detection in the alluvial valleys nearby. They would be supplemented by Schlumberger soundings to provide a high resolution definition of the conductive overburden that must be accounted for in interpreting deeper structure.

As previously discussed in Section 3.1.4 of this report, magnetotelluric soundings with their unique capability to penetrate to depths exceeding 3 km are the most desirable technique to examine the regional geoelectric structure. These regional data will provide important lithologic and structural constraints that will serve in the tectonic evaluation of the Yucca Mountain block.

## Radioactivity and Remote Sensing

Airborne and ground-based gamma-ray surveys will be evaluated for use in detecting buried faults in the site area (Activity 8.3.1.17.4.7.6). Buried faults must be detected by near-surface distribution of radioelements, possibly caused by percolation of meteoric waters or migration of radon or dissolved elements in a fault zone. Available airborne radioactivity data from the NTS (both classified and unclassified) will be evaluated. Hand-heid gamma-spectrometers will be tested along short (100 m) traverses across the Paintbrush Canyon, Bow Ridge, and Stagecoach Road faults. This is planned as a feasibility test; application of the method to site characterization will depend on the results. An airborne gamma-ray survey, if warranted, will be

performed simultaneously with the planned aeromagnetic survey (Activity 8.3.1.17.4.7.3).

## Seismic Reflection

Intermediate-depth reflection and refraction methods will be evaluated for application to site characterization, and possible applications of these methods will be planned in Activity 8.3.1.17.4.7.1. This is a planning activity only, which will review previous attempts and currently available technology, and assess the potential contribution of seismic methods to characterization of faults in the site area. This review will be separate from the review of the planned regional geophysical traverses (Activity 8.3.1.17.4.3.1), but both deep- and intermediate-depth objectives for seismic lines in the site area and vicinity (Jackass Flats, Crater Flat, northern Amargosa Valley) will be considered. Results from the Amargosa Valley seismic reflection test (Brocher et al., 1989) are applicable, possibly in different ways, to both reviews.

Tentative plans presently call for two deep-seismic lines across Yucca Mountain, to be conducted in a manner similar to the Amargosa Valley reflection test (Brocher et al., 1989). These lines will intersect with a nrrth-south reflection line in Crater Flat and other north-south geophysical profiles (refraction and MT) to the east and west of the site area (Figures 3.1-2, -3, and -4). Abundant information on the intermediate-depth structure was obtained from the Amargosa Valley test, but the structural setting of the upper 2-3 km at Yucca Mountain is relatively complex, and the same methods may not apply as well. This aspect of the planned lines across Yucca Mountain will be evaluated in the peer review planned for Activity 8.3.1.17.4.7.1, as will the objectives for deep- and intermediate-depth exploration within the site area and vicinity, the applicability of three-dimensional seismic methods, intersecting geophysical lines in the site area, and the applicability of VSP.

The report of previous attempts at intermediate-depth seismic reflection at Yucca Mountain (McGovern, 1983) recommends avoiding further application of reflection methods. The report identifies deficiencies in the 1980 and 1981 surveys that were addressed in the 1982 survey, which is represented as a thorough but unsuccessful application of available technology. However, Jones et al. (1987) believe that other techniques are now available and should be tried at Yucca Mountain. Possible improvements to the 1982 survey include (1) scoping studies using downhole recordings in available boreholes; (2) location of lines in simple structural settings, oriented perpendicular to structure; (3) use of the stack array concept (Jones et al., 1987) for noise suppression; and (4) geophone placement strategies including single phones and/or closely spaced groups to support statics analysis.

VSP surveys can and should be used with existing boreholes, prior to large-scale reflection surveys, to characterize intrinsic and apparent attenuation, mode conversion and splitting, reverberation, and the nature of reflections expected from the upper 2-3 km at the site. Different sources, source offsets, and surface receiver spreads are also used in conjunction with VSP geometry to evaluate acquisition parameters for surface seismic reflection. Previous reflection lines at the site were situated proximal to ridges, canyons, and known faults, and may not have had sufficient noise rejection capability to handle side-scattered energy. As pointed out by Brocher et al. (1989), an important difference between Vibroseis and impulsive sources is that the Vibroseis source is still active when the surface wave noise train reaches the geophones, because the Vibroseis source records are uncorrelated in time. Some potentially significant improvements in noise control between the Amargosa Valley test line (Brocher et al., 1989) and the 1982 survey at Yucca Mountain (McGovern, 1983) relate to wind monitoring and onsite processing capability.

It is apparent from the preceding comments that significant improvements may be possible, particularly with the use of VSP to reveal some of the mechanisms that affect reflection data from the site. As stated above, this topic will be considered as part of Activity 8.3.1.17.4.7.1 and the geophysics integration activity.

The high-resolution Mini-Sosie reflection method (see Section 2.5) may be applied to characterization of structures in the site area, if appropriate, based on evaluation of the technique in Activity 8.3.1.17.4.7.8. A number of shallow seismic reflection profiles will be used to study the position of possible marker horizons, at the site. The planned lines (see Section 3.1.4, and Figure 3.1-3) will be used to trace the subsurface extent of known structures (e.g., faults) where they are concealed by surface deposits, and to investigate the lateral continuity and structural features of reflecting horizons. Other high-resolution refraction or reflection methods may be tested, depending on results obtained from further Mini-Sosie work.

#### Seismic Refraction

An east-west refraction profile will be acquired across the Prow (Scott and Bonk, 1984) and Yucca Wash to explore for a lateral velocity contrast, possibly associated with changes in lithology caused by structural displacement or alteration processes, that could help to explain the large hydraulic gradient north of the site (Activity 8.3.1.4.2.1.2). This line will be located in the vicinity of planned geologic coreholes USW G-5 and G-6. Plans for this refraction line will be reviewed in the planned peer review of seismic methods (SCP Activity 8.3.1.17.4.7.1; see Section 3.2 of this report). The tentative location of this line is plotted on SCP Figure 8.3.1.4-6.

As stated above, an assessment of stratigraphic information needed from the site area will be considered along with tectonics in the planned review of the applicability of seismic reflection and refraction to intermediatedepth (<2 or 3 km) structural profiling (Activity 8.3.1.17.4.7.1). Past experience (described in Section 2.4) shows that the high-resolution upper-crustal refraction method as described by Ackermann et al. (in USGS, 1988) can produce interpretable, but relatively low resolution structural information. On the other hand, the lower energy, higher resolution surveys reported by Pankratz (1982) yielded signals of relatively low quality, which were difficult to reconcile with stratigraphic and velocity data from nearby boreholes. The underlying basis for these observations is undoubtedly complex, and will be taken up in the planned review.

## CHARACTERIZATION OF FRACTURES AND FAULTS IN THE REPOSITORY BLOCK

5. 1

Information on the frequency and subsurface distribution of faults and fractures is needed principally from the UZ, because the DOE plans to rely on natural barriers between the repository and the water table to demonstrate site performance. Accordingly, many of the methods used to detect and characterize subsurface fractures in the SZ are of secondary importance or are inapplicable. The limited range of methods which do not require water-filled boreholes, or water-filled fractures, will be thoroughly investigated during site characterization. Also, because of the unique aspects of fracture characterization in the UZ, a multidisciplinary approach is planned to compare different methods at the same location. At the USW UZ-9 complex of boreholes east-southeast of the site, and in the ESF, geophysical data will be acquired from regions of the rock mass for which fracture data are available from other techniques, including borehole packer testing and shaft wall mapping.

## Borehole Geophysics

Borehole television is known to produce useful information on fracturing in the U2; it is part of the standard suite of logs planned at Yucca Mountain. The borehole acoustic televiewer, resistivity, SP, and full-waveform acoustic logs will be used in the SZ, as discussed in the literature (for example, Paillet, 1980) and in Section 3.1.2. However, fracture data from the SZ are not generally applicable to the unsaturated portion of the repository block because of differences in lithclogy, structural context, and in situ stress distribution. Morever, future boreholes at Yucca Mountain will be largely restricted to the UZ as a target for exploration (in terms of numbers of holes, and spatial coverage). Thus the opportunities for use of more conventional SZ geophysical tools and extension of this database to the UZ will be limited.

Large-spacing EM and resistivity logs will be tested and evaluated for application to fractured and lithophysal zones, and possibly for characterizing anomalies detected by other surface and borehole geophysical methods. Application of these methods to site characterization, particularly for UZ characterization, will depend on the outcome of feasibility testing.

Borehole-to-surface resistivity and EM methods will be used depending on results from surface and borehole surveys. The borehole-to-surface resistivity technique has been applied at Yucca Mountain (Daniels and Scott, 1981), as discussed in Section 2.3, with results that are difficult to interpret geologically. Borehole-to-surface EM studies are discussed below.

#### EM Tomography

Current plans are to focus on application of existing technology that is suitable for characterization of faulting, fracture distribution, and trends in moisture content in the UZ at Yucca Mountain (Activity 8.3.1.4.2.2.3). Various borehole radar tools are available for use by the Project (e.g., from the USGS and Sandia National Laboratories). In addition, borehole radar tools have been developed for repository site characterization by the Swedish and German repository programs and could probably be made available for evaluation at Yucca Mountain. However, borehole radar is not expected to perform in the UZ at Yucca Mountain as it performs in crystalline rock or salt, because of the proliferation of moisture. Fractures and fault zones probably do not behave as radar reflectors, although major stratigraphic contacts may behave as such. Limited scoping analyses and field trials will be considered during site characterization to evaluate borehole radar and related radar methods.

The operating frequency of available borehole radars varies between about 20 and 80 MHz. Depending on the in situ resistivity, which can vary from 20 to 2,000 ohm-meters in the UZ, two-way radar penetration varies from roughly 10 to 100 m (D. Wright, personal communication). This type of performance is consistent with hole-to-hole penetration for imaging purposes it either the "c-series" boreholes or the planned USW UZ-9 complex.

VHF tomography is similar to crosshole radar, generally differing with respect to detection method and frequency. Ramirez and Daily (1987) reported a series of measurements made in welded tuff at the G-tunnel, using crosshole equipment operating at 200 MHz to image the rock mass both before and after a heater test was conducted. The length of the travel paths in this experiment ranged up to several meters. Results indicate that detectable changes in dielectric permittivity are associated with processes that are important for site characterization. The method is planned for use in the waste package (heater) test series in the ESF, and could possibly be used for mapping the extent of water injection from various hydrology tests, such as the planned Radial Boreholes Test in the ES-1 shaft. VHF tomography as used in G-tunnel produces images with signal strength and spatial resolution better than what can be expected from tools operating at lower frequency in widely spaced surface-based boreholes. The method might be used in the closely spaced, planned USW UZ-9 complex of boreholes; however, this application would not produce alterant data as acquired in G-tunnel and planne; for the ESF. Water injection testing in the UZ boreholes is only tentatively planned and will not be conducted until late in site characterization.

If crosshole VHF tomography results show that rock mass features detected in boreholes can be mapped between boreholes, and that penetration is useful, then the method may be extended to a borehole-to-surface configuration. With three-side access and raypath coverage of the plane intersecting parallel boreholes, the stability and resolution of tomographic algorithms are improved. Borehole EM or borehole-to-surface EM methods are also available and will be evaluated, as appropriate.

#### Vertical Seismic Profiling (VSP)

Planned VSP studies in the UZ at Yucca Mountain are discussed in detail in Section 3.1.1 above. Two studies are planned, one at the ESF and the other at the planned USW UZ-9 complex of boreholes. Each of these studies will use three-component sensors and a range of available sources will be considered, including vertical and shear vibrators, land air-gun, shear-impact sources, and explosives. Interpretation of VSP in boreholes (e.g., Activity 8.3.1.2.2.3.2) will be augmented by performing similar surveys in shafts and underground workings where more direct observations of fracturing and other rock mass characteristics can be obtained (Activity 8.3.1.4.2.2.5). If successful, the VSP approach to fracture and fault characterization in the repository block may provide a means of mapping the nature and distribution of fractures in the subsurface. Also, the method could be applied to performance confirmation, both in the sense of acquiring baseline data on the mechanical condition of the repository block, and for monitoring changes as a result of repository construction and operation.

## 3-D CHARACTERIZATION OF LITHOLOGY AND HYDROLOGIC STATE OF THE REPOSITORY BLOCK

## Borehole Geophysics

A standard suite of geophysical logs will be acquired from every deep borehole that penetrates the Topopah Spring Member in the site area (Activities 8.3.1.4.3.1.1, 8.3.1.4.2.1.3, and 8.3.1.2.2.3.2). In addition, logs are already available from 40 existing boreholes. Depending on the development of methods for correcting and interpreting various log responses from the unsaturated tuffs (as discussed in Section 2.7), these logs could become the most continuous record of lithologic and hydrologic variability available, with better spatial coverage of the repository block (vertical and lateral) than other types of data.

Commercial logs to be obtained as a standard suite from future boreholes in the UZ at Yucca Mountain include compensated density, induction, spectral gamma, epithermal neutron (single shielded detector), dielectric log (47 MHz), temperature, and caliper. In the SZ, the standard suite will include resistivity, SP, induced polarization, full waveform sonic, and the compensated neutron porosity log. The total magnetic intensity log, and possibly the magnetic susceptibility log, are indicative of certain horizons in the tuff sequence at Yucca Mountain and will become part of the standard suite of logs, although they may not be available commercially. It is anticipated that this full suite will adequately support lithologic interpretation and correlation.

The compensated density log will be used to characterize lithophysal zones where only cuttings are available, or where poor resolution exists on borehole television. Preliminary analysis shows that spikes prevalent on compensated density logs from Yucca Mountain are generally associated with borehole rugosity and not lithophysal porosity. The compensation algorithm for dry holes in volcanic tuff is presently being evaluated.

The induction and spectral gamma logs are expected to serve as indicators of smectite- and zeolite-rich intervals, and to identify key stratigraphic marker intervals at the top and base of major ash flows, which commonly show increased alteration. Preliminary indications are that the induction log can be used only qualitatively in much of the UZ, because the formation resistivity is so high that the signal is small and accurate resistivity values cannot be obtained.

The dielectric log, discussed in Section 3.1.1 of this report, will also be evaluated for detecting trends in hydrologic characteristics such as moisture content. Propagation of EM energy in the UZ at Yucca Mountain is dependent on pore geometry, saturation, and mineral alteration (which may also be related to moisture content).

The second objective of logging for three-dimensional characterization is to determine the distribution of rock characteristics (physical properties) within lithostratigraphic units in each borehole and then to map their distribution throughout the site. The physical property logs are essentially "computed logs" of such parameters as bulk density, porosity, water content, and alteration chemistry. The ability to compute these parameters depends on initial log quality, availability of core data, and development of algorithms to convert log quantities to the desired physical properties. These activities are part of the petrophysics and log analysis program planned for site characterization (SCP Studies 8.3.1.4.2.1 and 8.3.1.2.2.3). (Once the computed logs are available in a borehole, the physical properties must be averaged over vertical intervals which are consistent with lithological and hydrological zonation. Only then can these important parameters be mapped over the site area.) The application of geostatistical methods to modeling of the spatial variability of these log products will be investigated when the calculated logs become available.

#### 3.1.6 ENGINEERING GEOPHYSICS FOR SURFACE FACILITIES SITING

Various locations in Midway Valley, in the vicinity of Exile Hill, have been identified as possible sites for the repository surface facilities, particularly for the structure containing the waste handling and packaging system (see Chapter 6 of the SCP). As indicated in the issue resolution strategy for SCP Section 8.3.1.17, principal concerns for surface facilities siting, which will be addressed by geophysical exploration, are (1) avoiding fault displacement of more than a few inches under the waste handling structure; and (2) meeting probabilistic limits for peak ground motion, time histories, and response spectra. The associated primary performance goals are therefore to locate a site where the probability of total net fault displacement for the preclosure time period is less than 1 in 100, and where descriptors of predicted ground motion meet certain goals.

Investigations in Midway Valley are important for obtaining the following information: (1) detection and characterization of faults including those buried by alluvium, (2) assessment of the seismic potential of faults at or near the surface facilities sites, and (3) characterization of the dynamic properties of surficial deposits. Geophysical data are needed, although not required, for detecting and delineating fault offsets in the alluvium-bedrock contact and stratigraphic horizons within the alluvium. Such information can reveal where major faults emerge from the bedrock, and provide input for calculating the potential for slip associated with the surface expression of these faults. Shallow geophysical surveys can be used in conjunction with trenching across Midway Valley (Study 8.3.1.17.4.2) to trace the lateral extent of fault offsets that may be observed in the trench walls. Seismic studies will provide information on velocities; however, in situ measurements using borehole, crosshole, and surface-to-borehole methods are also needed where appropriate to investigate the detailed structure of velocity and other properties affecting the response of the waste-handling structure to ground motion.

#### TRENCHING AND GROUND PENETRATING RADAR IN MIDWAY VALLEY

Low-energy seismic reflection and refraction studies have been performed in the vicinity of Exile Hill (see Section 2.5) and along an east-west refraction profile across the central part of Midway Valley (Section 2.4). Interpretations based on these surveys (Reynolds, 1985; Pankratz, 1982) tend to be poorly constrained and are difficult to reconcile with data from nearby boreholes (RF-series boreholes near Exile Hill, and the UE25a#1 borehole near the line traversing Midway Valley). Other seismic methods are available that may be more effective for detecting and delineating faults, such as the use of shot holes and/or explosives for shallow work (A. Flint, personal communication) or the use of shear wave sources (see Section 2.4). However, current plans call for evaluation of ground penetrating radar (GPR) before proceeding with further geophysics for fault offset detection at the Midway Valley surface facilities candidate site area. Studies in Yucca Flat (about 50 km northeast of Yucca Mountain) have shown that GPR is sensitive to faulting in the upper few meters of alluvium (L. Shephard, personal communication), and may be observed down to a depth of 10 m or more.

One possible outcome of the planned work is that there exists a marker bed, such as the Bishop Ash bed of Izett et al. (1970) that is accessible to trenching, is not offset by faulting, and is of sufficient age to render the. probability of faulting in 100 yr insignificant (100 yr is the approximate duration of the operational period for the repository surface facilities in the current conceptual design; see SNL, 1987). Alternatively, it may be possible to reach the same conclusion using soils dating methods without relying on a marker bed. Thus it is at least conceivable that enough information could be collected to establish that the site meets the goals discussed above for faulting and ground motion, without mapping the geometry of the alluvium-bedrock contact. Additional seismic exploration in Midway Valley may be necessary depending on the results from GPR and trenching studies (Study 8.3.1.17.4.2).

## EXPLORATION OF THE ALLUVIUM-BEDROCK CONTACT

The SCP (Studies 8.3.1.17.4.2 and 8.3.1.14.2.3) includes plans to map the alluvium-bedrock contact, if it should be required. Two candidate areas for exploration are the surface faulting area in Midway Valley and the proposed ESF location by Coyote Wash. Seismic refraction would be used for near-surface targets, electrical resistivity would be used for target horizons up to a few hundred meters, and Mini-Bosie or another reflection method would be used for deeper targets. Shallow refraction with a sledgehammer source was reported by Rodriguez and Yount (in USGS, 1988) to penetrate up to about 18 m in alluvium. Prototype shallow refraction testing in alluvium-filled washes in the site area produced similar results (see Section 2.4). One of the common electrical resistivity methods could be used for deeper features up to roughly 100 m. The Mini-Sosie method may produce interpretable reflections from the alluvium-tuff bedrock contact, where it is deep enough for this method (about 100 m or more) although (as discussed in Section 2.5) such contacts have not been found to produce strong reflections at Yucca Flat (N. Burkhardt, personal communcation). The reflection results

reported by Reynolds (1985) for the Exile Hill area are not indicative of possible ture results for the reasons discussed in Section 2.5.

## IN SITU ENGINEERING PROPERTIES

There are two basic objectives for characterizing the in situ dynamic properties of soil and rock: (1) direct measurement of compressional and shear wave velocities (and attenuation if possible) to obtain in situ dynamic moduli and Poisson ratio; and (2) correlation of stratigraphic continuity between borings, as a basis for modeling the variability of dynamic properties. Crosshole methods using weight-drop methods, and borehole logging using standard tools for lithologic and physical properties evaluation, will be considered in Study 8.3.1.14.2.3 when specific needs are developed. Because the existing and planned exploratory holes in the Exile Hill vicinity do not penetrate the water table, the use of many of the standard logging tools depends on maintaining a fluid column during logging. The use of such methods will be evaluated with respect to fluid losses during drilling, expected losses during logging, and the sensitivity of site performance to fluid invasion at the candidate locations for repository surface facilities. Methods of geophysical logging in "dry" holes, such as those planned for Activity 8.3.1.4.2.1.3, will also be considered.

#### REFERENCES

- Barbier, M.G., 1983. The Mini-Sosie Method, International Human Resources Development Corp., Boston, Massachusetts.
- Bath, G.D., and C.D. Jahren, 1974. Interpretations of Magnetic Anomalies at <u>a Potential Repository Site Located in the Yucca Mountain Area, Nevada Test</u> <u>Site</u>, U.S. Geological Survey Open File Report 84-120, 40 p.
- Beydoun, W.B., C.H. Cheng, and M.N. Toksoz, 1985. "Detection of Open Fractures with Vertical Seismic Profiling," JGR, vol. 90, p. 4,557-4,566.
- Brocher, T.M., P.E. Hart, and S.F. Carle, 1989. Feasibility Study of the Seismic Reflection Method in Amargosa Desert, Nye County, Nevada, U.S. Geological Survey Open File Report 89-133 (in press).
- Cor, D.P., and D.A. Singer (eds), 1986. Mineral Deposit Models, U.S. Geological Survey Bulletin 1693, p. 379.
- Craig, R.W., and J.H. Robison, 1984. <u>Geohydrology of Rocks Penetrated by</u> <u>Test Well UE-25p #1, Yucca Mountain Area, Nye County, Nevada</u>, U.S. <u>Geological Survey Water Resources Investigations Report 84-4248</u>, 57 p.
- Craig, R.W., and K.A. Johnson, 1984. <u>Geohydrologic Data for Test Well</u> <u>UE25 p#1, Yucca Mountain Area, Nye County, Neval</u>, U.S. Geological Survey Open File Report 84-450, 63 p.

- Czarnecki, J.B., and R.K. Waddell, 1984. Finite-Element Simulation of Ground-Water Flow in the Vicinity of Yucca Mountain, Nevada-California, U.S. Geological Survey Water Resources Investigation Report 84-4349, 38 p.
- Czarnecki, J.B., and W.J. Oatfield, 1987. "Use of Drillers' Logs and Geophysical Surveys to Define the Hydrogeologic Framework of the Amargosa Desert, Southern Nevada" (abs.), Transactions, American Geophysical Union, EOS, vol. 68, no. 16, p. 302.
- deVoogd, B., et al., 1986. "Death Valley Bright Spot: A Midcrustal Magma Body in the Southern Great Basin, California?," Geology, vol. 14, p. 64-67.
- Evans, J.R., and J.J. Zucca, 1988. "Active High-Resolution Seismic Tomography of Compressional Wave Velocity and Attenuation Structure at Medicine Lake Volcano, Northern California Cascade Range," <u>Journal Geophys.</u> <u>Res.</u>, vol. 93, p. 15,016-15,036.
- Fitterman, D.V., 1982. <u>Magnetometric Resistivity Survey Near Fortymile Wash</u>, <u>Nevada Test Site, Nevada</u>, U.S. Geological Survey Open File Report 82-401, 28 p.
- Frischknecht, F.C., and P.V. Raab, 1984. "Time-Domain Electromagnetic Soundings at the Nevada Test Site, Nevada," <u>Geophysics</u>, vol. 49, no. 7, p. 981-992.
- Greenhaus, M.R., and C.J. Zablocki, 1982. <u>A Schlumberger Resistivity Survey</u> of the Amargosa Desert, Southern Nevada, U.S. Geological Survey Open File Report 82-897, 151 p.
- Hasbrouck, W.P., 1987. "Hammer-Impact, Shear-Wave Studies," in S.H. Danbron and S.N. Domenico (eds.), <u>Shear Wave Exploration</u>, Geophysical Development Series, vol. 2, Society of Exploration Geophysicists, p. 97-121.
- Hasbrouck, W.P., 1988. "Five Shallow Seismic Feasibility Surveys," <u>Expanded</u> <u>Abstracts</u>, Society of Exploration Geophysicists, 58th Annual Intl. Meeting and Exposition, Oct. 30 - Nov. 3, p. 286-289.
- Healey, D.L., and C.H. Miller, 1971. Gravity Survey of the Amargosa Desert Area of Nevada and California, U.S. Geological Survey Report USGS-474-136, 29 p. (available only from the J.S. Department of Commerce, National Technical Information Service, Springfield, Virginia, 22151).
- Healey, D.L., 1986. Borehole Gravity Meter Survey in Drillhole USW G-4, Yucca Mountain Area, Nye County, Nevada, U.S. Geological Survey Open File Report 86-205, 18 p.
- Healey, D.L., F.G. Clustom, and D.A. Glover, 1984. <u>Borehole Gravity Meter</u> <u>Surveys in Drillholes USW G-3, UE25 p#1, Yucca Mountain Area, Nevada</u>, U.S. Geological Survey Open File Report 84-672, 16 p.
- Izett, G.A., R.E. Wilcox, H.A. Powers, and G.A. Desborough, "The Bishop Ash Bed. A Pleistocene Marker Bed in the Western United States," Quaternary Research, Interdispl. Jour., vol. 1, no. 1.

- Jones, G.M., M.E. Blackey, J.E. Rice, V.J. Murphy, E.N. Levine, P.S. Fisk, and R.W. Bromery, 1987. <u>Survey of Geophysical Techniques for Site</u> <u>Characterization in Basalt, Salt and Tuff</u>, Weston Geophysical Corp., Westboro, MA, prepared for the U.S. Nuclear Regulatory Commission, NUREG/CR-4957, 139 p.
- Kane, M.F., and R.E. Bracken, 1983. Aeromagnetic Map of Yucca Mountain and Surrounding Regions, U.S. Geological Survey Open File Report 83-616.
- Kane, M.F., M.W. Webring, and D.K. Bhattacharyya, 1981. A Preliminary Analysis of Gravity and Aeromagnetic Surveys of the Timber Mountain Area, Southern Nevada, U.S. Geological Survey Open File Report 81-189, 43 p.
- Knuepfer, P.L.K., et al., 1987. "Crustal Structure of the Basin and Range-Sierra Nevada Transition from COCORP Deep Seismic-Reflection Profiling," <u>Geological Soc. of America Bull.</u>, vol. 98, p. 488-496.
- Mabey, D.R., 1963. "Complete Bouguer Anomaly Map of the Seath Valley Region, California," U.S. Geological Survey Geophysical Investigations Map GP-305, scale 1:250,000.
- McGovern, T.F., 1983. An Evaluation of Seismic Reflection Studies in the Yucca Mountain Area, Nevada Test Site, U.S. Geological Survey Open File Report 83-912, 58 p.
- Paillet, F.L., 1980. "Acoustic Propagation in the Vicinity of Fractures which Intersect a Fluid Filled Borehole," <u>Trans. SPWLA Annu. Logging Symp.</u>, 24th, DD1-DD33.
- Pankratz, L.W., 1982. Reconnaissance Seismic Fefraction Studies at Calico Hills, Wahmonie, and Yucca Mountain, Southwes: Nevada Test Site, Nye County, Nevada, U.S. Geological Survey Open File Report 82-478, 25 p.
- Ponce, D.A., 1981. Preliminary Gravity Investigations of the Wahmonie Site, <u>Nevada Test Site, Nye County, Nevada</u>, U.S. Geological Survey Open File Report 82-931, 2332 p.
- Ramirez, A.L., and W.D. Daily, 1987. "Evaluation of Alterant Geophysical Tom-graphy in Welded Tuff," JGR, vol. 92, n. B8, p. 7,843-7,853.
- Robison, J.H., <u>Ground-Water Level Data and Preliminary Potentiometric-</u> Surface Maps, Yucca Mountain and Vicinity, Nye County, Nevada, U.S. Geological Survey Water Resources Investigations Report 84-4197, 8 p.
- Robison, J.H., D.M. Stephens, R.R. Luckey, and D.A. Baldwin, 1988. Water Levels in Periodically Measured Wells in the Yucca Mountain Area, Nevada, 1981-1987, U.S. Geological Survey Open File Report 88-468, 132 p.
- Rush, F.E., W. Thordarson, and L. Bruckheimer, 1983. <u>Geohydrology of Test</u> <u>Well USW H-1, Yucca Mountain, Nye County, Nevada</u>, U.S. Geological Survey Water Resources Investigation Report 83-4032, 38 p.

- Scott, R.B., and J. Bonk, 1984. <u>Preliminary Geologic Map of Yucca Mountain,</u> <u>Nye County, Nevada, with Geologic Cross-Sections</u>, U.S. Geological Survey Open File Report 84-494.
- Senterfit, R.M., D.B. Hoover, and M. Chornack, 1982. Resistivity Sounding Investigations by the Schlumberger Method in the Yucca Mountain/Jackass Flats Area, Nevada Test Silc. Nevada, U.S. Geological Survey Open File Report 82-1043, 38 p.
- Serpa, L.B., B. deVoogd, L. Wright, J. Willemin, J. Oliver, E. Hauser, and B. Troxel, 1988. <u>Structure of the Central Death Valley Pull-Apart Basin from</u> <u>COCORP Profiles in the Southern Great Basin</u>, Geological Soc. of America Bull., vol. 100, p. 1,437-1,450.
- Smith, C., H.P. Ross, and R. Ediquist, 1981. Interpreted Resistivity and IP, Section Line W1, Wahmonie Area, Nevada Test Site, Nevada, U.S. Geological Survey Open File Report 81-1350, 14 p.
- Snyder, D.B., and W.J. Carr, 1982. Preliminary Results of Gravity Investigations at Yucca Mountain and Vicinity, Southern Nye County, Nevada, U.S. Geological Survey Open File Report 82-701, 41 p.
- Sutton, V.D., 1985. Data Report for the 1985 Seismic Refraction Experiment at Yucca Mountain, Beatty, and Vicinity, Southwestern Nevada, U.S. Geological Survey Open File Report 85-591, 66 p., 11 plates.
- USGS (U.S. Geological Survey), 1988. "Geologic and Hydrologic Investigations of a Potential Nuclear Waste Disposal Site at Yucca Mountain, Southern Nevada," U.S. Geological Survey Bulletin 1790, M.D. Carr and J.C. Yount (eds.), 152 p.
- Waddell, R.K., 1982. Two-Dimensional, Steady-State Model of Ground-Water Flow, Nevada Test Site and Vicinity, Nevada-California, U.S. Geological Survey Water Resources Investigations Report 82-4085, 62 p.
- Wynn, J.C., and E.H. Roseboom, 1987. "Rose of Geophysics In Identifying and Characterizing Sites for High+Level Waste Repositories," <u>J. Geophysical</u> Research, vol. 92, no. 88, p. 7,789-7,796.
- Winograd, I.J., and W. Thordarson, 1975. <u>Hydrogeologic and Hydrochemical</u> <u>Framework, South-Central Great Basin, Nevada-California, with Special</u> <u>Reference to the Nevada Test Site</u>, U.S. Geological Survey Professional Paper 712-C, p. C1-C126.

#### 3.2 GEOPHYSICAL INTEGRATION AND PRIORITIZATION OF ACTIVITIES

The objective of the geophysics integration activity as stated in SCP Section 8.3.1.4.1.2 is to increase the effectiveness of geophysical activities during site characterization by (1) systematic consideration of past results, and (2) analyzing how each planned survey addresses information requirements for site licensing. The integration activity will review and evaluate planned geophysical activities for the following:

- 1. Consistency with results from past surveys.
- 2. Direct or supportive uses of the data for site licensing.
- Identification of techniques that are likely to produce usable data.
- The need for the planned effort relative to alternative methods for obtaining the information.
- Scheduling of the planned effort with respect to other studies, and overall priorities for site characterization.

Some of these concerns, particularly items 1 and 3 above, have already been addressed by this report. The SCP program as described in Section 3.1 is generally consistent with past results summarized in Chapter 2, insofar as activities for which definite plans exist have sub-tantial likelihood of providing useable information. Where past results indicate that important methods such as seismic reflection may not produce useable information, the SCP contains plans to conduct organized reviews of the methods and their prospective application.

The status of activities to address other items in the list is discussed in this section, and in Sections 3.3 and 3.4. The editors intend for this report to represent the status of geophysical activities approximately one year after release of the statutory SCP. Accordingly, these sections of the report go beyond the starting point of geophysical integration, stating current working positions on (1) priority activities for the next 1 to 2 years (this section), (2) feasibility testing (see Section 3.3), (3) quality assurance for existing data (see Section 3.4), and (4) a geophysical activities integration study plan (this section). The positions are essential to coordinate activities that involve different investigators or organizations, to focus resources on characterization priorities, and to expedite further planning and integration. The positions will guide, but not control the site characterization program. The Site Characterization Technical Planning Basis (DOE, 1389) and Study Plans are the basis for controlled planning of specific site characterization activities.

Several topics are identified in SCP Section 8.3.1.4.1.2 for intensive review and evaluation. These are (1) subsurface tectonic structure and regional context for the site area, (2) characterization of fractures and fault zones in the repository block, (3) exploration of the large hydraulic gradient area, and (4) detection of potential natural resources. Another hay topic that has received recent attention is the distribution and volume of volcanic deposits. These five topics are addressed in the preliminary prioritization of geophysical activities that is presented below.

#### INTERFACES BETWEEN STUDIES

Recognition of possible interfaces between geophysical activities is provided to initiate a thorough approach to integration and prioritization. A matrix associating SCP Studies with geophysical methods (Figure 3.2-1), and a table associating planned data collection activities with activities that are sources of planning input (Table 3.2-1), are presented for this purpose.

Figure 3.2-1 is a matrix correlating categories of geophysical methods with the SCP studies that will collect and/or use the resulting data. In addition, the matrix contains letter codes describing the areal extent of the different types of geophysical data needed for each study. Figure 3.2-1 can be used to identify the studies that will implement or use data from a particular geophysical method, and the approximate area of coverag.

Table 3.2-1 contains a more detailed list of associations between activities involving geophysics. The table is organized in columns, such that the SCP Activities on the left-hand side should consider results or planning input from those on the right-hand side. Interfaces have been identified only for data collection activities, or activities directly affecting plans for geophysical investigations, thus neglecting activities such as synthesis and geologic modeling.

## NRC SITE CHARACTERIZATION ANALYSIS (SCA) CONCERNS

NRC concerns also form part of the basis for prioritizing near-term activities. The SCA (letter: Bernero-to-Rousso, with enclosure, dated 7-31-89) contains several comments that pertain directly to geophysical applications, and to the geophysics integration activity. The following discussion is not an official response to any of the SCA comments, but indicates how the comments are incorporated in the integration positions developed by this report.

Comment 32: Geophysics integration

This comment criticizes the integration of planned geophysical activities, and the SCP description of the geophysics integration activity (Section 8.3.1.4.1.2). The 3CP does not contain a geophysics program per se, because geophysical methods are described in the sections where they are applied. Chapter 2 and Section 3.1 of this report address this aspect of the SCP, reviewing the coverage, quality, and applicability of existing data, and relating past results to planned surveys for specific characterization objectives. The information presented in this report indicates that it would be unreasonable to plan a comprehensive program of geophysical data collection, using methods that have not been proven at the Yucca Mountain site or vicinity. In particular, it would be inappropriate

	Re	flect	Seis	smic   Re	ste.		1	Pote Fie	ntia		ds	g Radar	R	erno	te	Bo	reho I	ble (	Geo; i	onys I ø	KCS	2º
Study Plan 8.3.1.X.X.X	Deep (Regional)	Intermediate	Shallow	Upper Crustal	Shalkow	VSP & Crosshole	Aeromagnetics	Detailed Mag.	Detaded Gravity	Paleomagnetics	Geoelectric Methor	Ground Penatrating	LANDSAT	SLAR	Radiometric	Crosshole Gamma	Neutron Logging	IV & BATV	BHEM& WE	Gravity & Magnetic	Othe: BH Methods	Related SB Drillin Program
.2.1.3 Regional Flow (Czarnecki)	a b	a b	Ь		Antibulina P		a b	D	1		ь	d	ь				b	b		b	B	WT-holes; UZN-holes
.2.2.1 UZ Infiltration (Flint)			d		D							d	d			D	B	d	в		d	UZN-holes
2.2.3 UZ Percolation SB (Rousseau)						0										D	d	d	D		ď	UZ-holes; SD-holes
2.2.4 UZ Percolation ESF (Yang et al.)						¢											ø	с			c	MPBH activity
2.3.1 SZ Flow (USGS)		b		ь			b	b	b		b							ь			B	WT-holes
.2.3.2 SZ Hydro- chem. (Steinkampf)																				d	đ	WT-holes
.5.2.1 Quatern, Reg. Hydro, (Stuckless		C. C. Martin									b		b	b				b			B	WT-holes
4.2.1 3-D Strat. (Spengler)				b						b							BD	b d	BD	B	BD	G-holes, & others
4.2.2 3-D Structure (Spengler)		b				D				a b								b	bd	b	d	G-holes, & others
.4.3.1 Systematic Drilling (Rautmann)										d							d	d		d	BD	UZ-holes; SD-holes
8.1.1 Prob. Volcanic Eruption (Crowe)	a	a b		a b			a				b					-						
8.5.1 Volcanic Features (Crowe)							b	b	b	AB							and the se			5		V-holes
9.2.1 Natural Re- sources (USGS)	d	d					d	d			d		d		d	armenta.	. Constitution	hereine			d	G-holes: SD-holes
.14.2.3 Geotechnical (SNL)			Ċ		0			c				0						c			c	
15.1.5 Excavation Invest. (SNL)						0											ACCESSION OF THE OWNER OF					
17.4.2 Faulting, Surf. Fac. (Shephard)			c						BC LOOK IN	and and	Sectors, seaso	0		an a		Landara	in senior se					
17.4.3 Quat. Faults w/in 100 km (Fox)	A	A	b	AB			a	A	A	AB	A		a								ь	
.17.4.4 Proximal Faulting (Yount)			в		в						B			88 - 189-A								
17.4.7 Subsurface Faulting (Fox)	a	в	в	b			B	A BC	в	ab	Ь		b		в							anna annananna. Administration an ann
8.3.4.2.4.4 Engr. Barrier Tests (LLNL)																		C.C.ross	С			

NOTES:

1) a = Regional, b = Yucca Mtn & vicinity, c = Surface facilities, d = Repository block & vicinity.

(2) CAPITALIZED indicates actual data collection activities, vs. studies where geophysical data are used.

Figure 3.2-1. Matrix correlating categories of geophysical methods, with SCP studies wherein geophysical data will be collected and used.

SCP Activity:	Gp Application:		Input From:						
REGIONAL HYDROLOGY									
8.3.1.2.1.3.1	trace impermeable near-surface strata	<= <=	8.3.1.4.2.1.3	borehole geophysics					
8.3.1.2.1.3.2	borehole logging, non-Project wells	< * < * < *	8.3.1.4.2.1.3 8.3.1.4.2.2.3 logs	borehole geophysics BATV, flowmeter					
8.3.1.2.1.3.2	regional potentio- metric surface	<= <=	8.3.1.4.2.1.2	surface-based geophysics					
8.3.1.2.1.3.3	Fortymile Wash recharge studies	<pre>&lt; &lt; &lt;</pre>	8.3.1.2.2.1.2 8.3.1.4.2.1.3 8.3.1.2.2.2.3 8.3.1.2.2.3.2	natural infiltration study, geophysics applications borehole geophysics borehole eval. of faults/fractures site vertical boreholes study; geophysics applications					
UZ HYDROLOGY									
8.3.1.2.2.1.1	characterize hydrologic properties of surficial materials	<pre></pre>	8.3.1.4.2.2.3 8.3.1.17.4.2.1 8.3.1.17.4.4 8.3.1.17.4.7.8	borehole eval. of faults/fractures ground-based radar shallow refraction for fault char. Mini-Sosie tests					
8.3.1.2.2.1.1	hydrologic props. of surficial materials; airborne radar (SLAR)	<	8.3.1.2.1.3.2 ° 3.1.5.2.1.3	investigate hydro. significance of lineaments regional hydrology					
8.3.1.2.2.1.2	natu ` infiltration	< = = = = = = = = = = = = = = = = = = =	8.3.1.4.2.1.3	borehole Gp (calibration holes; analysis of wireline responses)					
8.3.1.2.2.3.1	matrix hydrologic properties from core/cuttings	<= <= <=	8.3.1.4.2.1.3 &8.3.1.4.2.2.4	use borehole geophysics to establish context					

Table 3.2-1. Interfaces Between Geophysical Activities, Arranged by SCP Activity (Page 1 of 6)

SCP Activity:	Gp Application:		Input From:	
8.3.1.2.2.3.2	crosshole gamma transmission	<= <= <=	8.3.1.2.2.1.2	experience acquired from infiltration monitoring program
8.3.1.2.2.3.2	VSP application	<= <=	8.3.1.4.2.2.5	VSP feasibility testing
8.3.1.2.2.4	hydrologic testing in the ESF	× × × × × × × × × × × ×	8.3.1.2.2.1 8.3.1.2.2.3.2 8.3.1.4.2.1.3 8.3.1.4.2.2.3 8.3.1.4.2.2.3 8.3.1.4.3.1.1	infiltration studies logging, fracture characterization techniques borehole geophysics crosshole methods systematic drilling program baseline log suite
SZ HYDROLOGY, S	SITE			
8.3.1.2.3.1	S2 well testing in the "c-wells" and other boreholes	+ + + + + + + + + + + + + + + + + + +	8.3.1.4.2.1.3 8.3.1.4.2.2.3 8.3.1.4.2.2.5	borehole geophysics borehole evaluation of faults, fractures VSP testing
SITE STRATIGRAM	PHY AND STRUCTURE			
8.3.1.4.2.1.2	surface-based Gp incl. planned refraction profile across northern YM		8.3.1.4.2.2.3 8.3.1.4.2.2.5 8.3.1.17.4.7.1 8.3.1.17.4.7.2 8.3.1.17.4.7.4	tests, surveys for borehole evaluation of faults/fractures VSP test results seismic review detailed gravity ground magnetics
8.3.1.4.2.1.34 8.3.1.4.2.2.3	borehole Gp & borehole evaluation of faults/fractures	<pre>&gt; &gt; &gt;</pre>	8.3.1.2.2.3.2 8.3.1.2.2.3.3 8.3.1.2.3.1.1 8.3.1.2.3.1.2	saturation, litho- logy, and fracture characterization Solitario Canyon horizontal hole Solitario Canyon study-saturated zone site potentiometric level study boreholes

Table 3.2-1. Interfaces Between Geophysical Activities, Arranged by SCP Activity (Page 2 of 6) Table 3.2-1. Interfaces Between Geophysical Activities, Arranged by SCP Activity (Page 3 of 6)

SCP Activity:	Gp Application:		Input From:		
8.3.1.4.2.1.3	borehole gravimetry and density logging	<- <-	8.3.1.17.4.7.2	detailed gravity	
8.3.1.4.3.1.1	systematic drilling program baseline log suite	<= <= <=	8.3.1.4.2.1.3 8.3.1.4.2.1.5	borehole geophysic magnetic strati- graphy	53
8.3.1.4.3.1.1	systematic drilling program borehole planning	<= <= <=	8.3.1.4.2.2.5	VSP feasibility testing	

ENGINEERING PROPERTIES OF SURFICIAL MATERIALS

8.3.1.14.2.3	engineering characterization of surficial materials	$\begin{array}{c} \land \land$	8.3.1.4.2.1.3 8.3.1.4.2.2.3 8.3.1.17.4.2 8.3.1.17.4.4 8.3.1.17.4.7.8 8.3.1.2.2.1.1	borehole geophysics borehole eval. of faults/fractures GPR testing at Midway Valley shallow refr. for fault char. Mir. Sosie tests Gp char. of hydro. props. of surficial materials
THERMAL/MECHANI	CAL ROCK PROPERTIES			
8.3.1.15.1.1.1	density, porosity and other props. from core samples	< + + + + < + + + +	8.3.1.4.2.1.3 &8.3.1.4.2.2.4	use borehole geophysics to establish context of samples
PRECLOSURE TECT	CONICS			
8.3.1.17.4.1.2	monitor current seismicity	<= <=	8.3.1.8.1.1	possibility of volcanic eruption

<- 8.3.1.8.5.1 characterize

volcanic features

<-

<-

SCP	Activity:	Gp Application:		Input From:	
8.1	1.17.4.3.1	regional Gp linas	<-	8.3.1.2.1.3.1	characterize channel
			<		Illi
			2.	0101414141016	Tartiary rocks
			2-	8.3.1.8.1.1	evaluate structural
			<-		controls on basaltic
			<-		volcanism
			<-	8.3.1.8.5.1	characterize
			<-		volcanic features
			<-	8.3.1.9.2.1.4	evaluate hydrocarbor
			<=		potential at the
			<=		site
8.3	.1.17.4.3.2	paleomagnetism	<=	8.3.1.8.5.1	studies to
		studies on faults	<-		characterize
		w/in 100 km	<=		volcanic features
			<*	8.3.1.2.1.3.2	investigate hydro.
			<=		significance of
			<-		lineaments
			<=	8.3.1.5.2.1.3	regional surface
		<=		ularoroâl	
8.3.1.17.4.3.5	LANDSAT thematic	1		characterize=	
		2	0.0.4.6.6.4.4	hydrologia	
		mapper investigation	2-		properties of
			2-		eurficial materiale
			<.	8.3.1.2.1.3.2	investigate
			<*	VIVIAIBINIVIE	hydro, significance
			<-		of lineaments
			<=	8.3.1.5.2.1.3	regional surface
			<-		hydrology
			<-	8.3.1.17.4.3.2	study of faults w/ir
			<-		100 km of YM
			<-	8.3.1.17.4.9.1	desert varnish
			<-		studies
		dependence			
0.3	/ . 4 . 4	detailed studies of	~ *	6.3.1.17.4.7.8	results from
		tautts in the IM	~	Mini-Sosi	Mini-Sosie tests
		region	S		
8.3	.1.17.4.7.1	review of inter-	<-	8.3.1.2.3.1.2	site potentiometric
1.1		mediate depth	<-		level-study of large
		seismic methods	<-		hydraulic gradient
			<-	8.3.1.8.5.1	data needs for
			<-		study of volcanic
			<-		features
			<-	8.3.1.9.2.1.4	data needs to assess
			<+		energy resources

Table 3.2-1. Interfaces Between Geophysical Activities, Arranged by SCP Activity (Page 4 of 6)

SCP	Activity:	Gp Application:		Input From:		
8.3	.1.17.4.7.2	detailed gravity	<=	8.3.1.8.1.1	evaluate structural	
			<-		volcanism	
			<=	8.3.1.9.2.1.2	geophysical	
			<-		appraisal of the	
			<=		site relative to	
			<		mineral resources	
8.3	.1.17.4.7.36	d'.ailed aeromag.;	<-	8.3.1.2.3.1.2	site potentiometric	
8.3	.1.17.4.7.4	ound magnetics	<-		level-large	
			<-		hydraulic gradient	
			<=	8.3.1.8.1.1	evaluate structural	
		<- <- 2.3.1		controls on basaltic volcanism characterize		
			831851			
			<-	VIVIAIVIVIA	volcanic features	
			<-	8.3.1.9.2.1.2	geophysical	
			<-		appraisal of the	
			<-		site relative to	
			<=		mineral resources	
			<-	8.3.1.17.4.7.6	airborne radio-	
			<-		activity survey la	
			<-		snare platiorm)	
8.3	.1.17.4.7.5	review of	<	8.3.1.2.1.3.2	characterize pre-	
		geoelectric methods	<-		Tertiary rocks	
			< **	8.3.1.2.3.1.2	site potentiometric	
			<."		level-data needs for	
			2		budraulia gradient	
			<	8.3.1.5.2.1.3	map regional water	
			<-		table variation	
			<-	8.3.1.8.5.1	characterize	
			<*		volcanic features	
			<-	8.3.1.9.2.1.2	geophysical	
			<-		appraisal of mineral	
			<		and hydrocarbon	
					resources	
8.3	.1.17.4.7.6	airborne and	<-	8.3.1.17.4.3.2	study of faults w/in	
		ground gamma-ray	<-		100 km of YM	
		tests, surveys	<-	8.3.1.17.4.2.1	raulting studies for	
			2		Surrace raciilties	

Table 3.2-1. Interfaces Between Geophysical Activities, Arranged by SCP Activity (Page 5 of 6)
SCP Activity:	Gp Appl.	ication:		Input From:	
8.3.1.17.4.7.7	thermal	infra-red	* * * * * * * * * * *	8.3.1.2.1.3.2 8.3.1.2.2.1.1 8.3.1.5.2.1.3	investigate lineaments with potential hydrologic significance hydrologic properties of surficial materials regional surface hydrology

Table 3.2-1. Interfaces Between Geophysical Activities, Arranged by SCP Activity (Page 6 of 6) to embark on a 3-D seismic reflection program, given that no reflection event has ever been positively identified from any surface or downhole seismic survey at Yucca Mountain (see Section 2.5). When planned feasibility tests and initial data collection activities have been performed and evaluated, a balanced approach to seismic exploration can be implemented.

Use of complementary geophysical methods has been incorporated into the program for objectives such as fault detection, characterization of volcanic features, and exploration of the large hydraulic gradient area (see Section 3.1). Several planned feasibility tests (see Section 3.3) will evaluate methods that could become additional complementary methods to address these objectives.

Comment 51: Geophysics integration, local and regional scales

The planned gravity, aeromagnetic, geoelectric, seismic reflection, and seismic refraction surveys described by th', report are the basis for an "integrated" geophysics program that can accomplish both regional reconnaissance and focused investigation of the site area. The teleseismic data collection and analysis activity described in Section 3.1 will also provide areally extensive reconnaissance. (Applications of seismic and geoelectric methods have yet to be reviewed as planned in SCP Studies 8.3.1.17.4.3 and 8.3.1.17.4.7). These data sets together are expected to improve understanding and interpretation of the isostatic gravity anomaly, depth to the Curie isotherm, crustal conductivity structure, seismic Moho, crustal and upper mantle velocity structure, and the geometry of upper- to mid-crustal reflectors and their assocation with structure at the surface. Implementation and possible elaboration of parts of this program, including 3-D seistic reflection, depends on the results from planned peer review, teasibility testing, and initial data collection and analysis.

Comment 52: Geophysical surveys for characterization of igneous activity

This report contains several new concepts for investigation of volcanic deposits and magmatic intrusions. The integrated program that will be elaborated in several different Study Plans, includes feasibility testing of gravity and ground magnetic methods for delineating known volcanic features, expanded coverage for the detailed aeromagnetic survey (based on the feasibility test results), detailed gravity and ground magnetic surveys over known or suspected volcanic deposits, and high-resolution teleseismic data collection and velocity modeling. This program will include data modeling and analysis, and will have high priority as discussed below.

Comment 53: Geophysical surveys for mineral and energy resource assessment

No geophysical investigations are currently planned which will specifically address data needs for natural resource assessment. Current plans adhere to the SCP which states or implies that geophysical data collected and interpreted for characterization of volcanism, tectonics, the large-hydraulic gradient, and fault detection/delineation will be evaluated, and are considered adequate for an initial assessment. Activities which will provide the data in question, particularly aeromagnetic surveys and seismic surveys, with the associated feasibility testing, interpretation, and peer review tasks, will be given high priority as discussed below.

The rationale for withholding more extensive geophysical exploration for natural resources includes: (1) needed scoping analyses and models for characteristic geophysical responses are as yet unavailable; and (2) uncertainty of geologic parameters that would be addressed by geophysical exploration in addition to that already planned for other activities, could be small relative to other uncertainties in natural resource assessment that site exploration cannot address. The latter rationale means that without an initial assessment based on new geophysical data, there is significant likelihood that no improvement in estimates of the probability of human intrusion would be gained from extensive site exploration for the natural resources program.

# Comment 59: Fault detection, repository surface facilities location

Geophysical methods appear to be of limited usefulness for locating fault offsets in alluvium prior to trenching. For reasons discussed in Sections 2.4 and 2.5 of this report, the alluvium-bedrock contact may not be a strong enough seismic reflector or refractor for reliable profiling. Signal strength and interpretability are further complicated by irregular contact geometry and interbedding in the alluvium. Useful contrasts in geoelectric properties have not been recognized. Certain techniques have not been tried, such as shallow geoelectrical soundings and low-energy seismic reflection (e.g., using shear w.ve sources), but do not seem likely to image fault offsets at the required resolution. The current strategy for characterizing faulting in Midway Valley relies heavily on trenching, but geophysics will be used if appropriate applications can be demonstrated (e.g., GFR and Mini-Sosie; see Sections 3.1.4 and 3.1.6).

Geophysical methods that may be useful for detecting the presence of small faults or associated fracture zones in the waste storage areas of the host rock, will be tested during the planned feasibility testing discussed below.

#### ACTIVITY PRIORITIZATION

The basis for preliminary prioritization of activities includes (1) resolution of uncertainty in applicability of geophysics to various planned activities in the SCP, (2) the schedule for site characterization as represented in the SCP, (3) the need to reduce uncertainty in technical plans for characterization activities, (4) NRC SCA Comments, and (5) other recent regulatory interactions. These influences are considered in the following discussion of priorities organized within specific topics. Also, it is important to recognize that a systematic effort to set priorities for

surface-based testing, in a manner that will provide data for evaluating siting criteria of 10 CFR 60.122, is underway at the time of publication of this report.

Subsurface Structure of the Yucca Mountain Site Area and Region

Geophysical data are needed to evaluate alternative conceptual models for tectonic structure at the site, for example, models predicting different geometry for major faults at depth. Present understanding of geologic structure below about 2 km depth at the site is based mostly on geologic inference, constrained by gravity interpretation, preliminary results from seismic refraction, and a few boreholes one of which penetrates the Paleozoic-Tertiary contact. Activities to better constrain subsurface structure beneath the site, and to establish the regional tectonic context of the structure, have high priority because of the many potential applications of the geologic model, notably in characterizing the saturated carbonate aquifer, and assessing the potential for natural resources. Geophysical exploration of subsurface structure at different scales is also needed to address NRC SCA Comment #51.

The first priority is to initiate peer reviews pertaining to the following activities:

Peer Review of Applicability of Regional Geophysical Traverses (Accivity 8.3.1.17.4.3.1) - This review will consider alternative regional structural hypotheses that can be tested by use of regional geophysical traverses, and evaluate the usefulness of available geophysical techniques for addressing these hypotheses. The assessment will make use of existing geophysical data collected from Death Valley, the Amargosa Desert, intervening ranges, and other venues. The importance of regional geophysical traverses, relative to intersecting traverses in the site vicinity, for detecting and understanding structures that could significantly affect site suitability, will also be assessed. It is anticipated that this review will provide guidance on technical issues that are of considerable importance to site characterization, such that it will be an independent peer review involving experts in regional tectonics, and geophysical exploration for crustal studies and resource development in the Great Basin.

Peer Review of Seismic Methods for Characterizing Yucca Mountain and <u>Vicinity</u> (Activity 8.3.1.17.4.7.1) - This review will consider the applicability of high-resolution upper crustal seismic refraction, and seismic reflection methods for structural profiling across Yucca Mountain and in adjacent allevium covered areas. The review will consider the available geophysical data from the site vicinity, and the objectives for seismic exploration for site characterization. It is anticipated that the review will recommend a phased exploration program, that includes specific recommendations on how and where to carry out the first phase and how to carry out subsequent phases based on interim results. The first phase of exploration would be implemented as the seismic reflection feasibility test described in Section 3.3 of this report. Because of past results from seismic reflection campaigns at Yucca Mountain (see Sections 2.5 and 3.1 of this report) there is a significant possibility that future surveys will produce no useful results from the site. It is therefore important to solicit special expertise for the review, particularly in the area of seismic reflection exploration of Great Basin volcanic terranes. The review will be an independent peer review involving experts in seismic reflection and refraction, VSF, and geophysical exploration (such as potential methods) in Great Basin volcanic terranes.

The Peer Review of Seismic Methods may also consider the need for VSP in the site area (e.g., borehole USW G-4) or in Crater Flat (e.g., borehole USW VH-1). At present, there have been no interpretable seismic reflection records obtained from the site area. Regardless of whether VSP is recommended by the peer review in regard to structural profiling, a VSP survey should be conducted and analyzed to determine the importance of layer velocities, reflection coefficients, attenuation, and other effects (e.g., mode conversion, splitting) in the tuff section, to the proposed feasibility test involving a reflection profile across Yucca Mountain.

The objective described above for VSP survey will be met by the VSP feasibility test that has already been planned for borehole USW G=4 (in support of SCP Activity 8.3.1.4.2.2.5). Planning documentation for this test has already received QA approval, and the major impediment to performance of the test is site access. The test has high priority because it will produce information that supports survey design and interpretation for structural profiling at the site. Applicability of the VSP test to seismic reflection profiling will be further reviewed informally by the geophysics integration activity (SCP Section 8.3.1.4.1.2).

Depending on the recommendations of the Peer Review of Seismic Methods, a seismic reflection feasibility test will be performed, involving a reflection line across Crater Flat, Yucca Mountain, Midway Valley, and (partially) Jackass Flats. A coordinated program involving downhole velocity surveys, VSP surveys, noise studies, or source studies may be recommended by the peer review. It is appropriate to make the reflection test contingent on the peer review because of the potential cost of the seismic line and associated testing activities.

Depending on the recommendations from the Peer Review of Seismic Methods, the planned reversed, east-west seismic refraction line across the northern part of Yucca Mountain (SCP Figure 8.3.1.4-6) will be acquired as planned for Study 8.3.1.4.2.1. This line may provide information on the presence of buried structure, such as an igneous intrusion, that could be a cause of the large hydraulic gradient.

Another priority will be to support the program of teleseismic data collection and inversion described in Section 3.1 of this report. This program will investigate lateral variability of the lower, middle, and upper crust in the Yucca Mountain region. The area of detailed investigation will include Yucca Mountain, Crater Flat, Jackass Flats, and the portion of the Amargosa Desert where volcanic deposits are evident from aeromagnetic data. For a similar purpose, data requirements for detailed Curie isotherm calculations for the Yucca Mountain region will be determined for use in scoping aeromagnetic data collection activities. The teleseismic inversion and magnetic continuation methods are important sources of information on the spatial distribution of crustal properties, that can be used to tie deep crustal features with surface features.

The geoelectric data scoping analysis identified in Section 3.3 will have high priority so that the resulting information can be assessed in the planned review of geoelectric applications in site characterization (SCP Activity 8.3.1.17.4.7.5). This latter review (described in the SCP) will be an internal Project activity which assesses the most current information on applicability of geoelectric methods to site characterization, and recommends specific surveys as appropriate. The review and recommendations are needed as soon as practicable, to assure that geoelectric methods are used effectively in site characterization. The review may be followed by appropriate updates to the SCP and Study Plans.

Finally, qualification of the data from the Amargosa Valley seismic reflection test line will have high priority. The qualification process represented by procedure AP5.9Q, and the candidate data sets for qualification, are discussed in more detail in Section 3.4 of this report.

#### Distribution of Volcanic Deposits

Geophysical characterization of the subsurface extent, and total volume of volcanic deposits in the northern Amargosa Desert including Crater Flat, is a high priority because of public awareness of relatively recent volcanism at the Lathrop Wells cone, because of NRC staff criticism of the approach taken by the Project toward volanic hazard assessment, and the sensitivity of that approach to definition of the domain of existing volcanic deposits. The work is also important for addressing NRC SCA Comment #52.

It may be appropriate to extend the coverage of the planned detailed aeromagnetic survey (Activity 8.3.1.17.4.7.4) to include the region where existing aeromagnetic data indicate the presence of surface and buried volcanics. The objectives for this coverage would be to provide additional constraint for estimating the total volume of these deposits, and for locating boreholes to sample buried volcanic rocks in the Amargosa Desert. However, there are technical factors which complicate the application of aeromagnetic survey to this purpose. The magnetic properties of the basalt and basaltic detritus may vary significantly throughout the deposits, with the possibility of strongly varying remanence. Also, it is likely that some of the deposits exhibit strong magnetic reversal.

Before embarking on an extensive program of data collection and analysis for volcanism studies, a feasibility test of gravity and ground magnetic methods will be conducted in the vicinity of particular known or suspected volcanic deposits in the area. The scope of this test will be limited to investigation of a feature or features that are likely to provide adequate information for evaluating the characterization approach using gravity and magnetics. Samples of material will be collected for magnetic analysis, and certain deposits will be more extensively sampled to assess the variability of magnetic properties. Ground surveys will be performed to evaluate whether the orientation and extent of feeder dikes can be detected geophysically, and to evaluate differences between ground and airborne surveys. Scoping calculations will be performed to determine the sensitivity of deposit volume estimates, borehole location strategies, and other applications based on different types of data, to uncertainty in the magnetic properties and geometrical approximations needed to model volcanic deposits.

Based on the results of the feasibility test, the area of coverage for the detailed aeromagnetic survey (SCP Activity 8.3.1.17.4.7.4) will be increased as indicated in Section 3.1 of this report, with a concomitant program of sampling, modeling, and interpretation.

The planned seismic reflection feasibility test will investigate whether (1) the known, buried basalts in Crater Flat produce interpretable reflections; and (2) whether seismic reflection can penetrate below such deposits. This evaluation will require that the feasibility test line be routed proximally to existing boreholes VH~1 and VH-2, and that downhole geophones be used to the extent practicable in VH-1 (which is available for this purpose).

Finally, the teleseismic data collection and inversion activity discussed in Sections 2.8 and 3.1 of this report, will be planned and implemented as a reconnaissance tool in the volcanism investigation.

Exploration of the Large Hydraulic Gradient Area

The cause or source of the large hydraulic gradient is important for assessment of postclosure repository performance. The altitude of the water table 1 to 2 km north of the site is roughyl comparable to, or greater than, the elevation of a portion of the conceptual repository. Additional information is needed to evaluate the credibility of the repository flooding scenario. It is anticipated that a major part of this information will come from borehole exploration and testing, and that preliminary findings from investigation of the large hydraulic gradient will be published early in site characterization (SCP Study 8.3.1.2.1.3, page 8.3.1.2-449). Accordingly, geophysical reconnaissance of the area, principally for use in siting exploratory boreholes, is a high priority. Such surveys are related to exploratory drilling planned for SCP Studies 8.3.1.2.1.3, 8.3.1.2.3.1, and 8.3.1.4.2.1.

Applicable reconnaissance methods include aeromagnetic (SCP Activity 8.3.1.17.4.7.2), seismic refraction (Activity 8.3.1.4.2.1.2), geoelectric soundings (Activity 8.3.1.17.4.7.5) and detailed gravity (Activity 8.3.1.17.4.7.2). As shown in the SCP, detailed gravity and aeromagnetic coverage will encompass the large hydraulic gradient area from Fortymile Wash to the western edge of northern Yucca Mountain. These surveys have high priority with respect to investigation of the large hydraulic gradient.

The planned east-west high-resolution, upper-crustal seismic refraction line across northern Yucca Mountain (SCP Activity 8.3.1.4.2.1.2) is needed to test for a velocity anomaly that could be associated with a structural origin for the large hydraulic gradient. Similarly, a program of magnetotelluric (MT) soundings along traverses perpendicular to the hydraulic gradient inferred from boreholes, is proposed as a feasibility test of the applicability of this method for detecting a possible structural origin (see Section 3.3 of this report). The proposed MT traverses will not depend on the performance or the outcome of the planned review of the applications of geoelectric methods to site characterization (SCP Activity 8.3.1.17.4.7.5) discussed above. This is because the existing geoelectric data are not adequate for assessing the applicability of these methods to the important objective of exploring the large hydraulic gradient area.

Finally, the feasibility testing program discussed in Section 3.3 also includes investigations to determine the applicability of detailed gravity methods to exploration of the large hydraulic gradient area and other features. As discussed in Sections 2.1 and 3.1 there is local variation of gravity in the area, but data acquisition and terrain correction accuracy are hindered by rugged topography and limited access. Several high-resolution gravity profiles will be collected along roads traversing the area. Acquisition and interpretation of these profiles will permit timely assessment of the importance of detailed gravity to the large hydraulic gradient investigation.

Characterization of Fractures and Fault Zones in the Repository Block

The nature of fracturing and fault zones in the repository block is important for understanding the hydrology of the UZ, for repository design, and for assessing the effect of these features on the integrity of emplaced waste containers. Seismic propagation is probably the principal geophysical response by which such features of the rock mass can be detected or characterized noninvasively. Depending on the amount of values water or mineral alteration that may be associated with moisture movement along fracture pathways, geoelectric response may also be indicative of such features. Because of the large distances separating boreholes over most of the site, the rugged surface conditions, and current uncertainty as to the geophysical response of fractures and fault zones at Yucca Mountain, surface-to-borehole techniques and large-spacing borehole logs are most appropriate for testing.

Current plans call for a VSP feasibility test in borehole USW G-4 using vibrator sources at multiple offsets (SCP Activity 8.3.1.4.2.2.5). The casing will be removed from this borehole prior to testing. Results from this test are important not only for conducting seismic experiments involving the exploratory shaft, but also for planning VSP surveys in the U2 hydrology boreholes (SCP Activity 8.3.1.2.2.3.2), specifying geophysical surveys to be performed in boreholes of the systematic drilling program (see SCP Activity 8.3.1.4.2.2.3), and design of the feasibility test of seismic reflection profiling across Yucca Mountain.

It is important to determine what methods will be used to characterize rock mass conditions throughout the repository block, so that the boreholes drilled for the UZ hydrology program and the systematic drilling program (comprising most of the planned penetrations) can be used for geophysical characterization prior to completion for other uses. Accordingly, the feasibility of large-spacing electromagnetic (EM) and electrical resistivity (ER) logs and borehole-to-surface surveys for fracture and fault zone characterization will be tested at an appropriate location (see Section 3.3 of this report).

## Exploration of the Proposed Exploratory Shaft Location

The DOE recently conducted a review of geological and geophysical data pertaining to structural geology at the proposed location of the Exploratory Shaft Facility (ESF) in Coyote Wash (reference: Technical Assessment Review of Geologic and Geophysical Evidence Pertaining to Structural Geology in the Vicinity of the Proposed Exploratory Shaft, YMP/90-2). The review activity focused on the likelihood of an unmapped fault within 100 ft of either shalt location. The Technical Assessment Review (TAR) produced several recommendations, including possible additional drilling and geophysical data collection. It was determined that needed information on the possible existence of a fault at the surface could be obtained by geologic mapping before and during construction of the facility. However, additional resistivity and EM lines were recommended in conjunction with shallow surveys to determine depth-to-bedrock in Coyote Wash, to investigate whether a buried fault (concealed by deposition of more recent Miocene ash flow deposits) exists at the location. The determination of whether there is sufficient likelihood of a buried fault to merit additional geophysical exploration, is poorly constrained by existing geophysical data and must be based principally on geologic inference. This determination will be solicited from the DOE, and, if appropriate, the recommended data collection and analysis activities will receive high priority.

### Assessment of the Potential for Natural Resources

In accordance with the SCP description of Study 8.3.1.9.2.1, no new geophysical data are currently planned for natural resource assessment for reasons given in a previous part of this section. However, natural resources assessement receives abundant attention from the public and from the NRC staff as indicated by NRC SCA Comment #53, so the activities which will furnish geophysical data for assessment of resource potential should be given schedule priority. The methods that have been identified for possible use in resource assessment are surface resistivity, induced polarization (IP), gravity, aeromagnetics and ground magnetics, seismic reflection and refraction, and magnetotellurics. (These have already been identified as high priority activities in conjunction with studies of subsurface structure for the tectonic model.) A key information need for resource assessment that will be addressed using the methods identified is the geometry of the Paleozoic-Tertiary contact. The applicable geophysical surveys are organized under SCP Study 8.3.1.17.4.7, but supporting activities should also include planned peer reviews (see above) and the program of feasibility testing and scoping analyses (see Section 3.3 of this report).

### Feasibility Test Plan - Geophysics Integration and Feasibility Testing

It is recommended that a test plan be prepared to provide necessary documentation and basis for approval of a planned program of geophysical feasibility and scoping analyses. The limited scope of the feasibility test ng program, limited impact of the activities on the site, and the objectives for feasibility testing, should permit relatively straightforward technical review and approval of the plan. Organizing the various feasibility tests under a single plan should also help to expedite review and approval, and is appropriate to ensure timely implementation of the tests.

## SUMMARY

The activity priorities discussed above are summarized in Table 3.2-2, which identifies high priority activities, actions that need to be taken, when, and by whom. Some of these actions do not require Study Plans, some are covered under existing Study Plans as indicated, and some are feasibility tests as discussed in the following section.

Table 3.2-2 Summary of High-Priority Geophysical Activities (page 1 of 2)

Action	Timing and Prerequi- sites	Responsibility	Study Plan 8.3.1.X.X.X	Prioritization Concern
A. (1) Peer Review Notice, for review of the applicability of regional geophysical traverses to site characterization	ASAP, per QMP-03-01	Project Office	N/A	Applicability of regional gephysical traverses such as those identified in SCP
(2) Peer review (SCP Activity 8.3.1.17.4.3.1)	Approval of Peer Review Plan	Project Office to initiate and sup- port; USGS partici- pation in review functions	N/A	Activity 8.3.1.17.4.3.1.
B. (1) Peer Review Notice, for review of Seismic Methods for Characterizing Yucca Mountain and Vicinity	ASAP, per QMP-03-01	Project Office	N/A	Successful seismic reflection methodology; structural profile across YM, Crater Flat, and Jackass Flat; characterize
(2) Peer review (SCP Activity 8.3.1.17.4.7.1)	Approval of Peer Review Plan	Project Office to initiate and sup- port; USGS partici- pation in review functions	N/A	volcanic deposits in Amargosa Desert
C. VSP feasibility test (Activity 8.3.1.4.2.2.5)	ASAP (Site access)	USGS (LBL)	N/A	Application to seismic reflection methodology; fracture/fault char, in repository block; plan ESF VSP; plan VSP in surface boreholes.
D. Seismic refraction line (Activity 8.3.1.4.2.1.2)	Peer Review of Seismic Methods	USGS	N/A	Structure of northern YM; structural cause for large hydraulic gradient.
E. Teleseismic data collection and inversion (associated with SCP Activities 8.3.1.8.1.1.3 and 8.3.1.17.4.1.2)	Study Plan 8.3.1.8.1.1 and 8.3.1.17.4.1	USGS	TBD	Structural reconnais- saince; correlate mid- and lower crust features with surface structure incl. volcanic deposits.
F. Geoelectric data interpretation and scoping analyses.	Feasibility test plan	USGS	N/A	Applicability of existing data and available techniques to investi- gation of large hydraulic gradient area; natural resource assessment.

Table 3.2-2 Summary of High-Priority Geophysical Activities (page 2 of 2)

Action	Timing and Prerequi- sites	Responsibility	Study Plan 8.3.1.X.X.X	Prioritization Concern
G. Feasibility test for detection and delineation of volcanic deposits by geophysical methods.	Feasibility test plan	USGS, with input from LANL	N/A	Volcanic hazard assessment.
H. Feasibility test of borehole and related methods for fracture/fault zone characterization in the UZ and SZ.	Feasibility test plan	USGS	N/A	Resolve applicability of geophysics to charac- terization of the repo- sitory block; require- ments on boreholes penetrating the block.
I. Detailed aeromagnetic survey (Activity 8.3.1 17.4.7.4)	Study Plan 8.3.1.17 4.7; feasibility testing	USGS, input from LANL	.17.4.7	Distribution of volcanic deposits; reconnais- sance of large hydraulic gradient area; mineral resource assessment; detailed Curie isotherm analysis.
J. Feasibility test of the use of MT traverses for exploration of the large hydraulic gradient area	Feasibility test plan	USGS	N/A	Structure of northern Yucca Mountain, as a possible cause for the large hydraulic gradient.
K. Feasibility test of seismic reflection for structural profiling across Yucca Mountain and vicinity.	VSP test; Peer Review of Seismic Methods; Feasibility test plan	USGS	N/A	Demonstrate applica- bility of seismic reflection for site characterization; structural profile across Yucca Mountain, Crater Flat, and part 1 Jackass Flat.
L. Feasibility test of the Mini-Sosie method for fault detection and characterization in the vicinity of Yucca Mountain.	Feasibility test plan	USGS	N/A	Determine extent to which method can be relied upon in site characterization.
M. Preparation and review of plan for geophysics integration and feasibility testing, corresponding to SCP Section 8.3.1.4 1.2.	TBD	TBD	N/A	Structure the integration activity; produce a single integrated geophysical feasibility test plan to facilitate review, approval, and implemen- tation.

2.0

#### 3.3 FEASIBILITY TESTS

Testing is needed to establish whether and how certain geophysical methods will work at Yucca Mountain, before the Project makes significant commitments to these methods (or to alternatives) in terms of budget or schedule. Successful geophysical exploration typically depends on adapting instrumentation, field procedures, and interpretive methods to a particular geologic setting. Useful results have already been obtained from several geophysical activities as discussed in Section 3.1 and summarized in Section 4.0. However, some important methods for exploring the Yucca Mountain site are developmental. Notably, although some surveys have failed in the past, it is too soon to abandon such powerful techniques as seismic reflection. An immediate program is needed to prove these techniques, and it would be best to conduct a portion of this effort as a timely program of feasibility testing.

The geophysical feasibility testing program is recommended by this report as an addition to the site characterization program. The nature of the recommended program would be similar to that of prototype testing in G-tunnel on the NTS, but geophysical tests would be performed at or near Yucca Mountain. If the recommended feasibility testing program is implemented, a scientific test plan would first undergo technical review and approval in accordance with Project proc dures.

It is recommended that the planning basis for feasibility testing be a Geophysical Feasibility Testing Plan. The plan will include evaluation of the potential adverse impacts on site performance from feasibility testing activities, and the potential for interference with other ongoing or future data collection activities.

A list of proposed feasibility tests is presented in Table 3.3-1. Although many of the methods to be tested are mentioned in SCP Sections 8.3.1.2, 8.3.1.4, and 8.3.1.17, feasibility tests are not identified in the TP except for the VSP feasibility test (8.3.1.4.2.2.5) and Mini-Sosie fersibility test (8.3.1.17.4.7.8). Integration of geophysical activities is bes served by consolidating the needed feasibility tests into a single plan. Approval for needed tests will be expedited because the activities are similar and can be evaluated together with respect to potential adverse impaces, test interference, site access, and environmental clearances.

The feasibility tests are mostly independent of one another, and represent little potential for adverse impacts or interference with other activities. They should be conducted as soon as practicable to be effective. Commencement of feasibility testing will be contingent on approval of the Geophysical Feasibility Testing Plan and completion of related requirements such as technical procedures.

In conjunction with feasibility testing, scoping analyses will be performed as indicated in Table 3.3-1. These may include some calculations, and are needed to evaluate characterization concepts not considered explicitly in the SCP, and to translate results from feasibility testing into specific plans for site characterization. An important scoping analysis will be to evaluate the need for long-term geophysical monitoring, to include construction of a geophysical observatory at Yucca Mountain. Scoping analyses will be further defined in the feasibility testing plan.

· 0

5

9

•

As.

Table 3.3-1. Geophysical Feasibility Tests

-

Iter	n	Description	Related SCP Studies/Activities
	FEASIBILITY	TESTS TO BE DESCRIBED IN STUDY PLAN 8.3	3.1.4.1.2
1.	TP logs	Evaluate IP logs and core data as an indicator of alteration, and acquire new IP logs from one or two existing uncased boreholes.	8.3.1.4.2.1.3
2.	Radioactivity methods	Ground radioactivity profiles will be run across known faults and other features, to evaluate the method for fault detection and the feasibility of airborne surveys for fault detection.	8.3.1.17,4.7.6
3.	Infrared remote sensing	Preliminary testing of airborne and satellite infrared sensing methods for investigating infiltration processes in surficial materials.	8.3.1.17.4.7.7
4.	Detailed magnetic profiles	Truck-magnetometer profiles will be run along existing roads crossing known faults to evaluate the usefulness of the method for fault detection.	8.3.1.17.4.7.4
5.	Calibration boreholes	Investigate the feasibility of calibrating borehole nuclear logging methods for the U2 by means of shallow calibration boreholes.	8.3.1.2.2.1.1 8.3.1.4.2.1.3
6.	Logging methods for UZ studies	Acquire and analyze a limited number of logs, including dielectric, nuclear magnetic resonance, and high-resistivity induction, for reliable estimation of water content in the UZ.	8.3.1.2.2.3.2
7.	Geoelectric data analysis	Analyze and reinterpret existing data, to support the planned review of the applicability of geoelectric methods for fault detection, charac- terizing the large hydraulic gradient area, and lithology variation in bedrock	8.3.1.17.4.7.5

Ito	m	Description	Related SCP Studies/Activities
ε.	Detailed gravity and magnetic methods	Obtain new detailed data and profiles to test methods for detecting concealed faults, and for investigating the large hydraulic gradient in the site area	8.3.1.17.4.7.2
9.	Geophysical monitoring scoping analysis	Evaluate needs, applicable methods, and potential costs for geophysical monitoring activities to begin during site characterization. Methods to be considered include continuous nano gravity monitoring, micro-seismic monitoring of the repository block, magnetic field monitoring, strain and tilt measurements, and construction of a geophysical observatory.	8.3.1.17.4.7 8.3.1.8.5.1
10.	Test of seismic reflection at Yucca Mountain	Seismic line across Crater Flat, Yucca Mountain, and Midway Valley. Primary objective to evaluate conventional seismic profiling methods for imaging the first 5 sec at and near Yucca Mountain. Depends on the outcome of the Peer Review of Seismic Methods (SCP Activity 8.3.1.17.4.7.1), and the VSP feasibility test (see above).	8.3.1.17.4.3.1 8.3.1.17.4.7.1
11.	MT and AMT traverses across large hydraulic gradient area	Evaluate MT methods for recon- naissance and characterization of specific features of the large hydraulic gradient area. Acquire a limited program of 10 stations along traverses perpendicular to the strike of the gradient inferred from boreholes. Support evaluation of geoelectric methods for SCP Activity 8.3.1.17.4.7.5.	8.3.1.2.3.1.2
12	Feasibility test of borehole methods for characterizing fracturing and fault zones in the UZ and SZ	Evaluate large-spacing ER and EM logs, for tracing the extent of features observed in boreholes. Extend methods to surface-to- borehole geometry as appropriate. Evaluate hydrophone VSP and heat-pulse flowmeter technology for S2 fracture characterization.	8.3.1.2.2.3 8.3.1.2.3.1 8.3.1.4.2.1.3 8.3.1.4.2.2.3

Table 3.3-1. Geophysical Feasibility Tests (continued)

.

i.

Related SCP Item Studies/Activities Description 13. Test of gravity Investigate a feature or features 8.3.1.8.5.1 and ground likely to provide sufficient infor-8.3.1.17.4.7 magnetic methods mation to evaluate the approach. for volcanism Collect and analyze samples; investistudies gate the relation of ground magnetics to existing aeromag. data. Explore for indication of feeder dikes. Perform scoping analyses to evaluate uncertainty in material properties and deposit geometry. 14. Test of tele-Test intermediate-period seismo- 8.3.1.8.1.1.3 seismic tomo- meters in the unconsolidated surface 8.3.1.17.4.1.2 graphy for material of Crater Flat, for the detecting purpose of recording compressional partial melt and shear wave arrivals, measuring detecting beneath Crater atcenuation, and measuring P-wave Flat receiver response functions. 15. Test of shallow Acquire profiles across marsh, 8.3.1.5.1.2.2 high-resolution lacustrine, playa, and volcanic seismic reflect- sequences at various locations 8.3.1.17.4.3 8.3.1.17.4.7 ion techniques throughout the Southern Great (i.e., Mini-Basin. Sosie or other syst (a) in support of paleoclimate studies FEASIBILITY TESTS DESCRIBED BY EXISTING PROGRAM DOCUMENTS 16. VSP feasibility Investigate the use of VSP to map 8.3.1.2.2.3.2 test subsurface fracturing and other features in the vicinity of boreholes. Testing is needed to (a) define the methodology and the UZ seismic response, (b) support planning certain characterization activities, and (c) constrain activity descriptions for impacts

Table 3.3-1. Geophysical Feasibility Tests (continued)

analysis and evaluation of test interference. (SCP Activity

8.3.1.4.2.2.5)

Ite	m	Description	Related SC Studies/Ac ivities
17.	Feasibility Test of the Mini-Sosie seismic re- flection method for fault detec- tion and characteri- zatic.	Evaluate application of Mini-Sosie method at or proximal to the site area. Acquire two preliminary traverses, to be used as the basis for deciding to proceed with seven or more additional profiles, or to apply the method more selectively.	8.3.1.17.4.7.8

Table 3.3-1. Geophysical Feasibility Tests (continued)

3.4 QA REQUIREMENTS FOR USING DATA FROM PAST GEOPHYSICAL ACTIVITIES

#### INTRODUCTION

Geophysical data are likely to be used as primary and secondary input for calculations and modeling activities to assess compliance with regulatory objectives. The DOE must be able to demonstrate in a license application that the applicable requirements of 10 CFR Part 60 have been fulfilled. Existing geophysical data and data analyses which have not been performed under a quality assurance (QA) program that meets the requirements of 10 CFR 60, Subpart G, but which will support the DOE's license application to construct and operate a geologic repository, will be qualified. This includes qualification of existing data that were collected under a QA program that did not meet Subpart G requirements. This section briefly describes steps that will be taken to assess whether existing geophysical data can be used in licensing.

Specific requirements for qualifying data are provided in the Yucca Mountain Project QA Plan (NNWSI 88+9, Appendix G), which is consistent with NRC guidance entitled: Qualification of Existing Data for High Level Nuclear Repositories (NUREG-1298). Many of the specific requirements are implemented in Yucca Mountain Project administrative procedure AP5.9Q (Qualification of Data or Data Analyses not Developed Under the Yucca Mountain Project Quality Assurance Plan).

If existing geophysical data are identified is primary information that will be, or may be, used in support of licensing the data may then be submitted to the qualification process defined b AP5.9Q. Existing data may already meet QA requirements if initially developed under a QA program; however, this must be demonstrated through the qualification process. As described in AP5.9Q, possible outcomes of the qualification process include recommendations that the data be qualified in whole or in part, or that further actions such as confirmatory testing or peer review be undertaken prior to qualification.

The QA program for site exploration has evolved from 1977 to present and has incorporated requirements in effect at the time work was performed. The QA program has been modified as new requirements were imposed, or adopted by the DOE. A QA program that meets 10 CFR 60, Subpart G, will be in place prior to the start of planned geophysical activities for site characterization. However, by the current definition such a program was not in place for collection and analysis of existing data. Therefore all existing data and data analyses are subject to qualification, as appropriate. Application of the qualification procedure (AP5.9Q) depends on whether the qualification criteria are met, and the processary approvals are obtained.

This section of the report contains a summary of the applicable requirements and a brief review of the qualification procedure. In addition, geophysical data sets considered likely to require qualification before licensing are identified, with a statement of rationale. Finally, one geophysical data set is identified as best suited for initial application of the qualification procedure AP5.9Q. The outcome of the initial effort will be used by the geophysics integration activity to determine the nature, priority, and appropriate scope for qualification of the other data sets. Note that existing geophysical data sets not specified below can also be qualified on an as-needed basis. This may include data developed prior to implementation of a 10 CFR 60, Subpart G, QA program by the DOE and its contractors; data developed outside the DOE repository program by energy or mineral companies, geophysical service contractors, national laboratories, or universities; and data published in technical or scientific publications.

### NRC-DEFINED DATA QUALIFICATION METHODS

The NRC has defined four acceptable alternative methods or combinations of methods in NUREG-1298. These methods are incorporated in NNWSI 88/9, Appendix G, and are implemented in AP5.9Q:

#### Method A. Peer Review of Data

A peer review will be a documented, critical review performed by two or more peers who are independent of the data or data analyses under review. Peer independence means that the peer (a) was not or is not involved as participant, supervisor, technical reviewer, or advisor in the work being reviewed; and (b) has sufficient freedom from funding considerations, employment, or business associations with program participants to ensure impartiality and to preclude the perception of conflict or vested interest.

## Method B. Use of Corroborating Data

Existing data may be used to support or substantiate other existing data. Inferences drawn for corroboration must be clearly identified, justified, and documented. The level of confidence associated with corroboration is related to the quality of the program under which corroborating data were developed, and the number of independent data sets. The extent of corroborating data appropriate for qualification purposes should be determined and documented on a case-by-case basis.

## Method C. Use of Confirmatory Testing

Testing may be conducted under a 10 CFR 60, Subpart G, QA program to reexamine the features of an existing data set. Such testing need not necessar'ly employ the same test method, provenance, or test conditions as the original work, but the applicability of confirmatory testing just be justified and documented. One example of confirmatory testing could be testing conducted with different methods and equipment, but which investigates subsurface structure at the location interest. The extent of confirmatory testing appropriate for qualification purposes should be determined and documented on a case-by-case basis.

#### Method D. Demonstrate Consistent QA Program

Demonstrate that a QA program consistent with the Project QA an (NNWSI 88-9) was used during specific activities that produced the existing geophysical data or data analyses.

Methods B, C, and D should be accompanied by a documented technical review by two or more qualified independent reviewers, to determine the quality of the geophysical data or data analyses, and the merit of any inferences used for qualification. Additional confidence and/or credibility will be achieved when a combination of Methods A through D is used.

#### THE QUALIFICATION PROCESS

.lanned application of the qualification process to geophysical data is described below in several steps:

- Geophysical data for the Yucca Mountain site and region veloped by the DOE and its contractors; or data developed outside the DOE repository program by energy or mineral companies, geophysical service contractors, national laboratories, or universities; or data published in technical or scientific publications, will be identified by Project Principal Investigators. This step is generally satisfied for existing geophysical data by Chapter 2 of this report, except for topical areas as noted in Section 1.0.
- 2. The geophysics integration activity (SCP Section 8.3.1.4.1.2) will review the existing data sets described in Chapter 2, and select for qualification any that are likely to be used for: characterizing natural barriers important to waste isolation, design of engineered structures important to waste isolation and/or preclosure radiological safety, or issue resolution in support of repository licensing. (The Amargosa Valley seismic reflection test line is identified below as the best candidate data set for the initial application of the qualification process to geophysical data.)
- 3. The geophysics integration activity will submit a Qualification Request to the Yucca Mountain Project Office, recommending application of one or more of the qualification methods described above to the selected data. The rationale for selection of data sets will be documented, including the reason why the data collection or analyses cannot be repeated, and an assessment of the existing data with respect to the qualification methods and attributes listed in procedure AP5.9Q.
- 4. The Omalification Request will be reviewed by the Director, Regulatory and Site Evaluation Division, Yucca Mountain Project Office, or designee in accordance with AP5.9Q. If the request is approved, a technical review or peer review, as appropriate, will be convened by the Director in accordance with the procedure. The outcome of the review will determine the status of data qualification. Documentation of the outcome, including all corroborating or confirmatory data, will become a QA record.

#### CANDIDATE DATA SETS FOR QUALIFICATION

A number of geophysical data sets are good candidates for use in licensing, but they differ with respect to the nature of data or data analyses for qualification. Two major, related discriminating factors are whether data processing contributed significantly to a product, and whether industry-standard commercial data collection or processing services were used.

Qualification of directly measured data is potentially more straightforward than review of data analyses. Reviewers will be required to consider measurement procedures, calibrations, record keeping, surveillance, environmental conditions, corroborating measurements, and confirmatory data, as appropriate. Examples of such data include most gravity and magnetic measurements, and survey or navigation control for these measurements. Other types of direct measurements may also be qualified, such as standard borehole geophysical logs, and ground motion recordings for seismic refraction surveys.

In general, geophysical data will be processed for use in licensing. Data are typically presented in processed maps and cross-sections. Geophysical models are presented as map or cross-section representations of earth structure, which require calculation of earth response. Data processing includes a wide range of corrections and enhancements varying in importance and complexity. For qualification of processed products, reviewers will be required to consider software QA where controls on the use of software could have a significant effect on quality. Examples of such products are terrain-corrected Bouguer gravity maps, composite aeromagnetic maps, seismic reflection sections, and velocity models from seismic refraction data.

In cases where industry-standard data collection or processing services have been performed by reputable geophysical contractors, there may have been proprietary or contractual restrictions on the implementation or verification of QA. For qualifying products of this nature, reviewers will determine the specific QA requirements that have not been explicitly addressed, and determine whether the measures actually taken constitute reliable and effective guality control that can be demonstrated in licensing.

Regional gravity and aeromagnetic maps are candidates for qualification because they represent accumulated data relevant to establishing the regional context for geologic structure and other aspects of the Yucca Mountain site. By qualifying the regional maps, a substantial portion of the available data will be addressed. Reacquisition of regional data would be prohibitive and unnecessary.

Potentially important gravity maps include Ponce et al. (1988), Kane et al. (1979), and Healey et al. (1980) from the reference list of Section 2.1 of this report. The reports of Snyder et al. (1981), Jansma et al. (1982), and Saltus (1988) contain principal facts for gravity stations in the Yucca Mountain region. Additional detailed gravity and magnetic data for the site area and vicinity will be collected during site characterization, and will confirm or supersede the existing regional data over a large area. Qualification of existing gravity measurements from the site area, such as

the borehole gravity surveys of Healey (1986) and Healey et al. (1984), will be deferred at least until the detailed data are available. Similarly, qualification of gravity measurements from the monitoring program of Harris and Ponce (1988) and from the survey of Zumberge et al. (1988) can be deferred until additional data are collected under the fully qualified QA program.

Several existing regional magnetic maps are candidates for qualification, including the 1:750,000 scale map of Hildenbrand and Kucks (1988), and the mosaic 1:250,000 sheets complete or in preparation (Saltus and Snyder, 1986; Saltus and Ponce, 1988). The 1:48,000 scale map of Kane and Bracken (1983) is also a good candidate because it covers a larger area than the planne detailed aeromagnetic survey (SCP Activity 8.3.1.17.4.7.4). Qualification if existing ground magnetic data from the site area and vicinity will be deferred at least until accuisition and analysis of the planned detailed aeromagnetic survey, and the planned feasibility test of gravity and ground magnetic methods for volcanism studies (see Section 3.3). Similarly, qualification of paleomagnetic data will be deferred until additional, corroborating or confirmatory measurements have been made during site characterization.

High-resolution upper-crustal seismic refraction lines are good candidates for qualification for several reasons. Structural profiles interpreted from these data currently constitute one of the most extensive and reliable data sets for use in tectonic studies. Refraction lines have already been run across Yucca Mountain, along Fortymile Wash, through Crater Flat, and across the Amargosa Desert, where seismic reflection lines are tentatively planned. Qualification of the ground motion recordings from these surveys (Sutton, 1984, 1985) will be straightforward, and qualification of processing (Ackerman et al., in USGS, 1988) can readily be addressed independently from that of recorded data. In addition, the reconnaissance survey reported by Hoffman and Mooney (1983), which used UNE sources and one high-explosive shot, will be examined. There appears to be no reason for qualifying the refraction data reported by Pankratz (1982), at least until corroborating detailed gravity and seismic reflection data are available, and until reinterpretation of the original refraction data has been performed using all available borehole control and velocity data.

Existing seismic reflection data are of little use for site characterization, except for the Amargosa Valle, est line (Brocher et al., 1989), as discussed in Section 2.5 of this report. The test line was acquired recently, with systematic quality ontrol, and is the best candidate for the first geophysical data set to be malified under procedure AP5.9Q. The line will probably eventually be used to the other reflection lines that are tentatively planned for the site area. The data set is a good choice because industry-standard data collection and processing services were provided by a geophysical contractor. The review will help to establish the basis for qualifying other existing data that have been collected or processed using industry-standard practices.

Borehole geophysical logs are a candidate data set for qualification, principally because many existing boreholes in the immediate vicinity of the site have been completed so that they can no longer be logged. These logs will probably be important in assessing lateral variability of rock characteristics or hydrologic conditions in the site area. Even for boreholes that can be reentered, relogging is not generally a viable alternative for moisture sensitive logs in the UZ. This is because uncased or uncemented boreholes are subject to drying and moisture movement associated with gas circulation. Efforts are currently underway to assemble, correct, replot, and analyze the existing borehole logs. A malification request can be formulated when documentation of these efforts is complete.

In summary, the seismic reflection test line of Brocher et al. (1989) is the best candidate for the first set to be submitted for qualification review. The necessary documentation of the data set is available, and the QA status of the activity is amenable to qualification of the data set for use in licensing. Preparation of a qualification request for review of the Amargosa Valley reflection test line is identified as a high-priority activity for geophysical integration in Section 3.2 of this report. Among the other data sets identified, the regional maps of potential field variation, the existing seismic refraction data, and the borehole geophysical logs are good candidates for future gualification review.

#### REFERENCES

- Ackermann, F.D., W.D. Mooney, D.B. Snyder, and V.D. Sutton, 1988. "Preliminary Interpretation of Seismic Refraction and Gravity Studies West of Yucca Mountain, Nevada and California," in Carr, M.D., and J.C. Yount (eds.), <u>Short Contributions to the Geology and Hydrology of a Potential</u> <u>Nuclear Waste Site at Yucca Mountain, Southern Nevada,</u> U.S. Geological Survey Bulletin 1790, pp. 23-33.
- Brocher, T.M., P.E. Hart, and S.F. Carle, 1989. Feasibility Study of the Seismic Reflection Method in Amargosa Desert, Nye County, Nevada, U.S. Geological Survey Open File Report, in preparation.
- DOE (U.S. Department of Energy), 1989. <u>Technical Planning Basis</u>: Site <u>Characterization, Yucca Mountain Site</u>, <u>Nevada Research and Development</u> <u>Area</u>, YMP/CC-0005, U.S. DOE, Las Vegas, Nevada.
- Harris, R.N. and D.A. Ponce, 1988. <u>High-Precision Gravity Network to Monitor</u> <u>Temporal Variations in Gravity Across Yucca Mountain, Nevada,</u> U.S. Geological Survey Open File Report 88-243, 19 p. (in press).
- Healey, D.L., 1986. Borehole Gravity Meter Survey in Drillhole USW G-4, Yucca Mountain Area, Nye County, Nevada, U.S. Geological Survey Open File Report 86-205, 18 pp.
- Healey, D.L., F.G. Clustom, and D.A. Glover, 1984. Borehole Gravity Meter Surveys in Drillholes USW G-3, UZ-25 p#1, Yucca Mountain Area, Nevada, U.S. Geological Survey Open File Report 84-672, 16 pp.

Healey, D.L., P.R. Wahl, and H.W. Oliver, 1980. "Bouguer Gravity Map of Nevada, Death Valley Sheet," Nevada Bureau of Mines and Geology Map 69, scale 1:250,000.

- Hildenbrand, T.G., and R.P. Kucks, 19°9. "Total Intensity Magnetic Anomaly Map of Nevada," Nevada Bureau of Mines and Geology Map 93A, scale 1:750,000.
- Jansma, P.E., D.B. Snyder, and D.A. Ponce, 1982. Principal Facts of Gravity Stations with Gravity and Magnetic Profiles from the Southwest Nevada Test Site, Nye County, Nevada as of January, 1982, U.S. Geological Survey Open File Report 82-1041.
- Kane, M.F., D.L. Healey, D.L. Peterson, H.E. Kaufmann, and D. Reidy, 1979. "Bouguer Gravity Map of Nevada, Las Vegas Sheet," Nevada Bureau of Mines and Geology Map 61, scale 1:250,000.
- Hoffman, L.R., and W.D. Mooney, 1983. A Seismic Study of Yucca Mountain and Vicinity, Southern Nevada: Data Report and Preliminary Results, U.S. Geological Survey Open File Report 83-588, 50 p., 1 plate.
- Pankrat , L.W., 1982. Reconnaissance Seismic Refraction Studies of Calico Hills, Wahmonie, and Yucca Mountain, Southwest Nevada Test Site, Nye County, Nevada, U.S. Geological Survey Open File Report 82-478, 25 p.
- Ponce, D.A., R.N. Harris, and H.W. Oliver, 1988. "Isostatic Gravity Map of the Nevada Test Site and Vicinity, Nevada," U.S. Geological Survey Open File Report 88-664, scale 1:100,000 (in press).
- Saltus, R.W., 1988. Gravity Dat. for the State of Nevada on Magnetic Tape, U.S. Geological Survey Open File Report 88-433, 20 p.
- Saltus, R.W., and D.B. Snyder, 1986. "Aeromagnetic Map of Nevada, Caliente Sheet," Nevala reau of Mines and Geology Map 89, scale 1:250,000; includes metric aeromagnetic map, scale 1:1,000,000.
- Saltus, R.W., and D.A. Ponce, 1988. "Aeromagnetic Map of Nevada, Las Vegas Sheet," Nevada Bureau of Mines and Geology Map 95, scale 1:250,000; includes merged aeromagnetic map, scale 1:1,000,000.
- SNL (Sandia National Laboratories), 1987. Site Characterization Plan--Conceptual Design Report, SAND84-2641, Albuquerque, New Mexico.
- Snyder, D.B., C.W. Roberts, R.W. Saltus, and R.F. Sikora, 1981. <u>Magnetic</u> <u>Tape Containing the Principal Facts of 64,402 Gravity Stations in the State</u> <u>of California, U.S. Geological Survey Report, 30 p.; available from</u> <u>National Technical Information Service, U.S. Department of Commerce,</u> <u>Springfield, VA 22161, NTIS PB-82168287.</u>
- Sutton, V.D., 1984. Data Report for the 1983 Seismic Refraction Experiment at Yucca Mountain, Beatty, and Vicinity, Southwestern Nevada, U.S. Geological Survey Open File Report 84-661, 66 p., 11 plates.

Sutton, V.D., 1985. Data Report for the 1985 Seismic Refraction Experiment at Yucca Mountain, Beatty, and Vicinity, Southwestern Nevada, U.S. Geological Survey Open File Report 85-591, 96 p., 17 plates.

Zumberge, M.A., R.N. Harris, H.W. Oliver, G.S. Sasagawa and D.A. Ponce, 1988. Preliminary Results of Absolute and High-Precision Gravity Measurements at the Nevada Test Site and Vicinity, Nevada, U.S. Geological Survey Open File Report 88-242, 29 p.

#### 4.0 SUMMARY AND CONCLUSIONS

The Yucca Mountain site is structurally complex, and has other aspects that complicate geophysical exploration, such as rugged topography and low seismic velocity near the surface. Because of these conditions and limitations on the scope of past activities, several important questions of hydrologic and tectonic significance have not been addressed by geophysicists. However, major results have been obtained and will be used as the basis for planned site characterization activities. (It is very likely that some existing data will be used for licensing, but specific data must be identified and qualified through the process described in Section 3.4, and cannot be definitely identified at present.)

Major results include crustal exploration using gravity and seismic refraction, which have provided information on the thickness of tuffs at Crater Flat, and depth to the Paleozioc-Tertiary contact associated with major features of the site (see Section 2.4). Buried volcanics have been discovered by magnetic surveys in the vicinity of the site and in the Amargosa Desert (Section 2.2). Intermediate-depth seismic refraction has been demonstrated at the site, and the use of shallow seismic methods to characterize faults has also been demonstrated (Sections 2.4 and 2.5). Geoelectric methods have been used in the Yucca Mountain region and site area for vertical sounding and profiling applications, with preliminary results that will be evaluated during site characterization (Section 2.3).

Major problems were encountered with high-resolution seismic reflection along the eastern flank of Yucca Mountain (Section 2.5). Definitive analysis of these results will require additional field work, so the SCP contains plans to review the past work and identify an approach for future surveys.

Other geophysical methods have become generally available since the previous work was performed, such as vertical seismic profiling (VSP), methods for surface seismic data acquisition and processing, and thermal infrared imaging. Because the success of these new methods will depend on site-specific factors that have not yet been fully assessed, the SCP contains plans to review the available capabilities, perform feasibility testing, and apply them to site characterization as appropriate.

Plans in the SCP call for gravity, magnetics, and refraction surveys with increased coverage and resolution relative to existing data. At the same time, feasibility testing and technical review are planned to assess methods for detailed shallow and intermediate-depth investigation. Reviews of seismic and geoelectric methods will assess past results, and produce recommendations for using these methods to augment gravity, magnetics, and seismic refraction. Additional feasibility testing may be recommended by these reviews, and is already planned to evaluate the Mini-Sosie method, VSP, various gravity and magnetics applications, and radiometric methods (see Section 3.3).

Feasibility studies in addition to those planned in the SCP, possibly leading to new characterization activities, have been identified in this report and will be considered by the geophysics integration activity. Changes to the SCP program that are generated by the integration activity will be subject to review and approval in accordance with extant procedures for controlling the SCP progress reports, Study Plans, and the Project technical baseline.

Contrasts in seismic velocity, density, electrical resistivity, and magnetic properties are needed to observe geophysical structure. The Paleozoic-Tertiary contact is marked by a contrast in density, magnetization, and possibly seismic properties. Much of the structure of interest lies within the Tertiary section, where the density, magnetic, seismic, and electrical properties of welded tuff contrast with nonwelded tuff. However, contrast can be muted even at major interfaces such as the alluvium-tuff, welded-nonwelded tuff, and Paleozoic-Tertiary contacts, where there may have been weathering or alteration such that the variation in geophysical properties is gradational.

Because of site-specific factors and the problems encountered with seismic reflection at the site, it is presently expected that each geophysical method will produce only a partial geophysical model. An integrated approach is therefore planned for major objectives including investigation of the large hydraulic gradient, characterization of the repository block, detection and delineation of faults, and structural profiling across the site area.

Within the SCP framework, many different activities depend on geophysical data. Integration among activities is needed to ensure that maximum benefit is obtained from geophysical surveys and interpretations (see Section 3.2), and to ensure that the characterization program provides timely information appropriate to support a repository license application.

#### 5.0 ACKNOWLEDGMENTS

In addition to the editors, the following USGS earth scientists contributed to subsections in Section 2 shown in parentheses: D.A. Ponce and R.F. Sikora (gravity); J.G. Rosenbaum and D.E. Champion (magnetics); D.P. Klein and D. Hoover (electric); W.D. Mooney (seismic refraction); J.R. Evans (tele= seismic); T.M. Brocher (seismic reflection); J. Friedman, M. Podwysocki, and J.A. Pitkin (remote sensing); and K. Clover and A. Flint (borehole geophysics). Section 3 on plans was pulled together largely by E. Hardin, and C. Fridrich of the Yucca Mountain Project Office. It is based on the Site Characterization Plan, but includes current information from available study plans by T.M. Brocher, J.S. Bergquist, S.F. Carle, W.E. Wilson, J. Rousseau, F. Sinton, J. Czarnecki, and others. This report has had the benefit of major reviews by R.J. Blakely, D.A. Ponce, D.V. Fitterman, T.G. Hildenbrand, B. Langer, and E.L. Majer. The maps for Sections 2 and 3 were produced by E. Ezra of EG&G Energy Measurements Group/Las Vegas, with assistance from D. Brickey of EG&G.

## APPENDIX A

Table of Contents for Volumes IV and V of the Site Characterization Plan (Section 8.0 to Section 8.3.1.17.5)

## TABLE OF CONTENTS

## VOLUME IV

Page

CHAPTER 8 SITE CHARACTERIZATION PROGRAM	
8.0 Introduction	8.0- 1
8.1 Rationale for the site characterization program	8.1- 1
8.1.1 The issues-based approach to planning site	
characterization	8.1- 1
8.1.1.1 Derivation, structure, and scope	8.1- 1
8.1.1.2 Application in the site characterization plan	8.1- 4
R 1 2 Teeno recolution strategy	8.1- 5
2 1 2 1 Teena identification	8.1. 5
0.1.2.2 Doutermanna allegation	8 - 5
0.1.2.2 Ferrormanoe arrovation in the termine	8 1- 11
0.1.2.5 Data correction and distypts	8 1- 14
0.1.2.4 ISSUE resolution documentation	0.1-17
6.1.2.5 Application of the issue resolution strategy	0.7- 71
8.2 issues to be resolved and information required during site	0 0 1
characterization	8.2- 1
8.2.1 Issues to be repolved	8.2= 1
8.2.1.1 Site-specific issues hierarchy	8.2-15
8.2.1.2 Other issuas	8.2- 15
8.2.2 Approach to issue resolution	8.2- 61
8.2.3 Issue tracking	8.2- 66
8.3 Planned tests, analyses, and studies	8.3- 1
8.3.1 Site program	8.3.1- 1
8.3.1.1 Overview of the site program: Role of alternative	
conceptual models	8.3.1.1- 1
8.3.1.2 Overview of the geohydrology program:	
Description of the present and expected dec-	
hudrologic characteristics required by the per-	
formance and design issues	8312-1
2 2 1 2 1 Torractionstion, Ctudice to provide a	
deveription of the regional hudralagia	
description of the regional hydrologic	0 0 1 0- 00
System · · · · · · · · · · · · · · · · · · ·	0.0.1.6 94
8.3.1.2.1.1 Study: Characterization of the mete-	
orology for regional hydrology	8.3.1.2= 91
8.3.1.2.1.1.1 Activity: Precipitation and meteor-	
ological mentoring	8.3.1.2- 97
8.3.1.2.1.2 Study: Characterization of runoff and	
streamflow	8.3.1.2-102
8.3.1.2.1.2.1 Activity: Surface-water runoff	
mortoring	8.3.1.2-103
8.3.1.2.1.2.2 Act ity: Transport of debris by	
severe runoff	8.3.1.2-112
8.3.1.2.1.3 Study: Characterization of the regional	
ground-water flow system	8.3.1.2-114
8.3.1.2.1.3.1 Activity: Assassment of the regional	STOTATE AND
hydrogelogia data neede in the	
nyarvyeoroyro data needo ru che	9 9 1 0-114
Satu Sa Zone	0.2.1.4"1.4

]

R.

- 7.

TABLE OF CONTENTS (continued)

		8.	3	.1	. 2	, 1	.3	. 2	Activity: Regional potentiometric-								
									level distribution and hydrogeologic								
									framework studies			8.1	3.	1.	2.	11	.8
		8.	.3	.1	.2	.1	.3	.3	Activity: Fortymile Wash recharge								
									study	Ξ.		8.	3.	1.	2-	12	25
		8	3	. 1	. 2	.1	. 3	. 4	Activity: Evapotranspiration								
				1	1.00	1			erudiae			8	2	1	5.	. 1 5	2.4
	0	5		- 5				0+	Dedutop			0.	••	* '	*		14
	0.	0	• *	1 64	**	17	1	ac	udy: Regional nydrologic system						~		
						1	1.1	sy	nthesis and modeling			8.	3.	4.	2.	113	57
		8	. 3	. 1	.2		4	.1	Activity: Conceptualization of								124
									regional hydrologic flow models			8,	3.	1.	2.	-13	37
		8	. 3	. 1	.2	. 3	4	.2	Activity: Subregional two-								
									dimensional areal hydrologic								
									modeling	1		8.	3.	1.	. 2.	-13	39
		8	.3	. 1	. 2	. 1	. 4	.3	Activity: Subregional two-								
									dimensional cross-sectional hydro-								
									logia modeling			0	2		0.		10
		0	0	1	0		÷ ,		togat moderning			Q .	0.	*	- 64		14
		9	• •	1.4				1.4	Activity: Regional chree-dimensional	•				į.			
							1	1	nyarologic modeling		97	8.	3.	1	4	-11	44
	0.	3	- 4	. 4		1	5	Ap	plication of results			8,	3.	1	. 2.	-1/	47
*	3.	1	. 2	14		I	nve	st	igation: Studies to provide a								
						de	930	ri	ption of the unsaturated zone hydro-								
						10	ogi	C	system at the site			8.	3.	1	. 2	-1	48
	8	.3	.1	.2	2.2		1	St	udv: Characterization of unsaturated-								
								zc	ne infiltration		22	8.	3.	1	2	- 1	57
		8	.3	. 1	. 2		2.1	. 1	Activity: Characterization of	18						-	
		10							hudrologia properties of surficial								
									Witheralde headereres it serviced			0			~		= -
		0	5	i i					Activity Puplication of activity		•	0.	5.	+	. 4	- 1	57
		Q	. 2				6		Activity: Evaluation of natural				U.				2.5
									inflitration	•	•	8.	3.	1	. 2	-1	64
		8	. 3		1.4		2		Activity: Evaluation of artificial								
									infiltration	×		8.	3.	1	. 2	-1	72
	8	.3	.1		2.2	2.	2	St	udy: Water rovement tracer tests								
								US	sing chloride and chlorine-36								
								me	asurements of percolation at								
								Y	icca Mountain			9	2	1	0	-1	70
		5	1	1	1.5	2	2	2.	Activity: "hlaride and chloring-36	*	1.	~	*	1	14		12
		~					** *		accevery, caround and chroting-20								
									measurements or percolation at								
			2					1	rucca Mountain	*	۰.	8	3		. 2	-1	80
	8	. 3			4.10	6.	3	St	tudy: Characterization of percolation								
								11	n the unsaturated zonesurface-based								
								S	tudy			8	.3	.1	.2	-1	81
		8	3.3	3.	1.	2.	2.	3.	1 Activity: Matrix hydrologic								
									properties testing		1	8	3	.1	. 2	-	83
		8	3.3	3.	1.	2.	2.	3.	2 Activity: Site vertical borehole	16							
									studies			2	2	4			00

The start of

S. S.

deres de

San San Ki

TABLE OF CONTENTS (continued)

horizontal borehole study	8.3.1.2-221
8.3.1.2.2.4 Study: Characterization of Yucca	
Mountain percolation in the unsaturated	
zone exploratory shaft facility study	8.3.1.2-232
8.3.1.2.2 4.1 Activity: Intact-fracture test in	
the exploratory shaft facility	8.3.1.2-238
8.3.1.2.2.4.2 Activity: Infiltration tests in the	
exploratory shaft facility	8.3.1.2-252
8.3.1.2.2.4.3 Activity: Bulk-permeability test in	
the exploratory shaft facility	8.3.1.2-271
8.3.1.2.2.4.4 Activity: Radial borehole tests in	
the exploratory shaft facility	8.3.1.2-281
8.3 1.2.2.4.5 Activity: Excavation effects test	
in the exploratory shaft facility	8.3.1.2-293
8.3.1.2.2.4.6 Activity: Calico Hills test in	
the exploratory shaft facility	8.3.1.2-300
8.3.1.2.2.4.7 Activity: Perched water test in the	
exploratory shaft facility	8.3.1.2-300
8.3.1.2.2.4.8 Activity: Hydrochemistry tests in	
the exploratory shaft facility	8.3.1.2-304
8.3.1.2.2.4.9 Activity: Multipurpose-borehole test-	
ing near the exploratory shafts	8.3.1.2-308
8.3.1.2.2.4.10 Activity: Hydrologic properties of	
major faults encountered in main test	
major faults encountered in main test level of the exploratory shaft	
major faults encountered in main test level of the exploratory shaft facility (ESF)	8.3.1.2-316
major faults encountered in main test level of the exploratory shaft facility (ESF)	8.3.1.2-316
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-322
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-322 8.3.1.2-323
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-323 8.3.1.2-323
<pre>Major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-323 8.3.1.2-323
<pre>Major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-323 8.3.1.2-323 8.3.1.2-334
<pre>Major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-323 8.3.1.2-323 8.3.1.2-334
<pre>Major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-323 8.3.1.2-334 8.3.1.2-334 8.3.1.2-334
<pre>Major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-323 8.3.1.2-334 8.3.1.2-334 8.3.1.2-334
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-323 8.3.1.2-334 8.3.1.2-334 8.3.1.2-337 8.3.1.2-341
<pre>Major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-323 8.3.1.2-334 8.3.1.2-334 8.3.1.2-337 8.3.1.2-341
<pre>major faults encountered in main test level of the exploratory shaft facility (ESF)</pre>	8.3.1.2-316 8.3.1.2-320 8.3.1.2-320 8.3.1.2-322 8.3.1.2-323 8.3.1.2-334 8.3.1.2-334 8.3.1.2-337 8.3.1.2-337 8.3.1.2-341

TABLE OF CONTENTS (continued)

											Page
		8		3.	1.	2	.2	. 8	. 2	2 Activity: Validation of conceptual and	
										numerical models of fluid flow	
	1					ŝ.				throigh unsaturated, fractured rock	8.3.1.2-345
	8	3			2.,	2	. 9		St	udy: Lite unsaturated-zone modeling	
			Ξ.						an	nd synthesis	8.3.1.2-350
		5		5 + 1	4.4	6	4	. 9		Activity: Conceptualization of the	
										unsaturated-zone nydrogeologic	0 0 1 0 051
		0			2	6	0			System	0.0.1.2-001
		0	* *	2.4.1	4.1	6	6	, y	. 4	Activity: Selection, development, and	
										testing of hydrotogic modeling computer	0 0 1 0-050
		0	١.			5	5	0	0	Detivity. Cimulation of the netural	0.0.1.4.000
		C		2.0.	4.4	fe 1	4		. 2	budrogoologia sustem	0 3 1 2-256
		0				0	0	6	4	1 Astivity. Stochastic modeling and	0.0.11.2-000
				6.913	• •	64.1			1	uncertainty analysis	8 3 1 2-350
		8	Ē.	1	1.	2	2	9	5	Activity: Site unsaturated=zone	
		1								integration and synthesis	8.3.1.2-361
	8	3			2.	2.	.11	0	A	application of results	8.3.1.2-362
8	. 3	1		2.	3		Inv	ve	st	igation: Studies to provide a	
						4	ie	SC.	ri	ption of the saturated zone hydro-	
						13	Loc	Ţì	C	system at the site	8.3.1.2-364
	8	573		l d	2.	3.	1		St	udy: Characterization of the site	
									sa	aturated-zone ground-water flow	
									sy	/stem	8.3.1.2-371
		8		3.	1.	2	.3	.1	.1	Activity: Solitario Canyon fault	
										study in the saturated zone	8.3.1.2-372
		8		3	1.	2	. 3	.1	. 2	2 Activity: Site potentiometric-level	
						Ľ				evaluation	8.3.1.2-375
		5		5	1.	2	. 3	.1	. 3	Activity: Analysis of single- and	
										multiple-well hydraulic-stress	
			÷.,			2				tests	8.3.1.2-383
		0	• •	2.4.	4 ×	4	2	e she	1.11	ACCIVITY: Multiple-well inter-	
		0				0	2	1		lerence testing	8.3.1.2-393
		0		1	4. 8	64	. 2	* *	• •	p Activity; resting or the C-hole	0 0 1 0 100
		¢	1	2	1	2	2	1	6	Activity, Well testing with	8.3.1.2-400
		1	1			-		1.14	1.5	concertative tracers throughout	
										the site	0 2 1 2-400
		-2		3	1.	2	.3	1	6	Activity: Mesting of the C-hole	0.0.1.400
		1								sites with reactive tracers	8 3 1 2-417
		8		3.	1.	2	.3	.1	. 8	Activity: Well testing with	91121212-721
					1	1		1		reactive tracers throughout the	
										site	8.3.1.2-423
	8	. 2		1.	2.	3	.2		St	tudy: Characterization of the	
									Se	aturated zone hydrochemistry	8.3.1.2-424
		8		3.	1.	2	.3	.2	. 7	Activity: Assessment of saturated-	
										zone hydrochemical data availability	
										and needs	8.3.1.2-425

# TABLE OF CONTENTS (continued)

																															Pa	ge
		8.3	.1	.2	.3.	2.	2	A	cti	iv:	ity	71		Hy	dr	00	he	mi	ca.	1 0	har	ac	te	r-								
								11	zat	tid	on	01	f	wa	te	r	ir	1.5	he	up	per	p	ar	t								
								0	5 1	the	a 3	at	tu	ra	te	d	zc	ne							1	1	8.	3.	1	.2	-4	26
		8 3	1	2	3	2.	3	A	at	iv	1+1	7.1		RA	ori	or	al	h	vd		hen	ic	al					Ő.				
		414					*		har		at a			at.	2-2	5			1								8	3	1	2	-4	31
		0 0			5	0	×	3.	na.	1 41			* *	G	**	he		.1					-	21	1	1	~ .					~~
		0.0	1.4	16		4 .	4	14	06.	TA?	103	(*		SÅ	116	116	:3-	. 5	0r	ad	u u ı	au	eu	Ξ.			6	5				22
								ZQ	one	9 1	nyc	lre	20	ne	ma	st	ry			. 1	1.1		1	1	1	*	0.	2.	-	4	- 4	20
	8.	3.1	2	.3	. 3	S	tud	iy	1 1	Sat	cur	tat	te	a	zo	ne	1	iyd	ro.	rođ	10	sy	st	em					12	÷.		
						S	ynt	:he	es:	15	ar	1d	m	od	el	in	g		18		÷						8,	3.	1	,2	-4.	34
		8.3		.2	.3.	3.	1	A	ot:	ivi	ity	11	. 1	Co	nc	ep	tu	al	iza	ati	on	of										
								Si	at	ura	ate	be	Z	on	e	fl	OW.	m	ode	215	Wi	th	in									
								2)	he	bo	our	hda	ar	ie	S	of	t	he	a	oce	ssi	.b1	e									
								er	nv:	ire	onr	ner	nt											1			8.	3.	1	. 2	-4	35
		8 3	1	2	3	3	2	A	-	iv:	++	2.4		De	ve	10	opr	an	ŧ,	f	fra	ot	ur	6								
		~				~ .	~	54	ati	uni	-1-	m	NA.	01									-	Π.			8	2	1	2	-4	36
		0 5		0	3	2	2	210		1	1.4.4	* *	1 'VA	11 0	1 -			1.		÷.,	£1.			. ^	1	1	~	~ .		1		~ ~
		9.0		1.60	1.2.1	2.	2	A		4 V .	103	( A .		ua.	4.0		.00	.10	1 1	**	1.4.6	w	1.	1	1							
								P	ati	ns,	1.3	111	1X	es	1	an	ia .	ve	100	336	162	W	10	nı	n							
								C.	ne	Sa	atu	113	at	ed	Z	on	10	0.7	t.	ne	acc	es	51	DT	e		1					
								eı	nv:	ird	: nn	ner	nt	. *		1						1.18			4	×.,	8.	3.	de la	. 2	-4	41
	8	3.1	2	.3	, 4	A	ppl	Lic	cat	cid	on	01	£	re	su	lt	S	1.0		1.	x . 1	1		÷		1.1	8.	3.	1.	.2	-4	44
8.	3.	1.2	1.4		Sch	ed	lule	e d	for	r t	the	5 6	ge	oh	yd	Irc	10	gy	p:	rog	ran	1					8.	3.	.1	,2	-4	45
.3.	1	3	OV	er	vie	W	of	tl	he	q	eod	che	em	is	tr	v	pr	:oa	rai	n:												
			De	SC.	rit	sti	on	0	f	the	e r	bre	es	en	t	an	nd	ex	Dec	ste	d e	reo	-									
			ch	em	ica	1	cha	ari	act	ter	ri	at	ie	8	re	au	117	be	bi	, t.	he											
			ne	rf.	0.00	nan	00	a	nd	di	0.01	m	n	ie	e 11	0.0											8	2	4	2		1
8	2	1.3	1		The	10.0		721		an	,	n ci		Ai	00		~			i da			1	1			×.	~ 1				
· · ·						20	mat	1.1.	~~	w. s. s				**	0		1.	**			hin		-									
							and L	- 1	UII OII	01	1	Y CL L	50	-	wi.	1911	14.5	1.00	¥ .	N d lu	- 1 -	1 6	ne									
					por	en	icia	44	ei	np.	190	Cer	ne	nc	- 5	101	122	ion	a	na .	alc	ng							4			
					pot	en	itla	a.1	Ξ.	101	W	pat	c n	S		1			1	κ.			1		È.		δ.	3.		. 3	*	30
	8	. 3		.1	1	S	stud	iy	n.	G	roi	ind	2-	wa	te	r	20	lem	13	cry	me	de	1	· *			8.	3	.1	. 3	-	37
	8	.3.1	3	.1	. 2	A	ppl	11	ca	tic	on	01	f	re	SU	ilt	S			Χ.	÷				1.		8.	3.	.1	.3	**	39
8.	3	1.1	3.2	- 1	Int	res	stig	gai	ti	on	1	St	tu	di	85	5 t	0	pr	OV.	ide	ir	fo	IM	a-								
					tic	n	on	m	in	er	ald	bg	Y,	p	et	rc	1.0	gy	1	and	ro	ock										
					che	mi	312	ry	W	it	hir	n f	ch	e	pc	te	ent	ia	1 .	emp	lac	em	ien	t								
					hor	riz	ion	a	nd	a	101	na	p	ot	er	iti	.a]	£	10	W D	ath	15					8.	3	.1	. 3	-	40
	3	.3.3	. 3	.2	.1	5	Stud	vb	:	M	ine	eri	al	00	TV.	T	set	ro	10	TV.	ar	d										
					-		her	ni	NT	rv.	0	£		an	ST	507		na	+ 1	uau wau	12						8	2	1	2		42
		0	2 1	2	2	1	1	A	~+.	- 1	1+1		**	De			~	110			+ 1 -	****	-		1	1.1	~	~	-			20
		0			1.64.1	1 alt 1	*	-	2	the Vie	44	Y ·		20	i la di	0.		440	D NA	o m h		120	ip.u	Y.			0	2		5		10
				-			~	0	I.	C EL	е .	1.01	po	2a	in	27	PL3	Lng	[M]	emp	er				*	•	0	3	• +	1.2	-	43
		8	5	3	. 4 .		4	A	Ct.	14	161	Y 1		MI	ne	era	1.1	<b>a</b> 1	st	<b>71</b> D	uti	.on	S									
								b	et	we	en	t	he	n	105	st	re	ock	a	nd	the	\$										
								8	cc	es	sil	01	e	er	ivi	LIC	onr	nen	t					1.4			8	3	.1	. 3	-	45
		8.3	3.1	3	.2	.1.	. 3	A	ct	iv	it	Y:		FI	ac	sti	ire	e m	in	era	100	TY					8	3	.1	.3	-	47
	3	.3.:	1.3	3.2	.2	01	Stud	dy	-	Н	ist	to	ry	¢	f	m	ine	era	10	gio	a	bc	ge	0-								
						0	chen	mi	ca	1	al	te	ra	63	.01	1 0	of	YU	100	a M	loui	nta	in	1 .			8	3	.1	. 3	-	49
		8.3	3.1	3	.2	.2.	.1	A	ct	iv	it	V:		Hi	st	:01	TV	of	m	ine	ra	Loc	ric									
								a	nd	a	eo	ch	em	ic		1	alt		at	ior	0	EY	uc	ca								
								M	0	nt	ai	n															8	2	1	3	-	49
									1.00.04	10.00		***						R 8				, ,					4	1.00	8 40			1. 1.

TABLE OF CONTENTS (continued)

Page

		8	. 3	. 1	. 3	.2	2.2.2 Activity: Smectite, zeolite,			
							manganese minerals, glass dehydra-			
							tion, and transformation	1	8.3.1.3-	52
	8	.3	.1	.3	. 2	.3	3 Application of results	1.1	8.3.1.3-	54
8	. 3	.1	. 3	.3		In	nvestigation: Studies to provide informa-			
						ti	ion required on stability of minerals and			
						al	lasses		8.3.1.3-	55
	8	.3	1	. 3	. 3	.1	1 Study: Natural analog of hydrothermal			
	1		1	1			systems in tuff		8.3.1.3-	56
	8	3	1	3	3	. 2	2 Study: Kinetics and thermodynamics of			
							mineral evolution		8.3.1.3-	59
		8	3	1	2	2	2 2 1 Activity: Kinetic studies of			
		×		t ele			zeolite and related framework			
							ciliate		8313-	60
		0	2		2		2 2 2 Bativity, Determination of ende	1	0.0.4.0	~~
		9	. 9	2.14			Dizie Autivity, Decentination of end			
							unumber ties sustaines for criticherto			
							lite, neurandite, albite, and		0 0 1 0-	61
					1		anaicime	×	0.3.1.3-	01
		8	. 3	- 4	. 3	. 3	3.2.3 Activity: Solid solution descrip=			
							tions of clinoptilolite-heulandite			~~
	Ú.	H.		1	1.		and analcime	× ( )	8.3.1.3-	03
	8	.3	-1	. 3	.3	.3	3 Study: Conceptual model of mineral			
							evolution	•	8.3.1.3-	64
	8	.3	.1	. 3		3.4	4 Application of results	¢	8.3.1.3-	65
8	.3	.1	.3	. 4		In	nvestigation: Studies to provide the			
						in	nformation required on radionuclide			
						re	etardation by sorption processes along			
						fl	low paths to the accessible environment	. 0	8.3.1.3-	65
	8	.3	.1	.3	.4	1.1	1 Study: Batch sorption studies		8.3.1.3-	68
		8	.3	11		3.4	4.1.1 Activity: Batch scrption measure-			
							ments as a function of solid phase			
							composition	11	8.3.1.3-	69
		8	.3	.1	1	3.4	4.1.2 Activity: Sorption as a function of			
		1	17	1			sorbing element concentrations			
							(isotherms)		8.9.1 3-	74
		R	2	1	1	2 4	4 1 3 Activity: Sorntion as a function of	1		1.4
		Ý	14	1.4			around-water composition		8313-	76
		0	2				A 1 A Notivitus Contion on particulator	10	0.0.7.0-	10
		0	. 0	* *		2.6.9	4.1.4 ACTIVITY: SOLDTION ON PARTICULATES		0 0 1 0-	~~~
			-				and colloids	ð	0.2.1.2-	11
		8	. 3			214	4.1.5 Activity: Statistical analysis of			-
							sorption data	۰.	8.3.1.3-	19
	8	.3	.1	. 3	. 4	4.2	2 Study: Biological sorption and			
							transport	4	8.3.1.3-	80
	8	.3	.1	.3	. 4	4.3	3 Study: Development of sorption models .		8.3.1.3-	83
	8	. 3	. 1	. 7	1.1	4.4	4 Application of results		8.3.1.3-	84
TABLE OF CONTENTS (continued)

Pade

## 8,3,1,3,5 Investigation: Studies to provide the information required on radionuclide retardation by precipitation processes along flow paths to the accessible environment . . . 8.3.1.3- 85 8.3.1.3.5.1 Study: Dissolved species concentration 8.3.1.3.5.1.1 Activity: Solubility measurements . . 8.3.1.3= 88 8.3.1.3.5.1.2 Activity: Speciation measurements . . 8.3.1.3-91 8.3.1.3.5.1.3 Activity: Solubility modeling . . . 8.3.1.3-92 8.3.1.3.5.2 Study: Colloid behavior . . . . . . . . 8.3.1.3-93 8.3.1.3.5.2.1 Activity: Colloid formation characterization and stability . . . . . 8.3.1.3-93 8.3.1.3.5.2.2 Activity: Colloid modeling . . . . . 8.3.1.3-94 8.3.1.3.5.3 Application of results . . . . . . . . . . . 8.3.1.3-94 8.3.1.3.6 Investigation: Studies to provide the information required on radionuclide retardation by dispersive, diffusive, and advective transport processes along flow paths to the accessible environment . . . . . 8.3.1.3- 95 8.3.1.3.6.1 Study: Dynamic transport column 8.3.1.3.6.1.1 Activity: Crushed tuff column 8.3.1.3.6.1.2 Activity: Mass transfer kinetics . . . 8.3.1.3-105 8.3.1.3.6.1.3 Activity: Unsaturated tuff columns . . . . . . . . . . . . . . . . 8.3.1.3-107 8.3.1.3.6.1.4 Activity: Fractured tuff column studies . . . . . . . . . . . . . . 8.3.1.3-109 8.3.1.3.6.1.5 Activity: Filtration . . . . . . . . 8.3.1.3-110 8.3.1.3.6.2 Study: Diffusion . . . . . . . . . . . . . . 8.3.1.3-112 8.3.1.3.6.2.1 Activity: Uptake of radionuclidas on rock beakers in a saturated system . . . . . . . . . . . . . . . 8.3.1.3-112 8.3.1.3.6.2.2 Activity: Diffusion through a saturated tuff slab . . . . . . . . . . 8.3.1.3-113 8.3.1.3.6.2.3 Activity: Diffusion in an unsatu-8.3.1.3.7 Investigation: Studies to provide the information required on radionuclide retardation by all processes along flow paths to the accessible environment . . . . . . . . . . 8.3.1.3-116 8.3.1.3.7.1 Study: Retardation sensitivity analysis . . . . . . . . . . . . . . . 8.3.1.3-118 8.3.1.3.7.1.1 Activity: Analysis of physical/ chemical processes affecting transport . . . . . . . . . . . . . . 8.3.1.3-119

 $\langle \rangle$ 

the state of the second

.

TABLE OF CONTENTS (continued)

Page

-

A. C. C. C.

	8	3.	1.	3	. 7	.1	.2	I	Act mot	tiv del	rit L o	y: f	Yu	Ge	a	Mo	mi un	cal tai	./g	and	hys in	sid	al -					ł				
	8.	3.	1.	3	.7	.1	.3		ca. Act	lou	11a	iti(	on	STI	ar	ISD	or	tr	nod	els	ai	nd		•	•	•	8.	3.	1.	3.	-1:	23
R	ą	1	2	7	2		S+	1	re.	lat	ed	1 51	up	po	ort	00		÷ ;		110	ah			,	ł.	•	8.	3.	1.	3.	-13	25
Ŭ		* '	ς,				of	1	abo	ora	ato	ry	00	iat	a	to	I	epo	)si	tor	y i	.IS	ans	-			•					
0	2	1	2	7	2		po	rt nl	10	810	oul	at.	10	ns	5 011	.1+	. '	1	۰.	• •			*	٠.	۰.	1	0.0	3.	+ •	30	- 1	20
0 2	1	4.2	0 0		- 9 7 6	17.0	AP.	pt.	100	acu	1.01	0.	÷	Le	100	d = 5 5 = 4	5	****		de '	+ 1.		•	*	1	•	0.	0.	+ '	2.	· + ·	22
0.0	1 4 1	~ .	9		T (1)	an	DL	191		nfr	1.1	ten	10	104.4	01		at	PL	dat.	ior	0.11	e f										
					ca	Se	ou	s	rai	die	onu	ic1	id	ies	8 2	10	na	f	low	Da	th	s †	0									
					th	e	ac	ce	SS	ib]	le	en	vi	re	onr	nen	t										8.	3.	1	.3	-1	34
8.3	1.	3.	8	.1		St	ud	v:		Gas	sec	sus	I	ac	iid	onu	101	ide	e t	rar	SD	ort		÷.,					-	*	-	
						ca	10	ul	at	ior	ns	an	d	me	eas	sur	en	ent	LS .						2		8.	3.	1	.3	-1	35
	8.	3.	1	. 3	. 8	.1	.1		Ac	ti	vit	v:		Pł	nvs	sic	al	t	can	SDC	rt											
									me	cha	anj	sm	19	ar	nd	ra	te	S=-	-re	tar	da	tic	on									
									me	cha	anj	sm	S	ar	nd	tr	an	SD	ort	Wi	th											
									re	ta	rda	ati	or	1,													8.	3.	1	.3	-1	36
	8.	3.	1	.3	, 8	.1	. 2		Ac	tit	vit	iy:		Ge	as	tz	an	sp	ort													
									me	asi	ure	eme	nt	:5								1	1				8.	3.	1	.3	+1	37
8		1	3	. 8	.2	ŧ.	Ap	pl	ic	at:	ior	0 1	f	re	es	ult	S	1.	51		ιù.	14	14				8.	3.	1	. 3	-1	38
8.3	.1	3	9		Sc	he	du	le	f	or	tì	ne	ge	906	ch	emi	.st	ry	pr	ogi	:am				÷.		8,	3.	1	.3	-1	38
.3.1	.4	1	QV.	er	V.	.ew	0	f	th	e :	roc	ck	C)	na	ra	ote	ri	.st	ics	pı	tog	rai	m :									
		3	9	sc	<b>I</b> 1	.pt	10	n	of	t	he	pr	es	sei	nt	ar	nd	ex	pec	tec	i r	oc	ĸ									
			ch	ar	ac	te	eri	st	10	S	rea	qui	re	ed	þ	YF	)ei	fo	rma	nce	a	nd										
0 0		1	10	51	gi	] ]	153	nie	5	1.	8			6	1	•			1	•					•		8.	3.	1	. 4	-	1
0.0		**	• •		44	176	est	19	at	10	n:	2	9	ve.	10	pme	ant	0	s I	in :	Int	eg	rat	cec								
					-			ing	P	10	gri	am	81	na	1	nce	egi	at	101	1 0:	r g	eo	-						d,	۰,		~ .
9	2	1	4		21	17.1	240	+	. G	+ + + +	4 V.	101		5					1.	1.				*	1	1	0.	3	1	. 4	1	24
	1.01			3.4		٠.	de		1 -	ng ng	'n	200	1.0.1	aw.	ψŅ	mer	16	01	ai	1 11	ace	gr	ace	ea			0	0	4			24
8	3	1	. 4	1		2	Ac		vi	tu	. 1	Tr	1+1	am	-	÷ .,		of.	inc	inn	1 1	1.	51	1	•	1.1	0	2	-	. 4	Ξ.	24
		1					ac		vi	+ 1	-		4.64.1	99		( she she	w 6.5		90	i you	uya		au				8	0	4	1		26
8.3	.1	. 4	. 2		II	nv:	est	ic	iat	io	n:	ċ	-	01	oa	ic	÷.	ram	AW	rk	of	-	he	1	1	1	•		1	. 7		20
					Y	uci	ca	Mo	bur	ita	in	Si	Lt	e													8	~	1	4		28
8	.3	.1	. 4	.2	2 . :	1	St	cue	iv:		Ch	ara	ac	te	ri	za	ti	on	of	th	e v	er	ti	cal	1		÷					
							ar	nd	18	ite	ra	1 0	11	st	ri	bu	ti	on	of	st	rat	i-	27									
							g	rat	phi	.0	un	its	S	wi	th	in	t	he	sit	te l	are	a		12		1.1	8	.3	.1	. 6	-	32
	8	. 3	.1		1.	2.	1.:	1	A	sti	vi	ty	:	S	ur	fa	ce	ar	d	sub	sur	fa	ce	11			12					
									st	ira	iti	gra	20	hi	c	st	ud	ies	0	ft	he	ho	st									
									ro	bok	t a	ind	S	uz	rc	un	di	ng	un	its							8	.3	.1	. 6	-	32
	8	. 3	.1		4.	2.	1,1	2	A	sti	.vi	ty	;	S	sur	fa	¢9	-ba	se	d g	eor	bhy	si	ca	1							
									SI	lrv	rey	S	•												.,		8	.3	. 1	4	- 1	41
	8	. 3	. 1		4.	2.	1.	3	A	cti	LVi	.ty	:	111	301	ceh	01	eç	jeo	phy	sid	cal										
									S	arv	rey	S		4		4								+		14	8	. 3	.1	4		57
	8		• •		4.	2.	4.4	4	A	cti	LV1	ty	1	114	et	IIO	ph	ys:	LCa	1 p	rop	per	sti	es								
									Ce	est	111	g	1														8	.3	. 1		-	8.0

4

-

TABLE OF CONTENTS (continued)

÷3.

Page

		8.	3	. 1	. 1	4.	2.	1.	5	A	ct.	iv	1++		N	Aai	m	04	10	0	ror	Ar	+ 1	00	a	nd									
				1		1	-				tr	at	ia	ran	hi	ic	0	or	ra	la	-ic	ne	~ ~							8	3	1	4		60
		8	2	1	1	4	2	2	6	Ť	nt	an	+ 91	tin	5	~~	f	101	201	hu	eid	al		1		1	1				~ ·	*			~~
		~ .					** *	* *	~		114	14	4 4 I	ine		~	•	90	P.P.		0+1	× SA ak								0	2	1		_	6.4
	p i	2	1	1		2	5	0	÷	Au		4 V	the.	100	2 4 4 7	-		1	1			+ 1	-	-			•	۰.	<b>*</b> 1	0,	2.	* '	4		04
	e .	2.	1 4	17	1.4	6 1	4	0	C U	ay	۰.,		na.	Lac	100	EL.	12	ac.	10	n ···	OL.	61	le	SC	ru	C-									
							~	~ 5	ur	ar	L	ea	cu.	res	5 ¥	N -	cn.	111	C.	ne	53	Le	a	re	a	1	1		•	0.	3.	+ 1	4	-	00
		0	. 3	1.4	• • •	8.1	4.	6.	+	A	.CE	14	10	Υ÷.,	. 4	se	01	og.	10	m	app	DIN	ığ_	OI	Z	on	al								
							~			1	ea	çu	re	S 1	.n	τ.	ne	P	a1.	nt.	pru	ISI	1 1	ur	Ι	۰.	4	•	•	8.	3.	1.	4	**	66
		8.	. 3			4.	2.	2.	2	A	ct	17	103	Y:		su.	rf	ac	9.	II	act	ur	e	ne	tw	or	ĸ								
										S	tu	di	es		8		. 1	. 1						٠.		4		•	•	8.	3.	1.	. 4	-	68
		8	. 3			4.	2.	2,	3	A	ct	iv	it	Y:	1	BQ.	re	ho.	le	e	SV.	Lua	iti	on	0	f									
										f	au	lt	S	anc	1 1	fr	ac	tu	re	S					. 1		۰.		а.	8.	3.	1	. 4	-	70
		8.	. 3	. 7		4.	2.	2.	4	A	ct	iv	it	Y:	4	Ge	01	og	ic	m	ap	pir	g	of	t	he									
										e	xp	10	ra	tor	y	S	ha	ft	a	nd	d	rif	its					e i	i.	8.	3.	1	. 4	-	74
		8	. 3	. 3		4.	2.	2.	5	A	ct	iv	it	y:	1	Sé	is	mi	c	to	mod	Jra	iph	v/											
										V	er	ti	ca	1 5	se:	is	mi	C	pr	of	11	Lac	1			1			11	8.	3.	1	. 4	-	79
	8.	3	.1	. 4	١.,	2.	3	S	stu	idy	:	T	hr	ee-	·d:	im	en	si	on	al	a	eoi	00	ric											
								n	nod	lel		1									1							ŗ.		8.	3.	1	4		81
		8	. 3			4.	2.	3.	1	P	ot	iv	it	v:	1	De	ve	10	pm	en	t d	f	a	th	re	è									
										c	lim	en	si	ona	11	a	80	10	ai	C	mod	101		f	th	0									
										9	it	0	ar	A.8					2-							· ·				0	2	1	4	-	91
	8.	3	.1	1.	4.	2.	4	2	hor	11	Ca	ti	on	01	e .	re	511	1+			1	÷.		1		1	٩.		•	8	2	1	4	-	80
8.	3.	1	4		3	1	nt	785		03	+ 1	00		De	217	01	00	ma	n+		÷ .			1	1	1	1	۰.		• •	~ '	*	1.13		0.6
							lin	ner	181	01	al	-	ind	010		04	- F.	00	6	ab		0 m+			4.4		2.1								
							+	4.4	10		me	01	+ 0	****		**	-	~~	~	A	a	201	.01			00				0	0	4			24
	8	2	1	Ϊ.,	4	ຊັ	1	6	2+1	1.4.		0.4	100	+ 3		**	0.0	-		1.	12					1.	1	1	•	0.	2	+	. 4	-	04
	~ .	~				~ .	*		one	141	44	-	120	her	114	44		a	44	40	-	101	1.0	57	51	Le.				0			1		0.7
		0			ŧ.	A	2	1	phe	1	h L d L de		su 	031	ur.	14	Ce	-	114	01	ma	5.20	on.	1	1	٠.	٠.	1	٨	8.	3	-	• 9	-	87
		0			**	n .	2	e de s	• •	1	300	11	110	Υ÷		SÅ	SC	em	at	10	a	r1.	173	ruĉ	Ŧ							U.			
	0	0									DIC	gi	am		1	. 1		1		1	1.1	κ.		1		۰.	*	÷,	1	8.	3	1	. 4	**	87
	0	- 2	* *	1	4.	3.	4		SEL	iaj	е.		inr	ee	-a	1.11	len	151	on	al	r	003	< <	ona	ara	10-					į.				
		~				2			cei	511	sta	.08	m	lodi	e1	S.				1.1						۰.		۰.	× 1	8,	3	1	. 4		00
		8		5.4	4 1	4	. 3	12	. 1	1	Act	:11	/1t	Ä:		De	Ne	10	pn	ler	t	of	in )	nre	e.	•									
										- 5	din	ner	151	on	al	n	100	iel	S	of	r	oci	K (	cha	ara	IC-	•								
											ter	11	sti	CS.	a	t	th	1e	re	pc	si	to:	ry	Si	ite	3		1		8	3	.1	. 4	-1	00
	8	. 3			4.	3	. 3	1	Ap	p1:	ica	ati	Lon	0	f	re	151	ilt	5		.,	14			4	1				8	3	.1	. 4	-1	02
8.	3	. 1	. 4	ł .	4	0.4	Sel	had	du.	le	fo	r	th	le	ro	ck	5	cha	ra	ict	er	is	ti	CS											
						1	or	og	rai	m																				8	3	.1	.1	1-1	.03

N ...

132 5

### TABLE OF CONTENTS

# VOLUME V

937 23

.Q.

	3.	1.	5	0	ve	rv	ie	W	of	the	c.	Lima	ate	P	ro	gra	am :	I	es	orig	tio.	n o	f							
				*	46	ur	6	6.5	a de ille	ice	COI	1017 I		ns	1	rec	ant	rec	1 0	t cr	ie									
1	R .	3	1	P	er 1	10	Im	ar	1Ce	and	on	9510	gn Stu	15	SU	es to	, n		ri di	· · ·		×. +			8	3.1	1.,	5=		1
					۰.	4	n.f	0.	mat	inn		ami	1 70	A	20	-	. F	* • ·	20	4	+ 00	~								
						-	ha	~	ALLIGA L			ant.	110		on d	1 4 1	ion	10	am		lict	01								
						5	ind.	112	10 1	188 99 19 1 m	i de de l	lici La	10	00	na	inter las a	LOU	5 1	10	brac	1100								0	-
		0				-	uc.		6 6	1111	ALC:	85			11	1	1	1	1			• •			0	2 .	* • •	27	0	2
		0,	0.	4.4	Q ,	4.4	*		scuc	ıy:		lara	act	er	ΊZ	at:	rou	03	: m	paer	m							1		
						2		. 3	regi	ona	11 (	2771	mat	e				1.1	1.1		1.1	1.1	1		8.,	3.	1.	5*	4	0
			ΰ.	3.	1.	5.	7.	4.	1	Act	:17:	rth	۱.,	Sy	no	pt:	LC	cha	ara	ctei	:1za	t10	n							
						5	1	١.		OI	rea	9101	nal	0	111	mat	te	1.13		1.1		÷	14		8	3.	1.	5-	4	0
		8.	3.	1.	5.	1.	2		stuc	iy:	Pi	aled	001	.im	at	ę :	stu	dy	t. (1)	lake	€,									
								Ĩ	play	7a,	ma	rsh	de	po	si	ts		κ.					4		8	3.	1.	5-	4	2
			8.	3.	1.	5.	1.	2	.1	Act	iv:	icy	٢.,	Pa	le	ont	tol	og:	i¢	ana]	lyse	s.	1.		8	3.	1.	5-	4	2
			8,	3.	1.	5,	1.	2	.2	Act	iv.	ity	1	An	al	ys:	1.8	of	th	e st	irat	i-								
										gra	aph	y=si	edi	me	nt	01	ogy	0	f m	arsi	1, 1	acu	15-							
										tri	ine	, a:	nd	pl	.ay	a	dep	OS:	its						8.	3.	1.	5-	4	7
			8.	3.	1.	5.	1.	2	.3	Act	:iv	ity	1	Ge	100	he	mic	al	an	aly	ses	of								
										lak	ce,	ma	rsh	1	an	id i	pla	ya	de	pos:	its				8.	3.	1.	5-	4	9
			8.	3.	1.	5.	1.	2	. 4	Act	ziv	ity	:	Ch	irc	no	100	ic	an	alva	888	of								
										la)	ke,	pl	ave	1,	an	id i	mar	sh	de	pos:	its	. A.			8.	3.	1.	5-	5	1
		8.	3.	1.	5.	1.	3		Stu	dv:	C	lim	ati	ia	in	lar	ica	ti	ons	of	ter	-						1	1	
									res	tria	a1	Dal	eoe	200	510	av									8	3	1	F	5	4
			8.	3.	1.	5	1.	3	.1	Act	tiv	itv	-	Ar	al	VS	is	of	Da	ck .	rat				~ .	w 1		~	×	
										mid	dda	ns							in m						R	2	•	s	5	Δ
			8.	3.	1	5	1	. 3	.2	Act	tiv	itu		Ar	าลไ	VR	ie	of		110					۰.	~ •		~	Ĭ	4
										Sat	mpl	69							2.4	***					2	2	4	s	5	6
			8.	3.	1	5	1	3	3	Act	tiv	itu	11	De		חדיד כ	ins	+ 4	'n	of	· ·	1			۷.	~ '	**	~	~	0
							15			VA	Tat	ari	nn.		1 1 1	ng+	A 1	-01		000	hine				0	5	4	c		-
		8	3.	1	5	1	4		Stu	du	2 L	nal	VE		0.4	F +	he	50	100	ONT	inne		•		0.1	2.1	* *	* 7	*	1
				-					man	tal	hi	et m	3 10.		£ ;	n ha	V.	1.0	n N	COLLY.	****									
									TOT	inn	****	000	+7	~ .			**	400	e r	io un	u a d d				0	5				
			2	2	4		14	٨	1 1	T.m				M	1		-		1	11			• •		0.	8.	÷ •	5-		3
			~		e an			17	1.00	+ 1	N P P	4 to Y	* *	1.10	v		M	UL.	50	*** ** **	PLOF	Set.				0			1	
			0	2	4			Å		to at	20	ale la la constante da la const La constante da la constante da	6.21	20.	1 141	e i e	TAIC	2 UII	Cal	1 1	edro	n	* 4		0.	3.	+ •	2-	5	14
					-		* *	1	. 6	22.0	in de V	TCA		0	LL.	110	14.		epc	SIL	S						1		1.	
			0	2						Illei 7 m	567	ng	10	-	ne	44	000	a M	our	ital	n ai	rea			8.	3.	д,	2-		0.0
			0	. 2		· 9	ų fei	1.8	. 2	AC	C2V	- 54	()	E.	01.	lan	i n.	lst	ory	o i	the	9					17	96		
									~	IU	500	MC	pun	ca.	ın	re	iår(	on	• •	. *	×		* *		8,	3.	1.	5-		12
		0	.0	• +	. 9	14	. 5		scu	dy:	E	ale	300	11	ma	ce*	pa.	lec	ent	1110	nmei	nta	1							
							4		syn	the	515	1	1	•	•	× 4		۰.	8.3		1.1		x 4		8,	3	1.	5-	1	16
			8	.3	. 1	. 5	. 1	. 5	1.1	Ac	tis	rity	11	D.	al	600	lin	nat	e-											
										pa	lec	penv	71r	on	me	nte	11	syr	ithe	esis					8.	3	1.	.5-	• 1	77
		8	.3	.1	. 5	.1	. 6		Stu	idy:	(	lhar	cac	te	ri	zat	:10	n c	f t	the	fut	ure								
									reg	rion	al	cli	ima	te	a	nd	en	vir	on	nent	s.	×.		1.5	8.	3	1	.5-	• •	78
			8	. 3	.1	. 50	.1	. 8	5.1	AC	tit	rity	1:	G	10	bal	C	lin	nate	e mo	del.	ing			8	3	. 1	. 5.	• 1	79
			8	.3	.1	.5	.1	. 6	5.2	Ac	stin	vity	7:	R	eg	ior	nal	0]	ima	ate										
										mo	de	Line	7												2	2	1	Е.		22

# TABLE OF CONTENTS (continued)

Page

			8.	3	.1.	5.	1	. 6	. 3	Activity: Linked global-regional						
										climate modeling	8.	3.	1.	.5-	• 8	34
			8.	3	.1.	5.	1	. 6	. 4	Activity: Empirical climate						
										modeling	8.	3.	1.	5-	• 1	16
		8	3	1	5	1	7		App	lication of results	8.	3.	1.	5-	. 8	38
		2	1	ŝ	2	-	-	170	at i	astion: Studies to provide the						1
0		(a) 1	* •	-	. *			Fo	2011	tion required on notential effects						
								10	Lina	Lion required on pocencial errocco						
						5	IC	2	ucu.	re climatic conditions on hydrologic	0	2		Ε.		0.0
		1		<u>.</u>	1	. (	ona	ar	act	eristics	۰,	2.	+ .		1.9	22
		8.	3.	1	. 5.	2	1		Stu	dy: Characterization of the	~	-				
									Qua	cernary regional hydrology	в.	3.	+ .	5-	•	13
			8.	3	.1.	5	.2	.1	.1	Activity: Regional paleoflood	1.5			5	1	
										evaluation	8.	3.	1.	.5-	• 5	14
			8.	3	.1	5	.2	.1	.2	Activity: Quaternary unsaturated						
										zone hydrochemical analysis	8.	3.	1,	.5-	• 5	96
			8.	3	.1	5	.2	.1	.3	Activity: Evaluation of past dis-						
								17		charge areas	8.	3.	1.	.5-	. 5	86
			R	3	1	5	2	1	.4	Activity: Analog recharge studies	8.	3.	1	5.	-10	6
			8	2	1	-	2	1	3	Activity: Studies of calcite and						
			0.	-	• •	~		• •		analine eilige wein dengeite	8	2	1	E	.11	0
		0	0			2	0		e	du. Characterization of the future	۰.	~.			**	
		۰.	0.	*	. 9	4	. 4		acu	ay: Gharacterization of the future						
									reg	ional hydrology due to climate						0
								~	cna.	nges	0.	2.	+ .	. 9 -		÷0
			З.	3	. 1	. 5	. 4	. 4	. 7	Activity: Analysis of future sur-						
										face hydrology due to climate						
										changes	8.	3.	1.	.5-	-10	-8
			8.	3	.1	. 5	. 2	.2	.2	Activity: Analysis of future						
										unsaturated zone hydrology due to						
										climate changes	8.	3,	1	. 5-	-13	19
			3.	3	.1	. 5	. 2	.2	.3	Activity: Synthesis of effects of						
										possible future recharge due to						
										climate changes on hydrologic char-						
										acteristics of the Yucca Mountain						
										saturated zone	8	2	1	s.,	-13	51
		R	3	1	5	2	2		ann	lication of regults	R.	2	1	E .	-1	22
	2	2	1	-		• •		-	Aul	a far the alimete program	0	20	1	5.	-1	53
	2.	2.	2				50	ne	aus	e for the crimate program	0.	9.			-++	20
	2.	+ 1	0		UV.	51	V1	ew	OI	the erosion program: Description of						
					cne	9	ru	cu	re	erosional rates required by the	~					
Υ.			12		pe:	rI	or	ma	nce	and design issue	8.	3	. +	. 0.	•	7
\$	3.	3.	1.	. 6	.1		In	ve	sti	gation: Studies to determine present						
							10	ca	tio	ns and rates of surface erosion	8.	3	.1	. 6.	•	6
		8	. 3	.1	. 6	.1	.1		Stu	dy: Distribution and characteristics						
									of	present and past erosion	8.	3	.1	. 6.	- 1	10
		8	3.:	3.	1.	6.	1.	1.	1	Activity: Development of a geo-						
										morphic map of Yucca Mountain	8.	3	.1	. 6.	-	10
			8	. 3	.1	. 6	.1	.1	. 2	Activity: Analysis of the down-						
										cutting history of Fortymile Wash						
										and its tributaries	8	2	1	. 6.		11

TABLE OF CONTENTS (continued)

	<i>.</i>			
- TE		-	-	100
		en 1	0.001	
		-	~	-
			- 10	

			8	.3.3	1.6	.1.1	. 3	Ac. ero	ivit; sion	y: at	An Yud	an	aly Mo	sis	of ain	hil	1510	pe		8.	3.	1.	6-	13
		8	3	1.6	6.1	2	Appl	lica	tion	of	Tes	aul.	ts			1.1		1.1		8.	3.	1.	6-	15
	0	5	1	6 1		T NATA	ma i /	n an ar an	00.	Lini	- Ant		1 0	++0	At D	n.F.	£ 194 1	170						
	0	1.4	1.16		6	TUNE		gara	Qui i	1. 1.	and	1 44	* •	***		1 m m		44.0						
						crru	19270	0 00	nait.	1011	5	3	oca	C10	ns e	and	race	:5						
						of e	rosi	ion	1.1		e ie		x		8 K.	$\mathbf{x} = \mathbf{x}$	19.10			8.	3.	1.	0-	15
		8	.3	.1.1	5.2	.1	Stud	dy:	Inf.	luei	104	of	fu	tur	e cl	lima	tic							
							cond	iiti	ons	on 1	Loca	ati	ons	an	d ra	ates	of							
							ATO	aion												8.	3.	1.	6-	18
			0	5 1	6	0.1	1	he'-	1 ++ 1 + 1		Carr				nd r	int n					1	1		
			0	19.19	1.10	1.6.1.4		AUL	de Vale.	11	ey:	46-44	004	5 6.	E E.	1910								
								eva	That:	101	OI.	211	pac	P Q	1 11	acur	e							
								C11	mati	c cc	ondi	127	ons	on	100	cati	ons			11			1.1	
								and	rate	es d	of €	ero.	510	n				1		8.	3.	1.	6=	18
		-8	.3	.1.6	6.2	.2	Appl	lica	tion	of	res	sult	ts		÷ .:			1.		8.	3.	1.	6-	19
	8	.3	.1	. 6.3	3	Inve	stic	rati	on:	Stu	idie	2.5	to	pro	vide	th th	8							
		1	17			info	mat	ion	ram	11 re	nd +	- 15	dat.	arm	ine	the	pot	en						
					-112	a	o F 4	Enne	+	£ \$			o At	ani	A 57	* 1 **		~~						
						in the state		Lect	5 01		- 41 5		ect	Calibra.	c al	e be she V	tcl	on		~			0	
						Loca	10101	ns a	nd ra	ates	5 01	c e	ros	lon	1	$\mathbf{x}_{i} \in \mathbf{y}$	1.1			8.	3.	4.	0-	19
		8	.3	.1.6	5.3	.1	stuc	dy:	Eva.	luat	ior	2 0	ft	he	effe	ects	of							
							futi	lre	tecto	onic	ad	sti	vit	y o	n er	cosi	on a	t						
							Yuco	ca M	ounta	ain	1							÷.,		8.	3.	1.	6-	22
			8	.3.3	1.6	.3.1	. 1	Act	ivit	0.5	SVT	hth	esi	s a	nd e	lata								
								17.0	1	in	A.4	+ 7.	0 11	mna	m+ /	£ £		- 0						
								- 1	1.44		wa.			III a		6 m m 1	u u u u u u u u u u u u u u u u u u u							
								- Q -	420 1	OI 1	supa	51Q	enc	e a	na i	aut	cruč	10	n					
								610	sion	at	Inc.	oca	MO	unt	ain	and					1			
								VIC	11116;	2					6 K.	x 8	$  _{\mathcal{T}}$			8.	3.	1.	6-	22
		8	.3	.1.1	6.3	.2	App]	lica	tion	01	res	sul	ts		6. X	1.1		11	÷.,	8.	3.	1.	6-	23
	8	.3	.1	. 6.4	4	Inve	stic	rati	on:	Pot	ent	tia.	1 e	ffe	cts	of	eros	io	n					
					1126	on h	vdre	10g	ic.	aeod	han	nic	al.	an	d re	ock								
						char	ante	arie	tice	2					0.01					8	2	1	6-	22
		0	2	18 12	e 1	4	Den	einen ser ser	Der.				14	1.		11				۰.	~ *	-		
		0	1.41	s de la l	81.2	1.4	0645	ayi	Devi	arol	Jiner	26	67	C	obte	-a+	repo	1 L C						
							CO 8	adar	855	cne	eri	réc	5.5	OI	eros	310n	on							
							the	hyd	rolo	gic,	, ge	90C	hem	ica	1, 3	and	rock	ç						
							chai	ract	eris	tic	s at	t Y	000	a M	ount	ain				8.	3.	1.	6-	27
		8	.3	.1.1	6.4	.2	i.ppl	lica	tion	of	res	sul	ts		1.1					8.	3.	1.	6-	27
	8	. 3	.1	. 6. 5	5	Sche	dule	e fo	r th	e e:	ros	ion	pr	our	am					8.	3.	1	6-	28
8	3	1	7	01	70 *	viau	1 of	PAA	k di	000	111+ 1	inn	nr	0.07	ami	De	ear	-	11	~ .	~ .	-	~	~~
~					1			1.00	n un	224.	4. 64 6. s	F (1.63	14	-94-	Casti e		0010	- 5						
				6.	1011	0r	TOCI	K al	2201		on i	red	nrt	ea	by i	ine					~	12		
1		1		P	sii	orma	ince	and	des	ıgn	153	sue	s.		1.1	+ +				8.	3.	1.	7-	1
8	. 3	11	47	.1	In	vest	igat	tion	: R	ate	s 01	fd	155	olu	tion	n of								
					Cr	ysta	allis	ne a	nd n	onci	ryst	tal	lin	e c	ompo	onen	ts							
					in	tuf	Ef .												1	8.	3.	1	7-	1
		8	. 3	.1 .	7.1	.1	App	lica	tion	04	TO	sul	te							8	3	1	7-	2
		0	2	· · ·	7 7	5	Soh	odul	0.00	d m	110	D = M	200		1					0	2 2	1	-7-	0
0	2	1			1 + 4	1. der	wwill	a la la la	e an	the still	-+	000	1105		* *.	1 1			×.,	Q.	4	-	10	4
0	. 2		.0	01	ver	ATEN	V OI	cne	pos	CCT(	osu:	r.e	cec	con	103	pro	gran	n:						
				De	esc	ript	:10U	10	Iutu	re	cect	con	10	pro	ces	ses	and							
				e	ven	ts I	requ:	ired	by	the	per	rfo	rma	nce	and	d de	sign	2						
				1	ssu	es														8.	3.	.1.	. 8-	1

TABLE OF CONTENTS (continued)

<pre>tion required on direct releases resulting from volcanic activity</pre>	8.	3.	1	. 8	.1		Ir	ive	stic	yatio	n: 3	Stu	dies	t	0 1	roi	rid:	e in	for	ma-						
from volcanic activity							ti	on	rec	nuire	d on	di	rect	r	ele	ase	s :	resu	lti	ng						
<pre>S.3.1.8.1.1 Study: Probability of a volcanic erup- tion penetrating the repository</pre>							fr	om	vol	cani	c act	tiv	ity									8	.3	1.	8-	48
<pre>tica penetrating the repository</pre>		8	.3	.1	. 8	.1	.1		Stud	iv:	Proba	abi	lity	0	fi	1 70	10	anic	er	up-						
8.3.1.8.1.1.1       Activity: Location and timing of volcanic events									tion	pen	etrat	tin	a th	ie	ret	losi	to	rv .		1.		8	.3	1.	8-	51
<ul> <li>volcanic events</li></ul>			8	3	1	. 8	.1	.1	.1	Acti	vity		Loca	ti	on	and	i t:	imin	a o	f						
<pre>8.3.1.8.1.1.2 Activity: Evaluation of the struc- tural controls of basaltic volcanic activity</pre>			Ť		1		1			volc	anic	eve	ents						1.1			8	.3	1.	8-	51
<pre>tural controls of basaltic volcanic activity</pre>			R	2	1	8	1	1	2	Acti	vitu	. 1	Eval	ua	tic	on d	ft	the	str	ue-						
activity8.3.1.8528.3.1.8.1.1.3Activity:Presence of magma bodies in the vicinity of the site8.3.1.88.3.1.8.1.1.3Activity:Probability calculations and assessment8.3.1.88.3.1.8.1.2Study:Effects of a volcanic eruption penetrating the repository8.3.1.88.3.1.8.1.2Activity:Effects of Strombolian eruptions8.3.1.88.3.1.8.1.2.1Activity:Effects of fydrovolcanic eruptions8.3.1.88.3.1.8.1.2.2Activity:Effects of strombolian eruptions8.3.1.88.3.1.8.2Investigation:Studies to provide informa- tion required on rupture of waste packages due to tectonic events8.3.1.86.3.1.8.2.1Study:Analysis of waste package rup- ture due to tectonic processes and events8.3.1.88.3.1.8.2.1.1Activity:Resement of waste package rupture due to igneous intrusion8.3.1.88.3.1.8.2.1.2Activity:Activity: Assessment of waste 			×			ľ				+1173	1 001	nt ri	ole	of	ba	sal	ti	c vo	lca	nic						
<pre>8.3.1.8.1.1.3 Activity: Presence of magma bodies in the vicinity of the site 8.3.1.8-55 8.3.1.8.1.4 Activity: Probability calculations and assessment 8.3.1.8-57 8.3.1.8.1.2 Study: Effects of a volcanic eruption penetrating the repository 8.3.1.8-59 8.3.1.8.1.2.1 Activity: Effects of Strombolian eruptions 8.3.1.8-59 8.3.1.8.1.2.2 Activity: Effects of hydrovolcanic eruptions 8.3.1.8-59 8.3.1.8.1.2.2 Activity: Effects of hydrovolcanic eruptions</pre>										anti	vitu		~~~	~ *								8	.3	1.	8-	52
<pre>in the vicinity of the site 8.3.1.8-55 8.3.1.8.1.1.4 Activity: Probability calculations and assessment 8.3.1.8-57 8.3.1.8.1.2 Study: Effects of a volcanic eruption penetrating the repository 8.3.1.8-59 8.3.1.8.1.2.1 Activity: Effects of Strombolian eruptions 8.3.1.8-60 8.3.1.8.2.1 Activity: Effects of hydrovolcanic eruptions</pre>			Q	2	1	0	4	1	2	ant i	v + + y	· ' ·	Drag	en.	-	of	mad	ma	hod	188	e de la					974
<pre>8.3.1.8.1.1.4 Activity: Probability calculations and assessment</pre>			0	. 0	1.4	. 0	1.4		1.0	in t	ha th	ini	nitu	0	f +	ha	eit	- 0	200			8	3	1.	8	55
<pre>and assessment</pre>			0	0		0	1		4	hat i	110 Y -		Drah	ah	414	+ + + +	04	1.0111	ati	000	100			-		
<pre>8.3.1.6.1.2 Study: Effects of a volcanic eruption penetrating the repository 8.3.1.8-59 8.3.1.8.1.2.1 Activity: Effects of Strombolian eruptions 8.3.1.8-59 8.3.1.8.1.2.2 Activity: Effects of hydrovolcanic eruptions 8.3.1.8-59 8.3.1.8.1.3 Application of results 8.3.1.8-60 8.3.1.8.1.3 Application of results</pre>			0	. 9	• *	. 0	14	1.1.1	· *	ACCI	ATCA		e t u u	er.	l'adar dan sa	Y	va.	4 4 44 4	aux	with a		0	2	1	ŝ	57
b.3.1.8.1.2Study: Effects of a voltamic endptionpenetrating the repository8.3.1.8eruptionseruptionseruptions8.3.1.8s.3.1.8.1.2.2Activity: Effects of Strombolianeruptions8.3.1.8eruptions8.3.1.8s.3.1.8.1.3Application of resultss.3.1.8.2Investigation: Studies to provide information required on rupture of waste packagesdue to tectonic events8.3.1.8due to tectonic processes andevents8.3.1.8events8.3.1.8events8.3.1.8events8.3.1.8s.3.1.8.2.1.1Activity: Calculation of the numberof waste packages intersected by afaultfaultfault8.3.1.8s.3.1.8.2.1.3Activity: Probability and rate offaulting8.3.1.8s.3.1.8.2.1.4Activity: Assessment of waste packagege rupture due to faulting8.3.1.8s.3.1.8.2.1.5Activity: Assessment of postclosureground motion in the subsurface8.3.1.8s.3.1.8.2.1.7Activity: Assessment of wastepackage rupture due to folding anddeformation1.1.8s.3.1.8.2.1.7Activity: Assessment of wastepackage rupture due to folding anddeformation1.1.8s.3.1.8.2.1.7Activity: Assessment of wastepackage rupture due to folding anddeformation1.1.8s.3.1.8.2.1.7Activity: Assessment of wastepackage rupture due to folding and <td< td=""><td></td><td></td><td></td><td></td><td>0</td><td></td><td></td><td></td><td>er</td><td>and</td><td>a5503</td><td>55114</td><td>enc</td><td>. 1</td><td></td><td></td><td></td><td>1 1</td><td>mit i</td><td></td><td></td><td></td><td></td><td></td><td>¥</td><td>¥ /</td></td<>					0				er	and	a5503	55114	enc	. 1				1 1	mit i						¥	¥ /
<pre>8.3.1.8.1.2.1 Activity: Effects of Strombolian eruptions</pre>		0	. 2	1.4	. 0	r ik	1.6		scuo	iy:	FILEC	LS	01	el o n	vo.	. Cai	140	eru	iput	ULL.		0	2	4	0_	5.0
<pre>s.3.1.8.1.2.1 Activity: Effects of Strombolian eruptions</pre>									pene	at at	ing (	ine	teb	03	1100	EY		a a amba	11.	1.1	1	0			0	22
<pre>eruptions</pre>			ø	13	• +	. 0		es le		ACCI	VICY		LIIE	Ct	5 0	or s	stic	ounc	++4	11		0	5		÷	5.0
<pre>8.3.1.8.1.2.2 Activity: Effects of hydrovolcanic eruptions</pre>			~	~				~	~	erup	Clons	5 .			12	3 3			11	1,1		0	. 9		0-	0.2
eruptions8.3.1.8.1.3Application of results8.3.1.4.8-608.3.1.8.1.3Application: Studies to provide information required on rupture of waste packages due to tectonic events8.3.1.8-628.3.1.8.2.1Study: Analysis of waste packages due to tectonic processes and events8.3.1.8-638.3.1.8.2.1Study: Analysis of waste package rup- ture due to tectonic processes and events8.3.1.8-668.3.1.8.2.1.1Activity: Assessment of waste package rupture due to igneous intrusion8.3.1.8-668.3.1.8.2.1.2Activity: Calculation of the number of waste packages intersected by a fault8.3.1.8-678.3.1.8.2.1.3Activity: Probability and rate of faulting8.3.1.8-688.3.1.8.2.1.4Activity: Assessment of waste pack- age rupture due to faulting8.3.1.8-698.3.1.8.2.1.5Activity: Nature, age, and rate of folding and deformation in the repository horizon8.3.1.8-718.3.1.8.2.1.7Activity: Assessment of waste package rupture due to folding and deformation8.3.1.8-728.3.1.8.2.2Application of results8.3.1.8-72			8	. 3		. ð	1.4	6	. 4	ACCI	ATCA	1	LIIE	Ct	SC	or t	iyai	COVO	1ca	nic				4	0	=
<pre>8.3.1.8.1.3 Application of results</pre>						5				erup	cions.	5 .	1.1.1			1.1	1.1	1.2		×	*	0		4	0-	00
<pre>8.3.1.8.2 Investigation: Studies to provide informa- tion required on rupture of waste packages due to tectonic events</pre>		S	.3		. 0	e à	13		Appi	licat	lon	DI	resu	ΠĽ	8	1.1	1.	+ . +	1	× . *		0	. 2	-	0-	04
<pre>tion required on rupture of waste packages due to tectonic events</pre>	8.	3	. 4	. 8	. 4		11	ive	stiç	jat 10	n: :	scu	ales	E	01	prov	1106	e in	lior	ma-						
due to tectonic events.8.3.1.8.1.8.1.8.638.3.1.8.2.1Study: Analysis of waste package rup- ture due to tectonic processes and events8.3.1.8-638.3.1.8.2.1.1Activity: Assessment of waste package rupture due to igneous intrusion8.3.1.8-668.3.1.8.2.1.2Activity: Calculation of the number of waste packages intersected by a fault8.3.1.8-668.3.1.8.2.1.3Activity: Probability and rate of faulting8.3.1.8-678.3.1.8.2.1.4Activity: Assessment of waste pack- age rupture due to faulting8.3.1.8-688.3.1.8.2.1.5Activity: Assessment of postclosure ground motion in the subsurface8.3.1.8-708.3.1.8.2.1.6Activity: Nature, age, and rate of folding and deformation in the repository horizon8.3.1.8-718.3.1.8.2.1.7Activity: Assessment of waste package rupture due to folding and deformation8.3.1.8-728.3.1.8.2.2Application of results8.3.1.8-72							51	.on	req	luire	a on	ruj	ptur	.6	OI	was	ste	pac	kag	es				44		
<pre>8.3.1.8.2.1 Study: Analysis of waste package rup- ture due to tectonic processes and events</pre>				1			dı	10	to t	ecto	nic e	evei	nts.	. *	1				1	×., ×	1	8	.3		8-	63
<pre>ture due to tectonic processes and events</pre>		8	. 3	- 1	. 8	. 2			Stud	iyi	Analy	ys1	s oi	W	ast	er	aci	kage	ru	P-						
events8.3.1.8.1.18.3.1.8.608.3.1.8.2.1.1Activity: Assessment of waste package rupture due to igneous intrusion8.3.1.8.608.3.1.8.2.1.2Activity: Calculation of the number of waste packages intersected by a fault8.3.1.8.678.3.1.8.2.1.3Activity: Probability and rate of faulting8.3.1.8.678.3.1.8.2.1.4Activity: Assessment of waste pack- age rupture due to faulting8.3.1.8.688.3.1.8.2.1.5Activity: Assessment of postclosure ground motion in the subsurface8.3.1.8-698.3.1.8.2.1.6Activity: Nature, age, and rate of folding and deformation in the repository horizon8.3.1.8-718.3.1.8.2.1.7Activity: Assessment of waste 									ture	due	tot	ceci	toni	Ç,	pro	oces	se	s an	d							
<pre>8.3.1.8.2.1.1 Activity: Assessment of waste package rupture due to igneous intrusion</pre>					i)				ever	its		4 . A	1. 1				11	5 ×	× -	4		8	.3	1.	8=	66
package rupture due to igneous intrusion			8	.3	.1	. 8	12	1	.1	Acti	vity	: 1	Asse	SS	mer	nt d	)Í I	wast	9							
<pre>intrusion</pre>										pack	age :	rup	ture	d	iue	to	igi	neou	15				1.1			
<pre>8.3.1.8.2.1.2 Activity: Calculation of the number of waste packages intersected by a fault</pre>			1			1				intr	usio	n .										8	.3	.1.	8=	66
of waste packages intersected by a fault			8	. 3	.1	. 0	18	2.1	.2	Acti	ATTA	:	Calc	ul	at:	lon	Of	the	nu	mbe	r					
fault8.3.1.8.2.1.38.3.1.8.2.1.38.3.1.8.2.1.38.3.1.8.2.1.48.3.1.8.2.1.48.3.1.8.2.1.48.3.1.8.2.1.48.3.1.8.2.1.58.3.1.8.2.1.58.3.1.8.2.1.58.3.1.8.2.1.58.3.1.8.2.1.58.3.1.8.2.1.58.3.1.8.2.1.68.3.1.8.2.1.68.3.1.8.2.1.68.3.1.8.2.1.68.3.1.8.2.1.68.3.1.8.2.1.68.3.1.8.2.1.68.3.1.8.2.1.68.3.1.8.2.1.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.78.3.1.8.2.										of w	aste	pa	ckag	jes	a in	nter	se	cted	i by	a						
<pre>8.3.1.8.2.1.3 Activity: Probability and rate of faulting</pre>										faul	t,	÷				11.5	÷	1	- x :		10	8	.3	. 1 .	8-	67
faulting			8	. 3	.1	, 8	.2	2.1	.3	Acti	vity	: :	Prob	ab	11:	Lty	and	d ra	ite	of						
<pre>8.3.1.8.2.1.4 Activity: Assessment of waste pack- age rupture due to faulting 8.3.1.8-69 8.3.1.8.2.1.5 Activity: Assessment of postclosure ground motion in the subsurface 8.3.1.8-70 8.3.1.8.2.1.6 Activity: Nature, age, and rate of folding and deformation in the repository horizon 8.3.1.8-71 8.3.1.8.2.1.7 Activity: Assessment of waste package rupture due to folding and deformation 8.3.1.8-72 8.3.1.8.2.2 Application of results 8.3.1.8-72</pre>										faul	ting					4		$\mathbf{x} \in \mathbf{x}$	18			8	.3	.1.	8-	68
age rupture due to faulting 8.3.1.8-69 8.3.1.8.2.1.5 Activity: Assessment of postclosure ground motion in the subsurface 8.3.1.8-70 8.3.1.8.2.1.6 Activity: Nature, age, and rate of folding and deformation in the repository horizon 8.3.1.8-71 8.3.1.8.2.1.7 Activity: Assessment of waste package rupture due to folding and deformation 8.3.1.8-72 8.3.1.8.2.2 Application of results 8.3.1.8-72			8	. 3	.1	. 8	.2	2.1	.4	Acti	vity.	: .	Asse	SS	smei	it d	of 1	wast	e p	ack						
<pre>8.3.1.8.2.1.5 Activity: Assessment of postclosure ground motion in the subsurface 8.3.1.8-70 8.3.1.8.2.1.6 Activity: Nature, age, and rate of folding and deformation in the repository horizon 8.3.1.8-71 8.3.1.8.2.1.7 Activity: Assessment of waste package rupture due to folding and deformation 8.3.1.8-72 8.3.1.8.2.2 Application of results 8.3.1.8-72</pre>										age	rupti	ure	due	a t	0	faul	Lti	ng i		• •		8	.3	.1.	8-	69
ground motion in the subsurface 8.3.1.8-70 8.3.1.8.2.1.6 Activity: Nature, age, and rate of folding and deformation in the repository horizon 8.3.1.8-71 8.3.1.8.2.1.7 Activity: Assessment of waste package rupture due to folding and deformation 8.3.1.8-72 8.3.1.8.2.2 Application of results 8.3.1.8-72			8	. 3	.1	. 8	.2	2.1	.5	Acti	vity	1	Asse	255	smei	nt c	of	post	clo	sur	e					
<pre>8.3.1.8.2.1.6 Activity: Nature, age, and rate of folding and deformation in the repository horizon 8.3.1.8-71 8.3.1.8.2.1.7 Activity: Assessment of waste package rupture due to folding and deformation 8.3.1.8-72 8.3.1.8.2.2 Application of results 8.3.1.8-72</pre>										grou	nd m	oti	on i	n	the	a st	ibsi	urfa	ice		1	8	.3	.1.	8-	70
folding and deformation in the repository horizon			8	.3	.1	. 8	. 2	2.1	.6	Acti	vity	: 1	Natu	ire	a, a	age,	a	nd r	ate	of						
repository horizon										fold	ing a	and	def	for	mat	tion	n in	n th	ie							
8.3.1.8.2.1.7 Activity: Assessment of waste package rupture due to folding and deformation										repo	sito	ry	hori	zc	n							8	1.3	.1.	8-	71
package rupture due to folding and deformation			8	.3	.1	. 8	.2	2.1	.7	Acti	vity	:	Asse	ss	sme	nt d	of 1	wast	e							
deformation										pack	age	rup	ture	3 0	iue	to	fo.	ldir	ig a	nd						
8.3.1.8.2.2 Application of results 8.3.1.8-72										defo	rmat.	ion										8	1.3	.1.	8-	72
		8	. 3	.1	. 8	. 2	. 2	2	Appl	licat	ion	of	resu	ilt	: 5							8	3.3	.1.	8-	72

11

TABLE OF CONTENTS (continued)

8.3	1.:	1.	8	3		T to B	nii	ve or d	100	ti re at	ga qu ur	ti ir at	on: ed	OI	St	iu ab	d.	i : ng vd	ser	tos	0	pi v	in: di	vi sa ue	de tu: to	in rat	fo	rma	i= nic								
	3.3	3.	1	. 8		e 3.	IV 1	eı	t co to s	s tu ec	dy to	: ni	10	na] pro	ys ys		< (1) (2)	0.5	f	t	he	e (	ef 7e	fe nt	ct: s :	s c on	fav	era	age		8.	3.	1	. 8		73	
									2	ep	08	it.	or	ý.				. 1				Ľ.		1			-17	1	171		8.	3.	1	. 8	-	82	5
		8.	33	. 1	•	8.	3	-	ι.	1	A V	ct ol	iv: car	ity	1:	or	A	nn ic	m	a1	. I	ore s e	db ev	ab en	il. ts	ity ir	0 1	f he									
		8.	3	. 1		8.	60		1.	2	C A	on ct	tro iv.	511 ity	.ed /1	1	a A	ress	ae	55	emi	ent		of	ť	he	ef	fed	ots	•	8.	3	1	. 8	-	82	2
											0	1	291	nec	pus nn	5 4	4	110	1	112	5.2.5 . + . 1	on.	5	an	a	VO1	.ca	nı	2		0	0				0 1	5
		8.	100	. 1		8	3		1.	3	A +	ct	iv	ity	71 71		F	au	11	10	1.01		ra n	te	·	re	icu	1= 			0.	3	*			0.	2
											1	at	10	6 7	n fr	fe	i A	*		n.	1	0.1	00	0	vr	k/ - k S		aun	80-		R	2	4	\$		8.	۵
		8.	3	. 1		8	. 3		1.	4	A	nt	iv	iti	0.1		E	F 4	- 42	** (**	- 12	0	F	fa	2+	+ 1 +	nor '	on	1	٠.	~ .	÷.		1			
											a	ve	ra	ae	£	11	IX	1	a	te	2.5							~		j.	8.	3	1	1	-	8	5
		8	3	. 3		8	. 3		1.	5	A	ct	iv	it	V:		A	55	5 Đ	55	SIN	en	È.	of	t	he	ef	fe	cts								
											0	f	fa	ul	Éir	ng	7	or	1	f.	lu	1	ra	te	S						8.	3	.1	18	3-	8	5
		8	3	1		ġ		1.	1.	6	A	.ct	iv	it	y:		U	p	Li	ft		ra	te	S	in	th	he	00	n=								
											t	ro	11	ed	a	re	a		÷			e i		ξ.			Ξ.	1		¢.	8.	3	. 1	. 1	3	8	7
		8	.3			8	2.23	1.	1	7.	P	ot	iv	it	Y1		A	SI	se	S	sm	en	t	of	t	he	ef	fe	cts								
											¢	f	fo	14	in	g,	ŧ.	u	51	1	ft	,	an	d	su	bs:	ide	nc	e								
			5								¢	n	f1	ux	r	at	:0	S	1		٢.		έ.,			۰.			$\mathbf{x}$	4	8	. 3	.1	. 1	8-	8	7
	<b>5</b> .	3	• -	1	5.	3	• 4		1	Sti	idy	11	A	na	ly	SJ	1.5	1	01		th	e	ef	fe	ot	01	٤.,										
									-	ceo	200	ma	C,	pr	00	61	55	e	5	81	nd	e	ve	nt	S	on										12	
		0	0			0	1	-		ana	anç	162	1	n	Wa	Ce	91	-	a a	D.	16	6	16	va.	101	on				۰.	8	. 3	. 1	• 1	8-	8	8
		0			* *	P	1.6	5.4	4	* +	1	101	11	15	YI		1	ne:	er 	m	a L	a	no	1.0	Dar	T1	BIT	•±0	-								
											1	n al la N M	W LTD	10	Tel	21	1 20				19	ne	100	S	70	Cr	151	.on	S		0						0
		8	2		1	R		2	2	2	1	1 144	10	144	1.7 1		2	0	000	0	C	va on	*	.QL 64	а 8 а	ho		4	ah e	1	0	- 2			0	0	0
		~		1							2	of	10	me	2 .	0		n	1.1	113	e i	00	0	OF		at	04.	++ 2	hie	2							
												110	ave	ti	on	8			-		52 th			1014	1	au	6.4	La	2.4.5		8	2			e	8	a
		8	. 3		1.	8		3.	2	.3	2	Act	iv	it	VI		3	Ls.	se	2.5	ST	en	t	of	e t	he		Ffe	at		~	***	1	1	÷ .	4	~
											6	of	st	ra	in	1	cì	18	ne	re	S	on		vat	er		abl	A									
												e16	sve	iti	on	15				10								Ξ.		ŵ.	8	.3	. 1		8-	9	1
		8		١.	1	8		3.	2	, 4	1	Act	it	it	y:		1	As	se	as	SI	ner	t	01	f t	he	0	ffe	ot							1	
											(	of	fo	bld	lin	Ig.	,	u	pl	11	ft	,	03	: 1	sut	si	der	nce									
											3	on	Wa	ite	ir-	·t.	al	01	e	.0	10	sve	t:	ior	n	τ.	ι.	ι,	1		8	.3	. 1	1	8-	9	2
		8		3.	1	. 8		3.	2	. 5	1	Act	cit	rit	Y:		1	Ef	fe	80	ts	3 5	f	fa	aul	lti	ng	or	1								
											1	wa	.er	:-t	ab	1	e	e	16	6 A	at	10	n								8	. 3		4.1	8-	03	3
		8	-	5.	1	. 8		3.	2	. 0		AC	213	71t	Y		-	As	Se	85	SI	ner	it	0;	ft	the											
											1	er	red	35	01		I	au	11	21	.nç	3 0	n	W	ate	97 -	tal	Dle	3					1			
											101	67	SAS	ata	.01	3				4	*		+		. 6			÷ ;			8			1.	8-	3	13

Page

6

0

TABLE OF CONTENTS (continued)

																																						Pa	g	e
	0	-		1	0			2							1								١.		e															
	0.			* *	0	2.9	2.4	9		0 L 4 G	unt i	Yi nni/	~	na	+Y	51	L 2 D 2	5	0		5	ne A		101	Le nt	00	SC	1		1										
										fr	act	- 13 44	6	ne	rm.	0.	2 2 2 2	0	1	1	17	·**	0	10.	of	D fo	ou.		000	3.1										
										00	100		0	he	4.410	0.0	a	2.18		6 14	3	9	****	1		10		6 Y .	0				R	-	e i	4	R		4	4
		8	3.	3	1		Β.	3	.3	.1		Act	iv	it	vi		1	A.s	s		S	the	int		of	+	he	6	ffe		ts	1	ř	8 14	1	* '	Ŷ		1	
		1								17		of	ic	me	ou	s	1	ir	it	ru	18	ic	n	8	on	1	000	a1	f	ra	0=									
												ture	e	pe	rm	ea	al	bi	1	it	v	2	and	1	ef	fe	ct	iv	ê		1									
												por	os	it	ie	s		١,								1					1		8		1.	1.	8		9	5
		8	8.	3	1		8.	3	. 3	.2	2	Act:	iv	it	y:		1	As	18	es	S	me	ent		of	t	he	e	ffe	ac	ts									
												of :	fa	ul	ti	ng	g	¢	n	1	.0	Ca	11	f	ra	ct	ur	e	pe:	rm	e-									
												abi	11	ty	a	nd	ż	6	f	fę	c	ti	V	8	po	ro	si	ti	es				8	1.5	3.	1	8		9	5
		~	8.	3	1		8,	3	. 3	. 3	3	Acti	iv	it	y:		1	As	SS	es	S	me	ent	5	of	t	he	e	ffe	ec	ts									
												of	st	re	SS	4	23	r	S	tı	a	in	1 (	on	h	yd	lro.	10	gi	0										
		l.	i.									proj	pe	rt	ie	S	\$	¢Í	5	th	1e	2	00	ck	m	as	S	÷ 1		÷.,	ŝ.	à.	8		١,	1	8		9	6
	8		3.	1	. 8		3.	4		Ap	opl	ica	ti	on	0	f	1	re	35	u1	t	S		•	κ.	÷	÷	s :	ĸ, i		۰.	x	8		3.	1	8	-	3	7
8	. 3	• •	1.	8	. 4		1	n	ve	st	tig	ati	on	11	S	ţı	u	dj	e	S	t	0	P	ro	Vi	de														
							3	.n.	10	III	nat	lon	I	eq	uı	re	ek	đ	0	n	C	na	an	ge	S	10	r	00	ĸ											
							0	le	00	ne	ema	cal	P	ro	pe	r	Ç.	16	53	1	:0	SI	11	51	ng	1	ro	m					1							
	0		5				1	e .	Ct	01	nic	pr	00	es	se	S	į.		•	1	1		1	1.	:	1		١.,	۲.	٤.,	۰.	٠	8	1.4	۶.	1	. 8	*	9	8
	0	***	2.,	*	1.0	1	4.	*		20	cua	Y:	-	ina	TÀ	5.	1	S	0	Ι.	0	De	8	eI	Ie	Ct	S	OI												
										10	ana	ham	0 4 4	pi	00	N S	5	St n/	25		11	D	6	ve	nt	5	on	-	oc	ĸ										. :
			R	2	1		Ŕ	۵	1	Ac	1	h ot	14	+4+		+	-	De No	24	0.0	10	3		1		1	in.	1	in a	*	*	3	0		5.4	+	. 0	7	- 4	+
		ľ	~ 1	Ĩ	1.14						*	in	* 1	ink a	3.0	0	~	сы (т)	50	m	10	11111	1.		01	0.4	.110	00	na.	119	ę									
												to	ic	me	00	2 0	~	11	nt	211	15	1	- 0 D	5.4	45	·~.	· br ab	60	u	ue			8		2	1	9		0	1
			8.	3	. 1		8	. 4	.1	1	2	Act	it	zit	vi	1		A	2.2	-	20	m	en	t.	of		ha	d				1		-		^				*
												of	mi	ine	ira	1		c)	ha	n	76		al	on	a	fa	ul	t	zo	ne	2									
												in	10	0,0	000		v	r															8		3	1	6	-	10	2
			8	3			8	. 4	.1		3	Act	in	vit	y:		1	A	SS	e	SS	sme	en	t	of	1	he	e	ff	ec	tt	3					1			
												of	fa	aul	.t	0	f	f.	se	t	¢	'n	t	ra	ve	1	pa	th	wa	v		1	8	1	3.	1	. 8	-	10	3
			8	3			8	. 4	.1		4	Act	11	vit	y:			A	s s	e	ss	sm	en	t	of	1	he	d	eg	Te	e									
												of	m	ine	era	11		C.	ha	n	ge	9	in	t	he	2 0	con	tr	01	10	be									
												are	a	re	esu	11	t	1	ng	r i	fı	:01	m	te	ect	:01	nic	al	1y											
												ind	iud	ced	d c	h	a	n	ge		iı	1	wa	te	r-	ta	abl	e												
				1								ele	V	ati	lor	15			•	x			•	÷									8		3.	1	. 8	3-	10	)4
	0	*	3.		+ 1	5 . 	4	. 4		A	ppl	10a	it:	101	10	)Í		I	es.	su.	11	S			÷	k	4		٠.	×	*	÷	8		3.	1	. 8	3-	10	5
0	. 3		+	, ö	*	5		In	Ve	25	110	rati	.01	nı	5	st	U	d	16	5	1	0	52	ro	DV3	de	e t	he												
								10	I		mat	100	4	rea	ju)	ŗ	9	a	1	Y	1	cn	e	ar	18.	.y	515	ê	ind											
								as	St	25	SME	int	5	nve	est	11	g	a	τı	.0	n	S	OI	1	ine	8 1	cec	tc	oni	C	5			5		i.	i.	d.	J,	
	Q	i.	2	4		ó	-	PI 1	0	JL	am			in.			1		1	1		1	1	٢.	-	4	4	*	1			ł	2	*	3	. 1	. 1	5-	10	05
	9		-	8.14	1	ο,	~	* **		0.4	0.04	iy :		un	are	30	10	9	1.1	. 2	a	22	00	1	IC	V	010	ai	110								٩.			
			8	-		1	8			1	eat 1	A of	10	17.4			. *	17	-	-			-	1	1.	1	11.			*		*	2	1 .	3 5			5 **	10	11
			Ř			1	2			1	2	Act		V == 1	LY Ly	4 4		× C	0.	10	d.	***	SI		11.	h	100	214	35	. *		1	5	2.2	20	• 4	*	5-	44	30
			8	1 1 1		1	R			1	3	Act		vi	L.			0 5	10	1	A	- 0	110	11	29.	10	566	111	11.				0	1.	00	• 4		0	4 .	16
			8	1 1 1		1	8		5	1	4	Act		vi	tv.			* C	-	20	h	AT	110	***	ry.	10	f	201	3 dates	0		.*	1	2.4	0			0	***	10
				-			1	1				Sec	au	en	Cel	S		-			**	set i	tude is		- 2	~	* *	1.00	-	.0			\$	2	2	1		A	1.	10
			8	. 2		1	8	. 5	5.	1.	5	Act	1	vi	tv		1	G	e	00	h	en	in	a	1	cv	cle	28	of	e *	1				-			-		h. 4
												bas	sa	lt	ic	1	70	51	C	an	1	C	fi	le	ld	S							3	8.	3	. 1		8-	1	21
																			-		-	-	-	-	-	-		1.0	100	100	1.1		-	11.1	100	1.18	1.0	100	1000	- 10 C

A-15

TABLE OF CONTENTS (continued)

Q.

Page

		8.	3.	1.	8.	5.	2	Stud	y:	Char	ac	te	riz	at		٩f	i	gneo	us									
					1			intr	usi	ve re	at	ur	es.		36	• •	1			1	1	8	3	. 1	. 8	-1	63	
			ο.	\$,	4.	8.	5.4	• •	ACT.	ivity	13	2.	val	luat	10	no	I	aept	n or							11		
			~						cur.	le te	mp	er	atu	ire	15	oti	en	m	1.1		•	δ,	3	. 1	. ŏ	-1	24	
			Β,	3.	+ •	δ.	5.2	. 4	Act.	TATEA	1	Ģ	hen	1108	a1	and	P	nysı	cal				2					
							1		ona	nges	ar	ou	ind	da)	kes.		1	4 4	×., ×		•	8.	3	.1	. 8	-1	25	
			8.	3.	1,	8,	5.2	.3	Act.	ivity	11	H	leat	: I.	LOW	at	X	ucca										
									Mou	ntain	a	ind	l ev	valu	lat	100	0	f re	gion	al								
									amb.	ient	he	at	f]	LOW	an	d 1	00	al h	eat			1.2						
			1	j.	1.1		201		flo	w and	oma	111	es	14								8.	3	.1	. 8	-1	27	
		8.	.3.	1.	8,	\$.	3	Stud	Y:	Inve	st	19	ati	lon	of	fo	ld	s in										
								Mioc	ene	and	yo	un	iger	r re	ock	SC	f	regi	on .	κ.		8.	3	.1	. 8	-1	29	
			8	3	. 1. ,	8,	5.3	.1	Act	ivity	11	E	[va]	luat	tio	nc	f	fold	s in									
									Neo	gene	ro	ock	5 0	of t	the	re	gi	on .				8	3	.1	, 8	-1	30	
		8	. 3	1	. 8 .	. 5 .	4	Appl	ica	tion	of	f r	est	ulti	S				6. 6		1	8	3	.1	. 8	-1	31	
	8,	3	.1.	8	. 6	01	Sche	dule	fo	r the	a p	008	stel	losi	ure	te	ot	onic	S									
						F	prog	ram	1.14			Υ.					1.					8	.3	.1	. 8	-1	31	
x	3.	1	. 9	(	DVE	71S	view	of	the	huma	an	in	ite:	rfe:	ren	ce	pr	ogra	m:									
				. 1	Des	SCI	ript	ion	of	poter	nti	al	. ft	utu	re	hun	an	act	ivi-	6								
				1	tie	e s	at	or n	ear	the	si	ite	2, 1	reg	uir	ed	by	per	-									
					foi	rma	ance	and	i de	sign	is	ssu	les	11.					1. 1			8	. 3	.1	. 9	-	1	
	8.	3	.1	9	.1	1	Inve	stig	ati	on:	St	: uc	iies	st	0 2	rov	7id	e th	e									
							info	rmat	ion	requ	ai :	- 80	d or	n n	ati	iral	p	henc	mena									
						2	and	huma	in a	ctivi	it.i	Les	s t)	hat	mi	ght	d	legra	ide									
							surf	lace	mar	kers	ar	nd	.nor	num	ent	5 .						8	.3	.1	. 9	-	15	
		8	.3	.1	. 9	,1	.1	Stud	iy:	An e	eva	alt	ia".	ion	of	na na	atu	ral										
								proc	ess	es th	hat	to	obu.	ld.	aft	ect	t t	he										
								long	-te	ITR ST	urv	vi	vab.	ili	ty	of	th	ie su	irfac	e								
								mark	ker	syst	em	at	t Y	uco	aN	lour	nta	in .				8	.3	.1	. 5	-	18	
			8	.3	.1	. 9	.1.1	1.1	Act.	ivit	v:	1	Svn	the	sis	5 01	f t	ecto	onic,									
									sei	smic	1 1	and	dv	010	an	ic 1	haz	ards	a dat	a								
									fro	in o'a'	hei	r :	sit	e c	har	ract	ter	izat	ion									
									act	ivit	ie	s		11								8	3	1		-6	19	
			8	.3	.1	. 9	.1.1	1.2	Act	ivit	v:	1	Sva	the	si	8 1	et	zalua	ation									
									of	the	ef	fe	cts	of	÷	ut u	re	eros	sion									
									and	i dep	05	it	ion	on	t	he	SUI	rviv:	abili	itu								
									of	the	ma	rk	er	SVS	tet	n a	+ 1	(noo)	3									
									Mou	intai	n			-1-			· ·					R		1			19	
		8	.3	1	. 9	.1	.2	App	lica	ation	0	+	TAS	mlt	9							R				a	20	
	8	.3	.1	0	. 2		Inve	Asti	ant	on:	Q	+ 11	die	e +	0	ora	vie	10 +1		• •						-		
	1			1			info	ormai	tion	n rea	mi	TA	do	n r	TA	RAN	+ ;	and	f 11+ 11									
							val	110 0	far	nerou		rii	ner	21	1	and		and		nd-								
							wat	ar ri	6801	17005	1	5 . b.d.:		a.	*	ante	1 9	and	grou.	1101-		0				0-	20	
		9	. 3	1	ó	2	1	Stin	AUI	Nat	11.9*	21	-		170				ont	· ·			•••	2 * *	• • •	9-	20	
		~						of	Yua	na Mo	11175	+ 0	in	N.	10	Cour	001	U M	ound			c				0-	27	
			ç		1 1	0	2	1 1	Bas	nd yeth	122.4	u ca	Gan	up.	e i	004	110	Y IN	evad	4 .			**	2 * •	4.4	2-	41	
							1. 40 1	an r ch	04	Vinne	2 .	Ma	000	nie	a i	n n	01	2262	amen	6								
									+ 1	1 000	an	+ + +	51	for	* *	11 1	ET.	110	11 60				1.5			0	20	
									1.111	- LILI	- <b>BOOK</b> (1997)		100					1 1 11								1000 1000		

# TABLE OF CONTENTS (continued)

		90
8.3.1.9.2.1.2 Activity: Geophysical/geologic		
appraisal of the site relative to		
mineral resources	8.3.1.9-	33
8.3.1.9.2.1.3 Activity: Assessment of the poten-		
tial for geothermal energy at Yucca		
Mountain, Nevada	8.3.1.9-	34
8.3.1.9.2.1.4 Activity: Assessment of hydrocarbon		
resources at and near the site	8.3.1.9-	37
8.3.1.9.2.1.5 Activity: Mineral and energy assess-		
ment of the site, comparison to		
known mineralized areas, and che		
potential for undiscovered resources		
and inture exploration	8.3.1.9-	39
8.3.1.9.2.2 Study. Wacer resource assessment of		
Yruca Mountain, Nevada	8.3.1.9-	42
8.3.1.9.2.2.1 Activity: Projected trends in local		
and regional ground-water develop-		
ment, and estimated withdrawal rates		
in southern Nevada, proximal to		
Yucca Mountain	8.3.1.9-	43
8.3.1.9.2.3 Application of results	8.3.1.9-	44
.3.1.9.3 Investigation: Studies to provide the infor-		
mation required on potential effects of		
exploiting natural resources on hydrologic,		
geochemical, and rock characteristics	8.3.1.9-	45
8.3.1.9.3.1 Study: Evaluation of data needed to		
support an assessment of the likelihood		
of future inadvertent human intrusion at		
Yucca Mountain as a result of explora-		
tion and/or extraction of natural		
resources	8.3.1.9-	46
8.3.1.9.3.1.1 Activity: Compilation of data to		
support the assessment calcula-		
tion of the potential for inad-		
vertent human intrusion at Yucca		
Mountain	8.3.1.9-	48
8.3.1.9.3.2 Study: An evaluation of the potential		
effects of exploration for, or extraction		
of, natural resources on the hydrologic		
characteristics at Yucca Mountain	8.3.1.9-	49
8.3.1.9.3.2.1 Activity: An analysis of the poten-		
tial effects of future ground-water		
withdrawals on the hydrologic system		
in the vicinity of Yucca Mountain,		
Nevada	8.3.1.9-	49
8.3.1.9.3.2.2 Activity: Assessment of initiating		
events related to human interference		
that are considered not to be suffi-		
ciently credible or significant to		
warrant further investigation	8.3.1.9-	. 50

MADID	0.27	(Crist	MUNITO	Inna	h mar all
IND LL	UL.	CON	TUNTO	(CONF	Tunned)

-

	Page
8.3.1.9.3.3 Application of results	8.3.1.9- 51
0.5.1.7.4 Schedule and milestches for Site Frogram 8.3.1.9 (human interference)	8 3 1 0- 51
8.3.1.10 Overview of population density and distribu- tion program: Description of population density and distribution required by the per-	0.0.1.5- 01
formance and design issues	8.3.1.10- 1
8.3.1.11 Overview of land ownership and mineral rights program: Description of status and plans for land ownership and mineral rights required by	
the performance and design issues	8.3.1.11- 1
8.3.1.12 Overview of the meteorology program: Descrip- tion of meteorological conditions required by	
the performance and design issues	8.3.1.12- 1
on regional meteorological conditions	8 3 1 12- 8
8.3.1.12.1.1 Study: Characterization of the	VI018146 V
regional meteorological conditions	8.3.1.12-10
8.3.1.12.1.2 Study: Plan for synthesis of Yucca Mountain Project meteorological	
monitoring	8.3.1.12-11
8.3.1.12.1.3 Application of results	8.3.1.12-13
o.3.1.12.2 Investigation: Studies to provide data on atmospheric and meteorological phenomena at potential locations of surface	
facilities	8.3.1.12- 14
8.3.1.12.2.1 Study: Meteorological data collection	
at the Yucca Mountain site	8.3.1.12- 16
8.3.1 12.2.1.1 Activity: Site meteorological	
8.3.1.12.2.1.2 Activity: Data summary for input	8.3.1.12= 16
to dose assessments	8.3.1.12-22
8.3.1.12.3 Investigation, Studies to provide date on	8.3.1.12= 24
the location of population centers relative	
the site	8 3 1 12- 25
8.3.1.12.3.1 Application of results	8.3.1.12-26
8.3.1.12.4 Investigation: Studies to provide data	VIUI2122 2V
on potential extreme weather phenomena and	
their recurrence intervals	8.3.1.12- 26
8.3.1.12.4.1 Study: Characterize the potential	
Sitreme weather phenomena and their	
8 1 1 2 4 2 Application of results	8.3.1.12= 27
8.3.1.12.5 Schedule for the meteorology program	0.1.1.12= 28
8.3.1.13 Overview of offsite installations and operations	0.0.1.12- 28
program: Description of the offsite installa-	
tions and operations required by performance	
and design issues	8.3.1.13- 1

-

TABLE OF CONTENTS (continued)

8		3.	1.	1	3.	1	1	inv	estigation: Determination of nearby								
							1	nd	istrial, transportation, and military								
							1	ns	callations and operations (nuclear and								
		0	5	4		2	1	non	NUCLEBR)		8.,	3.	1	1	3-	2	
	2	Q. +	9.	*	*	5.	+ 1	*	Activity: Identify hear-site		0	0			2	0	
		8	2	4	÷	2	1	2	Activity. Characterize muchast fuel		Q 4	2.	+	**	3-	2	
		~ •		-		~ .	* '		cycle facilities in the area		R	2	1	1	2	4	
		8.	3.	1	1	3.	1.	3	Activity: Characterize all nuclear		4.		-		~	1	
									facilities not associated with the								
									nuclear fuel cycle near the Yucca								
									Mountain site		8.	3.	1	.1	3-	5	
		8.	3	.1	. 1	3.	1.	. 4	Application of results		8.	3.	1	.1	3-	6	
8	3	3.	1	. 1.	3.	2		Inv	estigation: Potential impacts of nearby								
				2	ĩ.		4	ins	tallations and operations		8.	3.	1	.1	3-	6	
		8,	3	.1	. 2	3.	2	.1	Activity: Evaluate near-site								
							~	~	activities		₿.	3.	1	.1	3-	8	
		δ,	2	-		3.	4	. 4	Activity: Evaluation of the impact of								
									nuclear fuel cycle operations near the			~					
		8	2	1	4	2	2	5	lucca mountain site and Las vegas		ζ.	3.	+	• -	3-	9	
		~ .	~			~	-		notivity, Evaluate the impact of all								
									the nuclear fuel cucle near the Vucca								
									Mountain site		R	2	1	1	2-	10	
		8	. 3	.1	. 1	3.	2	. 4	Activity: Evaluate the impact of			Sec. 1		* **	-	ut V	
									ground motion from nuclear testing								
									activiti . at the Nevada Test Site		8.	3	.1	.1	3-	11	
		8	. 3	.1	. 1	3	. 2	.5	Applicat n of results		8.	3	.1	.1	3-	11	
	8.	3	.1	.1	3.	3		Sch	edule for offsite installations		8.	3	.1	.1	3-	11	
8	3.	1	12	4	0	)774	22	Vie	of the surface characteristics								
					1.00	r	Dġ	ran	: Description of the surface character-								
						.51	11	CS	required by the performance and design								
	à	2	4	1	1	1	su	es Tot	a ta		8.	3	.1	. 3	.4-	1	
	ę .	0	* .4	**	7	*		+ ~ *	estigation: Studies to provide the								
								10	stions of surface facilities		0				1	10	
		8	3	.1		4	1	1	Application of results	1	0 0	0 0		**	4-	18	
	8.	3	.1	.1	4	2	1	In	restigation: Studies to provide soil	•	0	. 0	• •	* *	.4-	20	
								and	I rock properties of potential locations								
								of	surface facilities		8	2	1	Ξ,	4-	. 27	
		8	.3	.1		14	.2	.1	Study: Exploration program	120	8	3	1		14-	31	
			8	.3		1.	14	.2	1.1 Activity: Site reconnaissance		8	.3	.1		14-	33	
			8			1.	14	.2	1.2 Activity: Preliminary								
									exploration		8	. 3	.1		14-	40	
			8	. 3		1.	14	.2	.1.3 Activity: Detailed exploration .		8	. 3	.1		14-	43	
		8	. 3	.1		14	. 2	.2	Study: Laboratory tests and material								
									property measurements		8	. 3	. 3		14-	45	
			8			۲.	14	. 2	.2.1 Activity: Physical property and								
									index laboratory tests		R	3	- 1	1	14.	. 45	

WH .

8

TABLE OF CONTENTS (continued)

Page

		8.3	3.	1.	14	.2.	2.	2	Activity: Mechanical and dynamic							
	0	2	1	1 4	2	0	0	+ A	aboratory property tests	8	. 3	• •	k . 1	14		48
	0.1	0	• •		1.60		*	inn	L' LTETA CEDED BUA CUBIGCELITA-	0				1.4		: 0
		0	2	d.	1.4	0	5	1 011	Interstation and a second the field	0				7.4.		00
		0	9.1	**	7.4	141	2.	*	Accivity: Physical property field							
									Costromonto	0	1			1 4	_	1
		0	2	4	14	5	2	2	Activity, Machanical property	0			* *	7.4		24
		Q	9.	**	7.4	• 6 •	2.	6	field tests	0				1.4		. 77
		0	2		1.4	5	2	0	LATIN Coophysical field	C	1.0		4.4	74		51
		0.	÷ •	* '	7.4	1.6.1	2.	2	verived, geobulatour treid					1.4		6.0
	à	2	1	1 4	0	4	2	nnl	astion of repulte	0		, 	**	14		60
8	0.00	1	14	+ 2	1.6	Sak	n nod	pp1	for the curface characteristics	c		2.4.	4.5	7.4		00
8.9	w .	* *	* 4	1.4		501	100	200	TOT THE SATTAGE CHURCHCETTSCTOP					1.4	_	61
2	1	15		00		via	.A+	of	thermal and mechanical reck	c	1.5.4	2.4.5	**	7.4	Ĩ.,	0 11
		**		nr nr	00	0 ====	110	e n	norman Description of thermal							
				an	d	mar	-+c	Di C	al rock properties required by the							
				ne	++	0 19	nan	100	and decide icense	c	1 1	2	1	15		
8	3	1.	15	1	**	Int	76.5	stia	ation: Studies to provide		1 * *		A. 1			*
						the	5 7	600	ired information for enstial							
						di	at y	ibu	tion of thereal and mechanical							
						Dro	DDe	rti	ad		2 1	2	1	15	_	22
	8.	3.	1.	15	.1	.1	0000	Stud	v: Laboratory thermal properties	ŝ	2.5	2	4.1	15	_	21
		8.	3.	1.	15	.1	.1	1	Activity: Density and porosity				***	**		~ *
			Ē.		1				characterization	5	2.1	2	1	15		21
		8.	3.	1.	15	.1	.1.	2	Activity: Volumetric heat capacity			1		**		~ *
								1	characterization	1	2 1	2	1.	15		24
		8.	3.	1.	15	.1	.1.	.3	Activity: Thermal conductivity	33.7						~ 1
									characterization	1	8.1	3	1	15	-	36
	8.	3.	1.	15	.1	2	3	Stud	y: Laboratory thermal expansion				1			~ ~
							t	test	ing	-	8.	3.	1.	15	-	38
		8.	3	1.	15	5.1	.2	.1	Activity: Thermal expansion					-		
									characterization		8.	3.	1.	15	-	38
	8,	.3.	1	15	5.1	3	1	Stud	y: Laboratory determination of					R.		11
							I	nech	anical properties of intact rock		8.	3.	1.	15	-	40
		8.	3	1	15	5.1	.3	.1	Activity: Compressive mechanical							
									properties of intact rock at base-							
									line experiment conditions		8.	3.	1	15	5-	40
		8	3	.1	.1	5.1	.3	.2	Activity: Effects of variable							
									environmental conditions on							
									mechanical properties		8.	3.	1	.15	5-	41
	8	.3	.1	.1	5.3	1.4		Stu	ly: Laboratory determination of the							
							)	meci	anical properties of fractures		8.	3.	1	.1	5-	43
		8	.3	.1	.1	5.1	. 4	.1	Activity: Mechanical properties							
									of fractures at baseline experiment							
									conditions	1.11	8.	3.	.1	.1	5-	43
		8	.3	.1	.1	5.1	. 4	.2	Activity: Effects of variable							
									environmental conditions on mechani-	•						
									cal properties of fractures		8.	3	.1	.1	5-	44

and at the

0

TABLE OF CONTENTS (continued)

																																							1	Pa	q	8
8		3.	1.	1	5.	1.	5		S	tu	d	71		E:	102	av	at	. i	01	n	in	nv	es	st.	ig	at	i	on	5	ċ.	j	į.	į.	8.	100		1.	1	5	÷	4	5
	1	8.	3.	1	.1	5.	1	con con		1	1	AC	ti ti	vi vi	ity	y:		03 10	hi	a f	t	ost	ion ra	nv at	er	ge	b	ce	ak.	01	it	•	۴.	8,	6.4		1.	1	63		4	6
	ŝ	8.	3.	1	. 1	5.	1			3	1	ro Ac	om	US V	it	V :	•	. 53	e	a	iei	nt.	ił	1	. 0	ri	f	 t	4			•		8.	644		1	1	5	-	4	B
											Ţ	ni	ni	nç	J		a.									r.			1		č.		۰.	8	-	١.	1	. 1	5	•	4	9
8		3.	1.	1	5.	1.	6		3 60	tu	id:	1:		II	n :	51	tı	a' '	t	he	E	mo	ma	30	ha	ni	.01	al						8		į.					5	5
		8.	3.	1	. 1	5.	1	. 6	5.	1	P	AC	ti	V.	it'	y:	Ì	H	le	at	e.	r	63	sp	er	in	ne:	nt	1	n		1	1				*		1	1		
											1	un	it	1	rsi	w1		1						ć.	•	÷	A		1		¢.,		4	8	1.0	\$ .	1	. 1	5	**	(31	5
		в.	3.	-		5	-	. 5	2 .	2	4	AC	11 DA	.V.	im	y: er	+	C	a	n	S	te	ε <b>Γ</b> .	• S	CB	16	9	ne	at	e:	r			8		2	1	. 1	5		5	۵
		8.	3	1	. 1	5	.1	. ŧ	ŝ.	3		AC	ti	V	it	y:		Y	lu	0	a	M	101	in	ta	11	n i	he	at	e	d		ľ.				1		1			
		0	0					1				b1	00	:k		•	х				. 1	1			1	1	ľ	1			٢.	۰.	٠	8		3.	1		5	-	5	7
		9.	8	1			• *	* 5	2.4	71		me	88	su:	re	y i me	n	ts	B	÷.		¢1.,		56	14		2		1				i.	8		3.	1	1	5		6	0
		8.	3	. 1	. 3	5	.1	. (	6.	5		Ac	ti	v	it	V:		ž	le	at	e	d	T	00	m	e	ED.	er	in	hei	nt			8		3.	1		5		6	2
8	3.	3.	1	.1	5.	1	. 7		S	ti	1d	V!		I	n	S	ti	u	m	e	ch	ar	i	ca	1	pi	20	pe	rt	i	es			8		3.	1	. 1	5	-	6	5
		8.	3	.1		5	.1	. 1	7.	1		Âc	tj	v	it	V		E	21	at	ce	1	0	ad	ir	à	t	es	ts	3	1		1	8		3.	1	. 1	5	-	6	5
		8.	3	. 1	. 1	15	. 1	1	7.	2		Ac	ti	v	it	v		F	RO	CI	<	ma	as	s	st	I	en	at	h													50
												ex	De	ar	im	e	it				١.							1	1	١.,			1,1	8		3.	1	1	15	-	6	8
1	Β.	3.	1	.1	5	1	. 8		S	t	ld	y:		I	n	s	t	u	d	le.	si	CI	1	ve	13	f.	ic	at	10	'n		2	1	8		3.	1		15	-	7	0
		8.	3	.1	1	15	. 3		8,	1		¢	ti	1V	it	y		1	Ēv	a	11	at	1	on	¢	f	m	in	in	na												
												me	th	ho	ds						×				á.							ι.		8		3.	1		15	-	7	1
		8.	3	.1		15	. 1		₿.	2		Ac	it;	iv	it	У	1	1	Mo	n	it	01	ri.	ng	Ċ	f	g	ro	u	nd	*											
												su	ipr	po	rt		sy	SI	te	m	5				1						i.	6		8		3.	1	1	15	-	7	2
		8	. 3	.1		15	. 1		8,	3		Ac	tt.	iv	it	y	1	1	Mc	n	it	01	ri	ng	1 4	ir	if	t														
												st	.al	D1	11	t;	Y		ŕ.	÷	j,		1		×.			1				×.		8		3.	1		15	5-	7	3
		8	.3	- 3		15			8	4		Ac	it;	iv	it	Y	*	ų,	Ai	.1	9	nui	al	it	У	8	nd	1	re!	nt	11	a•	•									
												ti	.01	n,	ex	(p	er	11	me	n	t		6	×	1	٩,				•				8	÷	3.	1		15	5-	7	4
	8 . 	1 62	.1	13	5	14	23	2	1	Jb	b1	10	a	21	on	1	of		re	2.5	11	.t.	5	÷	÷	18					÷.	÷	j.	8	4	3.	1		15	5-	1	6
8.	3.	1	. 4	5.	2		11	JV	e:	st	19	at	110	on	:		St	u	di	Le	S	t	0	pı	101	Vi	de	t	h	8												
							re	Pq	u.	11	ec	1.3	n	IO.	rn	na	tı	0	n	1	OI		sp	at	11	al	0	11.5	st	ri	bı	]										
							C ;	10	n	0	Ξ.	an	np.	10	int		st	ŗ	6:	55	10	an	¢,	ti	1e.	rm	al															
							C(	on	a.,	1t	10	ns	8	۲.,	4			Ċ.	*	Ŀ,			1	κ.	×	j,	1		۰.	1	٠			8		3	1	ł	11	5	1	7
	δ.	3	14	1.4	.0	16	1	£.		st	uc	Ϋ́Υ.		ç	ina	ar	ac	12	e:	<b>r</b> 1	Zā	31	10	n	0	Í.	ti	16	5	10	e						L					
			-	١.,	. 1			~	. 1	am	נם	e	36	60	ILI	re	55	3	00	on	di	Lt	10	ns	5	1			•	٤.	÷,	×		8		3	. 1	•	1	5-	96	18
		ŏ	. 3	• •		15	4.1	4.	+	• •		A	CC.	14	111	Y	1		AI	ne	10	15	11	C	. 00	tr	81	In														
												If	ec.	07	/ei	гy	6	5X	P	er	11	ne	nt	S	1	n	00	pre	9							1			i.	Ľ.	1	
		~										n	51	05	ł	1	1	1	1			٢.	٩.	×					*	£.	×	3		8		3	. ]	• •	1	5-		18
		ø	. 3	* *	kα	12	i e i	2.	+	• 4		A	ot	11	/11	су	1		0	Ve	I	00	re	1	st	re	55	5														
												6	xp	ei	:11	ne	nt	S		4.T	1	tn	9	e	xp	10	II	at	or	У									Ŀ			
	0	2			1					÷.,		S	na	IU	1.1	Ia	C3	1.4	+	сy		1	÷.,	*	1				٠.	÷.,					h	3	.,	• •	1	5*		80
	0	. 2	1.10		10	• 6		ke.		20	u.	Y	1	1	- h	aI	ac	10	e	11	Zi	10	10	n	0	I	01	ne	(I)	11	.e											
		0			1	11		5	-	dill a	u):	16	116	1	1116	e I	Illé	11		00	100	11	6.2	101	ns		3	•	1			. *		5	5.	3	• •	• •	1	0*		52
		0	1.4	1.8.1	÷ 1	4.5	2.4	÷ .	6			15	5	14	r d l	- 4	1		0 1	ul		d C	0	0	85	ec	1 1	ev	81	ua	a C	10	n									0.0
	ò	2			1 5		>	2		8.		0	*	cill t	ille).	10	111	5	-	116	1.4.1	ille 1 + C	-	C	QI	d.	. C.	10	ns		*	. *			2.1	3 6	2.4		+ 1	0."		54
	03	1.95	8.4		* *	5.6	4.10	10		123	12	de site	60	6 he a	a VI	4.4	4.	÷.	*	03	2.64	46	0	. 4		1	£	6		16		1.8		1	3 1	2		4. 6	*	2."	1	00

The second

MADID	00	COLT MELSIMO	(montinund)
TWDMP		CONTENTS	(continued)

8.3.1.15.3 Schedule fo	r the thermal and rock prop-	
erties prog	ram 8.3.1.15= 84	
8.3.1.16 Overview of pre	closure hydrology program:	
Description of	preclosure hydrologic charac-	
teristics and d	onditions required by per-	
formance and de	sign issues 8.3.1.16- 1	
8.3.1.16.1 Investigat:	on: Flood recurrence intervals	
and levels	at potential locations of sur-	
face facil:	ties 8.3.1.16- 7	
8.3.1.16.1.1 Study:	Characterization of flood	
potent	al of the Yucca Mountain site 8.3.1.16-10	
8.3.1.16.1.1.1 Act	ivity: Site flood and debris	
ha	ards studies	
8.3.1.16.1.2 Applic	tion of results R 3 1 16= 13	
8.3.1.16.2 Investigat	on: Location of ademiate	
Water supp		
8 3 1 16 2 1 Study	Innation of adamate water	
eunniv	for construction analytics	
alcent	and decommissioning of a minod	
(10)07	A disposal sustam at Vucca	
Mounts	n Novada 0 2 1 15-16	
8 3 1 16 2 1 1 bo	initian becoment of the sect	
VIVIAIAVIEIAIA AU	avery: Assessment of the cost,	
10	2 and T-12 for use of the alter	
	2 and 0=10 for use as the alter-	
iid.	ive water supply for a mined geo-	
10	nc disposal system at rucca	
0 0 1 16 0 1 0 3m	Intain, Nevada 8.3.1.16= 16	
0.3.1.10.2.1.2 AC	livity: Location of a primary	
wa	er supply for a mined geologic	
dl	sposal system at fucca Mountain,	
Ne	/ada 8.3.1.16-20	
0.3.1.10.2.1.3 AC	ivity: Location of alternative	
wa	ter supplies for a mined geologic	
d1	sposal system at Yucca Mountain,	
Ne	vada 8.3.1.16 · 22	
8,3,1,16,2,1,4 Ac	Livity: Identification and eval-	
ua	tion of potential effects of	
re	pository related withdrawals on	
th	e local flow system at Yucca	
Mc	untain, Nevada 8.3.1.16-24	
8.3.1.16.2.2 Applic	ation of results 8.3.1.16-25	
8.3.1.16.3 Investigat	ion: Ground-water conditions	
within and	above the potential host rock 8.3.1.16- 25	
8.3.1.16.3.1 Study:	Determination of the preclosure	
hydrol	ogic conditions of the unsaturated	
zone a	t Yucca Mountain, Nevada 8.3.1.16- 26	5

# TABLE OF CONTENTS (continued)

Page

8.3.1.16.3.1.1 Act sit the ist	ivity: Synthesis of data fro e program 8.3.1.2 to determin preclosure hydrologic charac- ics of the unsaturated zone a	m ie iter- it					
Yuc 8.3.1.16.3.2 Applica 8.3.1.16.4 Schedule fo	ca Mountain, Nevada		8.3	.1	.16	-	26 28
program . 8.3.1.17 Overview of pre of tectonic and	closure tectonics: Descriptic igneous events required by	, , n	8.3	.1	.16	-	28
performance and 8.3.1.17.1 Investigati information	design requirements	.red ould	8.3	.1	.17	-	1
affect repc 8.3.1.17.1.1 Study:	sitory design or performance Potential for ash fall at th	 1e	8.3	.1	.1	1-	51
site . 8.3.1.17.1.1.1 Act reg car	ivity: Survey literature arding Quaternary silicic vol ic centers in the western Gra	 .=	8.3	.1	.1'	-	54
Bas 8.3.1.17.1.1.2 Act	in		8.3	.1	.1	7	54
fal 8.3.1.17.1.1.3 Act tic	l thickness at the site livity: Assess potential par- le density and size distribu-		8.3	.1	.1'	7 -	54
tic	on of ash fall at the site .		8.3	1.1	.1'	7-	55
8.3.1.17.1.2 Applica 8.3.1.17.2 Investigation information could affect	ation of results lon: Studies to provide requ n on fault displacement that at repository design or	ired	8.3	3.1	.1	7 -	56
8.3.1.17.2.1 Study:	Faulting potential at the		8.3	3.1	.1	7	56
reposit 8.3.1.17.2.1.1 Act sui sit	tory	for	8.3	3.1	.1	7-	60
imj 8.3.1.17.2.1.2 Act di	portant to safety	for	8.3	3.1	.1	7	60
se	ct underground facilities .		8.	3.1	.1	7-	62
8.3.1.17.2.2 Applic	ation of results		8.	3.1	.1	7-	63
8.3.1.17.3 Investigat informatio could affe	ion: Studies to provide requ n on vibratory ground motion ct repository design or	ired that					
8.3.1.17.3.1 Study:	e	4 4 4 4 4	8.	3.1	.1	7- 7-	63 69
ea	rthquake sources		8.	3.1	. 1	7-	69

TABLE OF CONTENTS (continued)

Page

		8.	3	1	. 1	7.	3.	1.	2	Activi	ty:	Characte	erize 1	10,000-ye	ar		
										releva	nt se	suip ea: ismogen:	conquar le sour	ces ior		8.3.1.17-	72
	8.	3.	1	1	7.	3.	2	S	stud	v: Un	derar	ound nud	lear e	explosion			
								9	our	ces .						8.3.1.17-	74
		8.	3	1	. 1	7.	2	2	1	Antivi	tvi	Determit	he the	range of			
				1						INF SC	UTAAR	br⊘ le le la litule à	10 0110	range or		8 3 1 17-	75
		é.	2	d,		7	2	5	2	Ant ini	+ + + + + + + + + + + + + + + + + + + +	notormir	i i i	man		0.0.4.4.4/-	
		9.			1.4	1	-		*	undere	- y .	Decetumi	ie mari	Linuin			
										underg	round	nucreal	e exbre	ostou		0 0 1 10	
					-					source	:(\$)		1 1 1	1	• •	8.3.1.1/-	15
	ö .	3.	*		1	3	3	2	stud	iy: GI	ound	motion :	trom re	egional			
								e	art	nquake	s and	underg	round r	nuclear			
								6	rdxe	osions	1 a - a -	1 1 1 1	1.1.1	1. 1. 1. 1.		8.3.1.17-	76
		8.	. 3	.1		.7.	.3	.3,	1	Activi	ty:	Select (	or deve	elop			
										empiri	.cal m	odels fo	or eart	thquake			
										ground	i moti	ons				8.3.1.17-	76
		8.	.3	.1	.1	7	. 3	.3.	. 2	Activi	ty:	Select (	or deve	elop empi	r-		
										ical n	nodels	for gr	ound mo	otion fro	m		
										underg	round	nuclea	r explo	osions		8 3 1 17~	77
	8.	3	1	1	7	3	. 4	5	Stud	IV! E	facts	of loc	al site	e deology			1.11
				17		1		1	0000	urfand	and	aubauré.	and mai	e dectodã		0 0 1 17-	
		8	2	1	Ξ,	7	2	4	1	1 APril 1999	+	Datami		ciulis .	• •	0.0.4.4.4	11
		~	1.41	1.14		* 1		1.4.	• •	AUGLY.	r e y i	Decernir.	ne str	e errects			-
		0	-5	1					~	LIOM Q	ircuna	motion	record	aings .	х в	8.3.1.17-	18
		ö	. 0	1.34		1	. 0	. 9.	. 4	ACTIV:	rsA:	Model s	ita er:	iects usi	ng		
										the wa	ave pr	opertie	s of t	he local			
		1			2					deorod	TY .	1.1.1.1			1.1	8.3.1.17-	79
	8.	.3	,1	- 1	.7	. 3	. 5	-	Stud	iy: G:	cound	motion	at the	site fro	m		
								. (	cont	rolli	ng sei	smic ev	ents .			8.3.1.17-	80
		8	.3	. 1		17	.3	.5	.1	Activ:	ity:	Identif	y cont	rolling			
										seism	ic eve	ents	·			8.3.1.17-	80
		8	.3	. 1		17	.3	. 5	.2	Activ.	itv:	Charact	arize	around			
										motio	from	the co	ntroll	ing saien	in .		
										event	9		****		** ~	0 3 1 17-	0.1
	R	2	1		17	2	6	e h	C+ 110	Atra D	rahahi	listia	noi omi			0.0.7.1.	01
	Ť.		1		<b>H</b> . 1	1.4			0000	lunon	Lubaus	LTTDCTC	PETRUT	c nazaros			~~
		0			e.	,			ana.	TApes	1. 1. 1	* * * * *				8.3.1.1/-	. 82
		0	10		<b>k</b> +	41	. 2	.0	• .4	ACTIV	rtă:	Evaluat	e eart	nquake			
				d à						sourc	es .				× •	8.3.1.17-	82
		8	. 3		4.	17	.3	.0	. 2	Activ	ity:	Evaluat	e grou	ind motior	2		
										proba	biliti	ies				8.3.1.17-	- 84
	8	. 3	.]		17	.3	.7		App.	licati	on of	results				8.3.1.17-	- 85
8	.3	.1	.1	7	. 4		In	ve	sti	gation	: Pre	eclosure	tecto	nics data	3		· · · ·
							co	11	ect:	ion an	d ana.	lvsis .				8.3.1.17-	- 86
	8	.3	.1		17	. 4	.1		Stu	dv: H	istor	ical and	CUTTA	int			
									sei	smicit	v					83117.	. 87
		8	1	1	1	17	. 1	1.1	1	Activ	itv.	Compile	histo	ricel		0.0.11.11.	61
		~								oarth	make	rocond		riddi.		0 2 1 17	0.0
		e.			1	1 -	1	1 1	2	Jat in	guare	Monitor		* * * *		0.2.1.1.	- 00
		0	1 2 4		÷ •	* 1		2.2.4	1.54	ACCIV	TCAL	MONICOI	curre	9110			
										36150	LCITV	and the second sec				H 4 1 1 7 7.	- H G

TABLE OF CONTENTS (continued)

8.3.1.17.4.1.3 Activity: Evaluate potential for induced seismicity at the site . . . 8.3.1.17= 91 8.3.1.17.4.2 Study: Location and recency of faulting near prospective surface facilities . . . . . . . . . . . . . . 8.3.1.17-93 8.3.1.17.4.2.1 Activity: Identify appropriate trench locations in Midway Valley. . 8.3.1.17-93 8.3.1.17.4.2.2 Activity: Conduct exploratory trenching in Midway Valley . . . . 8.3.1.17- 95 8.3.1.17.4.3 Study: Quaternary faulting within 100 km of Yucca Mountain, including deep geophysical surveys in an east-west transect crossing the Furnace Creek fault zone, Yucca Mountain, and the Walker Lane . . . 8.3.1.17-105 8.3.1.17.4.3.2 Activity: Evaluate Quaternary faults within 100 km of Yucca Mountain earthquake of 1932 and its bearing on wrench tectonics of the Walker Lane within 100 km of the site . . . . . . . . . . . . . . . 8.3.1.17-127 8.3.1.17.4.3.4 Activity: Evaluate the Bare Mountain fault zone . . . . . . . 8.3.1.17-128 8.3.1.17.4.3.5 Activity: Evaluate structural domains and characterize the Yucca Mountain region with respect to regional patterns of faults and fractures . . . . . . . . . . . . 8.3.1.17-131 8.3.1.17.4.4 Study: Quaternary faulting proximal to the site within northeast-trending fault zones . . . . . . . . . . . . . . . 8.3.1.17-132 8.3.1.17.4.4.1 Activity: Evaluate the Rock Valley fault system . . . . . . . . . . . 8.3.1.17-133 8.3.1.17.4.4.2 Activity: Evaluate the Mine Mountain fault system . . . . . . 8.3.1.17-138 8.3.1.17.4.4.3 Activity: Evaluate the Stagecoach 8.3.1.17-139 fault system . . . . . . . . . . . . . 8.3.1.17-112 8.3.1.17.4.5 Study: Detachment faults at or proximal to Yucca Mountain . . . . . . 8.3.1.17-143

- 1

TABLE OF CONTENTS (continued)

	8		3.	1		17		4		5.	1	Activity: Evaluate the signifi- cance of the Miocene-Paleozoic contact in the Calico Hills area to detachment faulting within the								
	8		3.	1		17		4		5.	2	site area		8.	3.	1.	1	7-	14!	5
	8		3.	1	*	1.	1.	4		5.	3	Bare Mountain area		8.	3.	1.	1	7-	14	7
	8		3.	1		1	7.	4		5.	4	ment faulting		8.	3.	1.	1	7-	14	9
	6		3	1		1'	7.	4		5.	5	Range and Camp Desert Rock areas Activity: Evaluate the age of detachment faults using radiometric		8.	3.	1.	1	7 -	15	0
8	3		1.	1	7	. 1	4.	6			St	ages		8.	3.	1	1	7-	15	1
	8		3	. 1		1'	7.	4	•	6	si.1	te area		8.	3.	1	1	7-	15	4
	8		3	. 1		1	7.	4		6	. 2	faults at Yucca Mountain		8.	3.	1	1	7-	15	5
8	. 3		1	. 1	7		4	7			St	and known Quaternary faults udy: Subsurface geometry and con- ealed extensions of Ouaternary faults		8.	3.	1	. 1	7-	15	7
	60		171	.1		1	7	4		7	at	Yucca Mountain		8.	3	1	.1	7-	16	8
	8		3			1	7	. 4		7		within the site area		8.	3.	.1	.1	7*	16	9
	~	1.	3			1	7	. 4	۱.	7		of the site area	•	8	3	.1	. 1	7-	17	1
	~	3.	3			1	7	. 4	1.	7		survey of the site area 4 Activity: Detailed ground magnetic survey of specific features within	•	8	3	.1	.1	7-	17	2
	*	3	3		1.	1	7		-	7		the site area	•	8	3	.1	. 1	7-	-17	2
												within the site area		8	.3	.1	.1	7.	-17	13

Page

()

TABLE OF CONTENTS (continued)

Page

8.3.1.17.4.7.6 Activity: Evaluate methods to detect buried faults using gammaray measurements, and plan potential application of these methods within the site area . . . 8.3.1.17-175 8.3.1.17.4.7.7 Activity: Evaluate thermal infrared methods and plan potential applications of these methods within the site area . . . . . . . 8.3.1.17-176 8.3.1.17.4.7.8 Activity: Evaluate shallow seismic reflection (mini-sosie) methods and, if appropriate, conduct surveys of selected structures at and proximal to the site area . . . . 8.3.1.17-177 8.3.1.17.4.8 Study: Stress field within and proximal to the site area . . . . . . 8.3.1.17-179 8.3.1.17.4.8.1 Activity: Evaluate present stress field within the site area . . . . 8.3.1.17-180 8.3.1.17.4.6.2 Activity: Evaluate and test shallow borehole hydrofrac and triaxial strain recovery methods for the determination of in situ stress, and if appropriate, plan potential application of these methods within and proximal to the site . . . . . 8.3.1.17-182 8.3.1.17.4.8.3 Activity: Evaluate published and unpublished data on paleostress orientation at and proximal to the site and assess the relevance of these data to Quaternary tectonics . . . . . . . 8.3.1.17-183 8.3.1.17.4.8.4 Activity: Evaluate theoretical stress distributions associated with potential tectonic setting (wrench fault, normal fault, detachment fault setting, etc.) of the site . . . . . . . . . . . . 8.3.1.17-185 8.3.1.17.4.9 Study: Testonic geomorphology of the Yucca Mountain region . . . . . . . . . 8.3.1.17-186 8.3.1.17.4.9.1 Activity: Evaluate age and extent of tectonically stable areas at and near Yucca Mountain . . . . . 8.3.1.17-187 8.3.1.17.4.9.2 Activity: Evaluate extent of areas of Quaternary uplift and subsidence at and near Yucca Mountain . . . . 8.3.1.17-189

TABLE OF CONTENTS (continued)

5 GA -

Page

UNITED DE

	8				1'	7.	4.	9.	3	Activity: Evaluate variations in the nature and intensity of Quater- nary faulting within 100 km of	
										Yucca Mountain through morphometric	
										and morphologic analysis	8.3.1.17-191
8	. 3			17		4.	10		Sti	dy: Geodetic leveling	8.3.1.17-194
	8		3.1	1.	1	7.	4.	10	0.1	Activity: Relevel base-station	
										network, Yucca Mountain and	
										vicinity	8.3.1.17-194
	8	3.1	3.	1.	1	7.	4,	11	0.2	Activity: Survey selected base	
										stations, Yucca Mountain and	
										vicinity, using global position-	
					1		ŝ.	÷.	10	ing satellite	8.3.1.17-196
	1	5	3.	1,	1	7.	4.	1	0.3	Activity: Analyze existing	
										releveling data, Yucca Mountain	
0			2.		d i				~	and vicinity	8.3.1.17-197
0	1	5	4.1	+ 1	1	٩.	+-		SCI	dy: Characterization of regional	
			2		4	-	Å.	4	1 1	leral crustal movement	8.3.1.1/=198
	1	2.1.	ə ,	è i	*	1.4	4		***	Activity: Analyze lateral com-	
										op historical faulting reignicity	
										and trilatoration surveys	6 3 1 17-100
8		2	1	17	÷.,	a.	11	2	2+	and critaceration Surveys	0.3.1.1.=130
Y		e. a.				-			ev	nthaeie	0 3 1 17-100
	3	8.	3.	1.	. 1	7	4	1	2.1	Activity: Evaluate tectonic	0.0.1.1.1.25
				1	1			1	11	processes and tectonic stability	
										at the site	8.3.1.17-200
	j,	8.	3.	1	.1	7	. 4	.1	2.2	Activity: Evaluate tectonic	
										models	8.3.1.17-203
	i s	8.	3.	1	.1	7	ġ,	.1	2.3	Activity: Evaluate tectonic	
										disruption sequences	8.3.1.17-205
8		3.	1.	1	7.	4	.1	3	Ap	plication of results	8.3.1.17-206
3.	1	.1	7.	5		S	ch	ed	ule	for postclosure tectonics	
						p:	ro	gr	am		8.3.1.17-207

A-28

3.

1.\*



Abrams et al., 77 Achauer et al., 101, 102 Ackermann, 48 Ackermann et al., 47, 51, 52, 55 aeromagnetic(s), 17, 18, 19, 21, 23, 25, 42, 47, 55, 81, 108, 122, 130, 137, 141, 142, 144, 164, 165, 167, 168, 169, 171, 178, 184, 185 aeromagnetic profiles, draped, 17, 142 air-gun, 58, 66, 69, 115, 139, 247 aliasing, spatial, 65 Allmendinger, 73 Allmendinger et al., 71, 72 alluvial, 25, 41, 47, 61, 123, 124 alluvial cover, 42, 65, 117 alluvial units, 48, 55 alluvial valleys, 143 alluvium, 3, 11, 12, 33, 41, 42, 48, 58, 60, 64, 69, 70, 72, 94, 95, 108, 117, 123, 136, 149, 150, 165, 166, 190 alluvium, pre-Quaternary, 48 alluvium, Quaternary, 58, 61 alteration, 2, 3, 23, 27, 33, 42, 76, 81, 86, 92, 93, 112, 129, 145, 148, 149, 170, 177, 190 Amargosa Desert, 25, 42, 48, 58, 59, 107, 109, 110, 123, 124, 132, 139, 166, 167, 168, 185, 189 Amargosa River lineament, 126 Amargosa Valley, 51, 57, 70, 72, 137, 144, 168, 183, 185, 186 Amargosa Valley fault, 136 Amargosa Valley test line, 70, 72, 145 Ammonia Tanks Member, 12 Anderson, 48, 55, 92, 33, 95 anisotropy, 34, 115, 122 anomalies, 5, 11, 23, 25, 27, 33, 70, 81, 82, 101, 102, 109, 112, 70, 81, 82, 101, 102, 129, 137, 113, 118, 120, 122, 129, 137, aquifer(s), 166

Α

aquifers and aquitards, Paleozoic, 132 aquitards, 33 argillite, 12, 23, 119 arrival(s), 48, 52, 64, 101, 179 ash fall deposits, 3, 57 ash flow(s), 3, 11, 12, 24, 25, 27, 57, 92, 148, 171 Ash Meadows lineament, 126 attenuation, 65, 72, 112, 115, 116, 118, 144, 151, 167, 179 audiomagnetotelluric (AMT), 33, 34, 40, 122, 143

### B

Bailey et al., 77 Barbier, 60, 73, 151 Bare Mountain, 51, 52, 55, 61 Bare Mountain block, 61 Barry, 62, 73 basalt(s), 12, 17, 23, 25, 26, 27, 57, 99, 168, 169 basaltic volcapism, 102, 137, 138, 161, 162 basaltic volcanism, Quaternary, 102 basalts, Quaternary, 17, 23, 27 basement, 5, 11, 12, 27, 33, 34, 42, 48, 51, 58, 70, 99, 102, 140 basement, depth, 42 basement, Paleozoic, 11, 34, 42, 51, 58, 99, 102, 140 basement, pre-Cenozoic, 5, 11 basin(s), 3, 42, 58, 59, 60, 61, 92, 102, 132 basins, Tertiary, 58, 59 Basin and Range Province, 57, 71, 72 Bath, 23, 25, 28 Bath and Jahren, 23, 27, 28, 120, 126, 151 Bath et al., 27, 28 Beatty, 2, 48, 51, 59, 60, 75, 108, 131 Beatty scarp, 60, 61, 136 bedded tuff(s), 12, 26, 92

bedrock, 41, 65, 94, 95, 123, 124, 143, 149, 150, 165, 171, 177 Bernero, 156 Beydoun et al., 125, 151 Big Smokey Valley, 59 Birdwell, 61, 62, 85 birefringence, 64 Bish and Chipera, 86, 96 Bishop Ash, 150 Blakely, 27, 28, 109 borehole(s), 11, 24, 34, 45, 47, 51, 52, 55, 62, 85, 86, 89, 90, 91, 92, 93, 119, 129, 139, 142 borehole acoustic televiewer (BATV), 93, 112, 122, 125, 146 borahole control, 45, 55, 69, 120, 185 borehole gravimetry, 93, 140 borehole rugosity, 85, 86, 148 borehole television, 116, 93, 112, 125, 146, 148 boreholes, exploratory, 69, 125, 169 boreholes, c-series, 118, 124, 125, 147, 159 boreholes, RF-series, 150 boreholes, UE, 11, 24, 40, 47, 48, 85, 93, 95, 125, 150 boreholes, USW, 24, 62, 85, 93, 115, 116, 117, 118, 125, 142, 145, 146, 147, 167, 170 borings, shallow, 93 Bow Ridge fault, 62, 64, 143 Boynton et al., 27, 28 Boynton and Vargo, 18, 28 Brickey, 77 Brocher, 123 Brocher and Hart, 69, 73 Brocher et al., 59, 69, 70, 73, 132, 144, 145, 151, 185, 186 Bullfrog Member, 12, 24, 25, 26, 92 Burkhard, 58, 69, 150 Burson, 82 Busted Butte, 3, 11, 136 Byers et al., 11, 13

### C

caldera(s), 5, 11, 27, 99, 101, 102, 138 calderas, Miccene, 99, 102

calderas, silicic, 138 calibration, 12, 70, 81, 82, 94, 117, 124, 158, 177, 184 caliche, 48, 123 Calico Hills, 3, 23, 27, 42, 47, 110 Calico Hills tuffs, 12, 26, 92 Calico Hills unit, 3, 4, 57, 92 Caliente, 6, 75, 76, 77, 81 Cane Spring fault system, 136 carbonate(s), 3, 48, 66, 69, 120, 166 carbonate(s), Paleozoic, 3, 66 Carlin district, 75 Carpetbag fault, 60 Carr, 61 Carr et al., 11, 13 Carr and Yount, 28 Cenozoic, 11, 69, 119, 125, 131 cesium, 81 cinder cones, 27 clastic(s), 69, 139 clay(s), 3, 33, 93, 129 clipping, 65 cluster analysis, 137 cobalt, 81 COCORP Death Valley Survey, 71 Colorado School of Mines, 61, 62 Colorado p. teau, 71 common-depth-point (CDP), 60, 61, 62, 64, 69, 71 Connard et al., 109 Consortium for Continental Reflection Profiling (COCORP), 59, 70, 71, 72, 107, 132 contact(s), 24, 33, 42, 57, 66, 69, 70, 85, 94, 95, 115, 120, 123, 126, 132, 140, 142, 147, 149, 150, 165, 166, 171, 189, 190 cooling, 2, 3, 24, 25, 27 Cox and Singer, 129, 151 Coyote Wash, 42, 150, 171 Craig and Robison, 93, 96, 119, 151 Craig and Johnson, 119, 151 Crater Flat, 4, 11, 25, 48, 51, 53, 55, 58, 61, 65, 66, 101, 102, 107, 109, 111, 131, 136, 137, 138, 144, 167, 168, 169, 178, 179, 185, 189 Crater Flat Tuff, 12, 23, 24, 25, 26, 92 Crowe et al., 23, 27, 28

71, 72, 99, 101, 102, 123, 131, 132, 137, 138, 139, 145, 164, 166, 167, 168, 169, 185, 189 crust, lower, 47, 59, 72, 102, 138 crust, middle, 47, 72, 101, 102, 138, 167 crust, upper, 34, 48, 101, 102, 167 crystalline basement, 33, 70 Curie temperature isotherm, 17, 27, 109, 137, 164, 168 Czarnecki, 123 Czarnecki and Oatfield, 42, 45, 123, 152 Czarnecki and Waddell, 119, 152

### D

Daniels and Scott, 40, 45, 85, 96, 146 Daniels et al., 96 Darnley, 82 Davis, 83 Davis and Reinhardt, 81, 82 Dawson et al., 101, 102 Deal, 83 Death Valley, 6, 17, 48, 57, 58, 59, 66, 71, 72, 75, 77, 81, 101, 107, 109, 119, 131, 132, 100 Death Valley-Furnace Creek fault zone, 132 deconvolution, 60 delineation, 33, 42, 57, 137 density, 11, 12, 51, 52, 86, 89, 90, 91, 92, 93, 95, 116, 120, 129, 131, 132, 140, 142, 148, 149, 160, 190 depth of penetration, 34, 47, 55, 94, 143 detachment(s), 4, 27, 58, 69, 70, 71, 72, 132 detachment faults, 58, 131, 137 deVoogd et al., 71, 73, 132, 152 devitrification, 3, 93 dielectric, 89, 90, 116, 117, 147, 148, 177 dike, 26, 57, 71, 119, 169, 179 dipole-dipole, 33, 39, 42 discharge, 77

dolomitic basement, 11 Douglas and Millett, 85, 96 downhole surveys, 48, 52, 65, 69, 70, 116, 144, 164, 167, 169 downhole video camera, 93 Dresser-Atlas, 85 drill core, 23, 24, 26, 69, 137 Drill Hole Wash, 3, 42, 47, 62, 64, 95, 120 drillhole (see borehole) drilling, 11, 27, 48, 85, 86, 94, 95, 116, 117, 119, 122, 125, 139, 151, 159, 160, 169, 170, 171 Duval, 83 Duval and Pitkin, 83

#### E

El Dorado Mountains, 76 Eleana Formation, 27 electrical properties, 86 electrical resistivity (ER), 33, 89, 90, 110, 118, 122, 125, 150, 170, 178, 190 (spe also resistivity) electromagnetic (EM), 33, 39, 41, 75, 110, 112, 113, 118, 125, 139, 143, 146, 147, 148, 170, 171, 178 engineering seismic refraction profiles, 47 enhanced temperatures, 34 Erickson and Waddell, 93, 96 Eric).son et al., 93, 96 eruption(s), 17, 27, 57, 99 eruptions, silicic, 99 eruptive events, 92 Erwin et al., 18, 23, 29 Evans, 99 Evans and Smith, 101 Evans and Oliver, 102, 103 Evans and Zucca, 102, 103, 137, 152 Exile Hill, 47, 62, 64, 65, 66, 149, 150, 151 Exploratory Shaft Facility (ESF), 2, 95, 112, 115, 116, 146, 147, 150, 159, 171

```
explosive sources, 47, 48, 51,
57, 58, 59, 61, 62, 66, 69, 70,
72, 115, 130, 147, 150, 185
```

F

far-field, 115 fault(s), 3, 4, 5, 17, 23, 27, 33, 34, 39, 41, 42, 45, 48, 57, 58, 60-62, 64, 65, 69-72, 81, 82, 99, 109, 110, 115, 118-120, 123, 129-133, 136-140, 142-150, 155, 158-162, 164-166, 170, 177, 178, 180, 189, 190 fault(s), unmapped, 42, 64, 171 fault blocks, 69 fault detection, 33, 42, 57, 65, 143, 164, 165, 177, 190 fault delineation, 33, 165, 190 fault offset(s), 5, 57, 58, 60, 61, 136, 137, 140, 149, 150, 165 fault offset, total, 57 fault scarp, 61 fault slip, 122 fault traces, 81, 142 fault zone(s), 33, 57, 60, 110, 112, 118, 131, 132, 136, 143, 147, 155, 170, 178 faulted blocks, 64, 72 faulting, post-event, 60 Fe-oxide microcrystals, 25 feasibility testing, 1, 105, 110, 113, 117, 118, 120, 122, 123, 125, 136, 137, 143, 146 feeder dike, 71, 169, 179 filtering, frequency-wavenumber (f-k), 64, 65, 71 filtering, velocity, 62 Fitterman, 41, 45, 120, 152 Flanigan, 41, 45 flightline(s), 17, 19, 77, 81, 82, 137, 142 Flint, 94, 123, 150 flow, intraborehole, 93 flow, intraborehole, 93 fluid(s), 33, 34, 85, 86, 93, 95, Greenhaus and Zablocki, 42, 43, 125, 151 Fortymile Wash, 41, 48, 51, 53, ground motion, 5, 51, 130, 132, 119, 123, 124, 158, 169, 185 fracture locations and orientations, 93 fracture pathways, 95, 170 fracture pattern, 76, 132

Fran Ridge, 64, 117, 136 Fran Ridge/Alice Hill horst block, 48 Frenchman Flat, 48, 58, 69 Frischknecht and Raab, 39, 45, 120, 152 Funeral Mountains, 101, 131 Furgerson, 33, 34, 40, 45 Furnace Creek fault, 131 Furnace Creek fault zone, 131, 132

# G

G-tunnel, 147, 175 Galloway and Erickson, 93, 96 Garden Valley, 59 Geodata International, 83 gecelectric, 1, 12, 33, 34, 35, 37, 39-41, 42, 47, 55, 110, 122, 126, 129, 131, 139, 143, 162, 164, 165, 168, 169, 170, 177, 178, 189 geologic mapping, 24, 76, 81, 115, 140, 171 geologic units, 2, 11, 12, 24 geophones, 47, 48, 60, 62, 64, 65, 69, 70, 115, 116, 130, 144, 145, 169 geotemperature, 31, 95, 102 Gnojek and Prichystal, 82, 84 Goldfield, 6, 18, 75, 77, 81 graben, 102 Grapevine Mountains, 51, 131 Grasty and Darnley, 82, 84 gravity, 1, 5, 6, 7, 9, 11, 12, 13, 47, 51, 52, 55, 69, 70, 85, 93, 102, 108, 119, 120, 122, 123, 124, 126, 129, 130, 131, 132, 137, 139, 140, 141, 142, 159, 160, 162, 164, 166, 168, 169, 170, 171, 178, 179, 184, 185, 189 reat Basin of Nevada, 129, 166, 45, 120, 123, 124, 152 .85 137, 149, 150, 184, 185 ground roll, 62, 65 ground surveys, 82, 168 ground penetrating radar (GPR), 150, 160, 165 150, 160, 165

ground-water flow system, 119, 123, 124, 126 Guillou, 84

H

Hagstrum et al., 25, 29, 85, 86, 92, 96 Hamilton, 55, 131 Hardin, 42, 191 Harding, 65, 136 Harris and Fonce, 12, 13, 185, 186 Harris et al., 13 Hasbrouck, 64, 73, 123, 152 Hauser, 73 Hauser et al., 71 Healey, 13, 140, 152, 185, 186 Healey and Miller, 119, 124, 152 Healey et al., 6, 13, 14, 85, 93, 96, 152, 184, 185, 186 heat flow, 2, 5, 27, 95, 131 Hildenbrand and Kucks, 17, 29, 185, 187 Hildenbrand et al., 5, 6, 14, 17, 29 Hill, 18, 23, 29 Hoffman and Mooney, 47, 48, 56, 101, 103, 185, 187 Holocene, 48 Hoover et al., 34, 41, 42, 45 host rock, 3, 64, 92, 165 Howard and Danbom, 65, 73 Hunt, 77 hydrocarbon, 58, 59, 161, 162 hydrology, unsaturated zone, 94, 115, 158, 170 hydrothermal, 75, 76, 81, 111, 113, 129, 132

### I

igneous, 5, 33, 123, 132, 137, 164, 167 impedance, 69, 120 Indian Springs, 107, 108, 109 induced magnetization, 24, 25, 26 induced polarization (IP), 33, 39, 85, 89, 90, 91, 92, 110, 112, 126, 148, 171, 177 infiltration, 77, 93, 94, 95, 110, 117, 118, 123, 124, 158, 159, 177

instrument response, 86, 117, 124
interfaces, steep, 52
interference, 55, 64, 116, 124,
 175, 179
intrusion(s), 23, 33, 71, 72,
 119, 123, 137, 138, 164, 165,
 167
Iyer, 99, 103
Izett, 14
Izett et al., 150, 152

#### J

Jackass Flats, 48, 51, 69, 107, 109, 124, 136, 144, 167 Jansma et al., 14, 184, 187 Johnson and Ege, 14 Jones et al., 65, 105, 144, 153

#### K

Kane and Bracken, 23, 27, 29. 137, 153, 185, 187 Kane et al., 6, 11, 14, 27, 29, 119, 153, 184, 187 Keller, 14 Killeen, 82, 84 Klein, 34, 42 Klemperer et al., 71, 72, 73 Knepper, 77 Knuepfer et al., 72, 73, 131, 153 Koenigsberger ratio, 24, 26 Krohn, 78 Krohn et al., 78 Kruse, 78 Kruse et al., 78 Kruse and Raines, 78

### L

Lahond et al., 93, 96 Landsat MSS, 75, 76 Landsat thematic mapper, 75, 111, 132, 161 large hydraulic gradient area, 27, 107, 110, 162, 119, 120, 121, 122, 123, 139, 145, 155, 161, 162, 164, 167, 169, 170, 177, 178, 190 Las Vegas, 6, 66, 75, 76, 77, 81, 100 Las Vegas shear zone, 58, 66 Lathrop Wells, 18, 23, 59, 61, 69, 168 lateral contrasts, 42 lateral discontinuity, 57, 123 lavas, rhyodacite, 12 limestone, 12 limonitic surface materials, 76 lineaments, 76, 126, 132, 136, 158, 161, 163 listric faults, 69 Lithic Ridge Turf, 12, 26 Little Skull Mountain, 111 logging, 1, 85, 86, 89, 90, 93, 94, 95, 116, 117, 123, 124, 125, 149, 151, 158, 159, 160, 177, 186 Long Vailey, 101 Lovberg and Mose, 82 Lovborg et al., 84 Love waves, 64

.

# M

Mabey, 119, 153 magma, 122, 130, 137 magma bodies, 70, 72, 131, 137, 138 magma chamber(s), 33, 99, 101, 138 magma chambers, silicic, 99, 138 magma source, 102, 138 magmatic intrusion(s), 71, 164 magnetic(s), 12, 17, 23, 24, 25, 26, 27, 28, 41, 70, 86, 91, 108, 116, 117, 119, 120, 122, 126, 129, 130, 131, 132, 137, 139, 142, 148, 159, 160, 162, 164, 168, 169, 171, 177, 178, 179, 184, 185, 189 magnetic susceptibility, 23, 25, 27, 89, 90, 92, 116, 148 magnetite, 23, 27, 92 magnetotelluric (MT), 33, 34, 40, 109, 120, 121, 122, 130, 132, 137, 143, 144, 169, 170, 171, 178 mantle, upper, 99, 101, 138, 164 mapping, 11, 24, 25, 42, 47, 57, 59, 72, 76, 77, 75, 81, 115, 124, 130, 140, 146, 147, 148, 150, 171 Marvin et al., 12, 14

McArthur and Burkhard, 58, 69, 70, 74, 107, 136 McGovern, 61, 62, 63, 64, 65, 107, 144, 145, 153 McGovern et al., 58, 74 melt(s), partial, 99, 102, 179 Mesozoic, 27, 66 Mid Valley, 58, 69, 70, 107 Midway Valley, 33, 47, 48, 52, 62, 149, 150, 160, 165, 167, 178 Mine Mountain fault, 136 Mini-Sosie, 57, 58, 60, 61, 65, 107, 122, 133, 136, 137, 139, 145, 150, 158, 160, 161, 165, 175, 180, 189 Minster et al., 101, 103 Miocene, 11, 99, 101, 102, 132, 139, 171 modeling, 11, 23, 34, 39, 40, 93, 108, 113, 114, 122, 123, 149, 151, 156, 164, 169, 181 modeling, off-axis, 48, 70, 72 Moho, 47, 59, 70, 72, 99, 102, 138, 164 moisture, 85, 86, 92, 93, 94, 95, 112, 116, 117, 118, 123, 124, 146, 147, 148, 149, 170, 186 Mojave Desert, 71 Monfort and Evans, 99, 101, 103 Monitor Valley, 59 Mono Craters, 101 Mooney and Schapper, 47, 51, 56 Morelli, 14 Muller and Kibler, 85, 86, 97

### N

National Science Foundation, 70
National Uranium Resource
 Evaluation (NURE), 17, 18, 81,
 82
Nevada Test Site (NTS), 1, 3, 6,
 11, 17, 24, 35, 48, 57, 58, 66,
 69, 70, 81, 82, 85, 100,
 101, 119, 138, 143, 175
Niblack, 78
Nimick and Schwartz, 97
noise, 51, 55, 62, 65, 70, 71,
 72, 142, 144, 145, 167
nonwelded, 2, 3, 11, 25, 45, 86,
 92, 190

0

O'Leary et al., 78 O'Leary and Pohn, 79 O'Leary and Friedman, 78 ODEX drilling system, 94, 117 offset(s), 57, 61, 64, 120, 136. 137, 140, 149, 150, 165 oil and gas exploration, 66, 67 Oliver, 123 Oliver and Carle, 120 Oliver and Ponce, 17, 29 Oliver et al., 6, 11, 15 Osgood Mountains, 75 outcrop(s), 23, 25, 64, 101, 111

#### P

Pagany Wash, 123 Pah Canyon Member, 12, 25, 26 Pahranagat, 108 Pahrump Valley, 59, 65 Pahute Mesa, 11, 108, Paillet, 146, 153 Paintbrush Canyon fault, 48, 143 Paintbrush Canyon-Busted Butte fault zone, 136 Paintbrush Tuff, 3, 12, 23, 24, 25, 26, 27, 42, 45, 92 paleomagnetic, 17, 23, 24, 25, 111, 137, 135 paleotectonic history, 71 Paleozoic, 12, 42, 55, 69, 101, 111, 119, 123, 125, 126, 129, 132, 138 Paleozoic-Miocene contact, 132 Paleozoic-Tertiary contact, 66, 69, 70, 120, 166, 171, 189, 190 Pankratz, 47, 48, 49, 51, 52, 56, 120, 123, 145, 150, 153, 185, 187 partially welded, 3, 11 petrophysics, 86, 91, 114, 149 perching, 94 percolation, 109, 116, 118, 143 photogeology, 57 Pitkin, 82, 84 Pitkin and Huffman, 82, 84 Pitkin and Duval, 84 plate motions, 99 plutons, 5 Podwysocki et al., 79 Poisson ratio, 151

polarization, 33, 93, 126, 148, 171 polarization signatures, normal, 92 polarization signatures, reversed, 92 Ponce, 15, 23, 29, 126, 153 Ponce and Hanna, 119 Ponce et al., 6, 15, 184, 187 pore(s), 33, 34, 92, 149
porosity, 2, 3, 33, 86, 91, 92,
93, 116, 148, 149, 160 potassium, 81, 82 pre-Cenozoic, 5, 11 pre-Quaternary, 48 pre-Tertiary, 55, 71, 72, 125, 126 Precambrian, 12, 132 precipitation, 86, 94, 95, 118 properties, physical, 86, 92, 126, 149, 151 Prow Pass Member, 12, 24, 26, 92, Purdy and Rowan, 79 Purdy et al., 79

#### 0

quartzite, 12 Quaternary, 12, 17, 23, 27, 48, 58, 61, 102, 126, 131, 132, 136, 138, 139

#### R

radioactivity, 81, 143, 177 radioelement(s), 81, 82, 129, 143 radioisotopes, 81 Railroad Valley, 59 Raines, 79 Rainier Mesa, 3, 64 Rainier Mesa Member, 12, 26, 60 Rainier Mesa tuff, 60 Ramirez, 153 Ramirez and Daily, 147 ray-tracing, 48, 51 recharge, 119, 123, 124, 158 reflection Moho, 71, 72 reflections, intrabasin, 69 reflections, mid-crustal, seismic, 71, 72

reflector(s), 51, 61, 64, 65, 69, Sandia National Laboratories 72, 139, 147, 164, 165 remanent magnetization, 23, 24, 25, 26, 27, 92 remote sensing, 57, 75, 76, 110, 118, 129, 130, 131, 143, 177 repository block, 2, 25, 42, 102, 107, 139, 140, 146, 148, 155, 170, 178, 190 resistivity, 33, 39, 40, 41, 42, 45, 91, 92, 93, 110, 112, 113, 117, 119, 120, 122, 123, 124, 125, 126, 146, 147, 148, 150, 170, 171, 177, 190 (see also electrical resistivity) resolution, 55, 101, 115, 131, 137, 138, 142, 143, 145, 147, 148, 165, 189 resource exploration, 66 Reynolds, 64, 65, 150, 151 Robbins et al., 85, 93, 97 Roberts et al., 6, 15 Robinove, 79 Robison, 119, 153 Robison et al., 119, 153 rock alteration, 33 Rock Valley, 48 Rock Valley fault, 136 Rodriguez and Yount, 47, 48, 56, 136, 150 Rosenbaum, 24, 29 Rosenbaum and Rivers, 23, 30 Rosenbaum and Hudson, 24, 30 Rosenbaum and Schlinger, 25, 30 Rosenbaum and Snyder, 24, 25, 26, 30 Rosenbaum and Spengler, 25, 30 Ross and Lunbeck, 39, 46 Rowan et al., 80 Rowan and Wetlaufer, 80 Rowan and Purdy, 80 runoff, 86, 95 Rush et al., 93, 97, 119, 153

## S

Saltus, 5, 6, 15, 184, 187 Saltus and Ponce, 17, 30, 185, 187 Saltus and Snyder, 17, 30, 185, 187 San Francisco Mountain, 101

(SNL), 187 Sandy speculative survey, 59 Sass et al., 27, 30, 95, 97 saturated zone (SZ), 3, 85, 86, 90, 112, 116, 118, 119, 123, 124, 125, 126, 146, 148, 159, 178 Sawatzky, 80 scatter(ing), 34, 52, 61, 72, 112, 115, 145 Schlinger et al., 25, 30 Schlumberger sounding(s), 33, 34, 39, 42, 43, 123, 143 SCOLL and Bonk, 145, 154 Scott and Castellanos, 52, 56 Scott and Deenbaum, 24, 30 SCP Sections 8.3.1.2, 110, 175 8.3.1.2.1.3, 112, 169 8.3.1.2.1.3.1, 158, 161 8.3.1.2.1.3.2, 158, 161, 162, 163 8.3.1.2.1.3.3, 158 8.3.1.2.2.1, 112, 159 8.3.1.2.2.1.1, 107, 158, 160, 161, 163, 177 8.3.1.2.2.1.2, 117, 118, 158, 159 8.3.1.2.2.1.3, 117 8.3.1.2.2.2.3, 158 8.3.1.2.2.3, 107, 112, 149, 178 8.3.1.2.2.3.1, 116, 158 8.3.1.2.2.3.2, 107, 115, 116, 118, 147, 148, 158, 159, 170, 177, 179 8.3.1.2.2.3.3, 159 8.3.1.2.2.4, 112, 159 8.3.1.2.2.4.2, 112 8.3.1.2.3.1, 159, 169, 8.3.1.2.3.1.3, 124 8.3.1.2.3.1.4, 124 8.3.1.2.3.1.5, 124 8.3.1.2.3.1, 159, 178 8.3.1.2.3.1.1, 159 8.3.1.2.3.1.2, 159, 161, 162, 178 8.3.1.4, 175 8.3.1.4.1.2, 1, 105, 116, 126, 129, 130, 138, 155, 156, 167, 177, 183 8.3.1.4.2, 105 8.3.1.4.2.1, 139, 149, 167, 169 8.3.1.4.2.1.1, 139

8.3.1.4.2.1.2, 107, 122, 123, 145, 158, 159, 169 8.3.1.4.2.1.3, 85, 111, 112, 113, 117, 148, 151, 158, 159, 160, 177, 178 8.3.1.4.2.1.4, 114 8.3.1.4.2.1.5, 111, 137, 160 8.3.1.4.2.2, 135 8.3.1.4.2.2.3, 112, 113, 116, 122, 125, 146, 158, 159, 160, 170, 178 8.3.1.4.2.2.4, 158, 160 8.3.1.4.2.2.5, 107, 113, 115, 147, 159, 160, 167, 170, 175, 179 8.3.1.4.3.3.1, 116, 148, 159, 160 8.3.1.5.1.2.2, 139, 179 8.3.1.5.2.1.3, 158, 161, 162, 163 8.3.1.8, 4 8.3.1.8.1.1, 137, 160, 161, 162 8.3.1.8.1.1.3, 131, 137, 138, 179 8.3.1.8.2.1.3, 137 8.3.1.8.5, 139 8.3.1.8.5.1, 137, 160, 161, 162, 178, 179 8.3.1.8.5.2.1, 109, 137 8.3.1.9.2.1, 171 8.3.1.9.2.1.2, 162 8.3.1.9.2.1.4, 130, 161 8.3.1.14.2.3, 150, 151, 160 8.3.1.14.2.3.3, 108, 113 8.3.1.15.1.1.1, 160 8.3.1.17, 4, 149, 175 8.3.1.17.1, 137 8.3.1.17.4, 105 8.3.1.17.4.1.2, 137, 160, 179 8.3.1.17.4.2, 108, 149, 150, 160 8.3.1.17.4.2.1, 158, 162 8.3.1.17.4.3, 130, 137, 104, 179 8.3.1.17.4.3.1, 72, 107, 108, 109, 130, 131, 132, 137, 144, 161, 166, 178 8.3.1.17.4.3.2, 111, 161, 162 8.3.1.17.4.3.5, 111, 130, 132, 161 8.3.1.17.4.4, 107, 158, 160, 161 8.3.1.17.4.4.1, 136 8.3.1.17.4.4.2, 136

8.3.1.17.4.4.3, 136 8.3.1.17.4.4.4, 136 8.3.1.17.4.5, 137 8.3.1.17.4.7, 137, 138, 139, 164, 171, 178, 179 8.3.1.17.4.7.1, 70, 107, 123, 144, 145, 159, 161, 166, 178 8.3.1.17.4.7.2, 108, 122, 137, 139, 159, 160, 162, 11, 178 8.3.1.17.4.7.3, 108, 122 137, 142, 144, 162 8.3.1.17.4.7.4, 109, 122, 137, 142, 159, 162, 168, 169 177, 185 8.3.1.17.4.7.5, 110, 122, 123, 143, 162, 168, 169, 170, 177, 178 8.3.1.17.4.7.6, 110, 143, 162, 177 8.3.1.17.4.7.7, 110, 118, 163, 177 8.3.1.17.4.7.8, 107, 133, 136, 137, 145, 158, 160, 161, 175, 180 8.3.1.17.4.9.1, 111, 136, 161 8.3.1.17.4.12.1, 108 sedimentary, 11, 12, 33, 34, 40, 59, 62, 66, 92 Seisdata, 58, 59, 61, 62 seismic basement, 48, 70 seismic, intermediate-depth, 57, 66, 69, 70, 144, 189, 127, 135 seismic reflection, 12, 47, 55, 57, 58, 59, 62, 63, 67, 122, 123, 127, 130, 131, 133, 135, 136, 137, 138, 139, 144, 145, 150, 155, 164, 165-171, 175, 183-186, 189-190. seismic reflection, shallow, 60, 64, 65, 66, 107, 108, 109, 136 seismic reflection, shallow high-resolution, 57 seismic reflectors, shallow, 61 seismic refraction, 11, 12, 47, 49, 51, 52, 55, 120, 123, 124, 126, 130, 131, 136, 137, 139, 144, 145, 150, 164, 166, 167, 169, 171, 184, 185, 186, 189 seismic refraction, shallow, 47, 52, 107, 108, 123, 150, 158 seismicity, 131, 137, 160 Senterfit et al., 39, 46, 120, 154 Serpa, 58

Serpa et al., 59, 65, 71, 72, 74, 132, 154 Serra, 116 shear wave, 64, 65, 107, 115, 125, 139, 150, 151, 165, 179 Shephard, 150 Shoshone Mountain, 110 shot hole(s), 51, 52, 72, 150 signatures, 45, 92, 99 Silent Canyon, 138 Silent Canyon caldera, 11, 101 Site Characterization Analysis (SCA), 156 Site Characterization Technical Planning Basis, 105, 155 Skull Mountain, 51, 111 sledgehammer source, 48, 64, 139, 150 Smith and Ross, 39, 42, 45, 46 Smith et al., 120, 154 snowmelt, 85, 94 Snyder and Carr, 6, 11, 12, 15, 51, 52, 56, 119, 126, 140, 154 Snyder and Oliver, 23, 30 Snyder et al., 5, 15, 184, 187 Solitario Canuon, 102, 155 Solitario Canyon, 102, 159 sorting, 60, 61, 62 Southern Great Basin, 6, 100, 178 Southern Great Basin (SGB) Seismic Network, 33, 99, 100, 101 spectral discrimination, 76 spectral whitening, 60 Spence, 101, 103 Spengler et al., 85, 86, 97 Spring Mountains, 66 Stagecoach Road fault, 136, 143 statics, 70, 144 statics, residual, 61, 62 station(s), 1, 7, 9, 23, 33, 34, 62, 64, 71, 93, 99, 100, 101, 102, 139, 140, 142, 178, 184 Stauber, 101, 103 Stovepipe Wells, 107, 108, 109 stress field, 119 stress measurement(s), 2, 5, 122, 124, 131, 146 strike-slip, 61 structural rotation about a horizontal axis, 24, 60, 178 structural rotation about a vertical axis, 17, 23, 24, 27

8

subsurface, 4, 11, 12, 17, 42, 57, 70, 125, 136, 137, 145, 145, 166, 171, 179, 182 subsurface bodies, 42 subsurface characterization, 57 subsurface extension(s), 60, 132, 168 subsurface faulting, 61 subsurface geometry, 60, 70, 131, 138 sulfides, 112, 129 sulfide mineralization, 33, 112, 126 surface facilities, 47, 64, 108, 113, 142, 149, 150, 151, 162, 165 surface hydrology, 75 surface mapping, 11 surface outcrops, 23, 25 Sutton, 51, 56, 124, 154, 185, 188 Swadley and Boover, 61 swath-type profiles, 62 Swedish and German repository programs, 146 Syncline Ridge, 42 synthetic aperture radar (SAR) mosaics, 76

### T

Tarantula Canyon, 61, 65 Taylor, 101, 103 tectonic(s), 1, 2, 3, 5, 12, 47, 48, 66, 71, 105, 122, 129, 130, 132, 138, 139, 143, 145, 155, 160, 164, 166, 171, 185, 189 teleseismic, 5, 99, 101, 102, 137, 160, 167, 168, 169 teleseismic tomography, 1, 5, 99, 100, 137, 130 teleseismic velocity, 33 telluric soundings, 33, 34 telluric ratio, 34, 143 tensor impedance modeling, 34 Tertiary, 12, 25, 55, 120, 126, 161, 162, 166, 171, 189, 190 thermal infrared, 75, 77, 110, 118 chermal/mechanical, 160 Thirsty Canyon Tuff, 69 thorium, 81, 82 Tikaboo Valley, 59

Timber Mountain, 11, 12, 18 Timber Mountain Tuff, 23, 26, 60, 69 Timber Mountain caldera, 11, 27, 138 Timber Mountain-Oasis Valley calders complex, 102, 138 Timber Mountain-Silent Canyon, 99, 138 Tiva Canyon Member, 3, 12, 24, 25, 26, 27, 45, 48 Tonopah, 27 topography, 2, 17, 39, 42, 51, 57, 81, 86, 119, 139, 140, 142, 143, 170, 189 Topopah Spring, 18, 48 Topopah Spring Member, 3, 4, 12, 24, 25, 26, 45, 48, 64, 85, 92, 111, 148 Tram Member, 12, 24, 25, 26, 92 transport properties, 92 travel time(s), 51, 55, 60, 65, 69, 70, 72, 101, 132, 138 123, 127, 131, 132, 136, 142, 143, 144, 166, 169, 170, 178, 180 traverses, 12, 23, 37, 108, 109, Trench 14, 64 trenches, 61 trenching, 149, 150, 16 tuff(s), 2 2 trenches, mapped, 48 tuff(s), 2, 3, 4, 11, 12, 23, 24, 25, 26, 27, 42, 45, 51, 60, 64, 65, 69, 70, 86, 92, 93, 94, 99, 101, 102, 117, 119, 120, 125, 138, 140, 147, 148, 150, 167, 189, 190,

#### U

underground nuclear explosion(s) (UNEs), 46, 51, 60, 101, 185 United States Geological Survey (USGS), 2, 18, 26, 27, 30, 31, 40, 42, 47, 48, 51, 57, 58, 60, 61, 65, 69, 72, 74, 75, 76, 80, 81, 82, 85, 124, 126, 129, 131, 136, 145, 146, 150, 185 unsaturated zone (UZ), 3, 85, 86, 94, 112, 115, 116, 117, 118, 120, 123, 124, 125, 146, 147, 148, 158, 170, 177, 178, 179, 186

upper-crustal profile(s), 51, 52, 124 upper=crustal refraction, 47, 51, 53, 55, 123, 124, 137, 139, 145, 156, 169, 185 upper-crustal structure, 47, 55, 131 uranium, 17, 81, 82, 126 USGS Nuclear Hydrology Project, 86 USGS Remote Sensing Tape Library, 75 U.S. Department of Energy (DOE), 17, 18, 28, 29, 81, 83, 99, 105, 146, 155, 171, 181, 182, 183, 186 U.S. Nuclear Regulatory Commission (NRC), 99, 156, 165, 166, 168, 171, 181, 182

#### V

velocity, 47, 48, 51, 52, 55, 60, 61, 62, 64, 65, 66, 69, 70, 86, 91, 92, 93, 99, 101, 102, 115, 116, 120, 131, 138, 139, 145, 149, 151, 164, 167, 169, 184, 185, 189, 190 vertical seismic profiling (VSP), 107, 115, 116, 118, 125, 144, 147, 148, 159, 160, 167, 170, 175, 178, 179, 189 vertical seismic profiling (VSP), hydrophone, 125 VHF tomography, 118, 125, 147 vibrator source, 57, 60, 61, 62, 66, 69, 70, 72, 115, 116, 139, 147, 170 Vibroseis, 58, 59, 66, 71, 72, 107, 145 vitrophyre(s), 3, 69 volcanic(s), 23, 24, 25, 26, 27, 28, 34, 40, 47, 52, 55, 57, 61, 65, 75, 99, 102, 109, 120, 129, 132, 137, 139, 142, 148, 156, 160, 161, 162, 164, 167, 168, 169, 179, 189 volcanic cones, 61 volcanic dike, 119 volcanic eruption, 160 volcanic units, 24, 124, 129 volcanics, Tertiary, 42, 66, 124

volcanism, 101, 102, 130, 161, 162, 164, 168, 169, 179, 185 volcanism, Miocene, 11, 101, 102, 132, 139

#### W

Waddell, 119, 154 Wahmonie, 23, 42, 47 Walch, 101, 103 Walch and Phillips, 101, 103 Walker Lane, 34, 72, 76, 131, 132 Ward, 84 warping, 60 Wasatch Cordilleran hingeline, 58 water, meteoric, 118, 143 water table, 3, 69, 92, 93, 110, 111, 112, 115, 120, 123, 146, 151, 169 weapons testing, 66, 70 weight-drop source, 57, 151 welded (and welding), 2, 3, 4, 11, 25, 85, 86, 92, 117, 147, 190 White River Valley, 59 Whitfield et al., 93, 97 Windy Wash fault, 61, 65 Winograd and Thordarson, 119, 154 Wingate Wash fault, 71 Wright, 147 Wynn and Roseboom, 105, 154

### Х

x-ray diffraction, 86

#### Y

Yeates and others, 82, 84 Yucca Flat, 24, 58, 69, 111, 350 Yucca Flat basin, 60 Yucca fault, 60 Yucca Mountain Member, 12, 26 Yucca Mountain profile, 51 Yucca Valley, 27 Yucca Wash, 145 Z

zeolite(s), 3, 93, 129, 148
zeolitization, 93, 112
Zumberge et al., 12, 15, 185

en la

.